Earnings Management, Management Compensation, Managerial Ability and Market Competition

by

Malek Taisir Mohammed El Diri

Submitted in accordance with the requirements for the degree of

Doctor of Philosophy

The University of Leeds

Leeds University Business School

Accounting and Finance Division

Centre for Advanced Studies in Finance

July, 2016

Intellectual Property Statement

The candidate confirms that the work submitted is his own and that appropriate credit has been given where reference has been made to the work of others.

This copy has been supplied on the understanding that it is copyright material and that no quotation from the thesis may be published without proper acknowledgement.

Dedication

To the memory of my father.

Acknowledgements

I would like to thank my dear supervisors Kevin Keasey and Iain Clacher, and all the academic and administration members in Leeds University Business School, especially Mohammad Alhadab, Julio Pindado, Steve Toms, Alan de Ricquebourg, Francesco Vallascas, and Michelle Dickson. I am grateful to the feedback from the participants in the European Accounting Association Congress (2015, Glasgow), the participants in the International Accounting and Finance Doctoral Symposium (2015, Ljubljana), and the participants in the European Accounting Association Congress (2016, Maastricht). Finally, thanks to my beloved mother, brother, friends, colleagues and everyone else who supported me over the journey to accomplish this project.

Abstract

As a result of the agency problem, earnings management may take place due to the high contracting costs, shareholders' bounded rationalities, and information asymmetry. Therefore, three main groups of motives have been identified to explain earnings management behaviour at the contracting, capital market, and external levels. While the previous studies have individually examined those motives, this thesis provides evidence that they interact in determining earnings management behaviour.

The first empirical chapter of this thesis focuses on the contracting factors and examines the impact of earnings management on executive compensation conditioned on managerial ability. It finds that managers who utilize accrual earnings management receive higher compensation than those who undertake real earnings management. However, high quality managers are rewarded less for accrual earnings management and punished less for real earnings management.

The second empirical chapter examines the non-linear effect of market concentration as an external motive of earnings management. It documents that accrual earnings management increases in concentrated markets as the quantity of information decreases. However, the sophisticated real earnings management starts to substitute for discretionary accruals at higher levels of market concentration when the quality of information declines.

The third empirical chapter combines factors from the contracting and external motives. It examines the effect of market competition on the relationship between managerial ability and earnings management. The results show that in the face of increased competition, high quality managers manipulate earnings via accruals rather than more costly real earnings management.

Overall, the results of this thesis show that management compensation is a crucial factor in assessing the costs of earnings management at the firm level. An optimal level of market concentration exists and should be considered by the regulators. Finally, understanding how industry level factors influence managerial decisions at the firm level is essential to explaining earnings management behaviour.

Table of Contents

Dedication.	ii	
Acknowledgementsiii		
Abstract	iv	
Table of Con	ntentsv	
List of Table	esxii	
List of Figur	resxiv	
List of Appe	endices xv	
List of Abbr	eviationsxvi	
Chapter 1	Introduction	
1.1 Inti	roduction1	
1.2 Ma	nagement Compensation	
1.3 Ma	rket Concentration7	
1.4 Ma	nagerial Ability and Market Competition	
1.5 Res	search Questions	
1.6 Sig	nificance and Contributions10	
1.7 The	esis Structure	
Chapter 2	Definitions, Activities, and Measurement of Earnings Management 16	
2.1 Intr	roduction	
2.2 Det	finitions17	
2.2.1	Earnings Management Definition17	
2.2.2	The Difference between Earnings Management and Fraud	
2.2.3	The Difference between Earnings Management and Earnings Quality 22	
2.2.4	The Difference between Earnings Management and Impression	
Manage	ement	
2.2.5	The Difference between Earnings Management and Expectation	
Manage	ement	
2.3 Ear	rnings Management Activities	
2.3.1	Accrual and Real Earnings Management	
2.3.2	Earnings Smoothing	
2.3.3	Income Shifting	

2.3.4	Advanced Techniques	
2.3.5	The Trade-off between Earnings Management Activities	
2.4 Accrual Earnings Management Models		
2.4.1	The Ronen and Sadan Model	
2.4.2	The Healy Model	
2.4.3	The DeAngelo Model	
2.4.4	The Industry-Based Model	
2.4.5	The Jones Model	
2.4.6	The Modified Jones Model (Dechow et al., 1995)	41
2.4.7	The Forward-Looking Model	
2.4.8	The Competing-Component Model (The K S Model)	
2.4.9	The Cash-Flows Model	
2.4.10	The Cash-Flows Jones Model	
2.4.11	The Performance-Matching Model	
2.4.12	2 The Business Model	
2.4.13	B The Stubben Model	
2.4.14	The Single-Item Models	
2.4.15	5 The Distributional Approach	
2.4.16	5 Evaluation of Accrual Earnings Management Models	
2.5 R	eal Earnings Management Models	50
2.5.1	The Roychowdhury Model	50
2.5.2	The Gunny Model	
2.5.3	Evaluation of Real Earnings Management Models	54
2.6 S	pecial Measures of Earnings Management	55
2.6.1	Combined Models	55
2.6.2	Earnings Management Measures in the Financial Sector	57
2.6.3	Earnings Management in Qualitative Research	57
2.7 C	onclusions	
Chapter 3	Theoretical Framework	61
3.1 In	troduction	61
3.2 Fi	irm Theory	
3.3 A	gency Theory	

3.4	Rev	velation Principle	. 65
3.4	.1	Contracting Theory	. 67
3.4	.2	Bounded Rationality	. 71
3.4	.3	Information Asymmetry	. 72
3.5	Ear	nings Management Theories	. 75
3.5	5.1	The Costly Contracting Approach	. 76
3.5	5.2	The Decision-Making Approach	. 77
3.5	5.3	The Legal-Political Approach	.77
3.6	Cor	nclusions	. 79
Chapte	r4 I	Motives of Earnings Management	. 81
4.1	Intr	oduction	. 81
4.2	Ear	nings Management vs. Truth Telling	. 82
4.3	Cor	ntracting Motives of Earnings Management	. 84
4.3	8.1	Management Compensation	. 85
4.3	5.2	CEO Turnover	. 87
4.3	8.3	Managerial Ability	. 89
4.3	8.4	Corporate Governance System	. 92
4.3	8.5	Loans	. 96
4.3	6.6	Firm Characteristics	. 97
4.4	Cap	bital Market Motives of Earnings Management	. 97
4.4	.1	Stock Market	. 97
4.4	.2	Issuance of Equity	100
4.4	.3	New Listing and Cross-Listing	101
4.4	.4	Mergers and Acquisitions	101
4.4	.5	Insider Trading	102
4.4	.6	Management Buyouts	103
4.4	.7	Meeting or Beating a Benchmark	104
4.4	.8	Analysts	104
4.5	Thi	rd-Party Motives of Earnings Management	106
4.5	5.1	Industry	107
4.5	5.2	Industrial Diversification	109
4.5	5.3	Regulations	109

Political Environment and Country-Specific Policies	
Accounting Standards	
Tax Considerations	
Competitors	
Suppliers and Customers	116
Auditors	116
Other Gatekeepers	
onclusions	
Data and Research Methods	
troduction	
mple Construction	
ata Sources	
ne Estimation of Earnings Management	
Accrual Earnings Management	
Real Earnings Management	
ethodology	
onclusions	
Management Compensation: The Impact of Earnings Manag	gement and
l Ability	
troduction	
terature and Hypotheses Development	
Managerial Ability and Management Compensation	
Earnings Management and Management Compensation	
Managerial Ability, Earnings Management, and Management ensation	
ata, Variable Definitions and Descriptive Statistics	
Data	
Earnings Management Measures	
Managerial Ability Measure	
Compensation Variables	
Control Variables	
Descriptive Statistics	
	Political Environment and Country-Specific Policies

6.4	Methodology	150
6.5	Results	154
6.5	5.1 Managerial Ability and Management Compensation	154
6.5	5.2 Accrual Earnings Management and Management Compensation	156
6.5	5.3 Real Earnings Management and Management Compensation	158
6.5 Co	5.4 Managerial Ability, Accrual Earnings Management and Manageme	nt 158
6.5 Co	5.5 Managerial Ability, Real Earnings Management and Management ompensation	161
6.6 Perfo	Additional Analysis: Managerial Ability, Earnings Management and Opportunity	erating
6.7	Conclusions	164
Appe	endix A	166
Chapte	er 7 Market Concentration and Earnings Management: A Non-Linea	r
Relation	onship	168
7.1	Intoduction	168
7.2	Literature and Hypotheses Development	170
7.2	2.1 Market Concentration and Earnings Management	170
7.2	2.2 Market Concentration and Information Asymmetry	174
7.3	Data, Variable Definitions and Descriptive Statistics	176
7.3	3.1 Data	176
7.3	3.2 Earnings Management Measures	177
7.3	3.3 Market Concentration Measures	178
7.3	3.4 Information Asymmetry Measures	178
7.3	3.5 Control Variables	179
7.3	3.6 Descriptive Statistics	184
7.4	Methodology	187
7.5	Results	191
7.5	5.1 Market Concentration and Earnings Management	191
7.5	5.2 Market Concentration and Information Asymmetry	193
7.6	Additional Analysis: Governance Controls	195
7.7	Conclusion	196
Appe	endix B	198

Chapte	er 8 1	Earnings Management, Managerial Ability and the Role of	
Compe	titior	1	201
8.1	Intr	oduction	201
8.2	Lite	erature and Hypotheses Development	203
8.2	2.1	Earnings Management Behaviour	203
8.2 M	2.2	Managerial Ability, Market Competition and Accrual Earnings	204
8.2	2.3	Managerial Ability, Market Competition and Real Earnings Manageme 206	ent
8.3	Dat	a, Variable Definitions, and Descriptive Statistics	207
8.3	3.1	Data	207
8.3	3.2	Earnings Management Measures	208
8.3	3.3	Managerial Ability Measure	209
8.3	3.4	Competition Measures	209
8.3	3.5	Control Variables	210
8.3	3.6	Descriptive Statistics	213
8.4	Me	thodology	219
8.5	Res	sults	223
8.5 Ma	5.1 anage	Managerial Ability, Market Competition and Accrual Earnings	223
8.5	5.2	Managerial Ability, Market Competition and Real Earnings Manageme 225	ent
8.5	5.3	Comparison of Competition Measures	227
8.6	Ado	ditional Analysis	228
8.6 bet	5.1 tweer	Comparing the Effect of Managerial Ability on Earnings Management n High and Low Competition Settings	228
8.6	5.2	Earnings Smoothing, Managerial Ability, and Market Competition	232
8.6	5.3	Additional Control Variables	234
8.7	Cor	nclusions	235
Appe	endix	C	236
Chapte	er 9 (Conclusions	244
9.1	Intr	oduction	244
9.2	Sur	nmary of Results	245

9	9.2.1 Management Compensation: The Impact of Earnings Management and		and
Ν	Manag	erial Ability	246
ç F	0.2.2 Relatio	Market Concentration and Earnings Management: A Non-Linear	247
9	0.2.3	Earnings Management, Managerial Ability and the Role of Compet 248	tition
9.3	Im	plications	249
9.4	Liı	mitations and Future Research	250
10	Bibli	ography	252

List of Tables

Table 5.1 Summary of the samples used in the thesis 123
Table 5.2 The selection of estimator
Table 6.1 Calculation of all variables 145
Table 6.2 Expected and actual signs of the control variables 147
Table 6.3 Descriptive statistics 148
Table 6.4 Correlation table 149
Table 6.5 The effect of managerial ability on management compensation 155
Table 6.6 The effect of earnings management on management compensation
Table 6.7 The effect of earnings management and managerial ability on management compensation 160
Table 6.8 The effect of earnings management and managerial ability on future operating performance 163
Table 7.1 Calculation of the variables 181
Table 7.2 Panel A Expected and actual signs of the control variables - The effect on earnings management
Table 7.3 Descriptive statistics 184
Table 7.4 Correlation table 186
Table 7.5 The effect of market concentration on earnings management
Table 7.6 The effect of market concentration on information asymmetry 194
Table 8.1 Calculation of control variables 212
Table 8.2 Expected and actual signs of the control variables 213
Table 8.3 Descriptive statistics 214
Table 8.4 Correlation table 216
Table 8.5 Earnings management variables partitioned by competition and managerial ability 218
Table 8.6 The moderating role of market competition in the relation between managerial ability and accrual earnings management (AccrualEM)
Table 8.7 The moderating role of market competition in the relation between managerial ability and real earnings management (RealEM) 226
Table 8.8 Elasticity Indices 228

Table 8.9 The effect of managerial ability on accrual earnings management (Accrua	EM)
conditional on competition	230
Table 8.10 The effect of managerial ability on real earnings management (RealEM) conditional on competition	231
Table 8.11 The moderating role of market competition in the relation between managerial ability and earnings smoothing	233

List of Figures

Figure 2.1 Earnings management definition	21
Figure 2.2 The differences between earnings management, fraud, earnings quality, impression management and expectation management	26
Figure 3.1 The flow of the theoretical framework	67
Figure 3.2 The interaction between earnings management theories	80
Figure 4.1 The relation between the theoretical framework and earnings managemen motives	ıt 81
Figure 4.2 Earnings management motives	82
Figure 4.3 The motives of earnings management	. 120
Figure 4.4 A fourth group of earnings management motives	. 121
Figure 7.1 First assumption for the effect of market concentration on earnings management	. 173
Figure 7.2 Second assumption for the effect of market concentration on earnings	
management	. 174

List of Appendices

Appendix A Table I The effect of earnings management and managerial ability on management compensation
Appendix A Table II The effect of earnings management and managerial ability on management compensation
Appendix B Table I The effect of market concentration on earnings management. 197
Appendix B Table II The effect of market concentration on earnings management 198
Appendix B Table III The effect of market concentration on information asymmetry
Appendix C Table I The moderating role of market competition in the relation between managerial ability and accrual earnings management (AccrualEM)
Appendix C Table II The moderating role of market competition in the relation between managerial ability and real earnings management (RealEM)
Appendix C Table III The moderating role of market competition in the relation between managerial ability and accrual earnings management (AccrualEM)
Appendix C Table IV The moderating role of market competition in the relation between managerial ability and real earnings management (RealEM)
Appendix C Table V The individual effect of managerial ability on accrual earnings management (AccrualEM)
Appendix C Table VI The individual effect of managerial ability on real earnings management (RealEM)
Appendix C Table VII The individual effect of market competition on accrual earnings management (AccrualEM)
Appendix C Table VIII The individual effect of market competition on real earnings management (RealEM)

List of Abbreviations

GAAP	Generally Accepted Accounting principles
PPS	Pay-Performance Sensitivity
SFAC	Statement of Financial Accounting Concepts
ERC	Earnings Response Coefficient
MNC	Multinational Corporations
EPS	Earnings Per Share
SPV	Special Purpose Vehicle
SOX	Sarbanes-Oxley Act
KS	Kang and Sivaramakrishnan model
GMM	Generalized Method of Moments
OLS	Ordinary Least Square
R&D	Research and Development
SG&A	Selling, General, and Administrative
IE	Information Efficiency
FRREE	Fully Revealing Rational Expectations Equilibrium
EU	Expected Utility
PCAOB	Public Company Accounting Oversight Board
CEO	Chief Executive Officer
CFO	Chief Financial Officer
DMU	Decision Making Unit
DEA	Data Envelopment Analysis
IPO	Initial Public Offerings
SEO	Seasoned Equity Offerings

SEC	Securities and Exchange Commission
IFRS	International Financial Reporting Standards
UK	United Kingdom
US	United States
WG	Within Groups estimator
IV	Instrumental Variable
LTIP	Long Term Incentive Plan
NPV	Net Present Value
VIF	Variance Inflation Factor
LRT	Linear Restriction Test
FE	Fixed Effects estimator
SME	Small and Medium Enterprise
EI	Elasticity Index

Chapter 1 Introduction

1.1 Introduction

As a result of separating ownership from control in public companies, an agency problem appears between the principals (shareholders) and the agents (managers) (Holmström and Milgrom, 1987). Under this conflict in interests, managers might not reveal the truth because of the high contracting costs between managers and firms, shareholders' bounded rationalities that do not enable them to understand management actions, and the information asymmetry derived from the costly communication in the market (Milgrom and Roberts, 1992; Ronen and Yaari, 2008; Walker, 2013). In this sense, managers may exercise discretion over financial reporting that can take the shape of earnings management if performed under the umbrella of Generally Accepted Accounting principles (GAAP). Based on the previous theoretical perspectives, earnings management might be explained by three main theories that involve contracting theory, bounded rationality theory, and information asymmetry theory (Harris and Raviv, 1979; Fama, 1980; Strong and Walker, 1987; Walker, 2013). From these theories, modern research has introduced three different groups of motives to explain earnings management behaviour including contracting motives that arise from the deficiencies in the contract terms between the firm and its stakeholders, capital market motives that are related to the inefficiencies of stock markets, and third-party motives driven by external parties that influence the cost of communicating information in the market (Ronen and Yaari, 2008; Walker, 2013).

The earnings management literature has separately studied the previous groups of motives and identified a number of factors under each group. Among the contracting motives, management compensation, CEO turnover, managerial ability, corporate governance, and loans were identified (Godfrey et al., 2003; Boone et al., 2004; DeFond and Francis, 2005; Yu, 2008; Iatridis and Kadorinis, 2009; Laux and Laux, 2009; Demerjian et al., 2013b). To examine capital market motives, the literature has focused on the influence of the stock market, issuance of equity, new listing and cross-listing,

mergers and acquisitions, insider trading, management buyouts, meeting or beating a benchmark, and the effect of analysts on earnings management (Kothari, 2001; Lang et al., 2006; Efendi et al., 2007; Fan, 2007). Finally, from the external motives the literature has studied the impact of industry, industrial diversification, regulations, political environment and country-specific policies, accounting standards, tax considerations, competitors, suppliers and customers (Bagnoli and Watts, 2000; Goldman and Slezak, 2006; Jiraporn et al., 2006; Barth et al., 2008).

At the contracting level, management compensation is part of the contract structure between the firm and its managers that aims to link between firm performance and managerial payment to resolve the conflict of interests arising from the separation of ownership from control (Strong and Walker, 1987; Bergstresser and Philippon, 2006). Compensation has been frequently examined as an incentive of earnings management because managers might attempt to extract higher rewards by using earnings management to signal better performance (Healy, 1985; Cohen et al., 2008). However, compensation can be also viewed as an outcome of management behaviour as firms tend to reward or punish their managers based on the consequences of their activities (Adut et al., 2013; Dutta and Fan, 2014; Sun, 2014). The previous studies have emphasized the outcomes of earnings management activities to the firm but ignored the costs that managers might incur upon using accrual and real earnings management (Graham et al., 2005). While real earnings management is more costly to the firm because it contributes to losing cash flows and sacrificing future projects, it is also more likely to influence managers' future compensation. Understanding the impact of earnings management on managers' future benefits would assist in explaining their current behaviour e.g., the behaviour of high quality managers who select using more accrual and less real earnings management (Demerjian et al., 2013b).

At the external level, market concentration is considered as another motive of earnings management behaviour as it influences the communication of information in the market and ultimately contributes to moral hazard and adverse selection problems (Walker, 2013). While the previous studies have focused on a linear effect of market concentration on earnings management (Dalia and Park, 2009; Markarian and Santalo, 2010; Karuna et al., 2012; Datta et al., 2013), a non-linear effect is more plausible because of the concurrent opportunistic and disciplinary incentives in concentrated

markets. By the time earnings management is likely to occur because of the lack of information needed for monitoring performance, the presence of fewer competitors creates less communication pressure to manipulate earnings (Strong and Walker, 1987; Milgrom and Roberts, 1992). In this sense, the changes in the quantity and quality of information are likely to create a situation of uncertainty that explains the trade-off between earnings management activities (Hart, 1995; Arroyo, 2007). Analysing earnings management behaviour in concentrated markets would contribute to identifying an optimal level of market concentration that should mitigate the negative consequences of accrual and real earnings management.

Finally, the external and contractual motives can be combined to further understand earnings management behaviour. Within the earnings management literature, neither competition nor managerial ability has been found to have consistent effects on the earnings management behaviour of firms. Demerjian et al. (2013b), for example, examine the impact of managerial ability on earnings management and find less able managers undertake more earnings management. Conversely, Francis et al. (2008) find higher earnings management where managers have higher abilities. This conflicting evidence also occurs in the presence of competition with some studies finding a negative relation between competition and earnings management (Dalia and Park, 2009; Markarian and Santalo, 2010), while others document a positive relationship (Karuna et al., 2012; Datta et al., 2013). However, the effects of competition and managerial ability are unlikely to be independent. Therefore, understanding whether managerial ability explains the earnings management behaviour in the presence of differing levels of competition would contribute to highlighting the importance of simultaneously considering the different groups of earnings management motives when studying earnings management behaviour.

Overall, this thesis aims to examine the relations between the previously selected motives and earnings management, and subsequently studies whether the interaction between those contractual and external factors contributes to further understanding earnings management behaviour. At the contracting level, the thesis examines the influence of accrual and real earnings management on the future compensation of managers based on their different abilities. At the external level, the thesis studies the impact of market concentration on accrual and real earnings

3

management taking into consideration the changes in information quantity and quality in concentrated markets. Finally, at an external-contractual level, the thesis examines how market competition influences the amounts of accrual and real earnings managed by high quality managers. Earnings management was the underlying reason for some of the financial scandals of large companies like Enron, WorldCom, Adelphia, and Tyco in the early 2000s and ultimately resulted in the failure of those companies. These incidences have raised more concern about the role of financial reporting in reflecting the performance of firms (Giroux, 2004). Therefore, understanding the relations suggested in this thesis is expected to have implications to the regulators, policy makers, shareholders, investors, academics, and some of the gatekeepers e.g., auditors and analysts.

The remainder of this chapter is organized as follows. Section 1.2 explains the effect of earnings management on management compensation taking into consideration the variation in managerial abilities. Section 1.3 discusses the relationship between market concentration and earnings management based on the changes in information in the market. Section 1.4 presents the joint effect of managerial ability, as one of the contracting motives, and market competition, as an external factor, on earnings management. Section 1.5 summarizes the research questions. Section 1.6 explains the significance and contributions of this research. Finally, section 1.7 outlines the structure of the thesis.

1.2 Management Compensation

The first empirical chapter of this thesis (Chapter 6) examines the impact of earnings management exercised by managers at different levels of abilities on their future compensation. While the previous literature documents that managerial ability has a positive effect on accrual earnings management and a negative one on real earnings management, it explains the previous findings based on the consequences of earnings management on the firm (Demerjian et al., 2013b). From the firm perspective, accrual earnings management is considered less costly than real earnings management because it only allows for discretion over accrual accounting and subsequently unwinds in the next accounting period. In contrast, real earnings management involves actions that

have severe economic consequences to the firm's operating and stock return performance because they result in losing cash flows and ultimately sacrificing future projects (Graham et al., 2005; Merkl-Davies and Brennan, 2007; Cohen et al., 2008). Although accrual and real earnings management are likely to influence managers' welfare as well, the previous literature does not explain their costs from management perspective. Compensation is one of the major rents managers extract; hence creating a contractual motive that has been frequently examined as an incentive of earnings management behaviour (Healy, 1985; Gao and Shrieves, 2002; Cheng and Warfield, 2005; Graham et al., 2005; Bergstresser and Philippon, 2006; Burns and Kedia, 2006; Cohen et al., 2008; Carter et al., 2009; Feng et al., 2011; Oberholzer-Gee and Wulf, 2012). However, in this chapter, compensation is treated as an outcome of management rewards.

According to the principal-agent theory, the separation of ownership and control leads to specialized risk bearing and specialized decision skills in the organization (Holmström and Milgrom, 1987; Strong and Walker, 1987). This conflict in interests raises a situation of uncertainty due to the information asymmetry and moral hazard between the two parties. Therefore, an optimal structure of contracts is required that links managerial compensation to firm earnings in an optimal pay-performance structure and hence motivates managers to exert effort to enhance earnings and ultimately resolves the previous conflict (Jensen and Meckling, 1976; Harris and Raviv, 1979; Strong and Walker, 1987; Hart, 1995; Bergstresser and Philippon, 2006; Dutta and Fan, 2014).

Because of the previous situation of uncertainty, managers send signals in the reported earnings to improve their performance measures that are tied to their compensation plans (Strong and Walker, 1987; Carter et al., 2009). On the other side of the agency problem, shareholders deal with this situation of information asymmetry by screening the consequences of the information provided by managers before deciding on management compensation (Adut et al., 2013). As accrual earnings management is considered less costly from the firm perspective, it is also less likely to harm managers' future payments compared to real earnings management that is screened as a costly activity and thus might negatively influence managers' future compensation.

High quality managers are also motivated to use earnings management to signal their performance because shareholders might not be able to distinguish the quality of the reported earnings due to their bounded rationalities. Meanwhile, shareholders are less likely to detect or understand those signals because of the high contracting costs and the big knowledge gap between the more able managers and the less informed shareholders (Oberholzer-Gee and Wulf, 2012; Demerjian et al., 2013b; Walker, 2013). Consequently, while firms are more likely to pay higher incentives to managers with better abilities to compensate them for their superior efforts, a lesser impact of earnings management is expected on managers' future compensation (Hart, 1995; Stathopoulos et al., 2007; Baranchuk et al., 2011).

The previous studies that focus on management compensation as an outcome of earnings management behaviour are quite few. Empirically, Adut et al. (2013) document a negative relation between poor earnings quality and management compensation; hence showing that firms tend to punish managers for providing less informative earnings. The analytical modelling of Sun (2014) shows that when managers have more opportunities to manage earnings and shareholders are less likely to detect earnings management, the pay-performance sensitivity increases in order to eliminate managers' desire to manipulate earnings. Therefore, the positive association between earnings management and executive compensation reflects optimal contracting in this setting. In a similar way, the study of Dutta and Fan (2014) documents that when the cost of earnings management increases, e.g., under strong governance mechanisms, managers are motivated to exert more efforts which requires the firms to pay higher compensation. Thus, the study documents a positive relation between the cost of earnings management and managerial compensation.

This chapter contributes to the literature by examining the effects of both accrual and real earnings management on management compensation and explaining how these effects change when taking the different abilities of managers into consideration. To my knowledge, both questions have not been answered in the earnings management literature so far and would help in evaluating the costs of accrual and real earnings management at the management level along with the documented costs at the firm level. The answers to these questions would assist in drawing the regulators' attention to mitigating the costly activities of earnings management and emphasizing the role of compensation and managerial ability as governance mechanisms that enhance firm performance.

1.3 Market Concentration

The second empirical chapter of this thesis (Chapter 7) examines a non-linear effect of market concentration on earnings management based on two different perspectives of the revelation principle. On the one hand, the information asymmetry in concentrated markets results in the absence of common knowledge and thus allows earnings management as it becomes more difficult for the shareholders to understand or monitor management activities. On the other hand, the presence of fewer competitors decreases the chances of comparisons between firms; hence makes communication less costly and ultimately offsets the previous positive effect of market concentration on earnings management (Strong and Walker, 1987; Milgrom and Roberts, 1992; Walker, 2013). The studies in the literature tend to favour only one of the previous points of view at a time and, therefore, they present a linear relationship between market concentration and earnings management and document a positive effect in some cases and a negative one in the others (Dalia and Park, 2009; Markarian and Santalo, 2010; Karuna et al., 2012; Datta et al., 2013). However, the previous arguments suggest a contrasting effect of market concentration ranging between the opportunistic incentive and the disciplinary function and, therefore, this chapter introduces a non-linear relationship between market concentration and earnings management.

As earnings management behaviour in concentrated markets depends on information asymmetry, it may take the form of a moral hazard as shareholders face difficulties in monitoring performance or an adverse selection as managers solely have access to private information (Milgrom and Roberts, 1992; Boujelbene and Besbes, 2012). While the quantity of information is expected to decrease with market concentration, the information quality fluctuates due to the discretion exercised by managers over financial reporting (Gunny, 2010; Walker, 2013). Therefore, the changes in information raise a situation of uncertainty that is expected to explain earnings management behaviour in concentrated markets (Hart, 1995; Arroyo, 2007).

Few studies in the literature have viewed market concentration or information asymmetry in non-linear relations. Guo et al. (2015) introduce a non-linear quadratic effect of market concentration on earnings quality. They document that while earnings quality decreases at lower levels of market concentration as the firm faces the threat of losing its competitive advantage, the marginal benefit of earnings quality increases at higher levels of market concentration. In the same line, Bhattacharya et al. (2013) present a non-linear quadratic relationship between earnings quality and information asymmetry; hence considering information asymmetry as an alternative measure for the quality of disclosure (Bhattacharya et al., 2013). However, no study has examined the non-linear relation between market concentration and the different earnings management activities so far.

This chapter contributes to the literature by explaining the effect of market concentration on earnings management through a non-linear relationship based on the changes in information quantity and quality in concentrated markets. This shape of analysis has not been considered yet in the literature and would help in identifying the levels of market concentration where both accrual and real earnings management decline. As a result, it would assist regulators and policy makers in emphasizing the optimal range of market concentration by evaluating its consequences at different levels.

1.4 Managerial Ability and Market Competition

The last empirical chapter of this thesis (Chapter 8) studies how the interaction between motives at the external and contractual levels may influence earnings management behaviour. While high quality managers possess superior skills that allow them to extract better rents from their firms, they may avoid earnings management as they are able to evaluate its bad consequences on their future benefits (Francis et al., 2008; Demerjian et al., 2013b). At the same time, superior managers may operate firms in different business environments and thus different levels of competition may influence their motivations to manage earnings (Dalia and Park, 2009; Markarian and Santalo, 2010; Karuna et al., 2012; Datta et al., 2013). Therefore, this chapter examines whether market competition has an impact on the relation between managerial ability and earnings management.

Managerial ability is one of the contractual motives that might be viewed from different perspectives regarding its influence on firm performance and moral hazard between managers and shareholders (Francis et al., 2008; Demerjian et al., 2013b). According to the rent extraction hypothesis, the high contracting costs with more able managers, e.g., the need to write, negotiate, and renegotiate the contracts, result in establishing simplified general contracts that are more difficult to be monitored and thus give managers the opportunity to achieve more personal benefits (Hart, 1983; Walker, 2013). In contrast, the efficient contracting hypothesis implies that superior managers appreciate the value of maximizing firm wealth on their future benefits and, therefore, they provide a better quality of performance that contributes to aligning their interests with those of the shareholders (Hart, 1983; Walker, 2013). Therefore, the impact of managerial ability on earnings management is still debateable.

At the macro-level, market competition is one of the motives that is also expected to influence earnings management behaviour in two different directions (Dalia and Park, 2009; Markarian and Santalo, 2010; Karuna et al., 2012; Datta et al., 2013). Competition may play a disciplinary function in mitigating earnings management as it contributes to improving the information symmetry in the market and allowing better monitoring of management performance (Dalia and Park, 2009). In contrast, communicating more information allows the stakeholders to exert pressure on managers to imitate the aggressive behaviour of others in the same industry; hence aggravating adverse selection and motivating earnings management (Lieberman and Asaba, 2006; Walker, 2013). Accordingly, the influence of market competition on earnings management is also controversial.

The literature that studies managerial ability and market competition supports the previous points of view and documents contrasting results. While more able managers avoid real earnings management activities that are associated with high costs to their firms in the future, they prefer achieving personal benefits by managing accruals and providing lower earnings' quality (Francis et al., 2008; Demerjian et al., 2013b). Similarly, market competition drives more accrual and real earnings management because it encourages aggressive herding behaviour (Karuna et al., 2012; Datta et al., 2013). However, it may serve as an alternative governance mechanism as communicating more information in the market contributes to mitigating the conflict of

9

interests between management and shareholders; hence making it costlier to manipulate earnings in the same business environment (Holmstrom, 1982; Hart, 1983; Dalia and Park, 2009; Markarian and Santalo, 2010; Chhaochharia et al., 2012).

This chapter contributes to the existing literature by examining the effect of market competition on earnings managed by high quality managers. While the previous question has not been answered in the earnings management literature so far, it allows an understanding of the determinants of earnings management behaviour at two different levels by analysing the impact of industry factors on management decisions. The answer to this question would assist in drawing the regulators' attention to the simultaneous role of managerial ability and market competition as governance mechanisms that enhance firm performance and mitigate the agency problem.

1.5 Research Questions

Based on the previous arguments, the major objectives of this thesis can be summarized in the following research questions:

- 1. Do firms reward/punish their managers for using accrual and real earnings management through an increase/decrease in their future compensation?
- 2. Do firms reward/punish their high quality managers for using accrual and real earnings management in the same way they do other managers?
- 3. Do the levels of accrual and real earnings management activities change at different levels of market concentration?
- 4. Do the changes in information quantity and quality in the market explain the fluctuations in accrual and real earnings management at different levels of market concentration?
- 5. Does market competition influence the levels of accrual and real earnings managed by high quality managers?

1.6 Significance and Contributions

This thesis studies earnings management behaviour at two different levels - contracting and external - and subsequently examines whether factors at both levels interact in determining earnings management behaviour. While the previous studies in the earnings management literature have individually focused on the contracting, capital market, or external motives of earnings management at a time, they have ignored the interrelations between the motives of these different groups (Bagnoli and Watts, 2000; Kothari, 2001; Godfrey et al., 2003; Boone et al., 2004; DeFond and Francis, 2005; Goldman and Slezak, 2006; Jiraporn et al., 2006; Lang et al., 2006; Efendi et al., 2007; Fan, 2007; Barth et al., 2008; Yu, 2008; Iatridis and Kadorinis, 2009; Laux and Laux, 2009; Demerjian et al., 2013b). Examining the motives at different levels simultaneously allows for understanding their influence at the firm level while taking into consideration the strategic objectives and competitive advantages at the market level; hence extending the research benefits beyond the firm borders.

Starting at the contracting level, this thesis is the first to study the impact of earnings management on managers' welfare by examining the effect of accrual and real earnings management on management compensation in the future. While the previous studies focus on management compensation, and particularly performance-based payments, as an incentive of earnings management (Healy, 1985; Gao and Shrieves, 2002; Cheng and Warfield, 2005; Graham et al., 2005; Bergstresser and Philippon, 2006; Burns and Kedia, 2006; Cohen et al., 2008; Carter et al., 2009; Feng et al., 2011; Oberholzer-Gee and Wulf, 2012), few studies look at compensation as an outcome of management behaviour (Adut et al., 2013; Dutta and Fan, 2014; Sun, 2014). Therefore, this thesis provides new empirical evidence of firms punishing their managers for real earnings management but not for accrual earnings management. This evidence provides a better understanding of the costs of accrual and real earnings management from managers' perspectives and assists in implementing regulations that mitigate costly real earnings management activities.

The thesis also provides the first empirical evidence on the firm's response to earnings managed by managers with different levels of abilities. While the previous studies find that more able managers prefer using accrual than real earnings management (Demerjian et al., 2013b), the thesis finds that their compensation is negatively influenced by accrual earnings management and positively influenced by real earnings management. Nevertheless, real earnings management remains more costly to this type of managers as the firms generally punish this activity regardless of their

11

managers' abilities. In this sense, the thesis provides an explanation of earnings management behaviour of managers with high abilities and shows that their skills can contribute to enhancing firm performance in the future as far as they are appropriately rewarded.

Moving to the external level, the thesis examines the effect of market concentration on earnings management taking into consideration the opportunistic and disciplinary incentives in concentrated markets simultaneously. While the majority of the previous studies have focused on a linear positive or negative influence of market concentration (Dalia and Park, 2009; Markarian and Santalo, 2010; Karuna et al., 2012; Datta et al., 2013), very few ones have viewed its effect as non-linear (Guo et al., 2015). Therefore, this thesis provides the first empirical evidence of a non-linear impact of market concentration on earnings management where accrual earnings management increases at lower levels of market concentration while real earnings management dominates at higher levels. The findings contribute to identifying an optimal level of market concentration that needs to be emphasized by the regulators and policy makers to mitigate the negative consequences of accrual and real earnings management.

Furthermore, the thesis examines a non-linear effect of market concentration on information asymmetry measures taking into consideration the quantity and quality of information in the market. While the previous studies have linearly examined the effect of market concentration on information asymmetry (Ali et al., 2014; Dalia and Park, 2009; Markarian and Santalo, 2010), this thesis provides the first non-linear empirical evidence and documents that the fluctuation in the quality of information in concentrated markets is associated with the switching between accrual and real earnings management. The results assist in explaining the previous effect of market concentration quality in enlarging the optimal level of market concentration.

Finally, the thesis is the first study that examines the impact of managerial ability and market competition on earnings management and thus it contributes to the earnings management literature by introducing the combined influence of contractual and external motives. The previous studies have focused on the individual effects of managerial ability and market competition on earnings management and documented contradictory results (Holmstrom, 1982; Hart, 1983; Francis et al., 2008; Dalia and Park,

2009; Markarian and Santalo, 2010; Chhaochharia et al., 2012; Karuna et al., 2012; Datta et al., 2013; Demerjian et al., 2013b). However, this thesis provides new empirical evidence on the simultaneous influence of both factors on earnings management and shows that the behaviour of high quality managers changes under the pressure of market competition through using more accrual and less real earnings management. This evidence highlights the importance of understanding the consequences of the interaction between the different earnings management motives on firm performance.

1.7 Thesis Structure

This thesis is divided into nine chapters. Chapter 1 provides an introduction about this thesis and summarizes the motivation and main questions for each empirical chapter. In addition, it explains the significance of this thesis and the expected contributions of its empirical results. Finally, the chapter presents the structure that will be followed for the remaining parts of the thesis.

Chapter 2 explains the definitions of earnings management and its specific characteristics compared to fraud, earnings quality, impression management, and expectation management. It presents the different activities of earnings management and how they might be traded off in addition to the models of measuring accrual and real earnings management. Finally, the chapter discusses some special issues in measuring earnings management including the combined models, the specific measures in the financial sector, and the qualitative measures.

Chapter 3 provides a brief description of the firm and agency theories. It discusses the revelation principle and the relevant theories to explain its conditions including contracting theory, bounded rationality theory, and information asymmetry theory. Finally, it presents the three main theoretical approaches of earnings management that include the costly contracting approach, the decision-making approach, and the legal-political approach.

Chapter 4 provides a comparison between earnings management and truth telling. It explains the contracting motives of earnings management that include management compensation, CEO turnover, managerial ability, corporate governance, and loans. It also presents the capital market motives of earnings management that

13

consist of the effect of the stock market, issuance of equity, new listing and cross-listing, mergers and acquisitions, insider trading, management buyouts, meeting or beating a benchmark, and the role of analysts. Finally, the chapter discusses the third-party motives of earnings management including industry and industrial diversification, regulations, political environment and country-specific policies, accounting standards, tax considerations, competitors, suppliers and customers.

Chapter 5 discusses the data and methodology of this thesis. It explains the overall sample construction of the three empirical chapters and the sources used to collect data. It also presents the different models that will be used in the following empirical chapters to measure accrual and real earnings management. Finally, the chapter discusses the general basis for selecting the methodology of the thesis.

Chapter 6 represents the first empirical chapter of this thesis and examines the effects of accrual and real earnings management on management compensation in the future. It also studies whether managerial ability influences the previous effects of earnings management on compensation and thus explains the documented behaviour of high quality managers using accrual rather than real earnings management. Data, variable definitions, methodology and statistical analysis are discussed in detail in the chapter.

Chapter 7 is the second empirical chapter of this thesis and introduces a nonlinear effect of market concentration on earnings management based on two contrasting points of view in the theory. As the previous effect is likely to be determined by the asymmetry of information in the market, the chapter examines the changes in the quantity and quality of information in concentrated markets to explain that effect. Data, variable definitions, methodology and statistical analysis are discussed in detail in the chapter.

Chapter 8 is devoted to the last empirical chapter of this thesis and examines accrual and real earnings managed by high quality managers at different levels of market competition. In this sense, this chapter studies the mutual influence of contractual and external motives of earnings management compared to the majority of the studies in this area that emphasize the individual effects of earnings management motives at a time. Data, variable definitions, methodology and statistical analysis are discussed in detail in the chapter. Finally, chapter 9 concludes this thesis by providing a summary of the objectives and results of the three empirical chapters. It explains the implications of the thesis to the regulators, policy makers, shareholders, investors, gatekeepers, and academics. Furthermore, the chapter presents the limitations of this research and suggests some recommendations for future studies to take into consideration based on the results of this thesis.

Chapter 2 Definitions, Activities, and Measurement of Earnings Management

2.1 Introduction

This chapter starts by providing a summary of the definitions of earnings management in the literature and accordingly it suggests a four-stage process for a comprehensive definition of earnings management that includes its characteristics, conditions, activities, and targets. Following this approach, the chapter attempts to distinguish between earnings management and other concepts like fraud, earnings quality, impression management, and expectation management.

The chapter also explains the different methods of earnings management and how managers trade them off based on their different needs. It also discusses in detail the models to calculate the different activities of earnings management - particularly focuses on the measurement of accrual and real earnings management and evaluates each of the models introduced in this area. Finally, the chapter throws some light on special issues in measuring earnings management including the models that combine between more than one manipulation activity and the measurement of earnings management in the financial sector and qualitative research.

The remainder of this chapter is organized as follows. Section 2.2 explains the definitions of earnings management and its specific characteristics compared to fraud, earnings quality, impression management, and expectation management. Section 2.3 presents the different activities of earnings management and how they might be traded off. Section 2.4 explains the models of measuring accrual earnings management. Section 2.5 explains the models for measuring real earnings management. Section 2.6 discusses some special issues in measuring earnings management including the combined models, the specific measures in the financial sector, and the qualitative measures. Finally, section 2.7 provides a conclusion to the chapter.

2.2 Definitions

In this section I discuss some of the most common definitions of earnings management in the literature. Afterwards, I introduce my own definition of earnings management taking into consideration some of the deficiencies in the previous definitions of the literature. Because earnings management has been frequently confused with other concepts like fraud, earnings quality, impression management, and expectation management, I also discuss each of these definitions separately and distinguish them from earnings management.

2.2.1 Earnings Management Definition

No single definition exists for earnings management in the literature. Researchers have provided different explanations that mainly define earnings management as the manipulation of financial reporting to achieve specific targets. I present here some of the most common definitions of earnings management in a chronological order. Schipper (1989, P.92) focuses in her definition on the manipulation of external reporting to achieve private benefits like improving managers' compensation whereby,

"Earnings management means disclosure management in the sense of a purposeful intervention in the external financial reporting process, with the intent of obtaining some private gain as opposed to, say, merely facilitating the neutral operation of the process".

Healy and Wahlen (1999, P.368) also focus on changing financial reporting to mislead the stakeholders and achieve contractual benefits where,

"Earnings management occurs when managers use judgment in financial reporting and in structuring transactions to alter financial reports to either mislead some stakeholders about the underlying economic performance of the company or to influence contractual outcomes that depend on reported accounting numbers".

Mulford and Comiskey (2002, P.3) emphasize management discretion to meet earnings targets which may be set by internal or external parties. In this sense, their definition of earnings management is as follows: "The active manipulation of earnings toward a predetermined target, which may be set by management, a forecast made by analysts, or an amount that is consistent with a smoother, more sustainable earnings stream".

Phillips et al. (2003, P.493) provide a brief definition of earnings management that emphasizes the manipulation of accounting choices and operating cash flows as follows:

"Managerial discretion over accounting choices and operating cash flows".

Giroux (2003, P.280) also focuses on management discretion to achieve specific income targets. Therefore, he defines earnings management as:

"The operating and discretionary accounting methods to adjust earnings to a desired outcome".

In his second definition, Giroux (2004, P.2) emphasizes the incentives of management to manipulate earnings under different conditions and thus defines earnings management as follows:

"The planning and control of the accounting and reporting system to meet the personal objectives of management".

Ronen and Yaari (2008, P.5) provide a general definition for earnings management that focuses on the target of management to influence the interpretation of its reported earnings, whereby earnings management involves:

"Deliberate actions to influence reported earnings and their interpretation".

However, Ronen and Yaari (2008, P.27) later provide a comprehensive definition for earnings management that distinguishes between the two main activities to manipulate earnings – real vs. accrual - and shows that such activities are not necessarily bad all the times whereby,

"Earnings management is a collection of managerial decisions that result in not reporting the true short-term, value-maximizing earnings as known to management. Earnings management can be beneficial: it signals long-term value; pernicious: it conceals short- or long-term value; neutral: it reveals the short-term true performance. The managed earnings result from taking production/investment actions before earnings are realized or making accounting choices that affect the earnings numbers and their interpretation after the true earnings are realized".
Finally, Walker (2013, P.446) also emphasizes both the accrual and real earnings management in the following definition:

"The use of managerial discretion over (within GAAP) accounting choices, earnings reporting choices, and real economic decisions to influence how underlying economic events are reflected in one or more measures of earnings".

While some of the previous definitions are brief, others might better explain the meaning of earnings management. Overall, they emphasize how managers manipulate their earnings. Among the different methods of manipulation, the literature emphasizes the treatment of accruals by using different accounting principles (Baber et al., 2011; Walker, 2013), real economic decisions that influence cash flows (Graham et al., 2005; Roychowdhury, 2006), smoothing earnings to decrease their volatility over time (Coffee, 2003; Graham et al., 2005; Walker, 2013), shifting the classification of some items in the financial statements (McVay, 2006; Athanasakou et al., 2009; Barua et al., 2010; Walker, 2013), and other advanced methods like using derivatives and special purpose entities (Giroux, 2004; Petrovits, 2006; Feng et al., 2009). I will discuss these activities in detail in the next section of this chapter.

The previous definitions have also emphasized the motives that drive managers to manage earnings. There are different incentives of earnings management at the contracting level like compensation (Healy, 1985; Jensen and Murphy, 1990; Dechow and Huson, 1994; Laux and Laux, 2009), capital market level like improving the stock prices (Lev, 1989; Ramakrishnan and Thomas, 1998; Kothari, 2001; Gelb and Zarowin, 2002), and the external level like the influence of regulators and competitors (Jones, 1991; Cahan, 1992; Cahan et al., 1997; Kallunki and Martikainen, 1999; Bagnoli and Watts, 2000; Goldman and Slezak, 2006). Most of the empirical literature has emphasized examining the impact of these motives on earnings management as I will discuss in detail in chapter 4.

However, the previous definitions have ignored the conditions that allow management to manipulate earnings like managers' incomplete contracts that give them more flexibility in their decisions, stakeholders' limited capabilities to understand management sophisticated decisions, and the selective communication of information by management to stakeholders (Walker, 2013). Therefore, I define earnings

19

management as the within GAAP management discretion over external financial reporting by abusing some contracting deficiencies, stakeholders' bounded rationalities, and information asymmetry in the market, through some economic decisions, a change in the accounting treatment, or other sophisticated methods. The purpose of management is to present earnings in a way different (up or down) from what is known to them to achieve private benefits while misleading the stakeholders; although such discretion may not always be harmful to them. Figure 2.1 illustrates the previous definition in four stages.

In spite of the previous efforts to explain the definition of earnings management, it has been always confused with other concepts like fraud (Dechow and Skinner, 2000), earnings quality (Dechow et al., 2010), impression management (Merkl-Davies and Brennan, 2007), and expectation management (Das et al., 2011). Although my previous definition distinguishes earnings management from the previous activities in the 'What?' part of Figure 2.1, I explain all these concepts in detail in the following sections and show how they differ from the concept of earnings management.

2.2.2 The Difference between Earnings Management and Fraud

Based on the previous definition, earnings management may take place through the aggressive or conservative accounting within the Generally Accepted Accounting Principles (GAAP) that happens usually at the end of the fiscal year, e.g., decreasing or increasing the estimation of some provisions. It may also happen through some aggressive or conservative economic decisions that managers may take anytime during the fiscal year to influence cash flows, e.g., accelerating or delaying sales. Earnings management may result in increasing or decreasing the reported earnings. It can be considered pernicious if it contributes to minimizing firm value but beneficial if it allows signaling more information about the firm in the future. Nevertheless, earnings management does not involve any violation of the accounting principles (Dechow and Skinner, 2000; Ronen and Yaari, 2008).



Figure 2.1 Earnings management definition

Stakeholders' benefits

In contrast, fraud involves a violation of GAAP, e.g., ignoring some required provisions or recording fictitious or unrealized sales. Managers may commit fraud within or after the fiscal year to increase or decrease the reported earnings. Fraud generally follows aggressive earnings management behaviour; hence it is considered extremely aggressive compared to earnings management (Dechow and Skinner, 2000; Ronen and Yaari, 2008; Walker, 2013). In this sense, fraud is always harmful to the firm and its stakeholders.

2.2.3 The Difference between Earnings Management and Earnings Quality

According to the Statement of Financial Accounting Concepts No.1 (SFAC 1), earnings quality is related to providing information about firm performance that is relevant for decision making (Dechow et al., 2010). Therefore, earnings quality is different from earnings management because it measures the overall relevance of earnings for stakeholders to make decisions, whether that quality was related to management discretion or not. In contrast, earnings management focuses only on the discretionary aspect of financial reporting and does not consider factors outside management control, e.g., errors.

While there are specific models that measure earnings management which will be discussed in detail later in this chapter, there are three groups of proxies that measure earnings quality. The first group includes some external measures like the earnings restatements reported by the SEC. The second group represents the measures of earnings properties like earnings persistence, errors in the bad debt provision, and the mapping of accruals into cash flows according to the Dechow and Dichev (2002) measure that will be explained in detail later in section 2.4.9. Finally, the third group includes the measures of investors' responses to earnings like the earnings response coefficient (ERC) (Dechow et al., 2010; Demerjian et al., 2013a). High earnings quality is associated with low earnings restatements, high earnings persistence, lower probability of errors in the bad debt provision, better mapping of accruals into cash flows, and high investors' response to earnings announcements (Adut et al., 2013). However, there is no single appropriate measure for earnings quality as each proxy measures a specific aspect of the variable and supports different type of decision (Dechow et al., 2010).

2.2.4 The Difference between Earnings Management and Impression Management

Impression management is a group of strategies used by firms to opportunistically manage discretionary narrative disclosures and thus influence stakeholders' perceptions and decisions based on the asymmetry of information between the two parties (Merkl-Davies and Brennan, 2007). In line with this explanation, Clatworthy and Jones (2001, P.311) define impression management as a method "*To control and manipulate the impression conveyed to users of accounting information*". Similarly, Hooghiemstra (2000, P.60) defines impression management as "*A field of study within social psychology studying how individuals present themselves to others to be perceived favourabley*". In this sense, while earnings management emphasizes management discretion over the numerical part of the earnings component of the financial statements, impression management focuses on manipulating the narrative part in the financial reports.

There are seven strategies of impression management. The first strategy is manipulating the ease of reading in order to make the narratives more difficult to read and ultimately not easily understood by the stakeholders. Second, management may use rhetorical manipulation to make the language of the narratives more convincing to the readers. The first two strategies are mainly used by management to hide bad news. Third, thematic manipulation can be used to emphasize the good information in the narratives. The fourth strategy involves the use of visual and structural effects by manipulating the presentation like the order of the presented information. The fifth strategy is the use of performance comparisons where management benchmarks its performance to cases that relatively emphasize its better current achievements. Sixth, the choice of earnings numbers might allow management to selectively emphasize specific numbers in the financial reports that reflect its better performance. Finally, firms may use the attribution of organizational outcomes strategy to emphasize that all successes have resulted from the superior current management performance. The last five strategies aim basically to signal good news about management performance to the stakeholders (Merkl-Davies and Brennan, 2007).

Like earnings management, impression management involves managers' judgment in financial reporting to hide or signal information about the underlying economic performance of the firm that may ultimately mislead the readers of the financial reports and, therefore, result in short-term capital misallocation. Both activities are based on the assumption that a weak form of market efficiency exists resulting from the bounded rationalities of the different stakeholders. However, while earnings management activities involve the manipulation of accruals or real economic transactions that may influence the future cash flows, impression management relies only on the presentation of the narrative disclosures of the financial reports. In this sense, impression management can be considered as a more indirect way to influence stakeholders' decision making than earnings management. Nevertheless, it may involve high risk especially with the increase in the amount of narrative disclosures that occupy more than half of firms' annual reports and that are difficult to be monitored and regulated (Merkl-Davies and Brennan, 2007).

2.2.5 The Difference between Earnings Management and Expectation Management

Expectation management is often used to guide the perceptions of the different stakeholders whether explicitly through firm announcements to influence analysts' forecasts or implicitly by staying silent without advising the stakeholders about the truth of the underlying transactions. One common way of guiding expectations is the use of unaudited pro-forma earnings, particularly in the US where pro-forma statements are commonly announced. In this sense, firms may exclude or include specific items from these statements that may influence the forecasts of analysts about their overall earnings numbers (Walker, 2013).

According to the previous explanation, both earnings management and expectation management are used to meet earnings targets and avoid losses. However, while earnings management involves the manipulation of the reported earnings, expectation management emphasizes communicating specific information that

24

indirectly influences stakeholders' decisions. Das et al. (2011) find that the relationship between expectation management and earnings management is complementary. Managers even make more use of expectation management when earnings management activities involve higher costs; e.g., in the fourth quarter of the financial year when auditors' scrutiny increases. In addition, the market tends to discount the meet or beat premium when using both methods but the overall stock price generally benefits when a firm meets its earnings benchmarks (Das et al., 2011).

In summary, earnings management is the within GAAP manipulation of earnings for managers to achieve specific targets. However, it is essential to distinguish earnings management from other concepts like fraud, earnings quality, impression management, and expectation management. The main differences between these activities can be evidenced in their level of aggressiveness, violating the accounting principles, management control, and the numerical vs. narrative focus in the financial statements. Figure 2.2 summarizes the main differences between the five activities. As far as this thesis is concerned with the within GAAP discretion over financial reporting, I discuss earnings management activities in detail in the next section.

Difference	Earnings management	Fraud	Earnings quality	Impression management	Expectation management
Aggressive	Sometimes	Extremely	Sometimes	Sometimes	Sometimes
GAAP violation	No	Yes	Sometimes	No	No
Management discretion	Yes	Yes	Sometimes	Yes	Yes
Focus	Numerical disclosure (direct)	Numerical disclosure (direct)	Numerical disclosure (direct)	Narrative disclosure (Indirect)	Announcements (Indirect)

Figure 2.2 The differences between earnings management, fraud, earnings quality, impression management and expectation management

2.3 Earnings Management Activities

Earnings management mainly takes place by selecting a specific accounting treatment of certain transactions or by taking some economic decisions that may influence the cash flows, investments, or production of the firm. Both actions aim to improve earnings presented in the financial statements and, ultimately, stakeholders' interpretations of the accounting numbers. Other strategies for the within GAAP earnings management may involve earnings smoothing, classification shifting, and some advanced methods. However, while earnings management may take several forms, revenue recognition seems to be the most common area of manipulation (Ronen and Yaari, 2008). I explain each of the earnings management activities in detail in this section.

2.3.1 Accrual and Real Earnings Management

Until recently, the main focus of the literature was on accrual earnings management which involves the within-GAAP manipulation of accruals through the discretionary choices of accrual accounting, e.g., depreciation rates, inventory valuation methods, and bad debt calculation. Therefore, accrual earnings management does not influence the firm's underlying economics but involves the change in the accounting presentation of these economics. As accruals play a role in determining earnings, they ultimately influence the distribution of wealth between the stakeholders. In addition, management discretion over accruals allows conveying information about firms' future cash flows and, therefore, may result in decreasing the information asymmetry between management and shareholders. While accrual earnings management may influence the claims to cash flows, it does not influence firms' value which is basically determined by these cash flows (Walker, 2013). However, the reversal of accruals in the next accounting period creates a limitation on the subsequent use of accrual earnings management. The high magnitude of the previously managed accruals and the fast accruals' reversal impose more restrictions on the use of accrual earnings management in the same direction in the next accounting period (Baber et al., 2011).

In contrast, real earnings management has started to receive more attention since 2005, after the studies of Graham et al. (2005) and Roychowdhury (2006) which highlighted the importance of understanding such activity in evaluating management behaviour. Roychowdhury (2006, p. 336) defines real earnings management as,

"Management actions that deviate from normal business practices, undertaken with the primary objective of meeting certain earnings thresholds".

Real earnings management involves economic decisions like accelerating sales through more lenient credit terms and higher discounts to the clients (Cohen and Zarowin, 2010), timing the sale of long-term assets and investments in periods of low earnings (Bartov, 1993), overproduction to decrease the fixed cost per unit and ultimately the unit cost and the cost of sales (Chi et al., 2011), and manipulating discretionary expenses like research and development, advertising, selling and administrative expenses (Cheng, 2004; Osma, 2008). In this sense, real earnings management ultimately changes the free cash flows of the firm as it involves sacrificing some value maximizing activities and thus negatively influences its operating performance and stock returns in the future (Graham et al., 2005; Cohen et al., 2008; Kothari et al., 2016). However, Gunny (2010) documents a positive effect of real earnings management on future performance when it is used by a firm to meet some benchmarks that ultimately improves its reputation in the market. Because of the strong relation between accruals and free cash flows that are concurrently determined, both have to be considered in modelling earnings management to avoid endogeneity problems (Zang, 2012; Walker, 2013).

If the earnings management strategy attempts to increase earnings above the true level, it is referred to as a maximization strategy. On the other hand, if it attempts to reduce earnings below the true earnings, it is called a minimization strategy. Overall, because a firm's resources are limited, earnings maximization and minimization are likely to take a loop form. In other words, the current maximization will drive future minimization and vice versa. As companies tend mostly to maximize, rather than minimize, their earnings to improve their images to the stakeholders, maximization strategies have received more attention in the literature. Maximizing current earnings arises from consuming old earnings reserves or reducing those expected in the future. Overall, managers attempt to maximize their earnings to improve the market valuation of their firms' stocks, achieve better rewards, be able to renegotiate their contracts, and get a better shape compared to competitors (Healy, 1985; Demski and Frimor, 1999; Fischer and Verrecchia, 2004). In contrast, earnings minimization implies conservative reporting and hence it has not received lots of attention in the earnings management literature. Minimization results in shifting current earnings to future periods. One of these strategies is the "cookie jar reserves" when a firm decreases its current earnings to report higher numbers in the future when performance deteriorates (Giroux, 2004). The strategy of "taking a bath" is an extreme form of earnings minimization to report extremely low earnings which mostly happens when management does not expect any bonuses in the current period or attempt to meet high earnings' targets in the future (Scott, 1997; Levitt, 1998; Ronen and Yaari, 2008).

2.3.2 Earnings Smoothing

In addition to the previously mentioned two major strategies that management can use to manipulate earnings, other activities can be implemented to signal or hide information. Earnings smoothing is another way of management discretion which gives a chance for managers to decrease the volatility of their earnings and, therefore, influences the stakeholders' risk perceptions (Coffee, 2003; Walker, 2013). According to Graham et al. (2005), earnings smoothing is likely to be appealing to managers because it basically allows presenting the business in a more stable shape and hence it is perceived to be less risky by stakeholders.

According to the nature of the strategy, earnings smoothing can be classified into real and artificial types (Ronen and Yaari, 2008). Real earnings smoothing involves economic decisions related to firm's operating or investing activities that aim to decrease the volatility of earnings. Such activities are more complex and difficult to be detected; and thus may severely influence firm value in the future (Ewert and Wagenhofer, 2005). On the other hand, artificial earnings smoothing involves the accounting discretion to over- or understate firm's earnings; and hence may build on the economic decisions of the real smoothing strategies.

According to the consequences of the strategy, earnings smoothing can be classified into beneficial, neutral, and pernicious (Ronen and Yaari, 2008). Earnings smoothing might be beneficial when it improves the informativeness of earnings and allows the prediction of future earnings. This occurs when managers exert their best efforts and try to reduce the current high earnings numbers if they expect a decrease in future earnings, and vice versa. In this sense, earnings smoothing makes future earnings more predictable to shareholders especially under uncontrollable conditions, e.g., changes in accounting standards. In contrast, pernicious smoothing occurs when management opportunistically reports different earnings from those that are already known to them in order to hide bad current news. Such activity results in worse earnings in the future when firm performance does not improve; and thus takes the shape of current earnings maximization and subsequent earnings minimization (Dalia and Park, 2009). In this case, the opportunistic behaviour results in lower information quality and ultimately more volatility in stock returns and share prices (Markarian and Gill-de-Albornoz, 2012). Finally, neutral smoothing occurs when it does not influence the firms'

29

cash flows, and the market is well informed and rational to perceive it (Goel and Thakor, 2003).

As earnings volatility has a negative effect on stock price, firms tend to smooth earnings to avoid a decrease in their prices. However, although beneficial smoothing changes the valuation of the firms, it results in mitigating some of the misrepresentations in their share prices (Sankar and Subramanyam, 2001). Neutral smoothing, on the other hand, has no influence on share prices because the market already knows about it and incorporates it when discounting the firms' values (Goel and Thakor, 2003). Pernicious smoothing may positively influence the stock prices if the market is irrational and less informed. It may also succeed in rational markets under specific conditions, e.g., if it is accompanied with some other good news (Yaari, 2005).

The compensation contracts between shareholders and managers also influence earnings smoothing behaviour. Managers play the "timing game" in order to smooth their abilities to consume over time by shifting earnings from the periods when they receive high pay-offs to the periods when they expect lower ones, so that they guarantee receiving better compensation in bad times (Demski, 1998; Oyer, 1998). The duration of the contract may also motivate such behaviour as earnings smoothing is likely to occur in long-term employment contracts because they allow inter-temporal risk allocation (Allen, 1985). Managers may also "pull in" earnings to the current year if the firm performance is poor in order to guarantee the renewal of their contracts and avoid being terminated (Fudenberg and Tirole, 1995).

Finally, external parties may influence earnings smoothing behaviour. Firms may smooth earnings under the pressure of suppliers, customers, or competitors as the demand of one firm influences the earnings of the others (Graham et al., 2005). Regulators may interfere to mitigate earnings smoothing. However, for their role to succeed, they have to essentially focus on curbing pernicious earnings smoothing rather than eliminating the beneficial one, and hence contribute to improving shareholders' value.

2.3.3 Income Shifting

Firms may sometimes use classification shifting instead of accrual or real earnings management when managers are more concerned about net income rather than earnings' numbers. As a result, they transfer items from the operating activities to other activities and vice versa in order to improve the performance of the core business which is considered more important for the stakeholders, e.g., transferring small operating losses into the exceptional items (McVay, 2006; Athanasakou et al., 2009; Barua et al., 2010; Walker, 2013). Opposite to accrual earnings management, classification shifting neither changes the net earnings number nor reverses over time, and therefore it is a less costly activity than accrual earnings management (McVay, 2006). However, Athanasakou et al. (2009) document that the market rewards firms that achieve their targets by classification shifting less than those that genuinely perform the job.

In another form, income shifting may take place in multinational corporations (MNCs) by shifting income between the parent and its subsidiaries to avoid high taxes. In this setting, companies shift more income to their subsidiaries in low-tax countries and more debts to those in countries with higher statutory taxes to get better deductions on their interest depending on the tax enforcement level in each country (Beuselinck et al., 2015). Multidivisional firms may also manipulate the transfer prices between the divisions to avoid taxes and ultimately maximize net income (Martini, 2015).

2.3.4 Advanced Techniques

Recently, managers have started to use new complicated methods to manage earnings that are difficult for an average shareholder to understand. One of these means is the repurchase of firm stocks from the market which helps in improving the firm stock price as it indicates higher demand. At the same time, it results in an improved earnings per share (EPS) ratio because the profits of the year will be distributed over a lower number of outstanding shares, and hence it assists the firm in meeting analysts' forecasts (Hribar et al., 2006). In the same line, firms may also issue convertible bonds, rather than new shares, because they are excluded when calculating the diluted EPS and thus enable them to achieve higher EPS and reflect a better performance (O'Brien, 2005).

Another method is the use of financial derivatives like options, interest rate swaps, forwards and future contracts. Such instruments are structured on other transactions and therefore they are associated with sophisticated valuation and high risk. Derivatives can be used for speculation or for hedging to counterbalance the accompanying risk of other transactions. The gains and losses on some of these instruments are reflected in the income statement e.g., interest rate swaps for fair value hedge, while others are not e.g., the effective portion of interest rate swaps for cash flow hedge. Therefore, earnings management may take place when managers shift the related gains and losses between the two previous categories (Giroux, 2004).

A more complicated technique to manage earnings is the special purpose vehicles (SPVs) like the ones which were used by Enron before the 2001 scandal. Firms may establish an SPV as a separate legal entity to execute specific transactions like sales, loans, receivables' factoring, and transfer of assets. Therefore, an SPV allows firms to manage earnings especially that it stays off the balance sheet according to the US regulations as far as a third party owns 10% of its equity. However, strong corporate governance may mitigate the misuse of SPVs (Giroux, 2004; Feng et al., 2009). Similarly, firms may use corporate-sponsored foundations for their charitable contributions which allow for discretion over the amount of contribution from their earnings and flexibility in reporting them as an off-balance sheet reserve (Petrovits, 2006).

2.3.5 The Trade-off between Earnings Management Activities

Companies trade off accrual and real earnings management activities according to their relative costs, and hence accrual earnings management and real earnings management are commonly perceived as substitutes (Cohen et al., 2008; Bartov and Cohen, 2009; Baber et al., 2011; Zang, 2012). Cohen et al. (2008) document that firms switched from discretionary accruals to real earnings management after the release of Sarbanes-Oxley Act (SOX) in 2002. In this sense, the costs of accrual earnings management are different from those of real earnings management and thus managers tend to implement more real strategies when the costs of accrual activities are higher and vice versa. The costs of accrual earnings management include stakeholder scrutiny, audit quality, and

accounting flexibility (Zang, 2012). Analysts also influence accrual earnings management behaviour and encourage switching to real activities once their expectations fall (Bartov and Cohen, 2009). On the other hand, the costs of real earnings manipulation contain the competitive status in the industry, financial health, institutional ownership, and the tax consequences of manipulation (Zang, 2012). The detailed measurement of the two activities is explained in detail in the next section.

Among the aforementioned costs, stakeholder scrutiny has been emphasized in the various corporate governance codes that influence earnings management behaviour. The release of the Sarbanes-Oxley Act (SOX) in 2002 is considered among the most significant in the United States (US) and in many other countries. Generally, firms tend to replace accrual earnings management by real activities in periods following huge corporate failures as they become subject to detection by regulators and auditors due to the strict governance codes that aim to prevent the recurrence of such events and regain the public confidence in the market (Cohen et al., 2008; Bartov and Cohen, 2009; Nordberg and McNulty, 2013). While real earnings management activities remain less detectable because they need specialized knowledge that is not available for most of the stakeholders, they are considered more costly as they lead to sacrificing the long term objectives of the firm and hence negatively influence its future performance (Graham et al., 2005; Cohen et al., 2008; Kothari et al., 2016). Therefore, the increased scrutiny by regulators drives managers to utilize different forms of earnings management activities rather than entirely eliminating such behaviour (Badertscher, 2011).

The trade-off between accrual and real earnings management can be viewed as a sequential relationship. Managers focus on real earnings management activities throughout the accounting period because they have the flexibility of using the real earnings management activities that are based on their own decisions (Cohen and Zarowin, 2010). However, they tend to use more accrual earnings management after the period ends and before issuing the financial statements to achieve their targets as they miss the chance to manipulate the real activities to any further extent (Zang, 2012). Firms also utilize more accrual activities in periods when their stocks are overvalued in order to maintain the price for longer, but they may also switch to real earnings management based on the intensity and length of the overvaluation period (Badertscher, 2011). The trade-off is also noted in periods of seasonal equity offerings when the

utilization of one activity is positively correlated with the costs of the other (Cohen and Zarowin, 2010).

Managers may also trade-off accrual and real earnings management activities with other practices depending on their costs (Das et al., 2011; Haw et al., 2011). They can utilize expectation management by minimizing analysts' perceptions about firm performance to convince them that their expectations have been met (Bartov and Cohen, 2009). The nature of the relation between earnings management and expectation management depends on the constraints of using each activity (Das et al., 2011). Similarly, earnings management can be substituted with classification shifting in the financial statements. This technique is basically used to improve earnings from the major operations but does not tend to change the bottom line of the firm's earnings. Therefore, classification shifting is considered to be less costly than the other forms of earnings management activities, and can be used as a substitute when managers face strict constraints on managing their earnings (Haw et al., 2011).

Overall, there are different methods for firms to manage earnings that vary in their nature, timing, aims, costs, and consequences. Managers do not adhere to a specific technique and usually switch to the one which is more difficult to be detected, relatively less costly, and attains the short term goals of the firm. Although all these methods have to be taken into consideration to understand earnings management behaviour, the literature has mostly emphasized accrual and real earnings management. Therefore, I discuss the measurement of these two activities in detail in the next sections.

2.4 Accrual Earnings Management Models

The models of aggregate discretionary accruals have dominated the earnings management literature rather than those that measure a single component of accrual accounting choices, e.g., depreciation and inventory valuation, because aggregate models are more comprehensive in measuring accruals' manipulation. Although the Jones Model and the Modified Jones Model have been the most prominent in this area, other models were also introduced to solve for the numerous modelling problems. In this section, I discuss the different models that were introduced in the literature and explain their main advantages and disadvantages then provide an overall evaluation of all the attempts of measuring accrual earnings management.

2.4.1 The Ronen and Sadan Model

Ronen and Sadan (1981) only incorporate sales in their regression to calculate discretionary accruals compared to the Jones model that uses both sales and PP&E. Their model also takes classification shifting into consideration as another earnings management strategy. They measure how a company may use real, accrual, or classification smoothing to reduce the fluctuations in its reported earnings. To achieve this, they run their regressions at different levels of the income statement to calculate abnormal sales, abnormal extraordinary expenses, and abnormal ordinary income. Ronen and Sadan run the following regressions:

$$L_t = a_{0L} + a_{1L}t + u_t \tag{2.1}$$

$$OP_t = a_{0P} + a_{1P}t + a_{2P}u_t + s_t \tag{2.2}$$

$$X_{2t} = f_0 + f_{1t}t + f_{2t}u_t + q_t \tag{2.3}$$

Where L is the sales number from the income statement, OP is the operating income from the income statement, X_2 is the extraordinary expenses from the income statement, t is the time impact (i.e. year), u is the abnormal sales, s is the abnormal ordinary income, and q is the abnormal extraordinary expenses. By including the abnormal sales in the other two regressions, the model ensures controlling for any exceptional performance (e.g., a non-linear relation between performance and normal accruals). However, the model ignores the relations between working capital items and its regressands.

2.4.2 The Healy Model

Similar to the Jones model, Healy (1985) considers discretionary accruals as the difference between total accruals and normal (nondiscretionary) accruals. However, the model uses the average of total accruals over the last five years as a measure of normal accruals to reflect the long-term average. Therefore, Healy's model considers earnings management as a systematic event that repeats over time. Then, the model compares between the group of observations where earnings management is expected to be managed upwards and that where it is expected to be managed downwards based on the accounting flexibility available to management. The main contribution of Healy is defining normal accruals in the following way:

$$NDA_t = \frac{1}{n} \sum_{i=t-n}^{t} \frac{TA_i}{A_{i-1}}$$

$$(2.4)$$

Where *NDA* is the non-discretionary accruals, *TA* is the total accruals, A_{i-1} represents the lagged assets, *n* is the number of years over which the long-term accruals are averaged which has been set as 5 years by Healy (1985). Based on the previous equation, discretionary accruals are calculated as:

$$DA = TA - NDA \tag{2.5}$$

Where DA is the discretionary accruals and TA is the total accruals calculated as the difference between reported earnings and operating cash flows. However, as accruals reverse over time, the average normal accruals might be zero, and in this case discretionary accruals will be equal to total accruals. Therefore, some of the normal accruals might be mistakenly classified as discretionary in some years (Dechow et al., 1995).

2.4.3 The DeAngelo Model

DeAngelo (1986, 1988) studies consider last year's total accruals as a measure of normal accruals and hence any change in accruals this year compared to the previous year is considered discretionary. The model calculates non-discretionary accruals using the following equation:

$$NDA_t = \frac{TA_{t-1}}{A_{t-1}} \tag{2.6}$$

Where *NDA* represents the non-discretionary accruals, TA_{t-1} is the lagged total accruals, A_{i-1} represents the lagged assets. Similar to Healy model, DeAngelo calculates discretionary accruals according to the following equation:

$$DA = TA - NDA \tag{2.7}$$

Where DA is the discretionary accruals and TA is the total accruals calculated as the difference between reported earnings and operating cash flows.

Although there is high association between the DeAngelo model and the Healy model, the DeAngelo model reduces the serial correlation that might exist in Healy's model because it only takes the last year's total accruals rather than the last five years' total accruals into consideration when calculating normal accruals. Nevertheless, DeAngelo's assumption is still not valid all the time (Dechow et al., 1995).

2.4.4 The Industry-Based Model

Dechow and Sloan (1991) introduce an industry-based model to measure normal accruals because firms within the same industry face the same motives for discretionary behaviour. Therefore, the model uses the median of total accruals for all firms in a specific industry sector and in a specific year as a measure of non-discretionary accruals according to the following equation:

$$NDA_t = \gamma_1 + \gamma_2 median(TA_t)$$
(2.8)

Where *NDA* represents the non-discretionary accruals and $median(TA_t)$ is the median value of total accruals scaled by lagged assets for all firms in the same industry and in a specific year.

In this sense, earnings management will be the difference between a firm's accruals and those of other firms in the same industry. However, the assumption that all firms in the same industry operate under exactly the same conditions is weak (Dechow et al., 1995). Furthermore, if the other firms which do not have the motive to manage

earnings in the industry manage earnings, then earnings management measure will be biased upwards, and vice versa (Ronen and Yaari, 2008).

2.4.5 The Jones Model

The Jones model (1991) represents one of the most famous models in calculating accrual earnings management. It starts by measuring the following regression for the total accruals of the firms in the estimation period per year-industry:

$$\frac{TA_{it}}{A_{avg}} = \alpha + \alpha_1 \frac{1}{A_{avg}} + \beta_1 \frac{\Delta R_{it}}{A_{avg}} + \beta_2 \frac{PPE_{it}}{A_{avg}} + \varepsilon_{it}$$
(2.9)

The coefficients from the previous regression are used to determine the normal (nondiscretionary) accruals in the following regression in the event period for each firm within a specific year and industry group:

$$NA_{it} = \widehat{\alpha} + \widehat{\alpha}_{l} \frac{1}{A_{avg}} + \widehat{\beta}_{l} \frac{\Delta R_{it}}{A_{avg}} + \widehat{\beta}_{2} \frac{PPE_{it}}{A_{avg}} + \varepsilon_{it}$$
(2.10)

The residual (ε_{it}) from the second regression is the measure of normal accruals. Discretionary accruals are calculated as the difference between a firm's total accruals and normal accruals from the previous equation:

$$DA_{it} = \frac{TA_{it}}{A_{avg}} - NA_{it}$$
(2.11)

Where *TA* represents total accruals measured following the balance-sheet-approach or the cash-flow-statement-approach;¹ *A* represents total assets; *R* represents revenues; *PPE* represents the gross value of property, plant, and equipment; *NA* represents normal accruals; and *DA* represents discretionary accruals. An intercept is included and all variables are scaled by average total assets (A_{avg}) of the beginning and ending asset balances to avoid heteroskedasticity problems. A cross sectional model is used in the calculations for each year and each industry classified by its four-digit SIC code so that

¹ The balance sheet approach calculates total accruals as the change in current assets (except cash items) minus the change in current liabilities (except the current portion of long term debt) minus depreciation. On the other hand, the cash-flow-statement-approach measures total accruals as the difference between earnings before extraordinary items and discontinued operations and the operating cash flows. Hribar and Collins (2002) document that the cash flow statement approach is more reliable, and thus I follow this approach for calculating accrual earnings management in this thesis.

while taking the time influence into consideration the model controls for the economic variations between industries.

The model focuses on the changes in revenues as the main source of change in working capital accruals including accounts receivable, inventory, and accounts payable. Hence β_1 is expected to be positive as sales increase, but if sales are mostly in cash and purchases are in credit, a negative working capital appears that makes the coefficient negative. On the other hand, the gross property, plant, and equipment reflects the accruals related to total depreciation expense. Therefore, β_2 is expected to be negative as huge PP&E result in bigger depreciation expense. The impact of depreciation on regression (2.9) is usually more than that of current working capital accruals and hence it is excluded by some studies when applying the Jones model (Barth et al., 2001).

However, the Jones model does not take other expenses into consideration as explanatory variables although they influence total accruals. In this sense, the model suffers from the problem of omitted variables. Yaari et al. (2007) show that the omission of variables results in a model that does not explain discretionary accruals well and thus generates lower R^2 and F statistics. The omitted variables will be captured in the error term that might correlate with the existing explanatory variables and result in biased estimation of earnings management and subsequently biased results in any study that examines accrual earnings management as a dependent variable (Dechow et al., 1995; Young, 1999).

Meanwhile, measurement errors may influence the results obtained using the Jones model. When examining the effect of a specific factor on discretionary accruals, the correlation between the independent variable and the measurement error captured in the error term may result in biased inferences (McNichols, 2002). Similarly, some bias may rise because of the flexibility in calculations like using the balance-sheet-approach or the cash-flow-statement-approach in calculating total accruals. (Hribar and Collins, 2002). The Jones model also suffers from a simultaneity problem as account receivable influences the explanatory variables (ΔR) and the dependent variable (*TA*) at the same time.

The Jones model originally followed a time-series approach in the calculation of the coefficients in equation (2.9); and thus raised concerns about the stationarity of accruals over the estimation period. Therefore, a cross-sectional model has been mostly followed by the subsequent studies that mainly uses firms within the same industry and in the same time period (Yaari et al., 2007), or a group of matched firms (Kothari et al., 2005) or the performance of the industry in the previous years (Xie et al., 2003) to determine the level of normal accruals. However, the sample used to estimate normal accruals may itself include some firms that have managed accruals as well; which makes benchmarking inappropriate (Shivakumar, 2000). It is also not reasonable to assume the homogeneity of all firms in the same industry. In addition, small samples are ignored in cross-sectional analysis as they result in violating the assumption of Ordinary Least Square (OLS) regression regarding the normality of the error term (Bernard and Skinner, 1996).

The Jones model might detect discretionary accruals in some firms not because of earnings management but due to specific business conditions that drive a change in accruals like acquisitions, capital expenditure, discontinued operations, and sale of longlived assets (Ronen and Yaari, 2008). A similar problem appears in firms with extreme performance because they do not show a linear relationship between performance and normal accruals; thus the Jones model may end up categorizing high normal accruals as discretionary accruals. The same may occur in large random samples when firms manage earnings in different directions that will finally offset each other and result in an average normal accruals of zero which results in inappropriate benchmarking when determining the discretionary accruals (Dechow et al., 1995). In the three previous cases, there is a higher probability of type I error where the null hypothesis that earnings management did not occur might be rejected. To mitigate this problem, some studies add new explanatory variables to the model like acquisitions, capital expenditure, discontinued operations, and sale of long-lived assets (Hansen, 1999). Other studies exclude certain firms in the calculations like high-growth ones as they generally show high accruals which are not necessarily due to earnings management (Ye, 2006). Some studies even control for performance in the model by incorporating the return on assets or cash flows (Jeter and Shivakumar, 1999; McNichols, 2000).

More seriously, the Jones model may not allow detecting earnings management especially in small samples that generate high standard errors and thus increase the probability of type II error and ultimately accepting the null hypothesis that earnings management did not occur. However, as top journals tend to reject such papers, type II error seems more limited than type I error explained earlier (McNichols, 2000). To overcome this problem, large samples are recommended in a cross-sectional approach when using the model. However, large samples can hide the individual firm characteristics that might differ from the group and thus the homogeneity assumption is not valid (DeAngelo, 1986; Bernard and Skinner, 1996). Therefore, some studies add firm-specific characteristics to the model to overcome this homogeneity problem (Ye, 2006).

2.4.6 The Modified Jones Model (Dechow et al., 1995)

To mitigate some of the previous shortcomings of the Jones model, a number of studies have tried to modify it by adding the omitted variables, controlling for performance, taking the dynamic nature of accruals and cash flows into consideration, eliminating extreme observations, or applying different estimators to solve the model. Dechow et al. (1995) developed a new model from the Jones model (1991); which has been referred to as the modified Jones model (1995). It again starts by measuring the following regression for the total accruals of a firm in the estimation period:

$$\frac{TA_{it}}{A_{avg}} = \alpha + \alpha_1 \frac{1}{A_{avg}} + \beta_1 \frac{\Delta R_{it}}{A_{avg}} + \beta_2 \frac{PPE_{it}}{A_{avg}} + \varepsilon_{it}$$
(2.12)

The coefficients from the previous regression are used to determine the normal (nondiscretionary) accruals in the following regression in the event period:

$$NA_{it} = \widehat{\alpha} + \widehat{\alpha}_I \frac{1}{A_{avg}} + \widehat{\beta}_I \frac{(\Delta R_{it} - \Delta A R_{it})}{A_{avg}} + \widehat{\beta}_2 \frac{PPE_{it}}{A_{avg}} + \varepsilon_{it}$$
(2.13)

The equation, however, adjusts the change in revenues for the change in accounts receivable here to avoid any discretion in credit sales while calculating normal accruals; compared to the first equation where total accruals were measured (Cohen et al., 2008). Discretionary accruals are calculated as the difference between a firm's total accruals and normal accruals from the previous equation:

$$DA_{it} = \frac{TA_{it}}{A_{avg}} - NA_{it}$$
(2.14)

Where *TA* represents total accruals measured following the balance-sheet-approach or the cash-flow-statement-approach as mentioned above; *A* represents total assets; *R* represents revenues; *AR* represents accounts receivable; *PPE* represents the gross value of property, plant, and equipment; *NA* represents normal accruals; and *DA* represents discretionary accruals. All variables are scaled by average total assets (A_{avg}) of the beginning and ending asset balances to avoid heteroskedasticity problems. A cross sectional model is used in the calculations for each year and each industry classified by its four-digit SIC code so that while taking the time influence into consideration the model controls for the economic variations between industries.

Although the modified Jones model takes into account the possibility of manipulating accounts receivable in the event period, it ignores this fact in the estimation period; which leads to inconsistency in the calculations. This implies that the use of cash sales in both periods, rather than the event period only, is a better approach and can mitigate the simultaneity problem of having accounts receivable on both sides of the equation (Ronen and Yaari, 2008). Consequently, some studies in the earnings management literature have started to adopt the cross-sectional version of the modified Jones model that considers the adjustment for both the estimation and event periods rather than the time-series version of the modified Jones model that makes the adjustment in the event period only; e.g., Dechow et al. (2003) and Kothari et al. (2005). Compared to the Jones model, the modified Jones model overestimates discretionary accruals and therefore it is less likely to fall in type II error (Ronen and Yaari, 2008). However, the influence of measurement errors also applies on the modified Jones model.

2.4.7 The Forward-Looking Model

The forward-looking model developed by Dechow et al. (2003) also tries to solve some of the problems associated with the Jones model. In addition to excluding the change in accounts receivable from the change in revenues like the modified Jones model, it also excludes the discretionary part of credit sales by adding back the non-discretionary component calculated as the change in sales multiplied by the sensitivity of the change in accounts receivable to sales. The model also controls for lagged accruals that are expected to influence current discretionary accruals when they reverse. Finally, the model is forward-looking as it controls for the growth in sales that is expected to result in a growth in accruals as well and thus it avoids misclassifying such growth as earnings management. The forward-looking model is estimated per industry-year as follows:

$$TACC_{it} = \alpha + \beta_1 \left((1+k)\Delta Sales \Delta AR \right) + \beta_2 PPE + \beta_3 TACC_{it-1} + \beta_4 GR_sales_{it+1}$$

$$(2.15)$$

Where *TACC* is total accruals; k is the slope coefficient obtained from regressing the change in accounts receivable on the change in sales; $\Delta Sales$ is the change in sales from the previous accounting period; ΔAR is the change in account receivables from the previous accounting period; *PPE* is property, plant, and equipment; and *GR_sales* is the change in sales compared to the previous year. Discretionary accruals are again calculated as the difference between total accruals and normal accruals similar to the Modified Jones model. However, the model still suffers from the problems of simultaneity and measurement errors.

2.4.8 The Competing-Component Model (The K S Model)

As performance influences accruals, some models have attempted to control for specific performance characteristics that may result in classifying normal accruals as discretionary. Kang and Sivaramakrishnan (1995) introduce the competing-component model that incorporates more regressors to match the transactions and assets, e.g., revenues, expenses, and PP&E with their related working capital accruals, e.g., accounts receivable, accounts payable, inventory accruals, and depreciation. Identifying each of the previous items separately reduces errors that may arise because of the variation in the credit policies related to revenues and expenses. The model also considers the time impact on accruals by taking the lags of revenues and expenses in determining current accruals because previous year's credit policies will influence the current year's accruals. In addition, the model solves for the problems of endogeneity that arise from the simultaneity, omitted variables, and measurement error with a generalized method of moments (GMM) estimator that uses some instrumental variables rather than a pooled OLS estimator. Kang and Sivaramakrishnan (1995) document that the new estimator results in reducing type I and II errors. The competing-component model is as follows:

$$AB_{it} = \emptyset_0 + \emptyset_{AR} \left[\frac{AR_{t-1}}{REV_{t-1}} REV_{i,t} \right] + \emptyset_{APB} \left[\frac{APB_{t-1}}{EXP_{t-1}} EXP_{i,t} \right] + \emptyset_{DEP} \left[\frac{DEP_{t-1}}{GPPE_{t-1}} GPPE_{i,t} \right] + \varepsilon_{i,t}$$

$$(2.16)$$

Where AB_{it} represents the unmanaged accruals; AR is accounts receivable accruals; REV is revenues; APB is the aggregate of inventory accruals, other non-cash current asset accruals, and current liability accruals; EXP is the expenses related to the APB like cost of sales and selling and administrative expenses; DEP is depreciation expense; and GPPE is gross property plant and equipment. However, the model does not control for major performance measures like the return on assets or sales growth; which may give some space for misclassifying normal accruals as discretionary.

2.4.9 The Cash-Flows Model

Dechow and Dichev (2002) introduce the cash-flows model that controls for firm performance by incorporating its cash flows to measure how accruals map into these cash flows. As accruals are linked to the timing of cash recognition, cash flows of the past, current, and future periods are included in the explanatory variables to represent the probable returns of previous sales, current sales, and expected bad debt in the future, respectively. The model does not use accruals as the dependent variable as its calculation is based on cash flows and thus results in a simultaneity problem when cash flows variable is introduced as a control. Instead, the model uses the working capital as the regressand. The cash-flows models is as follows:

$$\Delta WC_t = b_0 + b_1 CF_{t-1} + b_2 CF_t + b_3 CF_{t+1} + \varepsilon_t \tag{2.17}$$

Where ΔWC is the change in working capital; *CF* is operating cash flows; and ε is the residual from the regression which is considered as the measure of accrual earnings management. However, the residual is likely to capture some non-discretionary aspects that might be related to firm characteristics like the volatility of the transactions and thus better represents the overall quality of earnings as discussed earlier in section 2.2.3 (Ronen and Yaari, 2008).

2.4.10 The Cash-Flows Jones Model

McNichols (2002) develops the previous model further by combining it with the original Jones model. She finds that the explanatory power of the model increases and thus confirms the importance of controlling for performance when measuring accrual earnings management. The McNichols model is as follows:

$$\Delta WC_t = b_0 + b_1 CF_{t-1} + b_2 CF_t + b_3 CF_{t+1} + b_4 \Delta Slaes + b_5 PPE + \varepsilon_t$$

$$(2.18)$$

Where ΔWC is the change in working capital; *CF* is operating cash flows; *PPE* is the gross property, plant and equipment, and ε is the residual from the regression which is considered as the measure of accrual earnings management. Although the residual is less likely to capture non-discretionary accruals in this case, the model suffers from simultaneity and measurement error problems.

2.4.11 The Performance-Matching Model

Kothari et al. (2005) introduce the performance-matching model that further develops the Jones model and the modified Jones model by controlling for firm operating performance. They match the discretionary accruals of firms in the sample with those of firms in a control group within the same industry. This way, they control for the effect of any extreme performance that results in a non-linear relation between performance and accruals and thus type I error, e.g., growth firms. In its simple form, the model considers accrual earnings management as the difference between a firm's accruals and its peer's accruals from the control group without running any regression. In a more sophisticated format, the model calculates accrual earnings management for each firm using the Jones model while incorporating the lagged return on assets ratio (ROA) as a control and then makes the comparison. As a result, discretionary accruals can be calculated in the linear-performance-matching model as follows:

$$DA_{it} = \frac{DA_{it}}{A_{avg}} - \frac{DA_{icont}}{A_{avg}}$$
(2.19)

Where DA represents discretionary accruals; A represents total assets; and *icont* represents a firm from the control group close to firm *i* in its return on assets (ROA) within the same year and industry sector. The performance-matching model shows

better explanatory power than the Jones model (Ye, 2006). However, a problem appears in identifying the matched group of firms as it is not always possible to obtain a homogenous relation between performance and accruals for two different groups. In this sense, selecting the inappropriate matched group will cause biased results. For example, selecting a matched firm that manages earnings may result in a type II error (Kothari et al., 2005).

2.4.12 The Business Model

Ye (2006) developed the business model which solves for more than one deficiency in the Jones model at once. The model controls for performance, takes the dynamic nature of abnormal accruals into consideration, adds the intensity of working capital to control for sales growth related to the short-term accruals, and finally incorporates depreciation to control for the impact of long-term accruals. This way, the model controls for firm-specific characteristics that were ignored in cross-sectional models. The business model looks as follows:

$$TA_{i,t} = \alpha_0 + \left(\frac{\beta_0 + \beta_1 \Delta REV_{i,t} + \beta_2 PPE_{i,t}}{A_{i,t-1}}\right) + \beta_4 ROA_{i,t-1} + \left(\beta_5 NCWC_{i,t-1} - \beta_6 \overline{NCWC}_{i,t}\right) + \beta_7 NCWC_{i,t-1} X\Delta REV_{i,t} + \beta_8 dep_{i,t-1} + \beta_9 dep_{i,t-1} PPE_{i,t} + \varepsilon_t$$

$$(2.20)$$

Where, *TA* is the total accruals; $\frac{\beta_0 + \beta_1 \Delta REV_{i,t} + \beta_2 PPE_{i,t}}{A_{i,t-1}}$ is the Jones model where ΔREV is the change in revenues, *PPE* is property, plant, and equipment, and *A* is total assets; *ROA* is the return on assets to control for firm performance; $NCWC_{i,t-1} - \overline{NCWC}_{i,t}$ represents the abnormal lagged accruals where *NCWC* is the non-cash working capital and \overline{NCWC} is the normal non-cash working capital calculated as the average non-cash working capital over the last three years; *dep* is the depreciation rate; and $dep_{i,t-1}PPE_{i,t}$ is the historical depreciation used to anticipate the depreciation expense for the current period. However, the model still suffers from an unsolved endogeneity problem due to the simultaneity between the dependent and explanatory variables.

2.4.13 The Stubben Model

Stubben's Model (2010) focuses on discretionary revenues which are the largest component of earnings in most firms (Stubben, 2010). Using revenues as an estimate of discretion reduces measurement error and makes the model less biased and more specified than other accrual models due to three reasons. First, Discretionary revenues reflect receivables' accruals, rather than aggregate accruals. Receivable accruals, in turn, are more directly related to revenues than other working capital accruals. Second, the model focuses on reported revenues rather than on cash revenues. While this results in understating discretionary revenues estimate, it is unlikely to overestimate discretion for firms that are less expected to collect their credit revenues by the year end like growth firms. Finally, the model examines receivable accruals for the fourth quarter separately because they are less likely to be collected before the year end. As a result, it prevents overstating discretion when the revenues of the fourth quarter are relatively high or understating discretion when the revenues of the fourth quarter are relatively low (Stubben, 2010).

$$\frac{\Delta AR_{it}}{A_{avg}} = \alpha + \alpha_I \frac{1}{A_{avg}} + \beta_I \frac{\Delta RI_{-3_{it}}}{A_{avg}} + \beta_2 \frac{\Delta RA_{it}}{A_{avg}} + \varepsilon_{it}$$
(2.21)

Where *AR* represents accounts receivable; *A* represents total assets; *R*1_3 represents revenues in the first three quarters; and *R*4 represents revenues in the fourth quarter. All variables are scaled by average total assets (A_{avg}) to avoid heteroskedasticity problems. A cross sectional model is used in the calculations for each year and each industry classified by its four-digit SIC code. The residual (ε_{it}) from the regression is the measure of discretionary revenues (*DiscRev*) that represents accrual earnings management (*AccrualEM*). However, the model ignores other working capital accruals that might be material in some firms depending on the nature of their transactions. In addition, by using a pooled OLS estimator, the model still suffers from a problem of endogeneity. Finally, it does not control for any measures of firm performance.

2.4.14 The Single-Item Models

Although the models of aggregate discretionary accruals have dominated earnings management literature as mentioned earlier, single-item models have been introduced in certain occasions. Based on materiality criterion, these models select one item of the income statement that can be manipulated within GAAP, like estimating the bad debt expense, reserves, or even aggregate accounts like total tax expense which includes deferred tax as well (McNichols and Wilson, 1988; Dhaliwal et al., 2004). This methodology allows better identification of discretionary accruals in a specific account. However, the discretionary part might be covered by the non-discretionary part if the latter is higher than the earlier. Furthermore, although I can accept the fact that total earnings management exists if discretionary accruals were detected in a specific account, the opposite may not apply and thus the results cannot be generalized to total accruals.

2.4.15 The Distributional Approach

A final methodology of measuring earnings management is by studying the distribution of earnings or earnings per share (EPS). This kind of methodology is referred to as the distributional approach. The approach assumes that unmanaged earnings follow a normal distribution and thus earnings management is measured by how each observation deviates from that normal distribution (Thomas, 1989; Burgstahler and Dichev, 1997). However, determining the boundaries of the normal distribution remains a subjective matter. In addition, having high or low earnings management.

2.4.16 Evaluation of Accrual Earnings Management Models

Overall, the majority of accrual earnings management models consider total accruals as the dependent variable, measured by using the balance sheet approach (working capital items) or the cash flow approach (the difference between earnings and operating cash flows). However, some models use a single item of the total accruals, e.g., Stubben's Model, because they want to emphasize the relative importance of this particular account among other accruals. While using total accruals allows for a more comprehensive measure of accrual earnings management, it is difficult to specify the degree of influence for each of the explanatory factors on each item within total accruals (McNichols and Wilson, 1988). The opposite applies on using a single accrual item as a dependent variable.

Different factors have been included within the explanatory variables of accrual behaviour. The most commonly used ones were revenues, PP&E, and expenses. Meanwhile, some models take the time effect on accruals' manipulation into consideration, e.g., the Ronen and Sadan Model, or incorporate the role of time in the form of the lagged accruals value, e.g., the Forward-Looking Model, the Business Model, and the Competing-Component Model. Furthermore, some models control for performance using sales growth (Forward-Looking Model), cash flows (Cash-Flows Models), return on assets (Performance-Matching Model and Business Model), abnormal sales (Ronen and Sadan Model), matching performance to a benchmark (Performance-Matching Model), or working capital items (Competing-Component Model and Business Model) to avoid misclassifying normal accruals as discretionary. However, some models still ignore this type of control, e.g., the Jones Model, Modified Jones Model, and Stubben's Model; and thus are more likely to fall in type I error.

For measuring accrual earnings management, some of the models do not even run any kind of regression, e.g., the Healy Model, DeAngelo Model, the Industry-Based Model, and the Distributional Approach. Instead, they consider accrual earnings management as the deviation of current accruals from the past or current accruals' values. Therefore, these models are less powerful in detecting earnings management because they do not consider the influence of the different firm operations and characteristics on the manipulation. On the other hand, almost all the models that run regressions use a pooled OLS estimator to solve for accrual earnings management. The Competing-Component Model is the only one that uses a system generalized method of moments (GMM) estimator which allows for resolving the endogeneity that arises because of the problems of simultaneity, omitted variables, and measurement errors.

Although the previous models have tried hard to provide reliable measures of accrual earnings management, they still suffer from different problems like the omitted variables, simultaneity between the dependent and explanatory variables, type I error due to ignoring performance effect, weak assumptions that result in biased measurement, and unsuitability for extremely small or large samples. Although some studies have documented high correlation between the different aggregate models of accrual earnings management (Guay et al., 1996), others have found that some models are better in measuring accrual earnings management than others (Dechow et al., 1995; Thomas and Zhang, 2000). As each model measures a different aspect of accrual earnings management, choosing the best measure is still a subjective matter that depends on the different needs of each researcher. Using more than one model is always a good approach in avoiding biased results, but developing more precise models remains one of the challenges for future research. For the measurement purpose of this thesis, I will select the Modified Jones Model which shows strong power in explaining total accruals and Stubben's Model which specifically emphasizes the receivable accruals related to discretionary revenues (Dechow et al., 1995).

2.5 Real Earnings Management Models

Before the introduction of the Roychowdhury Model in 2006, studies used to focus only on individual economic choices for measuring real earnings management activities and emphasized the manipulation in a specific industry sector, e.g., Baber et al. (1991) examine the manipulation of R&D expense in the manufacturing sector in the US. Afterwards, studies have started to implement more than one aspect of real earnings management in more than one industry sector. The most commonly used models that measure the overall real earnings management across industries and over time include the Roychowdhury Model and the Gunny Model.

2.5.1 The Roychowdhury Model

Based on the study of Dechow et al. (1995), Roychowdhury (2006) developed a model that represents one of the most commonly used measures of real earnings management in the non-financial sectors and consists of three components (Cohen et al., 2008; Walker, 2013). The first component represents the decrease in operating cash flows as a result of sales discounts and flexible credit terms in order to increase the sales volume and improve earnings in a specific period. Therefore, the model starts by establishing

the normal level of operating cash flows based on sales level and changes in sales according to the following regression.

$$\frac{CFO_{it}}{A_{avg}} = \alpha + \alpha_1 \frac{1}{A_{avg}} + \beta_1 \frac{Sales_{it}}{A_{avg}} + \beta_2 \frac{\Delta Sales_{it}}{A_{avg}} + \varepsilon_{it}$$
(2.22)

In the previous equation, *CFO* represents operating cash flows; *Sales* represents the sales for the current period; $\Delta Sales$ represents the change in sales during the current period. I scale all variables by average total assets (A_{avg}) of the beginning and ending asset balances to avoid any heteroschedasticity. I use a cross sectional model in the calculations for each year and each industry classified by its four-digit SIC code. The abnormal operating cash flows are calculated as the difference between the normal operating cash flows and the actual operating cash flows. For the resulting measure to reflect the upward earnings management, it is multiplied by -1.

The second component of real earnings management according to Roychowdhury model (2006) is the decrease in discretionary expenses in order to improve earnings and current cash flows. Discretionary expenses include research and development (R&D), advertising, and selling, general, and administrative (SG&A) expenses. The model starts by establishing the normal level of discretionary expenses based on sales level according to the following regression.

$$\frac{DiscExp_{it}}{A_{avg}} = \alpha + \alpha_I \frac{1}{A_{avg}} + \beta_I \frac{Sales_{it-I}}{A_{avg}} + \varepsilon_{it}$$
(2.23)

In the previous equation, DiscExp represents discretionary expenses that include research and development (R&D), advertising, and selling, general, and administrative (SG&A) expenses; $Sales_{it-1}$ represents the sales of the previous period which is used rather than the current period sales ($Sales_{it}$) to avoid the mechanical decrease in discretionary expenses in case of managing sales upwards during the current period. I scale all variables by average total assets (A_{avg}) of the beginning and ending asset balances to avoid any heteroschedasticity. I use a cross sectional model in the calculations for each year and each industry classified by its four-digit SIC code. The abnormal discretionary expenses are calculated as the difference between the normal discretionary expenses and the actual discretionary expenses. For the resulting measure to reflect the upward earnings management, it is multiplied by -1. The third component of real earnings management according to Roychowdhury model (2006) is the increase in operating costs like the increased inventory production in order to decrease the cost of goods sold and ultimately improve earnings. As the fixed costs are split over a bigger number of units, the unit cost will decrease. The production costs, however, will still be high relative to sales and result in an overall decrease in operating cash flows. The model again starts by establishing the normal level of production costs based on sales level and changes in sales according to the following regression.

$$\frac{Prod_{it}}{A_{avg}} = \alpha + \alpha_1 \frac{1}{A_{avg}} + \beta_1 \frac{Sales_{it}}{A_{avg}} + \beta_2 \frac{\Delta Sales_{it}}{A_{avg}} + \beta_2 \frac{\Delta Sales_{it-1}}{A_{avg}} + \varepsilon_{it}$$
(2.24)

In the previous equation, *Prod* represents the production costs which includes the costs of goods sold (*COGS*) in addition to the change in inventory (ΔInv) during the period; *Sales* represents the sales for the current period; $\Delta Sales$ represents the change in sales during the current period; $\Delta Sales_{it-1}$ represents the change in sales during to the previous period. I scale all variables by average total assets (A_{avg}) of the beginning and ending asset balances to avoid any heteroschedasticity. I use a cross sectional model in the calculations for each year and each industry classified by its four-digit SIC code. The abnormal production costs are calculated as the difference between the normal production costs and the actual production costs.

However, the model still suffers from the problem of omitted variables. Furthermore, the assumption that the sample used to measure normal real earnings management is by itself normal cannot be valid all the time as it may include some firms that manipulate earnings, and thus benchmarking to this group is not appropriate and the results might be biased. The model is also not suitable for extremely small samples due to violating the OLS assumption regarding the normality of the error term. The same applies to extremely large samples where the individual characteristics of the firms are hidden.

2.5.2 The Gunny Model

Gunny's Model (2010) for real earnings management measures four types of real earnings management activities. These include decreasing discretionary research and

development expense (REM_{RD}) as in equation (2.25), decreasing discretionary selling, general, and administrative expense (REM_{SGA}) as in equation (2.26), timing of fixed asset sales to report gain (REM_{Asale}) as in equation (2.27), and overproduction (REM_{Prod}) to cut prices or decrease the cost of goods sold (COGS) as in equation (2.28) (Gunny, 2010).

$$\frac{RD_t}{A_{avg}} = \alpha_0 + \alpha_1 \frac{1}{A_{avg}} + \beta_1 M V_t + \beta_2 Q_t + \beta_3 \frac{INT_t}{A_{avg}} + \beta_4 \frac{RD_{t-1}}{A_{avg}} + \varepsilon_t^{R\&D}$$
(2.25)

$$\frac{SGA_t}{A_{avg}} = \alpha_0 + \alpha_1 \frac{1}{A_{avg}} + \beta_1 M V_t + \beta_2 Q_t + \beta_3 \frac{INT_t}{A_{avg}} + \beta_4 \frac{\Delta S_t}{A_{avg}} + \beta_5 \frac{\Delta S_t}{A_{avg}} *DD + \varepsilon_t^{SG\&A}$$
(2.26)

$$\frac{GainA_t}{A_{avg}} = \alpha_0 + \alpha_1 \frac{1}{A_{avg}} + \beta_1 M V_t + \beta_2 Q_t + \beta_3 \frac{INT_t}{A_{avg}} + \beta_4 \frac{ASales_t}{A_{avg}} + \beta_5 \frac{ISales_t}{A_{avg}} + \varepsilon_t^{Asset}$$
(2.27)

$$\frac{PROD_t}{A_{avg}} = \alpha_0 + \alpha_1 \frac{1}{A_{avg}} + \beta_1 M V_t + \beta_2 Q_t + \beta_3 \frac{S_t}{A_{avg}} + \beta_4 \frac{\Delta S_t}{A_{avg}} + \beta_5 \frac{\Delta S_{t-1}}{A_{avg}} + \varepsilon_t^{Production}$$
(2.28)

The previous equations are initially used to calculate normal real earnings management, where RD represents R&D expense; SGA represents selling, general and administrative expense; GainA represents income from asset sales; PROD represents COGS plus change in inventory; A represents total assets; MV represents the natural logarithm of market value calculated as the number of common shares outstanding multiplied by the share price; Q represents Tobin's Q calculated as the sum of the market value of common shares, the book value of preferred shares, long term debt and current debt divided by total equity and liability; *INT* represents internal funds calculated as the sum of income before extraordinary items, research and development expense, and depreciation and amortization expenses; DD represents an indicator variable that reflects the sticky cost behaviour for the intentional reduction in SG&A when the demand drops, which equals 1 when total sales decrease between t-1 and t, and zero otherwise; ASales represents the sales of long-lived assets; and ISales represents the sale of long-lived investment. To keep the relation between GainA, ASales, and ISales monotonic in equation (2.27), I make all their signs negative when GainA is negative. I scale all variables by average total assets (A_{avg}) of the beginning and ending asset

balances to avoid any heteroschedasticity. I use a cross sectional model in the calculations for each year and each industry classified by its four-digit SIC code. The residuals ($\varepsilon_t^{R\&D}$, $\varepsilon_t^{SG\&A}$, ε_t^{Asset} , $\varepsilon_t^{Production}$) from the regressions represent the measures for the R&D, SG&A, fixed assets' sale, and production components of real earnings management (REM_{RD} , REM_{SGA} , REM_{Asale} , REM_{Prod}) respectively. Imultiply the first and second residuals by negative one so that cutting the discretionary expense reflects an increase in real earnings management.

For the purpose of obtaining an overall estimate for real earnings management in this thesis, and similar to what I have done earlier when applying the Roychowdhury model, a principal component analysis with varimax rotation is performed (Demerjian et al., 2013b). This step prevents the inclusion of highly correlated variables in the measure. Although the model incorporates more variables to explain each measure of real earnings management, it suffers from the problem of endogeneity due to the simultaneity between the dependent and explanatory variables. Meanwhile, it is vulnerable to error in the measurement of any of the included variables. Similar to Roychowdhury Model, measuring normal real earnings management does not provide a valid benchmark as it may include some firms that already use real earnings management. Furthermore, the model would not be suitable for extremely small or large samples.

2.5.3 Evaluation of Real Earnings Management Models

Very few models have been developed to measure real earnings management compared to those that measure accrual earnings management. The models of Roychowdhury (2006) and Gunny (2010) seem to be the only ones in this area. Both models emphasize management decisions to manipulate discretionary expenses and production to improve earnings. However, while the Roychowdhury Model takes into consideration some of the managerial decisions related to manipulating sales transactions, the Gunny Model emphasizes those that allow for manipulating the sale of assets and investments.

Although the Gunny Model incorporates more variables to explain its dependent variables than the Roychowdhury model, both models still suffer from endogeneity due to the problems of simultaneity, omitted variables, and measurement error. Therefore,
applying an OLS regression is not the optimal way to solve the previous models and it is essential to use alternative estimators, e.g., a generalized method of moments (GMM) estimator. In addition, both models suffer from some inappropriate assumptions and may generate problems when applied on small or large samples.

Similar to the case of accrual earnings management, selecting a model to measure real earnings management remains a subjective matter depending on research needs. However, using more than one model is recommended and therefore I apply both of the previous models in this thesis because they measure different aspects of real earnings management. Meanwhile, future research has to take into consideration introducing new models that assist in solving some of the problems encountered by the previous models.

2.6 Special Measures of Earnings Management

To complete the previous discussion, I also cover some special issues encountered in the literature regarding the measurement of earnings management. Some of the models in the earnings management literature have tried to solve for the mutual effect between accrual and real earnings management. In addition, specific models have been introduced to measure earnings management in the financial sector because of its characteristics that require special attention in measurement. Finally, some attempts to measure earnings management have been evidenced in the qualitative research. All these topics are covered in some detail in this section.

2.6.1 Combined Models

Zang (2012) introduces a model that tackles the endogeneity problem between accrual and real earnings management. The model considers the sequential nature of the two activities - while real earnings management takes place within the financial year, accruals' manipulation is generally performed after the year end. Therefore, the level of accrual earnings management depends on the previously manipulated real activities, but not the other way round. In this sense, real earnings management is introduced as an exogenous variable when solving for accrual earnings management equation but accrual earnings management is excluded when solving for real earnings management equation. The model emphasizes the trade-off between accrual and real earnings management activities based on their costs and benefits that vary across firms according to the following two equations:

$$RM_{t} = \beta_{0} + \sum_{k} \beta_{l,k} Cost \text{ of } RM_{k,t} + \sum_{l} \beta_{2,l} Cost \text{ of } AM_{l,t} + \sum_{m} \beta_{3,m} Control_{m,t} + u_{t}$$

$$(2.29)$$

$$AM_{t} = \gamma_{0} + \sum_{k} \gamma_{l,k} Cost \text{ of } AM_{k,t}$$

$$+ \sum_{l} \gamma_{2,l} Cost \text{ of } RM_{l,t} + \gamma_{3} Unexpected RM_{t} + \sum_{m} \gamma_{4,m} Control_{m,t} + v_{t}$$

$$(2.30)$$

Where RM_t represents real earnings management measured using the Roychowdhury model (2006); AM_t represents accrual earnings management measured using the Jones model (1991); *Cost of RM* is the cost of real earnings management which include the competitive status in the industry, financial health, institutional ownership, and the tax consequences of the manipulation; *Cost of AM* is the cost of accrual earnings management which include audit quality, public scrutiny, and accounting flexibility; *Control_{m,t}* include predetermined firm characteristics; and *Unexpected RM_t* is the estimated residual from the first equation (u_t).

Although Zang (2012) has introduced the first model that considers the simultaneous measurement of accrual and real earnings management, it strictly relies on the assumption that both activities are sequential (Walker, 2013). However, firms are not prohibited from manipulating accruals within the accounting period. Furthermore, the model ignores the trade-off with other activities like earnings smoothing and classification shifting. The model also ignores the dynamic nature of earnings management by only examining the impact of the unexpected portion of real activities on the levels of discretionary accruals in the same year. However, it does not consider the missed chances to manipulate earnings in prior years on the current levels of earnings management. Incorporating the trade-off behaviour over time. Finally, some elements which may also influence earnings management behaviour are missing in Zang's model; e.g., managerial ability which influences the costs of both accrual and real activities.

2.6.2 Earnings Management Measures in the Financial Sector

The financial sector has its unique accounting and financial practices and is subject to distinct regulations. Therefore, managers in this sector have different motivations to manipulate earnings than those of managers in other industry sectors. Accordingly, different proxies are used to measure earnings management in the financial sector compared to other industries. One of the most commonly used ones is income smoothing through loan loss provision which represents the difference between the amount of the loan loss estimated and the amount actually required to set the provision. Earnings management in the financial sector can be also measured by earnings smoothing calculated as the correlation between the change in cash flows from operations and the change in total accruals or the volatility of earnings relative to the volatility of cash flows. Finally, benchmark beating can be used to measure earnings management in the financial sector, e.g., meeting or beating analysts' forecasts or reporting small profits or losses compared to the previous year (Imperatore and Trombetta, 2015; Martinez et al., 2015; Vlad, 2015).

2.6.3 Earnings Management in Qualitative Research

Although earnings management involves lots of behavioural perspectives, very few studies in the literature have examined it qualitatively. The most famous study in this area is the study of Graham et al. (2005) which performed a survey of 400 executives in addition to 20 interviews from the US firms. The study mainly finds that most executives in the sample are likely to use real earnings management than accruals' manipulation in order to achieve their earnings benchmarks. Executives are also more inclined to use earnings smoothing because it involves lower risk and tend to use aggressive smoothing strategies under the pressure of meeting earnings targets. The study also documents that earnings are more important numbers to the outsiders than revenues or cash flows; and prior earnings and analysts' forecasts are important benchmarks for management to meet in order to avoid negative market reactions. Similar findings were observed in a smaller scale survey and interviews in the UK performed by Choi et al. (2006) where analysts reported that UK executives are more likely to use real rather than accrual earnings management.

Another survey was performed by De Jong et al. (2014) and covered 306 analysts in addition to interviews of 21 analysts from the biggest investment banks in the world to examine the influence of earnings management on investors' reactions. Their results show that analysts believe in the influence of earnings management on firms' stock prices; and therefore firms manage earnings to meet earnings benchmarks and boost firm value. To achieve the previous target, managers prefer to increase earnings by reducing discretionary costs although it may contribute to destroying the value of the firm in the future. However, analysts believe that stock repurchase is a better choice for management to influence investors' perceptions.

Finally, Nelson et al. (2002) perform a qualitative study to examine the relation between earnings management and audit quality. To achieve this target, they use a questionnaire to collect data from 253 auditors in one of the big 5 audit firms in the US. Their results show that auditors do not request their clients to adjust for their misstatements that are structured to conform to precise standards e.g., manipulating leases and consolidations, or unstructured under imprecise standards e.g. manipulating reserves. Accordingly, auditors are likely to request an adjustment when there is a material unstructured misstatement under a precise standard.

Given all the limitations of qualitative research such as the respondents' bias and the low response rate; the previous findings have shifted the attention of modern research in earnings management to some of the manipulative activities that have underlying economic consequences. For future research, using such qualitative methodology is recommended besides the widely used quantitative methods to obtain new insights in the earnings management literature.

2.7 Conclusions

This chapter has attempted to provide a comprehensive definition of earnings management taking into consideration its characteristics, conditions, methods, and targets. Identifying the characteristics of earnings management allows for distinguishing it from fraud which involves a violation of GAAP, earnings quality that might be influenced by factors unrelated to management discretion, impression management that involves the narrative part of the annual reports, and expectation management that focuses on firm announcements.

The within GAAP managerial discretionary methods over financial reporting include accrual earnings management, real earnings management, earnings smoothing, income shifting, and other advanced techniques. Each of the previous methods has its unique features and hence managers tend to trade-off between them according to how their features fit into their needs. Although the earnings management literature has mostly focused on one method per study, it is more meaningful to analyse more than one of these methods at a time when examining earnings management behaviour.

Lots of models have been introduced for measuring accrual earnings management compared to very few ones for measuring real earnings management which is a relatively new topic in the literature. Almost all the models in both areas suffer from a problem of endogeneity due to the simultaneity between the dependent and explanatory variables, the omission of some of the independent variables, or errors in measuring any of the variables. In addition, the models sometimes make weak assumptions that lead to biased results and they are not always suitable to be applied in any context. Using more than one model may help in mitigating some of the previous problems, but researchers still have lots of flexibility in choosing between the different models; which may sometimes result in 'research management of earnings management'.

The previous models for measuring earnings management discussed in sections 2.4 and 2.5 cannot be applied to the financial sector due to its unique accounting characteristics. Therefore, specific measures have been developed in the literature for this task. The only model that takes the mutual impact of accrual and real earnings management is that of Zang (2012). However, it suffers from some problems that create a need for developing new models to solve for the trade-off between more than one earnings management activity at a time. Finally, the earnings management literature has been dominated by quantitative research although the qualitative methodology has already made significant contribution to the literature. Therefore, qualitative studies are worth considering by the current researchers in the field.

In this chapter, the conditions that allow earnings management to occur have been briefly described. However, each of the identified conditions has a theoretical basis in contracting theory, bounded rationality theory, or information theory. Therefore, the following chapter discusses in detail how earnings management theories have evolved based on each of these theories.

Chapter 3 Theoretical Framework

3.1 Introduction

This chapter establishes the theoretical framework of earnings management by focusing on the agency aspect of firm theory that best explains management opportunistic behaviour due to the separation of ownership from control in public companies. It also discusses the model developed by Walker (2013) to explain earnings management through the violation of the revelation principle and identifies the main conditions that allow for this behaviour. Then, the chapter discusses in detail the theory behind each of those conditions based on contracting theory, bounded rationality theory, and information asymmetry theory.

The chapter also discusses the three major theoretical approaches to explain earnings management based on Ronen and Yaari (2008). These include the costly contracting approach that is based on contracting theory, the decision-making approach that is based on information asymmetry theory, and the legal-political approach that takes into consideration the theories of contracting, bounded rationality and information asymmetry altogether. The chapter finally explains how all these theories contribute to explaining earnings management behaviour.

The remainder of this chapter is organized as follows. Section 3.2 provides a brief description of firm theory. Section 3.3 explains the agency aspect of firm theory. Section 3.4 discusses the revelation principle and the relevant theories to explain its conditions that include contracting theory, bounded rationality theory, and information asymmetry theory. Section 3.5 presents the three main theoretical approaches of earnings management that include the costly contracting approach, the decision-making approach, and the legal-political approach. Finally, section 3.6 concludes the chapter.

3.2 Firm Theory

A firm is considered as a black box that contains participants with conflicting objectives who aim to maximize their utilities (Jensen and Meckling, 1976). Therefore, a firm needs to organize the control and power between the different parties in the relationships within and outside its boundaries. As a result, the theory of the firm has been developed to provide a comprehensive view of the firm taking into consideration different factors at the agency, contracting, and property rights levels. To explain the internal and external relations, the modern perspective of firm theory emphasizes some behavioural aspects; e.g., it explains managerial performance through the motivation of managers to control the firms without owning them (Fama, 1980). For the purpose of this thesis, I emphasize the agency perspective of the firm because it assists in explaining earnings management behaviour as discussed in the following section.

3.3 Agency Theory

The agency theory, also referred to as the principal-agent theory, was established by Holmström and Milgrom (1987) based on the previous work of Jensen and Meckling (1976). It assumes that there are two parties involved in the employment contract: the agent (manager) and the principal (shareholder). The main problem between the principal and the agent is a problem of information. As the manager is more informed than the shareholders, a situation of information asymmetry arises which gives a chance for adverse selection by management. The different motivations and attitude towards risk by the different decision makers leads to an incentive problem that also contributes to the agency problem and referred to as moral hazard (Harris and Raviv, 1979). Adverse selection and moral hazard give rise to the problem of coordination in making decisions regarding what information is reported, how it is communicated, and who makes the decision. Thus, shareholders delegate the task of decision making to the managers while trying to maintain some control over their performance.

The different preferences and objectives of the principals and agents also contribute to the agency problem. The separation between ownership and control leads to specialized risk bearing and specialized decision skills in the organization. In this sense, principals generally prefer high payoff and are indifferent to the agent's effort. On the other hand, agents possess the decision making skills regardless of their personal wealth and thus prefer gaining high payoffs and expending less effort. Each party aims to maximize its utility function and ultimately the overall agency payoff depends on the agent's effort (Strong and Walker, 1987). The difference in the intensity of the agency problem results in different behaviours of management ranging from obedience to self-interest and further to opportunism that involves earnings manipulation (Giroux, 2004).

Furthermore, shareholders and managers differ in their interests to emphasize firm horizons when making their decisions. Shareholders may focus on the short- or long-term horizons because they are heterogeneous in their rationalities and ways of thinking. Therefore, some of them can get more benefits by maximizing firm's shortterm value like speculators, while others' interests are linked to maximizing firm's longterm value like long-term investors. Similarly, managers may focus on the long-term horizons of their firms as they are concerned about their careers in the long-run and, thus, they care about their reputations which will enable them to enter into more rewarding contracts in the future. In this sense, managers are motivated to maximize firms' long-term values which emphasize shareholders' value. On the other hand, managers may focus on the short-term horizons, so that they can extract better compensation by maximizing firms' short-term value and, therefore, they invest in short-term projects to signal a high-quality performance while ignoring shareholders' value. A post-horizon problem arises sometimes when managers aim to reflect better current performance that would open more chances for their future employment after retirement or after they leave their current firms. Consequently, when shareholders and managers focus on different aspects of the firm's horizons, a conflict of interests appears that contributes to the agency problem (Ronen and Yaari, 2008).

As a result of the agency problem, agency costs arise like the monitoring costs incurred by the principal, bonding costs incurred by the agent, and the residual loss represented by the decrease in principals' welfare as a result of their conflicting interests with the agents (Jensen and Meckling, 1976). However, the agency costs can be controlled by mechanisms like the hierarchical control systems in the organization, managerial labour market penalties, outsider experts in the boards of directors, the relative performance evaluation of the manager compared to other agents within the

same industry, and incentives like the performance-based rewards (Strong and Walker, 1987).

Managerial incentives have received most of the attention among the control mechanisms of the agency problem, especially that they are dependent on different factors. On the one hand, there is an inverse relation between pay performance sensitivity (PPS) and firm risk and, hence, an efficient contract has to combine between incentives and risk sharing. The previous relation was emphasized in the studies of Lambert and Larcker (1987), Garen (1994) and Aggarwal and Samwick (1999). On the other hand, the study of Demsetz and Lehn (1985) documents a positive relation between the pay performance sensitivity and firm risk. They find that firm risk increases under higher levels of uncertainty leading to more discretion in managerial decisions which makes them more difficult for shareholders to understand. As a result, shareholders offer more incentives to the managers to deal with this information gap. The previous relation was also emphasized in the studies of Core and Guay (1999, 2001, 2002) and Prendergast (2000, 2002). In this sense, the use of managerial incentives to mitigate the agency problem depends on how the shareholders view the relation between firm risk and the incentives.

Determining the type of managerial incentive is also dependent upon the risk attitude of management. Managers who are risk averse are associated with less information uncertainty about their decisions. Therefore, they require less monitoring as shareholders possess enough information to evaluate their performance. As a result, their incentives would be based more on their current salaries. On the other hand, managers who are risk takers are associated with more information uncertainty in their decisions, which increases the information problem. Therefore, their incentives would rely more on their delayed compensation as the firm obtains more information in the future (Eaton and Rosen, 1983).

Furthermore, determining management incentives depends on the costs of contracting like the monitoring cost. If the shareholders can observe the performance of the managers, then fixed wage contracts are optimal for the contracting between the two parties. However, if the agent's performance is difficult to be observed, then it would be better for the two parties to engage in an incentive scheme that allows for the trade-off between optimal incentives and optimal risk sharing.

Overall, although viewing earnings management as a beneficial activity may imply a resolved agency problem because it is used to convey firm private information according to the stewardship theory (Barney, 1990; Donaldson, 1990; Davis et al., 1997; Jiraporn et al., 2007; Gunny, 2010), earnings management is mainly perceived as a conflict of interest between management and shareholders. The problem originates from the information asymmetry between the principal and the agent, the difference in risk attitudes of the two parties or their focus on different aspects of the firm horizons. Managerial incentives have been the most commonly introduced mechanism to mitigate the agency problem. However, inappropriate incentives may lead to a productivity crisis because managers who are unsatisfied with their incentives may continue using the same old production tools to avoid any new investments. While the resulting high profits allow them to extract more compensation and get better reputations to further develop their careers, they can have a negative impact on firms' future productivity.

3.4 **Revelation Principle**

The revelation principle represents the ideal situation where no agency problem exists. In such a situation, the outcome of the game when all information is revealed can be generalized to situations with restrictions on information. Therefore, in the presence of information asymmetry, the revelation principle assumes that privately informed managers achieve more benefits from revealing the truth as they avoid any subsequent penalties on misreporting bad results (Ronen and Yaari, 2008). At the same time, shareholders would maximize their expected utilities as managers are motivated to act for improving firm performance (Milgrom and Roberts, 1992). In this sense, the revelation principle helps in solving the conflict between principals and agents and emphasizes truth telling equilibrium where the truthful revelation of private information by managers leads to maximizing the utility functions of all players in the game.

For the truth telling equilibrium to take place, the revelation principle assumes the following sequence of events. First, principals design the truth inducing contracts then agents do their efforts to meet these contracts, which are not observable by the principals. When the economic profits of these efforts are achieved, the agent produces a report about the economic results he/she obtains. Finally, the audit report is issued, the agent gets his/her payment based on the contract terms with the firm, and the principals receive the residual share (Ronen and Yaari, 2008).

However, the revelation principle does not apply when any of the following four conditions is violated (Walker, 2013). The first condition is violated when contracting costs are high, especially with the need for regular renegotiation to co-ordinate between management and the large number of shareholders. This renders contracts imperfect and difficult to enforce; and ultimately results in the use of simplified contracts with general terms and conditions (e.g., the use of GAAP to prepare the financial statements) that give more chances for earnings' manipulation (e.g., the within GAAP discretion by management). Second, the revelation principle is violated because of shareholders' imperfect rationalities especially under conditions of market uncertainty, which makes them unable to take wealth maximizing decisions all the time. The third condition of the revelation principle is violated when there is no common knowledge, in the form of a management information system, which is well known to shareholders and enables them to understand management actions. Finally, the revelation principle is violated when managers find that communicating with shareholders is costly due to the presence of third parties like regulators, competitors, and tax authorities. Any of the previous violations of the revelation principle results in aggravating the agency problem and is sufficient to give managers the chance to manage earnings (Walker, 2013).

Although the revelation principle provides a solid explanation for the motives of earnings management, it does not take into consideration the cost variation of earnings management activities. Therefore, the revelation principle cannot explain the different motivations of management to use more than one activity to manage earnings at a time or to trade off these activities based on their consequences. To improve the quality of research, the cost of each activity has to be considered simultaneously based on the four previous conditions to explain earnings management behaviour.

The previous four conditions of the revelation principle are based on three major aspects of the economic theory that include contracting, bounded rationalities and information asymmetry. Each of these theories is going to be discussed in detail in this section. Then, I show how earnings management theories have been developed from these three theories in the following section. The overall flow of the theory discussion in this thesis is displayed in Figure 3.1.





3.4.1 Contracting Theory

Contracting theory views the firm as a nexus of contracts that identify the rights and obligations of the different parties involved in relations with the firm (Fama, 1980). The need for contracts rises because of the uncertainty in situations of information asymmetry or complete absence of information, which lead to variation in the costs and benefits of information to decision makers. Furthermore, the different attitudes towards risk by the different decision makers lead to an incentive problem, which is referred to as a moral hazard. The situation of uncertainty together with the incentive problem can

be mitigated by an optimal structure of contracts between the firm and the other parties that allows for the optimal sharing of risk (Harris and Raviv, 1979).

The written contracts attempt to solve the conflict of interests that mainly appears in the relations between the firm and its executives on the one hand and the firm and its creditors on the other hand. In this sense, two main types of contracts are identified: reward contracts between the firm and its executives that separate between the control role of management and the risk bearing role of the owners, and financial contracts between the firm and its creditors that specify their subsequent rights and obligations upon borrowing and lending (Walker, 2013). Each party is assumed to possess sufficient rationality and thus uses logic and experience in collecting information and evaluating its costs and benefits when signing a contract (Salop, 2015).

In line with the previous discussion about the role of managerial incentives in section 3.3, the reward contracts represent one of the most common ways to mitigate the agency problem by motivating managers to improve shareholders' wealth through their compensation (Garen, 1994). Reward contracts assume a situation of uncertainty due to the imperfect information about the agents' performance, which makes it difficult for the principal to observe the agent. Therefore, high managerial incentives may improve the communications between managers and shareholders leading to less false information (Strong and Walker, 1987). An optimal reward contract involves a fixed payoff to the principal and a residual payoff to the agent. In this sense, the utility functions of both parties can be improved by sharing risk; hence solving moral hazard problem.

Managerial incentives in the reward contracts depend on managers' performance, which is likely to be determined by their abilities to make better decisions and take more risk (Baranchuk et al., 2011). Managers' performance is evaluated using different kinds of measures which mainly include the security market performance measures (e.g., security market returns) and accounting measures (e.g., return on equity). The informational properties of these measures play an important role in determining managers' compensation in their contracts with the firms. The less the noise and the higher the sensitivity of the measures to managerial actions, the more reliable they are in establishing the compensation contracts. The informational properties of the performance measures change over time and thus time plays a role in determining the reward contracts. According to Harris and Holmstom (1982), firms prefer to be engaged in full insurance contracts (e.g., fixed wages) with their managers in the short run because of information uncertainty about their abilities. The uncertainty arises in a single period agency model because the nature of managers' actions would lead to ambiguity in evaluating their efforts and thus ending with imperfect information about their performance (Lambert and Larcker, 1987). However, in the long run, firms can assess the labour marginal product of their mangers because the performance measures of prior periods, referred to as memory, provide more information about management performance in a multi period agency model (Lambert and Larcker, 1987). The better information raises the need for partial insurance contracts (e.g., variable wages) that help companies in avoiding inter-firm mobility of their managers, especially the high quality ones who might be targeted by competitors (Hayes and Schaefer, 1999).

In addition to taking the firm's own performance into account, reward contracts also rely on the relative performance evaluation of a manager compared to others in the industry. The strategic interaction with the competing firms, however, contributes to decreasing the relative performance-based incentives. In this sense, the relative performance evaluation contract helps in softening the competition among firms by decreasing managers' exposure to risk while increasing the probability of overall lower industry returns. In contrast, the strategic interaction between firms in highly competitive markets becomes more difficult and thus compensation is less based on the relative performance evaluation. Therefore, competition leads to more managerial efforts to improve firm efficiency thus mitigating the principal-agent problem (Aggarwal and Samwick, 1999).

The nature of the contracts, in turn, influences management performance. Contracts which involve fixed rewards drive less managerial efforts because mangers have low incentives to improve performance. Thus the principal will carry all the risk; which may result in low payoff to the principal. On the other hand, forcing contracts lead to more managerial efforts as they involve high monitoring to the agent's performance. They ultimately lead to high payoff to the principal but cannot be considered a practical solution for the agency problem as they assume the availability of

69

perfect information that always allows the principals to observe the agent (Strong and Walker, 1987).

However, contracts are incomplete as they cannot cover all the rights and obligations for all parties involved in the current and all possible future conditions. Therefore, contracting always involves some costs like the contract structuring, contract monitoring and contract bonding, in addition to the residual loss of output as the cost of contract's enforcement may exceed its benefit (Strong and Walker, 1987). Contracts also involve the costs of thinking, negotiating, writing, and renegotiating contingencies that are expected to happen in the long run based on the changes in economic conditions (Hart, 1995).

Due to the previous contracting costs, the hold-up problem arises when the firm gets engaged in long-term agreements with other parties who may not fulfil their duties in the future regarding price or quality and thus use their power to bargain. The hold-up problem is also influenced by the bounded rationalities of the parties in a contract that render them unable to carefully evaluate the utility consequences of their actions, and ultimately create the need for renegotiating the contract. Similarly, the changes in information asymmetry may affect the hold-up problem. Consequently, the hold-up problem may result in changing firms' decisions when signing contracts.

A number of ideas have been suggested to solve the hold-up problem. Competition can mitigate such problem because any party can switch to another supplier/client when it realizes that it is bounded with inappropriate terms (Hart, 1995). The allocation of the rights to control, according to the property rights theory, may also solve the hold-up problem as it specifies some of the rights that are usually ignored in the contracts based on a clear definition of the firm boundaries. In this sense, defining the firm as a legal fiction with a group of contracting relations makes it difficult to distinguish between the relations inside and outside of the firm and thus may mix the behaviour of the firm with the behaviour of the market (Jensen and Meckling, 1976). Furthermore, developing contingent contracts that specify all possible future conditions or the use of cost/revenue sharing contracts among the parties of the contract sound impractical solutions to the hold-up problem.

Overall, contacts have been developed to solve the conflict of interests related to the information asymmetry and moral hazard. Among the number of contracts that a firm might sign, optimal reward contracts have been emphasized to encourage managers to improve productivity by sharing risk. The relation between the reward contracts and management performance is mutual – while management performance is an important determinant of the reward contracts' structure, the nature of the contract shapes management performance as well. Because contracts are incomplete, contracting costs arise between the firm and its management and, therefore, encourage the use of simplified general contracts that give management the chance to hide information from the stakeholders (Walker, 2013).

3.4.2 Bounded Rationality

According to the agency theory, individuals exhibit rational behaviour and, therefore, utilize an expected utility approach in decision making by ranking the alternative actions according to their expected outcomes and ultimately selecting the best alternative that maximizes their objective utility functions. The expected utility approach assumes that rational individuals have a minimum level of consistency in their choices and will ignore any situation with equal consequences when selecting the optimum action. It also assumes that tastes are independent of beliefs and vice versa (Strong and Walker, 1987).

The concept of rationality is strongly related to the ideas of individualism and equilibrium. Individualism underlies the selective behaviour of the individuals to maximize their objective functions subject to their own constraints. Meanwhile, rationality is complementary with the equilibrium situation within the group, so that all the individuals' choices are consistent and implementable under the availability of different information to each individual in the group (Strong and Walker, 1987).

Based on the previous argument, all stakeholders take into consideration the possibility of management opportunistic behaviour and factor its costs opposite to its expected benefits. In this sense, the rational abilities of the shareholders can offset the high costs of contracting like thinking, negotiating, renegotiating, and writing; hence drive efficient contracting and mitigate opportunistic behaviour (Walker, 2013).

However, individuals suffer from cognitive limitations and partial mental capabilities to evaluate the different events in the exogenous environment (Strong and

Walker, 1987). Furthermore, decision makers may not obtain all information they require about the available alternatives. Therefore, under the need to make the decision of selecting only one of the alternatives, they are likely to face a situation of uncertainty. In this sense, individuals' complete rationality is questionable and they are more likely to possess bounded rationality instead.

The bounded rationalities of the shareholders and their inabilities to completely understand management actions create a weak form of market efficiency where some of these actions are not reflected in the share prices (Merkl-Davies and Brennan, 2007). The bounded rationalities also contribute to developing incomplete contracts because the shareholders cannot foresee all the contingencies, determine the course of action for each contingency, or abide by the contract terms as they may need to renegotiate them in the future. Under such conditions, managers may behave opportunistically through imperfect commitment, hold-up problem, and signalling private information to achieve personal benefits (Milgrom and Roberts, 1992). Consequently, they may use their judgement to alter the financial statements and manipulate the reported earnings (Ronen and Yaari, 2008).

To summarize this part of the theory, an ideal case might be assumed if the individuals act rationally in order to maximize their utility functions based on their individual needs while maintaining equilibrium with the other players in the market. However, in reality individuals suffer from bounded rationalities due to the limitations in their cognitive and mental capabilities. Consequently, the bounded rationalities of the different stakeholders contribute to a weak form of market efficiency and creating incomplete contracts that may give management the chance to avoid revealing all information to the stakeholders.

3.4.3 Information Asymmetry

Information plays an essential role in the capital market functioning as it contributes to minimizing the levels of uncertainty by assisting to detect or forecast the different possible situations in the market. Accordingly, information allows the decision maker to take an action strategy in the form of a combination of different choices that increase his/her expected utility function. Meanwhile, the effect of information on the utility function is based on the individual's ability to utilize that information and revise his/her action according to information available. In this sense, information contributes to the welfare of the society by improving the decisions of the different parties in the market.

Information efficiency (IE) exists in ideal markets that are characterized by perfect competition where no individual has power over information and thus all individuals receive the same input. In such an environment, market prices fully reflect and react instantaneously to the information available in the efficient market and, therefore, a fully revealing rational expectations equilibrium (FRREE) is achieved (Strong and Walker, 1987).

However, markets cannot be considered completely efficient as management prepares the financial statements that shareholders have to rely on for part of their information. The resulting information asymmetry can take one of the two forms: moral hazard and adverse selection. Moral hazard happens because shareholders do not possess the information to monitor management and assess whether it works for maximizing their firms' values; resulting in different attitudes and motivations for each party. On the other hand, adverse selection happens because managers have access to private information that allows them to make decisions for their own benefits and ignore shareholders' value (Walker, 2013). Therefore, a condition of imperfect information in the market can be assumed where information is neither absent nor complete.

Under such uncertainty, shareholders who are at different levels of sophistication will not be able to make their decisions based on rationality. Meanwhile, the uncertainty gives managers the chance to obfuscate their failures or send the messages they wish to deliver to the outsiders. Accordingly, signaling and screening take place when there is informational asymmetry between the insiders (e.g., management) and the outsiders (e.g., investors). While the insiders possess superior information, the outsiders are imperfectly informed about product quality and firm performance. Accordingly, the outsiders pay the prices that reflect the quality they perceive about the firm and its products; which drives the insiders to offer different qualities to the different outsiders (Walker, 2013).

Signaling occurs when the market does not distinguish between the performance of a good firm and that of a bad one. Investors would consider reporting bad financial results as a consequence of the auditors' efforts and thus would take their value with high confidence. On the other hand, they might consider reporting good financial results as a consequence of audit failure and thus discount the auditee's price. In this sense, the price of a good performer will be understated and that of a bad performer will be overstated. This will stimulate good firms to signal messages to the market based on a self-selection principle (e.g., paying dividends) to inform the outsiders about the quality of their products and thus improve their prices. The above informational asymmetry can result in high quality firms to withdraw from the market as the outsiders do not distinguish the quality provided and thus do not pay the expected prices. The overall result is lower quality and prices in the market as the poorest quality products remain, ending up in a market break down. However, if good firms are more, the higher prices in the market would drive bad firms to manage earnings in order to signal better performance (Ronen and Yaari, 2008).

Signals can be non-dissipative when the outsiders have the ability to observe the insiders' quality subsequent to a contingent contract. Here, both parties share the risk, thus signalling cost is inversely proportional with quality. On the other hand, signals can be dissipative if the outsiders cannot observe the insiders' quality and thus only the outsiders carry the risk.

The most common example of signalling is firm's capital structure that reflects management choice of debt and equity and thus can be used for signalling firm value. In this sense, management choses capital structure that maximizes its expected utility and thus takes into consideration both the increase in firm value and the risk of bankruptcy when deciding on the signalling equilibrium. Another example of signalling is firm's dividends policy that reflects firm profitability and the investment and financing plans for the future. Once more, management choses dividends policy that maximizes its expected utility and takes into consideration both the increase in firm value and the subsequent adjustment penalty when deciding on the signalling equilibrium.

On the other hand, screening is another consequence of the information asymmetry between insiders and outsiders. The less information available to the outsiders creates an incentive for them to sort firms of different quality products. A common example is the credit rationing by banks where they sort firms by using loan agreements according to their credit risk. With higher firm risk, banks charge higher interest and request more securities to maximize their returns and offset any risk in case

74

a firm's projects fail to achieve the required returns. However, under informational asymmetry, banks attempt to charge optimal interest rates in order to balance between risk and return, thus they avoid the extremely risky or safe clients (Strong and Walker, 1987).

Information asymmetry and the resulting signalling and screening may influence the contracting between the firms and the managers. In situations where information is available, individual managerial reward schemes are optimal as managers' efforts can be easily evaluated. However, with imperfect information, relative managerial performance reward schemes are expected as they provide more chances for adjustment according to the uncertainty in the environment. In this case, the optimal reward structure includes an incentive level (e.g., basic salary), a risk component (e.g., bonus), and some flexibility based on the level of uncertainty in the environment (Nalebuff and Stiglitz, 1983).

Overall, information theory assumes an ideal case of information efficiency where all players in the market are equally informed and, thus, prices fully reflect market information. However, incomplete efficiency exists in reality because of the information asymmetry between the firms and the stakeholders. The resulting uncertainty drives the good performers to signal information that emphasizes their quality and bad performers to signal information that enables them to stay comparable to others. On the other hand, the uncertainty drives the stakeholders to screen information in order to make their decisions. The interaction between information signalling and screening determines the market utility equilibrium.

3.5 Earnings Management Theories

The theories of contracting, bounded rationality, and information asymmetry discussed above have contributed to developing three main approaches to explain earnings management behaviour according to Figure 3.1. In all of these approaches, I focus on the conflict of interests between shareholders and management while excluding other stakeholders, e.g., lenders, employees, etc. This is because managers generally attempt to meet the expectations of other stakeholders as far as they are in line with shareholders' interests. The three approaches are explained in the following sections.

3.5.1 The Costly Contracting Approach

The costly contracting approach supports the contractual view of the firm emphasized in the positive accounting theory. It implies that contracting is the major motive for earnings management behaviour because of its high costs. Contracts between shareholders and managers attempt to solve the conflict of interests between the two parties who behave rationally to achieve their own benefits. However, individuals cannot expect all contingencies in the future because of their bounded rationalities. As a result, contracts are imperfectly adjustable because they are incomplete; and individuals may exhibit opportunistic behaviour to fulfil the conditions of the contract when an unforeseen contingency occurs.

The costs of contracting and renegotiating contracts, however, can be avoided by using some accounting measures. Earnings are usually used as performance measures by shareholders to monitor management behaviour; thus mitigate the conflict of interests. Nevertheless, the costly contracting approach assumes that shareholders have full knowledge of the underlying economic conditions of the market and, therefore, there is no informational asymmetry between shareholders and managers. In this sense, contracts are expected to adjust optimally to economic changes; hence accounting is considered irrelevant and earnings do not have an intrinsic value. Therefore, opportunistic behaviour applies only to contracts that are un-adjustable to economic conditions or that are costly to renegotiate; which creates a limitation to the costly contracting approach.

Accordingly, earnings management in the costly contracting approach can be defined as an opportunistic behaviour in choosing an accounting treatment to achieve specific target numbers in firms' contracts; which results in the deterioration of firm's value. If this behaviour, however, leads to the maximization of firm's value, it is considered economically efficient rather than opportunistic (Ronen and Yaari, 2008).

3.5.2 The Decision-Making Approach

The decision-making approach focuses on the decision makers who design the contracts in order to achieve their interests, and emphasizes the value of earnings in decision making. Thus, the approach assumes that a firm is an interaction of decision makers. According to the game theory, decision makers are rational and set their preferences as ranked alternatives in the form of utility functions, while each decision maker aims to maximize his/her own expected utility (EU).

The maximization of the individual's expected utility depends on the decision taken by other decision makers to establish an equilibrium situation. Achieving the equilibrium relies on information to make decisions that require estimating future earnings or require assessing future risk. Information comes in the form of earnings and accounting numbers; which create a common knowledge structure for decision makers. The decision-making approach, however, assumes that decision makers are not fully informed about the underlying economic events and thus emphasizes the demand for information. In this sense, the different decision makers cannot perfectly predict management opportunistic behaviour.

Therefore, even if the financial statements provide useful information to shareholders, they would not make a difference unless they are relevant in decision making and have economic consequences. As shareholders are considered rational, the opportunistic behaviour takes place with an explicit or implicit consent from the shareholders. Earnings management would be good if it has no effect on decision making. The opposite applies if earnings management is bad and, therefore, the harmed parties will take actions to prevent earnings management as far as the benefits of these actions are greater than their costs (Ronen and Yaari, 2008).

3.5.3 The Legal-Political Approach

The legal-political approach emphasizes the current legal systems in the US and the UK. It considers the weaknesses in shareholders' rights as the main source of earnings management. Compared to the previous approach, it assumes that the shareholders are neither powerful nor knowledgeable to make decisions and, therefore, they cannot design complete contracts to maximize their utilities.

Although shareholders possess the rights to firm assets and cash flows, they do not have high power over management performance or the major decisions of the firm, e.g., dividend payments or the power over directors' performance. In contrast, management possesses the right to make decisions for managing assets and, therefore, earnings management arises when shareholders cannot effectively direct management especially with the presence of poor governance systems. In this sense, the legal-political approach emphasizes the importance of improving the legal systems at the levels of investor protection, ownership structure, and legal enforcement in order to prevent earnings management. One of the most prominent regulations in the modern accounting scene is the Sarbanes-Oxley Act (SOX) of 2002 which had a noticeable influence on improving the governance systems. Although such a regulation comes at a cost in the form of investment in governance systems, it provides benefits in the form of a decline in earnings management behaviour which, in turn, contributes to improving the quality of financial reporting and restoring public confidence in the market (Ronen and Yaari, 2008).

In addition, the legal political approach assumes that shareholders are not knowledgeable of the economic conditions and, therefore, a situation of information asymmetry exists. Accordingly, they cannot direct management with the limited amount of information provided to them; hence management opportunistic behaviour becomes more likely. In this sense, shareholders are considered ignorant rather than rational; and accounting information would be more valuable.

According to the legal-political approach, earnings management can have good and bad implications. In the good sense, earnings management is a way to please the shareholders by providing summary information about the firm which, in turn, enhances firm value. In the bad sense, earnings management results from moral hazard, poor governance, and the shareholders' ignorance, thus devastating the value of the firm (Ronen and Yaari, 2008).

In summary, each of the previous approaches has explained earnings management behaviour by emphasizing a major cause of earnings management while ignoring the others. The costly contracting approach emphasizes the contracting costs, the decision-making approach focuses on the availability of information for decision making, and the legal-political approache pays more attention to the lack of the legal power for shareholders to monitor management performance. However, the previous approaches are based on some invalid assumptions e.g., the symmetry of information in the costly contracting approach and the rationality of shareholders in the decision-making approach. The legal-political approach seems to be closer to reality as it takes shareholders' lack of knowledge and power into consideration. Consequently, to obtain a comprehensive theory of earnings management, it has to be viewed as an interaction between the nature of contracts, information asymmetry, and the individuals' bounded rationalities.

3.6 Conclusions

This chapter has established the theoretical framework for this thesis. Earnings management reflects the conflict of interests between the principal (shareholders) and the agent (managers), and thus the main theory that explains this phenomenon is the agency theory. Earnings management is the opposite of truth telling; hence there has to be a violation of the revelation principle for earnings management to occur. The violation may happen because of the costly contracting between the management and the shareholders based on contracting theory, the limitations in the abilities of the shareholders to understand management actions based on bounded rationality theory, or problems in communicating information to the shareholders based on information asymmetry theory.

There are no distinct boundaries between the previous theories as they might interact to determine earnings management behaviour. While complete contracting allows the shareholders to overcome some of their cognitive limitations in making decisions, the shareholders' bounded rationalities contribute to establishing incomplete contracts. Similarly, information enables shareholders to improve their decisions but selecting the appropriate alternative depends on the abilities of the shareholders to understand and use information. Finally, while the degree of contract completeness influences the level of information asymmetry, the available information contributes to determining the structure of the contracts. The previous relations are depicted in Figure 3.2 below.



Figure 3.2 The interaction between earnings management theories

All these theories have contributed to establishing the three earnings management approaches that include the costly contracting approach, the decision-making approach and the legal-political approach. However, while the first two approaches focus on a specific aspect of the theory, the third approach is more comprehensive in considering the different parts of the relevant theory when explaining earnings management behaviour. Furthermore, from the previous three theories, researchers have derived three main groups of motives that include the contracting motives, the capital market motives, and the third-party motives. The next chapter is going to discuss these motives in detail.

Chapter 4 Motives of Earnings Management

4.1 Introduction

Based on the discussion in the previous chapter, the three main theories that explain earnings management suggest three main groups of motives for this phenomenon as shown in Figure 4.1. According to contracting theory, contractual motives exist based on the contract terms between the firm and its stakeholders that are linked to firm performance. Bounded rationality theory implies the inefficiency of the market and thus suggests the existence of capital market motives that influence firms' stock values. Finally, external (third-party) motives arise from parties that have current or future interests in the firm and thus interfere in the way it communicates information to the stakeholders (Ronen and Yaari, 2008; Walker, 2013). Under each of the previous three groups, different motives have been identified in the literature. They are summarized in Figure 4.2 and will be explained in detail in the subsequent sections.



Figure 4.1 The relation between the theoretical framework and earnings management motives

Contracting	Capital Market	External
Motives	Motives	Motives
Management compensation	Stock market	Industry
CEO turnover	Issuance of equity	Industrial
Managerial ability	New listing and cross-listing	diversification
Corporate governance	Mergers and acquisitions	Regulations
Loans	Insider trading	Political environment and country-specific policies
Firm characteristics	Management buyouts	
	Meeting or beating a benchmark Analysts	Accounting standards
		Tax considerations
		Competitors
		Suppliers and
		customers

Figure 4.2 Earnings management motives

The remainder of this chapter is organized as follows. Section 4.2 compares between earnings management and truth telling. Section 4.3 explains the contracting motives of earnings management. Section 4.4 presents the capital market motives of earnings management. Section 4.5 discusses the third-party motives of earnings management. Finally, section 4.6 concludes the chapter.

4.2 Earnings Management vs. Truth Telling

Earnings management is the opposite of truth telling and, therefore, it sounds helpful to learn about telling the truth first in order to understand the motives of earnings management later. Managers may have incentives to avoid earnings management when they expect more benefits and less costs subsequent to their true earnings' disclosures (Verrecchia, 1983; Hellman, 1999). Meanwhile, the stakeholders have to be able to verify management disclosures and make decisions that protect their interests (Dye, 1985).

Firms also tend to report the truth and communicate their private information in situations of separating equilibrium. This happens when the good firms in the market are less than the bad ones so that the market will not be able to distinguish the difference in performance especially under the imperfect audit. In this case, a good-performing firm tends to separate itself from bad-performing firms by incurring some costs e.g., paying dividends. In this sense, the stock price of the good firm after incurring such cost will be higher than its price when the market does not distinguish its performance. On the other hand, the stock price of a bad firm that does not incur any cost is already higher than its expected price even if it does not manipulate its earnings. Under such equilibrium, both types of firms are motivated to report the truth (Ronen and Yaari, 2008).

Furthermore, truth revelation happens in case of signal jamming equilibrium when the market knows and understands the signal properly (Stein, 1989). In this case, if the firm starts by manipulating earnings, the market follows by discounting the firm price with the same amount of the inflated earnings (Elitzur, 1995). Although the situation involves earnings management, firms are motivated to reveal the truth and thus they do not mislead the stakeholders. In this sense, equilibrium may occur when firms use mixed strategies between truth reporting and misrepresentation depending on the cost and benefit of each strategy (Ronen et al., 2006).

According to the revelation principle, a manager is motivated to tell the truth when the compensation benefit from revealing the bad news is the same or higher than his/her expected outcome when misrepresenting earnings to communicate better news (Milgrom and Roberts, 1992). Even if they are expecting to extract rents from earnings management, managers tend to avoid the subsequent penalty of misrepresenting the financial reports. This equilibrium is likely to happen under the presence of auditors who may reveal earnings management as far as the shareholders themselves cannot monitor management actions (Ronen and Yaari, 2008).

In addition, firms prefer to tell the truth to avoid losing credibility in the market when the misrepresentation is detected subsequently. Even when not detected, such misrepresentation may drive third parties to react to the communicated information in an unfavourable manner, e.g., the aggressive response of a competitor (Ronen and Yaari, 2008). Furthermore, the cooperation with third parties is not always possible to enable a firm to manage its earnings, e.g., for a supplier to manage earnings, it needs the cooperation of the buying firms by purchasing goods in specific quantities or at specific prices. In other instances, management might get motivated to tell the truth because of the conflicting interests between the external parties it has to deal with, e.g., a firm might choose to manage earnings upwards to discourage new firms from entering the market while it may need to manage earnings downwards in face of the labour unions' demands. Finally, mandatory regulations may enhance truth telling, e.g., the audit-related rules imposed by the Sarbanes Oxley Act (SOX) and the Public Company Accounting Oversight Board (PCAOB). In response to such regulations, management may tell the truth when it expects higher pay-off from its good performance compared to the penalties it might face from violating the regulation through managing earnings (Ronen and Yaari, 2008).

Overall, truth telling is dependent on management perception of its benefits and costs. In this sense, management is likely to avoid earnings management when it expects more benefits from telling the truth, e.g., better compensation, higher stock prices, superior credibility in the market, milder third party reactions, or less regulatory penalties. In contrast, earnings management occurs when management does not expect high benefits from telling the truth under the three groups of motives identified earlier and which are going to be discussed in the next three sections of this chapter.

4.3 Contracting Motives of Earnings Management

As discussed in the previous chapter, the principal-agent relationship involves a conflict of interests between management and shareholders. As shareholders cannot monitor management performance, they elect a board of directors to act as an agent to the shareholders and a principal to the management. However, managers are averse to work and risk and thus incentive contracts are designed to mitigate this conflict. Similarly, conflicts in interests may appear in the contracts between the firm and the other stakeholders. As a result, factors incorporated in the internal or external contracting of the firm are expected to influence the motivation to manage earnings (Ronen and Yaari, 2008).

4.3.1 Management Compensation

If management objectives were aligned with shareholders' objectives, there would not be any incentive for earnings management. However, the objectives of management and shareholders in reality are different because they may want to emphasize different horizons when making their decisions. Shareholders and managers also carry different risks – while the shareholders bear the risk of their equity in the firm, managers are generally risk averse because the risk involved with their human capital is not diversifiable. As a result, managers enjoy benefits without bearing any risk; which creates an agency cost against shareholders' desire.

To avoid the previous cost, shareholders design compensation schemes that align their interests with those of the management (Laux and Laux, 2009). Therefore, they link management compensation to shareholders' equity using equity-based compensation, e.g., stocks and options that change the risk-taking behaviour of managers. Even though, compensation may still drive managers to manage earnings because their incentives depend on these earnings. In addition, managers enjoy limited liability and can influence the decision of compensation which is made by the board of directors and not directly by the shareholders. Finally, mangers tend to sell their stocks and options immediately compared to shareholders who aim to keep them for the longrun (Ronen and Yaari, 2008). Under such circumstances, management opportunistic behaviour is still probable even if they are compensated with valuable rewards.

Management compensation is determined by some measures that provide signals to the shareholders about management performance (Healy, 1985; Jensen and Murphy, 1990; Dechow and Huson, 1994). Earnings is one of the commonly used measures that reflect the effort made by management during the current accounting period. However, they may not reflect some managerial decisions that influence future cash flows (Basu, 1997; Barclay et al., 2005).

Conventionally, management compensation includes a fixed payment in the form of salaries. Performance-based payments in the form of bonuses, stock grants, and stock options has been gradually incorporated in management compensation packages recently (Perry and Zenner, 2001; Balsam et al., 2003; Jensen et al., 2004). Opposite to the performance-based payments, salaries are generally not perceived as incentives of earnings management (Gao and Shrieves, 2002).

As bonuses create an incentive to manage earnings, managers try to keep the earnings between the minimum and maximum limits set by the board of directors to get eligible to receive bonuses (Healy, 1985). Evidence of discretionary accruals is, however, obtained when earnings fall below the minimum limit (Gaver et al., 1995; Holthausen et al., 1995). Gao and Shrieves (2002) also confirm the positive effect of bonuses on discretionary accruals. However, long-term bonus plans seem to mitigate earnings management (Richardson and Waegelein, 2002).

Equity-based compensation comes in the form of stocks and options. Stock compensation can take the shape of stock grants, restricted stock grants², phantom stocks and stock appreciation rights³, and performance units⁴ (Ronen and Yaari, 2008). Options are instruments granted to the managers to exercise after paying certain fees. Managers only exercise their options when the share price becomes more than the exercise price. In this sense, options are granted to the managers to motivate them to improve firm value by making benefit from the increment between the stock price and the exercise price (Coles et al., 2006; Ronen and Yaari, 2008).

Equity-based compensation aims to increase a manager's utility with the increase in the firm's equity value. Nevertheless, the role of equity-based incentives in aligning the interests of managers and shareholders, and hence their effect on earnings management, are controversial. In the short run, equity-based compensation is expected to stimulate earnings management, so that managers can cultivate the benefits of an

 $^{^{2}}$ Restricted stock grants are stocks granted to a manager with restriction on their sale. The manager can only sell this type of stocks when a specific condition is met, e.g., after a specific period of time passes or a specific goal is achieved.

³ Phantom stocks and stock appreciation rights link between management compensation and stock prices by rewarding managers in cash, and to a lesser extent by distributing stocks. The two types of incentives mainly differ in terms of settlement dates and dividends payments.

⁴ Performance units give a reward promise to a manager in the form of stocks if the firm achieves a specific level of performance.

increase in stock prices. Meanwhile, the high stock values of the current year increase shareholders' expectations in the long run and thus may drive managers to avoid earnings management; a phenomenon referred to as the ratchet effect (O'Connell, 2004).

In support to the first point of view, Gao and Shrieves (2002), Cheng and Warfield (2005), Bergstresser and Philippon (2006), and Feng et al. (2011) find a positive effect of equity-based incentives on discretionary accruals, so that managers can show better performance to the outsiders and meet analysts' expectations. Burns and Kedia (2006) document that options are the only component of management compensation which is positively related to accounting restatements because the expected increase in management's wealth from exercising options overcomes any punishments that may occur when earnings management is detected. Options also give a manager a chance to pool with other managers in order to minimize the market reaction to bad information. Cohen et al. (2008) also find a positive effect of management's unexercised options on accruals' manipulation, especially in the period preceding the release of the Sarbanes-Oxley act.

In line with the second point of view, Bauman et al. (2005) document that management compensation in the form of stock options drive managers to implement guidance to analysts in order to meet their forecasts. In this sense, managers seem to guide analysts by reducing their expectations rather than manipulating accruals. Coles et al. (2006) document that before the issue of options, managers tend to avoid discretionary accruals in order to keep stock prices as low as possible and, therefore, decrease the exercise price which is determined at the grant date. Managers may even make use of income-decreasing discretionary accruals to achieve the previous target (Baker et al., 2003; Balsam et al., 2003).

4.3.2 CEO Turnover

As the turnover process involves the predecessor CEO and the successor CEO, it is essential to understand the differences in the incentives of each of them to manage earnings. The predecessor CEO can get more benefits from inflating earnings, so that he/she hides any bad performance, extracts more earnings-based benefits in his/her last year onboard, and creates better chances for moving to another job (Hazarika et al., 2012). On the other hand, the successor CEO gets more benefits from deflating earnings, usually by taking a big bath, so that he/she decreases shareholders' expectations that allow reflecting better future performance (Geiger and North, 2006; Ronen and Yaari, 2008). In this sense, the successor CEO establishes low benchmarks for his/her performance in the future based on the current year's bad results attributed to the poor performance of the predecessor CEO, especially if the later is not moving to serve on the board of directors (Murphy and Zimmerman, 1993; Pourciau, 1993; DeFond and Jiambalvo, 1994; Godfrey et al., 2003; Florou and Conyon, 2004).

The non-routine CEO departure is generally associated with poor performance compared to the routine (peaceful) departure. In such case, shareholders tend to punish management for making bad decisions by replacement and thus the CEO is suddenly forced to leave the firm because of the negative implications of performance on the stock price (Warner et al., 1988; Brickley, 2003; Lehn and Zhao, 2006). The frequency of senior management turnover is higher in distressed firms that suffer from default on debt, bankruptcy, or debt covenant violations (Gilson, 1989; DeFond and Jiambalvo, 1994). As a result, poorly performing managers may manipulate earnings upwards to postpone the announcement of bad performance and, thus, avoid being ousted as long as possible (Fudenberg and Tirole, 1995). In this sense, poor performance is perceived as a motive of management discretionary behaviour and, therefore, drives higher management turnover (Murphy and Zimmerman, 1993; Desai et al., 2006). In contrast, Pourciau (1993) document that departing managers manage earnings downwards in their last years of service because of the high monitoring of the poorly performing firms.

Even in the case of peaceful termination, a retiring manager manages accruals upwards in the year of departure to improve the short-term performance of the firm, so that he/she can get higher bonus or a chance of employment in the board of directors (Reitenga and Tearney, 2003). Real earnings management is also evidenced as CEOs might decrease research and development (R&D) and capital expenditures during their last years of service (Dechow and Sloan, 1991; Florou and Conyon, 2004).

When the nature of the departure (routine vs. non-routine) is ignored, evidence of downward earnings management is observed in the year of CEO change (Godfrey et al., 2003). Ahmed et al. (2006) provide more specific results and document that the new managers tend to manage accruals upwards in the current year when future performance is expected to be bad, while they manage accruals downwards in the current year if the firm performance is expected to be good, so that they create some reserves to enable them to secure their jobs (Ahmed et al., 2006).

Overall, the CEO turnover is positively related to earnings management. Earnings management is more evidenced when managers are at the beginning of their tenure as they are more expected to leave their firms. Earnings management is also expected in firms with old managers as they are not highly concerned about their future in the market (Ronen and Yaari, 2008).

4.3.3 Managerial Ability

Different managers have dissimilar styles in running their firms as they vary in their accounting preferences, disclosure choices, tax positions, and selecting and implementing corporate policies and earnings management strategies (Bertrand and Schoar, 2003; Bamber et al., 2010; Dyreng et al., 2010; Ge et al., 2011; Graham et al., 2012). Those differences arise from managers' personal characteristics such as age, education, inborn capabilities, personalities, the tendency to take risk, individual beliefs, military service, early life experience, career background, communication, interpersonal, and execution skills (Bertrand and Schoar, 2003; Bamber et al., 2012; Kaplan et al., 2012). The overall result is different managerial abilities that contribute to the variation in performance among firms at the operating, investing, financial, and organizational levels (Bertrand and Schoar, 2003; Jian and lee, 2011; Malmendier et al., 2011).

Several measures have been introduced to quantify managerial ability such as firm size, past abnormal performance, compensation, tenure, media mentions, and manager fixed effects⁵ (Demerjian et al., 2012). However, they were always criticized for the high noise arising from the influence of other firm characteristics. In contrast, the measure of Demerjian et al. (2012) has been the most credible so far because it reflects

⁵ According to Bamber et al. (2010), the manager fixed effect is estimated as the residual from regressing management forecast characteristics (e.g., forecast frequency, precision, and bias) on the specific determinants of voluntary disclosure (e.g., change in EPS, R&D expenditure, market to book ratio, the quality of governance, market value of equity, number of analysts, litigation risk, etc.), after excluding the firm fixed effect and time influence.

the role of the management in enhancing firm efficiency after excluding the impact of other factors that may also interfere with firm performance (Demerjian et al., 2012). The new measure calculates the ability of a manager relative to others within the same industry and focuses on managerial operating and investing skills. It correlates well with the old measures of managerial ability mentioned earlier (Demerjian et al., 2012).

The measure of Demerjian et al. (2012) represents how efficiently managers generate revenues from the available firm resources. They divide the process of quantifying managerial ability into two steps. In the first step, they consider each firm as a decision-making unit (DMU) that converts inputs into outputs. Inputs include net property, plant, and equipment (PPE), net operating leases (OpsLease), net research and development expenditure (R&D), purchased goodwill (Goodwill), other intangible assets (OtherIntan), cost of inventory (CoGS), and selling, general, and administrative expenses (SG&A). Firm revenues (Sales) are the only output in the calculation of managerial ability. Subsequently, Demerjian et al. (2012) perform Data Envelopment Analysis (DEA) to calculate firm efficiency as shown in equation (4.1):

Firm Efficiency=
$$\frac{\sum_{i=1}^{s} u_i y_{ik}}{\sum_{j=1}^{m} v_j x_{jk}}$$
(4.1)

In equation (4.1), s represents the outputs; m represents the inputs; u represents the output weight; v represents the input weight; y represents the output quantity; x represents the input quantity; and k represents a number from 1 to n. To obtain the weights of inputs and outputs (u, v), Demerjian et al. (2012) use the optimization model that appears in equation (4.2). They then divide the firm efficiency scores they get from equation (4.1) by the highest number obtained in each industry, so that they get an efficiency measure ranging from zero to one.

$$max\theta = (u_1 Sales).(v_1 CoGS + v_2 SG \&A + v_3 PPE + v_4 OpsLease + v_5 R \&D + v_6 Good will + v_7 Other Intan)^{-1}$$
(4.2)

In the second step, Demerjian et al. (2012) exclude some firm specific characteristics from the firm efficiency score obtained from the previous step. They believe that such firm attributes enhance or impede the performance of managers. Therefore, they eliminate those characteristics in order to reach a score that purely represents managerial ability. For this purpose, they execute a Tobit regression model in equation
(4.3), so that the remaining residual (ε_i) is the ultimate measure of managerial ability. Finally, they take the decile ranks of these measures by industry-year to obtain better comparability and to avoid the effects of outliers.

Firm Efficiency_i=
$$\alpha+\beta_1 \ln(\text{Total Assets})_i+\beta_2 \text{Market Share}_i$$

+ $\beta_3 \text{Free Cash Flow Indicator}_i+\beta_4 \ln(\text{Age})_i$
+ $\beta_5 \text{Business Segment Concentration}_i$
+ $\beta_6 \text{Foreign Currency Indicator}_i+\text{Year}_i+\varepsilon_i$ (4.3)

Nevertheless, the Tobit model of Demerjian et al. (2012) does not eliminate all firm specific characteristics that influence firm efficiency. Some of the variables that are expected to interfere with the measure include corporate governance, firm auditor, ownership, and employee turnover. Finally, the measure mainly focuses on core business activities by incorporating the operating and investing aspects of the firm. However, it ignores other firm activities like the financing, social, and environmental aspects.

As discussed above, managerial ability influences firm performance and thus it is expected to have an impact on earnings management. Two plausible approaches are suggested in this regard. On the one hand, more able managers tend to avoid earnings manipulation in order to safeguard their precious reputations in the market (Demerjian et al., 2013b). Instead, they invest their high competencies in improving firm operations and mitigating financial distress (Demerjian et al., 2013a). In this sense, better managerial ability is associated with high earnings quality to allow the managers to maintain their reputation-based compensation premium; which is in line with the efficient contracting hypothesis (Bertrand and Schoar, 2003; Jian and Lee, 2011; Graham et al., 2012). On the other hand, superior managers may invest their high skills in using more earnings management with the purpose of maintaining their personal welfare. Managers' private benefits include increasing the value of their stock compensation and drawing the attention to their performance in the labour market; which is in line with the rent extraction hypothesis (Jian and Lee, 2011; Demerjian et al., 2013b).

Following the first explanation above, several positive consequences of managerial ability can be identified. Superior managers are expected to show fewer

subsequent financial restatements, less errors in accruals, better earnings persistence and enhanced mapping of accruals into operating cash flows (Demerjian et al., 2013a). Their role may even become more evident in the future as they are generally associated with enhanced subsequent business performance, better stock market responses to their decisions, and lower exposure to litigation (Jian and Lee, 2011; Kaplan et al., 2012; Demerjian et al., 2013a). Managerial ability also moderates the negative relationship between equity financing and future abnormal returns (Demerjian et al., 2012). Even if superior managers engage in earnings manipulation, they seem more likely to use the accrual activities rather than the costly real practices. Such consequences are attributed to the high knowledge of superior managers in their business conditions, more experience with market competitors, and higher capability to understand and implement accounting standards. Accordingly, better managers use their abilities to accomplish both their own benefits and maintain stakeholders' welfare at the same time (Graham et al., 2005; Demerjian et al., 2013b). In contrast, managerial ability may have some negative consequences in the future - although not frequently documented. According to Francis et al. (2008), more reputable managers do not necessarily produce better disclosures (Francis et al., 2008). On the contrary, earnings quality may decrease with the increase in managerial ability (Demerjian et al., 2013a). The contrasting consequences of managerial ability raise some dispute around how superior managers invest their talents and the purpose of their different decision.

4.3.4 Corporate Governance System

The studies that have examined the effect of the corporate governance on earnings management have mainly documented that the quality of corporate governance influences the quality of earnings. In this sense, strong governance mechanisms can generally decrease earnings management behaviour. However, some studies find that corporate governance can stimulate earnings management because managers are under more pressure to improve firm value (Siregar and Utama, 2008). In the following subsections, the major components of the governance system are discussed which include the ownership structure of the firm, board of directors, and the audit committee.

4.3.4.1 Ownership

The shareholders of a firm can be institutional owners or individual retail owners (Koh, 2003, 2007; Zang, 2012). Both types of owners use earnings information to make their decisions. However, the institutional owners are more sophisticated and, thus, more able to detect earnings management and beat the individual owners (Shang, 2003). Meanwhile, the investment horizons of the institutional shareholders influence earnings management behaviour. Short-term institutional owners may drive firms to manage earnings in order to buy shares at a lower value or sell their investments at a higher price (Lakonishok et al., 1991; Yu, 2008). On the other hand, long-term institutional owners have more influence on firm decisions because they hold their shares for more than a year. Therefore, they use earnings information to monitor management performance and prevent earnings management behaviour that might be discounted by the market. As a result, long-term institutional owners serve as gatekeepers that reduce the noise associated with earnings compared to the short-term institutional owners (Bushee, 1998; Koh, 2005; Srivardhan, 2009).

Based on their listing status, public firms are less inclined to manage earnings than private firms because of the monitoring by the stock market and the higher governance requirements (Prencipe and Bar-Yosef, 2011). However, some studies document the opposite because private firms are more closely monitored by their owners (Burgstahler et al., 2006). Therefore, the public listing of a firm cannot guarantee a better earnings quality as the stock market may sometimes create pressure on the listed firms to manage earnings (Jeong-Bon and Cheong, 2006; Givoly et al., 2010).

Furthermore, family ownership may stimulate earnings management because of the less visible performance to the public and the higher probability of collusion between the owners (Siregar and Utama, 2008; Jaggi et al., 2009; Haw et al., 2011; Prencipe and Bar-Yosef, 2011). Therefore, family-owned businesses are expected to receive less monitoring than other firms. However, the lower agency problem in such firms may mitigate earnings management (Dechun, 2006).

4.3.4.2 Board of Directors

The board of directors is the body charged in governance for maintaining the interests of the shareholders. To achieve the previous target, the board of directors monitors management using the available information (McAnally et al., 2008; Jaggi et al., 2009; Prencipe and Bar-Yosef, 2011). However, information is provided by management which makes the board's monitoring role difficult to be achieved. Furthermore, management may have some control over the appointment of the board members and thus both parties may collude in managing the earnings of the firm (Boone et al., 2004).

The emphasis of the Sarbanes-Oxley act (SOX) on the role of the board of directors in improving firm performance has resulted in an overall increase in board size in public companies (Linck et al., 2006). Although this increase can be beneficial in some aspects, it may have some negative consequences. On the bright side, a larger board may result in lower levels of earnings management because of the higher number of independent and more experienced directors who are more able to monitor management activities (Chtourou et al., 2001; Xie et al., 2003). On the dark side, larger board size may drive more earnings management because of the high cost and time required for information flow between the firm and the directors. It also results in a bigger free-rider problem where the directors tend to rely on each other due to the distributed responsibility (Lipton and Lorsch, 1992; Blair, 1995; Yermack, 1996). In this sense, the board of directors is less effective because it basically serves as part of the public firm's prestige to impress investors (Hermalin and Weisbach, 2003). In addition, the more diverse and conflicting demands of a larger board may impose more pressure on managers to manipulate earnings (Aggarwal and Nanda, 2004).

The effect of the board independence on earnings management is also debatable. On the one hand, more independent directors can be associated with better earnings quality because they do not have direct employment benefits to collude with management compared to the internal directors (Vancil, 1987; Blair, 1995; Klein, 2002; Peasnell et al., 2005). In this sense, independent directors are in a better position to monitor management and take critical decisions, e.g., the restructuring and layoffs, and thus serve in aligning the interests of shareholders and management (Weisbach, 1988; Perry and Shivadasani, 2005; Visvanathan, 2008). On the other hand, more independent directors can stimulate earnings management because they are less knowledgeable about

94

firm-specific operations; hence result in higher communication costs (Yermack, 2004). In addition, no pure independent board members exist because management frequently interferes in the nomination of those members (Monks and Minow, 2004).

Other board attributes, e.g., the multiple directorships, position-duality, number of board meeting, directors' equity holdings, age and tenure, and staggered boards, may also influence earnings management behaviour. Multiple directorships, in the form of a single director holding a similar position in the boards of other companies, contribute more experience to the firm and allows for more connections; hence result in better earnings quality (Conyon and Read, 2006). A position-duality happens when a board chairman holds the position of the CEO at the same time. As a result, he/she obtains more power and control over the firm that may allow for more earnings management (He et al., 2003). The number of board meetings indicates the efforts made by the directors in monitoring management performance which may prevent earnings management (Jiraporn et al., 2007). Long-term equity holdings of the directors may align their interests with shareholders, but short-term holdings may encourage the collusion of the directors with the management (Perry, 2000; Chtourou et al., 2001). Age and tenure of a board member indicate more experience and familiarity with firm transactions and thus enhance better monitoring. However, they result in closer relations with management that may allow collusion to occur (Niskanen, 2005). Finally, staggered boards appear when the directors are elected partially each year; resulting in better earnings quality because they do not suffer from the takeover risk (Zhao and Chen, 2008).

4.3.4.3 Audit Committee

The audit committee is part of the board of directors and plays an important role in corporate governance that has been emphasized in most of the regulations, e.g., SOX. Overall, the empirical evidence documents that a strong audit committee provides high earnings quality. However, different attributes of the audit committee contribute to determining earnings management behaviour. The independence of the committee's members, their financial and governance expertise, and the higher frequency of meetings contribute to mitigating earnings management. However, while the high tenure implies more experience, it may allow the collusion with management. Similarly,

the larger size may result in including more experienced members in the committee but higher cost of communication. The terms of the directors' equity holdings may also align their interests with the shareholders or drive them to collude with the management (Klein, 2002; DeFond and Francis, 2005; Zhang et al., 2006). On the other hand, Peasnell et al. (2005) document that the audit committee does not influence earnings management behaviour or even does not create any market response (Peasnell et al., 2005; Anderson et al., 2006). Finally, the high quality of the internal audit results in less earnings management (Prawitt et al., 2009).

4.3.5 Loans

Before taking a loan, firms might be motivated to manipulate earnings upward in order to avoid reporting low earnings (Iatridis and Kadorinis, 2009). This way, they aim to get a better evaluation by debtors about their credit worthiness and, consequently, receive debt at a lower cost according to the debt hypothesis (Das and Shroff, 2002). Firms may also manipulate earnings downward because entering a loan agreement means a long term commitment that requires firms to meet the debtors' expectations over time. Therefore, firms would be motivated to report conservatively in order to mitigate the conflict between shareholders and creditors; hence keep the expectations of creditors and shareholders as low as possible (Ahmed et al., 2002). On the other hand, debt contracts may include restrictions on how management reports earnings (Smith Jr, 1993). In this sense, firms may not be motivated to use earnings management before taking a loan (DeAngelo et al., 1994).

After taking a loan, the probability of earnings management depends on the financial health of the borrowing firm and the flexibility of the lender to renegotiate the debt (Zang, 2012). Financially distressed firms are more likely to manage earnings when there are difficulties in renegotiating debt according to the debt-covenant hypothesis (Hassabelnaby et al., 2005). As a result, the borrower may manage earnings upward to relax the debt covenants or manage earnings downward to influence the future renegotiation of the debt (Jaggi and Picheng, 2002).

4.3.6 Firm Characteristics

Firm attributes related to the contracting between the shareholders may also influence management behaviour. Although some studies document that the size of the firm has an impact on its earnings management behaviour (Lee and Choi, 2002), others document no influence (Siregar and Utama, 2008). It is difficult to distinguish between earnings management behaviour of small and large firms because both types of firms face pressure to manage earnings in order to maintain them at an optimal level (Moses, 1987; Albrecht and Richardson, 1990; Michelson et al., 1995; Lee and Choi, 2002). Earnings management is expected in small firms because they have less predictable operations and less diversified businesses. Similarly, managers of large firms have opportunities to manipulate earnings because of their huge number of transactions and complicated operations (Demerjian et al., 2013b). Another firm attribute that may influence earnings management is its operating cycle. Longer operating cycles increase the levels of uncertainty and therefore, the potential for earnings management (Dechow and Dichev, 2002). Finally, business complexity in the form of the number of business segments and the frequency of foreign transactions may drive managers to use more earnings management, and particularly real earnings management because it is less likely to be detected than discretionary revenue manipulation (Karuna et al., 2012).

4.4 Capital Market Motives of Earnings Management

This group of motives involves factors that drive earnings management through their impact on the firm's stock price. If the stock price fully reveals information about the firm's underlying economic value, then the capital market motives are not expected to have any influence on earnings management. However, because of the noisy nature of the capital markets, stock prices do not completely reflect the firms' values; hence earnings management is likely to occur. (Ronen and Yaari, 2008).

4.4.1 Stock Market

Earnings are associated with stock prices and, therefore, managers are motivated to manage earnings in order to improve stock prices and raise more capital (Lev, 1989;

Ramakrishnan and Thomas, 1998; Kothari, 2001; Gelb and Zarowin, 2002). As a result, earnings management creates a difference between the stock price of a firm and its fundamental economic value⁶ which is unknown to most of the shareholders (Ronen and Yaari, 2008; Beyer, 2009). The previous effect is evidenced in inefficient markets where discretionary accruals influence stock prices and, therefore, managers may opportunistically manipulate accruals to improve the capital market valuation of their firms (Guay et al., 1996; Subramanyam, 1996). Badertscher (2011) finds that with longer periods of overvaluation, more total earnings management is evidenced with a tendency to switch from the within-GAAP to the non-GAAP accounting discretion.

The market reaction to the reported earnings is measured by the earnings response coefficient (Feltham and Jinhan, 2000). In cases of unmanaged earnings, a linear earnings response coefficient with a unit slope is expected (Sankar, 1999). The reaction of the market to earnings management, however, depends on the ability of the market to detect earnings management which, in turn, relies on the firms' disclosures, investors' sophistication, and the market optimism (Shivakumar, 2000; Balsam et al., 2002; Coles et al., 2006). Although the chance that the market learns about earnings management attempts after the announcement of earnings always exists, detecting earnings management may sometimes take years (Karpoff et al., 2008). Furthermore, the market needs time to collect more information before it reacts to the manipulation. Earnings management may even go undetected and, thus, the market may over-value firms that manage their earnings upwards and undervalue those that manage their earnings downwards (Chambers, 1999). The tendency to detect earnings management increases with the sophistication of the investors and decreases in optimistic environments where the investors more readily believe the reported earnings (Daneshfar et al., 2009; Coffee, 2003; Roychowdhury, 2006).

If the market detects the manipulation, it negatively prices any upward earnings management and positively prices the downward attempts. The response of stock prices is expected to be less than a unit slope in the case of myopic earnings management that emphasizes the short-term horizons (Sankar, 1999). Similarly, Feltham and Jinhan (2000) find that the earnings response coefficient is lower with more noise associated in

⁶ The fundamental economic value is calculated as the discounted dividends expected to be distributed by the firm in the future.

communicating the earnings. While the market discounts any good news when it doubts the occurrence of earnings management, Ronen et al. (2003) document that the reaction to the negative earnings surprises is higher than that to the positive ones. Furthermore, investors consider firms that meet analysts' forecasts with zero or small earnings as a signal of earnings management and, therefore, punish management with a lower earnings response coefficient (Lin and Shih, 2006).

Management may also use earnings smoothing to positively influence their share prices and reduce their volatility. However, the market responds negatively to the lower quality of information when earnings smoothing is used opportunistically and thus it discounts them to end up in more volatile share prices. In this sense, the investors demonstrate a sophisticated response to earnings smoothing by incorporating their perceptions about the risk of the different management activities when determining shares' values (Markarian and Gill-de-Albornoz, 2012).

Overall, the results of the previous studies show that the response of the stock market to earnings management is debatable between the 'accruals anomaly' stream and the 'cost of capital' stream (Walker, 2013). According to the 'accruals anomaly' stream, accrual earnings management may allow managers to beat the market because it results in low correlation between accruals and cash flows and ultimately drives the mispricing of accruals by the stock market. In this sense, earnings management can be used as a method to positively influence market prices and ultimately improve stock returns (Sloan, 1996; Xie, 2001; Collins et al., 2003; Beneish and Nichols, 2005; Chan et al., 2006; Pincus et al., 2007; Soares and Stark, 2009). On the other hand, the 'cost of capital' stream is in line with the efficient markets hypothesis and assumes that markets are smart enough to discount earnings management. In this sense, the cost of capital depends on earnings quality i.e. accruals' quality is a priced risk factor. Although most of the studies in this regard have emphasized the efficient markets hypothesis (Francis et al., 2004; Francis et al., 2005; Lambert et al., 2007; Francis et al., 2008; Ogneva, 2012), other studies document insignificant results (Core et al., 2008; McInnis, 2010).

4.4.2 Issuance of Equity

Initial public offerings (IPO) and seasoned equity offerings (SEO) can create incentives for managers to manipulate earnings. Earnings at IPOs are important because some investors may request earnings information before purchasing stocks and subsequently will compare firms' future performance to the IPO earnings (Ronen and Yaari, 2008). At the same time, IPOs are associated with high information asymmetry between management and shareholders because little information is known about the firms before they go public (Cheung and Krinsky, 1994). In addition, IPOs might be perceived as a final stage for the early investors who want to convert their investments to cash (Li and Zhou, 2006). All the previous reasons may drive managers to manipulate earnings in order to influence the valuation of their firms in the market (Fan, 2007). Mainly, high quality firms are expected to send signals that reflect their superior performance because they might be underpriced (Fan, 2007). The selection of a specific type of earnings management at the IPOs depends on the relative importance of the financial statements' items for each firm. Science- and internet-based firms are more inclined to manipulate their research and development costs, while asset-based firms are more likely to exert discretion over sales' accruals (Singer, 2007). Overall, IPOs may motivate both accrual and real earnings management (Darrough and Rangan, 2005; Lo, 2008).

On the other hand, earnings can be considered invaluable in the case of IPOs because they might not reflect a firm's future performance and, therefore, the market values the firms based on other criteria e.g., cash flows (Cheng and Firth, 2000; Bartov et al., 2002b). Meanwhile, an IPO might be perceived as a first stage to raise capital for future growth. Furthermore, reporting misleading earnings may drive shareholders to sue the firms once the manipulation is detected (Ronen and Yaari, 2008). In this sense, IPOs would not create an incentive for management to manipulate earnings. On the contrary, they would demotivate earnings management behaviour so that firms maintain their credibility and keep their future financing opportunities necessary for their growth, especially under the higher monitoring of IPO firms (Li and Zhou, 2006; Ball and Shivakumar, 2008).

SEOs also have a dual effect on earnings management. Firms manage earnings around SEOs in order to improve their stock prices and, therefore, increase the wealth of

the current owners (Teoh et al., 1998; Cohen and Zarowin, 2010). On the other hand, a number of restrictions appear around SEOs including the high analysts' coverage, information requirements by the market and the auditors, and the possibility of being sued when the manipulation is detected (Teoh et al., 1998). Therefore, the market discounts earnings management, especially when real earnings management is used, because it expects bad performance after the SEOs (Teoh et al., 1998; Shivakumar, 2000; Cohen and Zarowin, 2010). The previous factors drive the firms to avoid earnings management in the periods of SEOs. Overall, there is a general trend of a drop in stock prices after SEOs because investors consider an SEO as a way of raising capital due to bad financial situation or stocks being overpriced (Brazel and Webb, 2006).

4.4.3 New Listing and Cross-Listing

New listings influence earnings management in two opposing ways. Firms listed in a new market have incentives to manage earnings in order to improve their stock prices; although such aggressive behaviour may result in negative returns in the future (Lin, 2003; Lang et al., 2006). From another point of view, firms may avoid earnings' manipulation due to the restrictions they face in the new markets where they are listed (Ronen and Yaari, 2008).

Earnings management appears more vividly in firms listed in non-US markets as they have less restrictions in terms of investor protection and legal requirements. Even if a firm is listed in the US at the same time, the SEC conventions cannot replace the local regulations in the firm's home country (Lang et al., 2006). Firms usually select the time of peak performance when they attempt to list in a new market. At this point, they are more likely to use earnings management in order to improve their stock prices and get more cash infusion (Ndubizu, 2007).

4.4.4 Mergers and Acquisitions

The acquirer firm aims to improve its stock price in order to decrease the number of stocks transferred to the target firm in case of stock mergers and acquisitions; hence avoids the dilution of the current shareholders' ownership. In this sense, acquirers select

to manage earnings in order to avoid bad results that might drive the market to discount their stock prices. Acquiring firms also tend to manage earnings to sell their shares at a higher price so that they can finance a cash acquisition with a lower cost (Erickson and Wang, 1999; Efendi et al., 2007). On the other hand, the length of the negotiation process between the acquirer and the target firm may prevent earnings management behaviour because it gives a chance for the target to understand the acquirer's transactions and detect the manipulation, especially the discretionary accruals that reverse in the next accounting period. Therefore, target firms are required to perform their due diligence in evaluating the acquirer's financial statements; otherwise they might get subsequently sued by their shareholders (Erickson and Wang, 1999; Gong et al., 2008).

The target firms may also manage earnings to improve their stock prices and, consequently, get a higher value for the acquisition transaction. This particularly applies to hostile takeovers where targets attempt to avoid the acquisition (Easterwood, 1997). On the other hand, the target firms may be less inclined to manage earnings because the price of the transaction is mostly determined by the bargaining power of the acquirer and the target rather than the reported earnings of the target firm (Erickson and Wang, 1999).

4.4.5 Insider Trading

According to the Security and Exchange Act of 1934, insiders are defined as parties that own more that 10% of any class of firm's equity. The definition includes any employees or beneficial owners in the firm like lawyers and accountants. The act considers trading based on private information illegal if such information has material consequences on decision making (Ronen and Yaari, 2008). Since then, the Securities and Exchange Commission (SEC), the Congress, and the courts in the United States have been attempting to prevent insider trading especially before critical corporate announcements e.g., earnings' announcements (Bainbridge, 2001).

Managers may achieve self-benefits by making use of their access to private information and, thus, they buy stocks if their values are expected to increase in the future while sell those whose values are expected to decrease (Lakonishok and Lee, 2001; Brochet, 2016). In this sense, insider trading is likely to motivate earnings management to meet analysts' expectations and, thus, managers can sell their stocks at better prices (Beneish and Vargus, 2002; Hochberg et al., 2003; Park and Park, 2004; McVay et al., 2006; Sawicki and Shrestha, 2016). Elitzur and Yaari (1995) document that the effect of insider trading on earnings management is based on how the market reacts to earnings management. If managers expect the market to react positively, they buy more shares at the current cheaper prices and manage earnings upwards to increase their wealth. They may also sell their shares at the current higher prices and subsequently manage earnings downwards. The opposite applies if the managers expect the market to react negatively to earnings management. Therefore, insider trading is a good tool to predict the direction of earnings' manipulation (Elitzur and Yaari, 1995).

On the other hand, insider trading is sometimes perceived as an efficient contractual arrangement, which is part of management compensation scheme, to maximize the welfare of both the managers and the shareholders (Dye, 1984; Seyhun, 1992; Noe, 1997; Bainbridge, 2001; Bolton et al., 2006). In this sense, insider trading mitigates the agency problem as it allows private information to be reflected in the stock prices (Manne, 1966). In this line, Boyer et al. (2003) emphasize the ethical, rather than the opportunistic, aspect of insider trading. They suggest that the relation between insider trading and earnings management depends on the degree of optimism/pessimism of the managers about the firm performance. Optimistic managers tend to manage earnings upwards and still buy more stocks. Pessimistic managers, however, manipulate earnings downwards while selling more shares (Boyer et al., 2003). Their findings suggest the need for selective, and not prohibitive, regulations to control the insider trading phenomenon (Manne, 1966).

4.4.6 Management Buyouts

Mangers may attempt to purchase the firm they are working for. A conflict of interests appears between managers and shareholders as the managers want to pay less while the shareholders want to receive more (Ronen and Yaari, 2008). Therefore, managers are expected to manage earnings downwards before the buyout, so that they can purchase the firm at a lower price. In this line, Wiedman and Marquardt (2002) document that

managers tend to defer revenue recognition before the buyout announcement. Wu (1997) find that managing discretionary accruals downwards before the announcement of the buyout bid drives stock prices down; which is not observed in the case of third party takeover. Therefore, management buyouts raise concerns about the negative consequences on the shareholders' interests (Perry and Williams, 1994).

4.4.7 Meeting or Beating a Benchmark

Managers may get motivated to manage earnings under the incentive of meeting or beating a benchmark. Based on the degree of information asymmetry, the firm and the market implicitly agree on the benchmarks. Benchmarks can take the form of zero earnings, prior years' earnings, or analysts' forecasts (Xue, 2003). Managers generally try to avoid reporting losses (Burgstahler and Dichev, 1997). They also prefer to show an increase in their earnings compared to a similar prior period and, therefore, even profitable firms would have the incentives to manage earnings (DeAngelo et al., 1996; Ayers et al., 2006; Barua et al., 2006; Roychowdhury, 2006; Myers et al., 2007; Cohen et al., 2008; Osma and Young, 2009). In addition, managers try to meet analysts' forecasts because they reflect the general market expectations that will give the firms more credibility in the market if they are met (Graham et al., 2005).

The market takes into consideration meeting the benchmarks when valuing the firms (Burgstahler and Dichev, 1997; Barua et al., 2006). Firms that meet their benchmarks are valued at different valuation models than those which do not. Therefore, meeting or beating a benchmark is expected to have a positive influence on firms' stock prices (Durtschi and Easton, 2005). Because of the important role of the analysts as gatekeepers in preventing market deception by earnings management, their effect on earnings management is going to be discussed separately in the next section.

4.4.8 Analysts

Analysts are gatekeepers that monitor management performance in order to protect the shareholders' welfare. Consequently, they reduce the agency costs and improve the public confidence to invest in the markets (Ronen and Yaari, 2008). Analysts' coverage

is negatively associated with information asymmetry and, thus, it is used as a measure of information asymmetry variable in some studies (Houston at al., 2008). They play an important role in simplifying the complicated information reported by management to the users of the financial reports and providing relevant analysis about the future for decision making (Brennan and Hughes, 1991).

Analysts' coverage has been increasing over time leading to a more influential role in the stock market (Francis et al., 2004). Analysts' forecasts influence market prices as they improve the efficiency of communicating the financial information and, thus, allow a closer view by the investors (Barth and Hutton, 2000; Brennan and Tamarowski, 2000; Shroff at al., 2004). Therefore, investors prefer to invest in firms that are covered by more analysts (O'Brien and Bhushan, 1990; Brennan and Hughes, 1991). The response of the stock market to earnings' announcements revised by analysts is higher than the response to other strategic business information because analysts provide information that can be directly used in firm valuation models (Bagnoli et al., 2005).

Analysts are expected to provide accurate forecasts because the clients pay commissions to the brokers and investment banks where analysts work (Ronen and Yaari, 2008). In this sense, analysts use their professional experience in analyzing the financial information and identifying management incentives and, therefore, providing valuable information in their forecasts to the different users (Gu and Chen, 2004; Barton and Mercer, 2005; Lin and Shih, 2006). As a result, analysts mitigate earnings management because they are able to detect opportunistic behaviour and discount inflated earnings, especially in environments where transparent financial disclosures exist (Ke, 2001; DeGeorge et al., 2004; Brown, 2004; Yu, 2008).

Astonishingly, most of the literature provides evidence on the positive effect of analysts on earnings management. Analysts' may face pressure from their employers or clients to make optimistic forecasts (Francis and Philbrick, 1993; O'Brien et al., 2005). They are also limited to the amount of information provided by the management. In this sense, analysts are more inclined to tolerate earnings management or even collude with the management by reducing their expectations to be easily met; hence reflecting better market values (Bartov et al., 2002a; Griffin, 2003). Under the previous incentives, analysts may not report overestimated accruals that might reduce firms' future earnings.

105

They may even not distinguish between discretionary and nondiscretionary accruals when making their forecasts. Consequently, analysts do not always discount earnings management and thus may ultimately provide biased forecasts (Bradshaw et al., 2001; Abarbanell and Lehavy, 2003; Ahmed et al., 2005).

Accordingly, meeting analysts' expectations is more subjective compared to meeting other benchmarks (e.g., prior years' earnings), and thus firms may not only use earnings management but also expectation management to achieve their opportunistic targets (Barua et al., 2006; Kross et al., 2011). Expectation management allows for guiding the perceptions of the analysts whether explicitly through firm announcements or implicitly by staying silent without advising the analysts to review their forecasts. Das et al. (2011) find that the relationship between expectation management and earnings management is complementary. Managers tend to manage analysts' expectations when earnings management activities involve higher costs (Das et al., 2011). Similarly, management may use classification shifting to meet or beat analysts' forcasts (McVay, 2006; Athanasakou et al., 2009; Yun et al., 2010). However, Athanasakou et al. (2009) document that the market rewards firms that manipulate their accounts' classification less than those that genuinely achieve their targets.

4.5 Third-Party Motives of Earnings Management

The relations of the firm with the external parties may shape its earnings management behaviour to be able to maintain its future interests with them. Third parties need the accounting information to make decisions related to directly entering into business transactions with the firm or indirectly revising the structure of the environment where the firm is operating. In this sense, third parties influence the firm's strategies in managing its resources to generate earnings (Ronen and Yaari, 2008). The third-party motives are also referred to as the external motives of earnings management in this thesis.

4.5.1 Industry

Firms within the same industry generally have similar legal and financial incentives, and thus they tend to take into consideration the behaviour of the other firms in that industry when making their decisions (Kallunki and Martikainen, 1999; Popp et al., 2003; Othman and Zeghal, 2006). Consequently, earnings management activities of a firm are correlated with the levels of comparable activities in the same industry (Bagnoli and Watts, 2000). In contrast, different industries develop based on different resources that ultimately determine their distinct structures and characteristics and contribute to the variation in performance and profitability between them (McGahan and Porter, 1997; Popp et al., 2003). Eventually, the variation in managerial incentives between the different industries results in the variation in the type and extent of earnings management activities used in each sector (Aharony et al., 2000; Gu et al., 2005). In this sense, industry may have an impact on discretionary accruals as well as real earnings management activities (Roychowdhury, 2006; Toniato et al., 2006; Sundvik, 2013).

The accounting choices available to the managers in an industry may allow the use of earnings management practices specific to each sector. For example, firms in the manufacturing sector have the opportunity of manipulating the amount of units produced. Overproduction as a real earnings management strategy would lead to a lower cost of goods sold and ultimately improve net revenue figures (Gu et al., 2005; Roychowdhury, 2006). Under different conditions, firms in the retail sectors are expected to have higher receivable balances and, therefore, may be subject to more bad debt manipulation (Gu et al., 2005).

In the same way, the technology associated with specific industry sectors may create incentives for the firms to manage earnings. Industries which share chemical or physical technologies tend to use less real earnings management activities because of the information shared with the competitors that imposes a restriction on earnings manipulation (Bagnoli and Watts, 2010). In contrast, industries with less similar technologies, such as services, are more likely to smooth their earnings since it is more difficult for other firms to obtain knowledge about their unique transactions (Bagnoli and Watts, 2010). Overall, high-tech firms exhibit better earnings quality due to industry specific factors like litigation, the funding needs, and the bigger investments (Kwon et al., 2006).

107

Regulation and state-protection also play an important role in determining earnings management behaviour in the different industries; hence differences in earnings manipulation are evidenced between regulated and unregulated sectors. In this line, lower levels of earnings management are expected in highly regulated industries as there is less space for management discretion in selecting and using the accounting principles (Gu et al., 2005). High-tech firms represent an example of a highly regulated industry sector that exhibits more conservatism in financial reporting (Kwon et al., 2006). The same applies to the state-protected industries due to the high supervision and distinctive treatment by the regulatory bodies (Aharony et al., 2000).

Furthermore, differences in earnings management appear between core and peripheral industries as the core sectors are less likely to manage earnings than the peripheral sectors (Sun and Rath, 2009). Core sectors represent industries with more economic and political influence like the construction, manufacturing, and extraction industries in the US. The enhanced earnings quality in such industries is attributed to the lower competition and environmental uncertainty, along with the higher profitability, unionization, capital investments, and more sophisticated regulations (Albrecht and Richardson, 1990).

While the previous studies focus on a dual-industry view, others implement a multiple perspective to examine the effect of industry on earnings management. Toniato et al. (2006) consider the impact of industry on accrual earnings management activities in Brazil and document that most of the industry sectors do not have a significant impact on discretionary accruals. On the other hand, Sundvik (2013) applies the same idea on the industries in Finland and identifies some differences in using discretionary accruals between the different sectors. The results of the two studies reflect the differences in industry characteristics and earnings management behaviour between the different economies.

Overall, industry reflects firms' strategic objectives and competitive advantages; hence plays a major role in determining managerial choices and earnings quality (Demerjian et al., 2013a). However, all of the studies in this area of research focus on accrual earnings management. Furthermore, although the multiple-economy approach followed by Toniato et al. (2006) and Sundvik (2013) allows the revelation of each

108

industry's behaviour separately, the specific industry characteristics need to be examined in order to explain their influence on earnings management.

4.5.2 Industrial Diversification

The operations of industry-diversified firms are complex for the shareholders to understand and thus create more information asymmetry between the managers and the shareholders and more chances for earnings management. On the other hand, diversified firms may not need to manipulate earnings because they have different sources for earnings and cash flows. In addition, the correlation between accruals of the divisions in the different industries makes it difficult for top management to decide on the overall result of earnings management activities that may cancel each other. In this line, Jiraporn et al. (2006) provide empirical evidence on the low association between industrial diversification and earnings management.

4.5.3 Regulations

Although regulation is generally perceived as a mechanism that limits opportunistic behaviour (Graham et al., 2005), it may have the opposite effect (Goldman and Slezak, 2006). While it controls financial disclosures, regulation may prohibit firms from disclosing their private information and thus end up in violating the revelation principle (Ronen and Yaari, 1993). In the same line, while regulations generally aim at enhancing management incentives to improve performance and firm value, such incentive may create a pressure to manage earnings and thus reduce firm value (Cheng et al., 2011). The emphasis of regulations on the role of auditors can also mitigate earnings management, but it contributes to a decrease in the value of the firm as it encourages management conservative behaviour (Goldman and Slezak, 2006).

Firms may manage earnings in different ways under the pressure of regulation. Regulatory investigations or new regulatory projects drive firms to manage discretionary accruals downwards in order to demotivate the regulators from issuing strict regulations that might harm the firms (Jones, 1991; Cahan, 1992; Cahan et al., 1997). Poorly performing firms in regulated industries, however, manage earnings upwards by decreasing their reserves and allowances to avoid any intervention that might result in losing some of their benefits or even getting their businesses closed (Petroni, 1992; Gaver and Paterson, 2000; Gray and Clarke, 2004). Firms may also respond to the regulatory power by manipulating their non-operating activities (Chen and Yuan, 2004). In contrast, the lack of regulation can drive firms to manage earnings upwards due to the absence of protection in the market (Aharony et al., 2000).

The Securities and Exchange Commission (SEC) in the US is the main body that regulates the capital market and the listed firms' affairs (Giroux, 2004). One of the most famous and recent regulations is the Sarbanes-Oxley act (SOX) that followed the Enron scandal in 2002. SOX incorporated lots of emphasis on corporate governance including the independence of the board of directors, the audit committee, and the external auditor (Hossain et al., 2011; Krishnan et al., 2011). While such regulatory intervention seems to decrease accrual earnings management and the opportunistic non-GAAP disclosures, firms tend to use other methods to achieve their opportunistic targets like real earnings management or the abuse of special items (Ewert and Wagenhofer, 2005; Cohen et al., 2009; Heflin and Hsu, 2008; Kolev at al., 2008; Bartov and Cohen, 2009; Hutton et al., 2009; Baber et al., 2011; Badertscher, 2011). Ghosh et al. (2010), however, document that SOX did not have any significant influence on earnings management behaviour; hence they raise concerns about the context and conditions of applying the different studies in this area.

4.5.4 Political Environment and Country-Specific Policies

Firms are inclined to manage earnings under political pressure. For example, oil firms managed earnings downwards during the Second Gulf War in order to avoid any unfavourable interventions after the sudden increase in petrol prices (Han and Shiing-wu, 1998). The distinct policies of each country also influence earnings management behaviour. Firms in countries with strong investor protection policies are involved in less earnings management compared to those in other countries (Leuz et al., 2003; Lang et al., 2006; Haw et al., 2011; Gopalan and Jayaraman, 2012; Houqe et al., 2012). Firms in countries with common laws use less earnings smoothing than firms in countries with

code laws⁷ (Gassen et al., 2006). However, Gaio (2010) document that the differences in policies between countries do not influence earnings quality as it is basically attributed to the differences in individual firm characteristics.

4.5.5 Accounting Standards

Although the voluntary adoption of the International Financial Reporting Standards (IFRS) has resulted in a better quality of earnings, the bright side of the effect needs to be interpreted with care (Barth et al., 2008). In this sense, the researchers have to take into consideration that the adoption of IFRS allows firms to have a fresh start. Therefore, making a fair comparison between the pre-adoption period and the post-adoption period might be impractical (Walker, 2013).

In contrast, examining the influence of the mandatory adoption of IFRS in 2005 on earnings management is difficult because of the continuous changes in the economic conditions before and after the adoption, e.g., the financial crisis in the late 2007. In addition, the mandatory adoption of IFRS has provided the firms with the chance to clean-up their financial statements before the adoption (Garcia-Osma and Pope, 2011; Walker, 2013). The empirical evidence shows that the influence of the mandatory adoption of IFRS differs in different countries. While no influence was noticed in Australia, a significant increase in accrual earnings management has occurred in the French firms (Jeanjean and Stolowy, 2008). Meanwhile, the mandatory adoption of IFRS depends on the enforcement regime in a particular country, e.g., the adoption of IFRS in the UK has resulted in less accrual earnings management because of the strong enforcement regime (Ipino and Parbonetti, 2011). It has been also noticed that income smoothing has increased in the European Union firms after the adoption of IFRS in 2005 because of the clean-up that was taken by most firms prior to the adoption (Callao and Jarne, 2010; Capkun et al., 2016).

Similar to the regulatory intervention effect on earnings management, the adoption of IFRS has driven the trade-off between the different earnings management methods. Ipino and Parbonetti (2011) document that while the adoption of IFRS in the

⁷ While the common laws develop based on the cases experienced in the courts of a specific country over time, code laws represent systematic legislation enforced by law.

UK has resulted in a decrease in accrual earnings management, firms have started to use more real earnings management and earnings smoothing (Ewert and Wagenhofer, 2005). Furthermore, managers develop the skills to manipulate the new accounting standards over time; hence there is a continuous need for updating the accounting regulation (O'Brien, 2005).

4.5.6 Tax Considerations

Firms may manage earnings downwards to decrease their tax expenses which are calculated based on their earnings' numbers (Boynton et al., 1992). In this sense, taxation mitigates the upward earnings management and, thus, decreases the pernicious manipulation (Eilifsen et al., 1999). Furthermore, taxation mitigates aggressive earnings management as huge differences between book income and tax income can be easily noticed by the tax authorities and the stock market. Therefore, while the managers aim to achieve tax benefits, they take into consideration the costs of earnings management they will subsequently incur (Badertscher et al., 2006).

To achieve tax advantages, firms manage earnings through their deferred tax allowances that are kept open until the last moment before the financial reporting to the public. Therefore, the allowances allow managerial judgment to manipulate the tax expenses so that firms can meet their earnings' targets (Bauman et al., 2001; Kumar and Visvanathan, 2003). In addition, the flexibilities in tax laws give a suitable chance for earnings management behaviour. For example, allowing income adjustment for tax purposes drives firms to manage earnings downwards, while reducing tax rates over specific periods in the future drives firms to shift earnings forward (Gramlich, 1991; Maydew, 1997).

4.5.7 Competitors

Market competition influences earnings management behaviour as firms in the same industry tend to take each other's performance into consideration when manipulating their earnings (Kallunki and Martikainen, 1999; Bagnoli and Watts, 2000). Therefore, incentives of the opportunistic behaviour exist within the same sector so that a firm

manages its earnings when the expected costs of this behaviour are lower than those of the rival firms (Bagnoli and Watts, 2010). While some of the studies perceive competition as a motive of earnings management (Karuna et al., 2012; Datta et al., 2013), the opposite results have been also documented (Dalia and Park, 2009; Markarian and Santalo, 2010).

The relationship between competition and earnings management can be interpreted based on two different perspectives in information theory. From one point of view, market competition is associated with more information and thus allows the stakeholders to make comparisons between the firms. Therefore, managers face more pressure to improve performance relative to their rivals and more incentives exist for managerial myopia and earnings management (Karuna et al., 2012; Datta et al., 2013). From another point of view, the less information asymmetry raises the managerial concerns about market punishment due to the higher monitoring by the stakeholders (Dalia and Park, 2009). In this sense, market competition moderates the agency problem and contributes to minimizing managerial incentives to manage earnings in order to protect their firms and individual reputations (Tinaikar and Xue, 2009; Bagnoli and Watts, 2010; Datta et al., 2013).

In addition, the impact of competition on earnings management can be explained from two different points of view based on profit volatility. Market competition increases price elasticity and, therefore, makes profits more volatile. As a result, the cost of borrowing, the required rate of return, and the audit fees are expected to increase leading to more incentives for the firms to manage earnings in order to avoid such costly outcomes. At the same time, it becomes difficult for the stakeholders to understand management sophisticated activities or build expectations under the higher levels of uncertainty; hence profit volatility may create less pressure on management to manipulate earnings (Kole and Lehn, 1999; Christie et al., 2003; Tinaikar and Xue, 2009).

Furthermore, managerial incentives can explain the effect of market competition on earnings management from two different points of view. On the one hand, firms in competitive markets are likely to hire more skilled managers and offer better incentives to maintain a competitive advantage (Hubbard and Palia, 1995; Kole and Lehn, 1999; Karuna, 2007). In this sense, competition enhances the disciplined behaviour of management. On the other hand, the lower rewards to the managers in competitive markets compared to those in less competitive industries e.g., monopolies, may discourage innovation and productivity and stimulate indifference and demoralization (Karuna, 2007). Therefore, competition may be perceived as a motive of earnings management.

Various proxies have been used to measure market competition in the industrial organization and accounting literature e.g., product substitutability, market size, entry barriers, pricing power, and market concentration (Dalia and Park, 2009; Markarian and Santalo, 2010; Karuna et al., 2012; Datta et al., 2013). Product substitutability reflects the amount of similar products produced by rival firms; market size represents the number of customers demanding those products in the market, entry costs reflect the hurdles in front of the new entrants to go into the market, and market pricing power implies the firms' abilities to set higher price-cost margins (Tinaikar and Xue, 2009; Karuna et al., 2012; Datta et al., 2013). While higher product substitutability and bigger market size reflect more intense competition, higher entry costs and greater pricing power imply less competition in the industry. In contrast, while higher market concentration implies less competition, it might result from the excessive competitive forces that drive less competent companies to exit the market (Markarian and Santalo, 2010; Karuna et al., 2012; Datta et al., 2013).

The previous measures have been used by different studies to examine the effect of market competition on earnings management. Datta et al. (2013) focus primarily on pricing power and its impact on accrual earnings management. They find that firms with higher pricing power exhibit less discretionary accruals because it is a cushion that can absorb any revenue shocks and, therefore, relieves management from the need to manage accruals. In addition, firms with higher market pricing power have less pressure to meet market expectations or hide information from the other competitors. Datta et al. (2013) conclude that the mechanisms of pricing power and discretionary accruals are used as substitutes. Overall, the results emphasize a positive impact of market competition on accrual earnings management.

Karuna et al. (2012) focus on other measures of competition which include product substitutability, market size, and entry costs. Their study is the only one, to my knowledge, that covers both accrual and real earnings management activities in its scope. They extend the results of Datta et al. (2013) by documenting a positive impact of market competition on both accrual and real earnings management. According to their explanation, competition enhances managerial incentives to improve firm efficiency and thus may drive earnings management. However, the reverse result is achieved when using market concentration as a measure of competition because of the imprecise meaning of market concentration relative to the other competition measures. In particular, Karuna et al. (2012) believe that high competition drives less competent companies to exit; ending up with a more concentrated market in the long run. Therefore, although higher concentration intuitively implies less competition, it can be sometimes associated with high competition. As a result, the study of Karuna et al. (2012) claims that market concentration does not reflect a precise meaning of competition.

The study of Markarian and Santalo (2010) uses market concentration measure to examine the relation between market competition and discretionary accruals from a different perspective. It uses information as a basis for explaining the previous relation and documents that competition mitigates discretionary accruals because of the decreased information asymmetry in the market. According to their explanation, earnings management becomes costlier when the stakeholders can access more information about the firm and its competitors. Similarly, Dalia and Park (2009) examine the impact of market concentration on accrual earnings management and document that competition discourages discretionary accruals. According to Dalia and Park (2009), information in competitive markets allows the stakeholders to punish the firms engaged in the manipulation. However, the study is exclusively performed in the manufacturing sector in the US and focuses on accrual earnings management and earnings smoothing while entirely ignores real earnings management activities.

Other measures of competition exist at the individual firm level e.g., market leadership, and the international market levels e.g., foreign markets' entry barriers. At the firm level, a market leader in the industry tends to use more real earnings management because of its high competitive status, specialized knowledge, bargaining power, and economies of scale (Zang, 2012). At the international market level, competition also tends to enhance earnings management behaviour as the imposed

pressure from foreign suppliers and the low price-cost margins result in more profit uncertainty (Tinaikar and Xue, 2009).

Overall, the role of market competition has been perceived from the bright side as enhancing firm performance to attain a competitive advantage in the market. However, it might be viewed from the dark side as one of the motives of earnings management because it results in lower profits to the existing firms, stimulates harmful signals from the rivals, and motivates hiding private information from the stakeholders e.g., new entrants, suppliers, customers, and investors (Fudenberg and Tirole, 1986; Bolton and Scharfstein, 1990; Ettredge et al., 2002; Dharan, 2003; Karuna et al., 2012). Competition may even result in aggressive forms of earnings management which take the shape of structured transactions or huge restatements e.g., the cases of Enron, WorldCom, and Xerox Corporations (Dharan, 2003; Ronen and Yaari, 2008).

4.5.8 Suppliers and Customers

Earnings management is also determined by the extent to which the firm depends on its suppliers in providing the raw materials and customers in purchasing its final products. A firm that extensively relies on specific suppliers or customers is more inclined to manage earnings in order to influence their perception of its performance (Raman and Shahrur, 2008; Ronen and Yaari, 2008). As the relationships between the firm and its suppliers and customers last over the long horizons, its reputational concerns may also drive upward earnings management so that it can meet their expectations in the long run (Bowen at al., 1995). Overall, further research is recommended in this area.

4.5.9 Auditors

Auditors provide an attestation function that gives credibility to the financial statements and thus they are important gatekeepers for maintaining earnings' quality and protecting the shareholders' welfare. The role of the independent auditors has been emphasized by the SOX through its Public Company Accounting Oversight Board (PCAOB) that regulates their work. To maintain their independence, the regulation has assigned the responsibility of dealing with auditors to the audit committee instead of management. In this sense, auditors are expected avoid earnings management to protect their reputations and avoid the costly litigation.

However, auditors may fail to detect and report earnings management because of some uncontrollable factors like the increased complexity of business transactions in the modern firms (Ronen and Berman, 2004). In addition, audit failure may occur because of the auditor's fear of losing its clients, especially under the growing competition in the audit market (Antle and Nalebuff, 1991; Nelson et al., 2002, 2003). In this sense, auditors are not independent anymore and may act for the interest of management who directly pays their fees and determines their employment terms (Abdel-Khalik, 2002). The situation gets worse when the auditor highly relies on a specific client in its income or when its non-audit services make a major source of its income e.g., the consulting services that have been increasingly provided by the auditors over time (Coffee, 2003; Walker, 2013). Therefore, auditor's service has been always criticized for targeting money instead of aiming at protecting the shareholders' welfare (Branson, 2006).

Different audit-related factors influence earnings management behaviour e.g., the auditor's opinion, quality, fees, effort, tenure, experience, and turn over. The auditor's opinion may provide an idea about the occurrence of earnings management, e.g., the auditor is expected to give a qualified opinion after incidences of earnings management to drive a negative market response (Lennox, 2005). The auditor's quality also influences earnings management as it is generally associated with better earnings quality. For example, big audit firms have more economies of scale and valuable reputations that motivate more due diligence (Danos and Eichenseher, 1981; Coffee, 2003; Kim et al., 2003; Francis and Wang, 2004; Larcker and Richardson, 2004; Francis and Wang, 2008). However, while high quality auditors do not accept discretionary accruals, they may allow real earnings management as a substitute (Cohen and Zarowin, 2010; Chi et al., 2011). Overall, the auditor's quality depends on the business environment e.g., stronger investor protection policies assist the high quality auditors in assuring high quality earnings (Francis and Wang, 2008). The audit fees are negatively associated with earnings management while the non-audit fees are positively associated with earnings management; hence the emphasis of SOX on increasing the audit fees and avoiding the non-audit services can be justified (Frankel et al., 2002). Auditor effort in the form of more working hours and skeptical attitude contributes to mitigating earnings

117

management (Caramanis and Lennox, 2008; Cohen et al., 2008; Chen et al., 2012). Audit tenure and the industry experience result in better earnings quality attributed to the extensive knowledge about the client (Gul et al., 2009). Meanwhile, long tenure results in closer auditor-auditee relations and thus may violate the auditors' independence requirement (Davis et al., 2006). Auditors leaving their audit firms and moving into managerial positions in one of their clients are more likely to manage earnings; hence a cooling-off period has been required by SOX (Lennox, 2005). Finally, the resignation of the current auditor and assigning the job to a new one may imply the occurrence of earnings management (DeFond and Subramanyam, 1998; Davidson et al., 2006).

Overall, although auditors are one of the critical gatekeepers that monitor firms' performance and protect shareholders' interests, the auditing standards have ignored their role in mitigating earnings management. Instead, they only focus on assuring the fair presentation of the financial statements according to the generally accepted accounting principles (GAAP) and reporting material misstatements whether due to fraud or error in the audit report. As a result, auditors are not expected to detect and report earnings management because it is within GAAP and thus does not violate the accounting standards. In this sense, the matter of reporting earnings management by the auditors remains subjective although it may materially influence decision making.

4.5.10 Other Gatekeepers

The press is one of the most influential gatekeepers in the market and thus contributes to mitigating earnings management. It provides stakeholders with information about accounting manipulation, fraud, and changes in the regulations and, therefore, it assists them in making their decisions (Borden, 2007). Turner (2001) reports that the Securities and Exchange Commission (SEC) may rely on the press more than it does on its own investigations to learn about cases of accounting restatements. As a result, the revelation of bad news about a firm in the press e.g., fraud or earnings management attempts, will negatively influence its stock price (Foster, 1987).

Lawyers have been criticized for losing their influence as gatekeepers and transforming their profession to a routine job recently (Branson, 2006). They basically

play a role in structuring the contracts and thus establishing the relations between the firm and the external or internal parties (Giroux, 2004). However, lawyers' role can be more influential in mitigating earnings management if they provide advice on those contracts to the less knowledgeable shareholders and thus contribute to maximizing firm wealth.

Employee unions are also important gatekeepers in preventing earnings management. Firms negotiating their contracts with the employees have incentives to manage earnings downwards so that they can convince the employee unions to accept lower payments (Liberty and Zimmerman, 1986; Bowen at al., 1995). However, restrictions on earnings management may exist when a firm negotiates its employment contracts because the employees are able to expect the earnings and the employee unions usually hire experts who assist in making their decisions (Liberty and Zimmerman, 1986). A poorly performing firm is less likely to face a pressure from its employees and the involved unions because they already have low expectations based on its poor performance (Liberty and Zimmerman, 1986; Peltier-Rivest, 1999).

Finally, investment banks are responsible for arranging the new share issues of the firms in addition to hiring analysts to provide their forecasts and recommendations (Giroux, 2004). In this way, investment banks play an important role as gatekeepers to mitigate earnings management. Credit rating agencies play a similar role by rating the firms according to their credit worthiness and thus contribute to decreasing the information asymmetry in the market.

4.6 Conclusions

This chapter has summarized the motives of earnings management that have already been identified in the literature. Based on the theories of earnings management discussed in the previous chapter, three groups of motives have been identified. First, contracting motives arise from the conflict of interests in the contracts between the firm and the other stakeholders. Second, capital market motives exist because of the inefficiency of the stock markets to reflect firms' intrinsic values. Finally, third-party motives appear due to the influence of external parties on firm performance. Overall, the previous motives are influential when the managers expect to achieve more benefits from earnings management than those from revealing the truth.

The three groups of earnings management motives are not completely distinct because more than one theory may explain a specific motive. For example, while auditors are considered as a third-party motive because it is related to the influence of an external party on the firm, auditors are involved in contracts with the firm to avoid any conflict of interests. Similarly, although the role of analysts is included in the capital market motives because they influence firms' stock values, they are also considered as external parties that may exert pressure on the firm to manage earnings. For the purpose of this thesis, I have classified each of the previous motives based on its major role in determining earnings management behaviour that has been identified in the literature. The overlap between the three groups of motives appears in Figure 4.3.





Finally, the three main theories that have explained earnings management behviour so far i.e., contracting theory, bounded rationality theory, and information theory, shape firm performance which, in turn, plays a mediating role in determining earnings management behaviour. Therefore, I suggest identifying firm performance as a fourth group of motives for earnings management according to Figure 4.4. The group includes factors like firm profitability and social responsibility. While less profitable firms are motivated to manage earnings to avoid reporting losses, more profitable firms may also manage earnings to meet the expected dividends, improve their share prices, and beat benchmarks (McVay, 2006; Roychowdhury, 2006; Kerstein and Rai, 2007; Cohen et al., 2008; Daniel et al., 2008; Iatridis and Kadorinis, 2009). In addition, firms that perform ethically at the social and environmental levels are more likely to report earnings with better quality (Patten and Trompeter, 2003; Kim et al., 2012). The next chapter explains the data and methodology of this thesis.





Chapter 5 Data and Research Methods

5.1 Introduction

This chapter constructs the data and methodology for the thesis to examine the influence of some contracting motives (executive compensation and managerial ability) and external motives (market concentration and market competition) on earnings management. It explains the procedures followed to select the sample and presents the sources used to collect data, models used in estimating accrual and real earnings management, and the methodology that will be adopted in the empirical analysis.

The remainder of this chapter is organized as follows. Section 5.2 explains the overall sample construction of the thesis. Section 5.3 presents the sources used to collect data. Section 5.4 explains the different models that will be used in the following empirical chapters to measure accrual and real earnings management. Section 5.5 discusses the general basis for selecting the methodology of the thesis. Finally, section 5.6 concludes the chapter.

5.2 Sample Construction

For the empirical work of Chapter 6, I select all the firms in the United States that have the required data for calculating the measures of managerial ability, management compensation, and earnings management. However, the Execucomp database only provides compensation data starting from 1992. Furthermore, I use the number of board meetings among the controls for corporate governance, which is only available until 2006. As a result, Chapter 6 only covers the period from 1992 to 2006 with a sample of 6,974 observations.

To perform the empirical analysis of Chapter 7, I include all the firms in the United States that have the required data for calculating the measures of market concentration, information asymmetry, and earnings management during the time period between 1989 and 2011. The time period begins in 1989 because the cash flow statement has been adopted since 1988 and a minimum of one year for cash flow data is needed as a basis for the calculation of earnings management metrics. The analysis ends in 2011 when the work on this thesis has started. As a result, Chapter 7 ends up with a sample that consists of 25,119 observations only.

Finally, the sample of Chapter 8 includes all the firms in the United States that have the required data for calculating the measures of earnings management, managerial ability, and competition for the same previous period from 1989 to 2011. As a result, the sample ends up with 66,695 observations. The samples used in the three empirical chapters of this thesis are summarized in Table 5.1.

Empirical chapter*	Major variables	Sample size	Start date	Reason	End date	Reason
Ch 6	Managerial ability Management compensation Earnings management	6,974	1992	Compensation data availability on Execucomp database	2006	Availability of governance variables
Ch 7	Market concentration Information asymmetry Earnings management	25,119	1989	Cash flow statement availability	2011	Not applicable
Ch 8	Managerial ability Market competition Earnings management	66,695	1989	Cash flow statement availability	2011	Not applicable

 Table 5.1 Summary of the samples used in the thesis

The table presents the differences in sample construction between the three empirical chapters of the thesis based on the major variables, the sample size, the start date, the reason of selecting the start date, the end data, and the reason of selecting the end date. Each of these samples will be discussed individually in the related chapters. * indicates the chapter number in the thesis.

In all of the previous samples, and following Cheng et al. (2011), I exclude regulated industries such as banks, credit institutions, brokers, insurance, real estate, holding companies, and investment firms. These industries have their unique accounting and

financial practices and are subject to distinct regulations. Therefore, managers in these industries have different motivations to manipulate earnings than those of managers in other industry sectors.⁸ In addition, I exclude firm-years where accounting changes, merger and acquisition activities, or discontinued operations occur.⁹

Furthermore, following prior literature I exclude any industry with fewer than six observations for each SIC code in a specific year to ensure sufficient data exists to calculate earnings management measures and the Ordinary Least Square (OLS) assumption regarding the normality of the error term holds (e.g., Rosner, 2003; García Lara et al., 2005; Kothari et al., 2005; Athanasakou et al., 2009; Iqbal et al., 2009). For that purpose, I follow the SIC classification of Fama-French (1997). As the models introduced in the thesis are dynamic, I ensure that information is available for at least five consecutive years for each firm over the study period (Miguel et al., 2004).

5.3 Data Sources

I use different data sources that include the Annual Compustat, Quarterly Compustat, and Historical Segments Compustat to collect data from the financial statements i.e., statement of financial position, income statement, and cash flow statement. I use the IBES database to collect data related to the analysts and their forecasts. Execucomp database is used for management compensation data. Finally, I use the dataset made available by Demerjian et al. (2012) for the estimates of managerial ability.¹⁰ The sample construction of each of the empirical chapters is discussed in detail in Chapter 6, 7, and 8.

⁸I exclude firms with the following SIC codes: $4000 \le SIC \ge 4900$ and $6000 \le SIC \ge 6300$.

⁹ According to McNichols (2002), I specifically exclude firm quarters or years with non-blank values for accounting changes cumulative effects (ACCCHGQ_FN), or merger and acquisition activities (ACQMETH_FN), or discontinued operations (DOQ_FN) in the Compustat database. ¹⁰ Data are obtained from the following link:

https://community.bus.emory.edu/personal/PDEMERJ/Pages/Home.aspx.

5.4 The Estimation of Earnings Management

5.4.1 Accrual Earnings Management

Stubben's Model (2010) is primarily used for measuring accrual earnings management in this thesis. The model focuses on discretionary revenues which are the largest component of earnings in most firms (Stubben, 2010). Using revenues as an estimate of discretion reduces measurement error and makes the model less biased and more specified than other accrual models due to three reasons (Demerjian et al., 2013b). First, discretionary revenues reflect receivables' accruals, rather than aggregate accruals. Receivable accruals, in turn, are more directly related to revenues than other working capital accruals. Second, the model focuses on reported revenues rather on cash revenues. While this results in understating discretionary revenues estimate, it is unlikely to overestimate discretion for firms that are less expected to collect their credit revenues by the year end e.g. growth firms. Finally, the model examines receivable accruals for the fourth quarter separately because they are less likely to be collected before the year end. As a result, it prevents overstating discretion when the revenues of the fourth quarter are relatively high or understating discretion when the revenues of the fourth quarter are relatively low (Stubben, 2010). Discretionary revenues are estimated using the following cross sectional OLS regression for each industry-year group with at least 6 observations.

$$\frac{\Delta AR_{it}}{A_{avg}} = \alpha + \alpha_1 \frac{1}{A_{avg}} + \beta_1 \frac{\Delta RI_3_{it}}{A_{avg}} + \beta_2 \frac{\Delta R4_{it}}{A_{avg}} + \varepsilon_{it}$$
(5.1)

In equation (5.1), AR represents accounts receivable; A_{avg} represents average total assets calculated as the average of total assets at the beginning of the year plus total assets at the end of the year; $R1_3$ represents revenues in the first three quarters; and R4 represents revenues in the fourth quarter. All variables are scaled by average total assets (A_{avg}) to avoid heteroskedasticity problems. The residual (ε_{it}) from the regression is the measure of discretionary revenues (*DiscRev*) that represents accrual earnings management (*AccrualEM*). The measure is decile ranked for better comparability and to avoid outliers biasing the results.

In addition to using Stubben's Model (2010) as the main proxy for accrual earnings management based on the previously mentioned reasons, the Modified Jones'

Model is also used as a corroborating measure (Cohen et al., 2008).¹¹ Again, a cross sectional model is used in the calculations of this measure for each industry-year group with at least 6 observations. Finally, the decile ranks are taken for better comparability and to mitigate for the effect of the outliers. The measurement of the Modified Jones' Model has been already explained in detail in section 2.4.6.

5.4.2 Real Earnings Management

To estimate real earnings management, the model of Gunny (2010) is used as it captures more aspects of real earnings management activities compared to other models (Gunny, 2010; Demerjian et al., 2013b). The model measures four components of real earnings management that include reducing discretionary research and development expense (REM_{RD}) as in equation (5.2), decreasing discretionary selling, general, and administrative expense (REM_{SGA}) as in equation (5.3), timing of fixed asset sales to report gain (REM_{Asale}) as in equation (5.4), and overproduction (REM_{Prod}) to cut prices or decrease the cost of goods sold (COGS) as in equation (5.5) (Gunny, 2010). The following cross sectional OLS regressions for each industry-year group with at least 6 observations are used to estimate real earnings management proxies.

$$\frac{RD_t}{A_{avg}} = \alpha_0 + \alpha_1 \frac{1}{A_{avg}} + \beta_1 M V_t + \beta_2 Q_t + \beta_3 \frac{INT_t}{A_{avg}} + \beta_4 \frac{RD_{t-1}}{A_{avg}} + \varepsilon_t^{R\&D}$$
(5.2)

$$\frac{SGA_t}{A_{avg}} = \alpha_0 + \alpha_1 \frac{1}{A_{avg}} + \beta_1 M V_t + \beta_2 Q_t + \beta_3 \frac{INT_t}{A_{avg}} + \beta_4 \frac{\Delta S_t}{A_{avg}} + \beta_5 \frac{\Delta S_t}{A_{avg}} * DD + \varepsilon_t^{SG\&A}$$
(5.3)

$$\frac{GainA_t}{A_{avg}} = \alpha_0 + \alpha_1 \frac{1}{A_{avg}} + \beta_1 M V_t + \beta_2 Q_t + \beta_3 \frac{INT_t}{A_{avg}} + \beta_4 \frac{ASales_t}{A_{avg}} + \beta_5 \frac{ISales_t}{A_{avg}} + \varepsilon_t^{Asset}$$
(5.4)

$$\frac{PROD_t}{A_{avg}} = \alpha_0 + \alpha_1 \frac{1}{A_{avg}} + \beta_1 M V_t + \beta_2 Q_t + \beta_3 \frac{S_t}{A_{avg}} + \beta_4 \frac{\Delta S_t}{A_{avg}} + \beta_5 \frac{\Delta S_{t-1}}{A_{avg}} + \varepsilon_t^{Production}$$
(5.5)

¹¹ The results using the Modified Jones' Models are presented in the appendices of this thesis. The inferences are the same as those I make from the Stubben's model.
In the previous equations, RD represents R&D expense; SGA represents sales and general admin (SG&A) expense; GainA represents income from asset sales; PROD represents cost of goods sold (COGS) plus change in inventory; A represents total assets; MV represents the natural logarithm of market value calculated as the number of common shares outstanding multiplied by the share price; O represents Tobin's Q calculated as the sum of the market value of common shares, the book value of preferred shares, long term debt and current debt divided by total equity and liability; *INT* represents internal funds calculated as the sum of income before extraordinary items, research and development expense, and depreciation and amortization expenses; S represents total sales; DD represents an indicator variable that reflects the sticky cost behaviour for the intentional reduction in SG&A when the demand drops, which equals 1 when total sales decrease between t-1 and t, and zero otherwise; *ASales* represents the sales of long-lived assets; and ISales represents the sale of long-lived investment. To keep the relation between GainA, ASales, and ISales monotonic in equation (5.4), all their signs are made negative when GainA is negative according to Gunny (2010). All variables are scaled by average total assets (A_{avg}) of the beginning and ending asset balances to avoid any heteroschedasticity. A cross-sectional model is applied in the calculations for each year and each industry classified by its four-digit SIC code. The residuals ($\varepsilon_t^{R\&D}$, $\varepsilon_t^{SG\&A}$, ε_t^{Asset} , $\varepsilon_t^{Production}$) from the regressions represent the measures for the R&D, SG&A, fixed assets' sale, and production components of real earnings management (REM_{RD} , REM_{SGA} , REM_{Asale} , REM_{Prod}) respectively. However, the first and second residuals are multiplied by negative one so that cutting the discretionary expense reflects an increase in real earnings management. Finally, the four measures are decile ranked for better comparability and to avoid the effect of the outliers.

Using the sample of 66,695 observations identified above that includes all data required to calculate the four components of real earnings management, I follow Demerjian et al. (2013b) in performing a principal component analysis with varimax rotation to obtain an overall estimate for real earnings management. This step prevents

the inclusion of highly correlated variables in the measure.¹² As a result, I obtain two real earnings management factors according to the following equations:

$$REM_{SGA_Prod} = (0.22REM_{RD}) + (0.64REM_{SGA}) + (0.22REM_{Asale}) + (0.64REM_{Prod})$$
(5.6)

$$REM_{RD_AssetSale} = (0.66REM_{RD}) + (0.25REM_{SGA}) + (-0.65REM_{Asale}) + (-0.25REM_{Prod})$$
(5.7)

The first factor represents discretionary reduction in SG&A expenses and overproduction to cut prices or to decrease the cost of goods sold. The second factor reflects the discretionary reduction in R&D expense and the sale of fixed assets to report gains. As the first factor explains most of the variance in the dataset, it is used for the discussion of real earnings management (*RealEM*) results in this thesis.¹³

In addition to using the Gunny's Model (2010) as the main proxy for real earnings management, the Roychowdhury's Model (2006) is used as a corroborating measure for real earnings management (Cohen et al., 2008).¹⁴ Again, a cross sectional model is used in the calculation of this measure for each industry-year group with at least 6 observations and the decile ranks are taken for better comparability and to mitigate for the outliers' impact. The measurement of the Roychowdhury's Model has been already explained in detail in section 2.5.1.

5.5 Methodology

To solve the models introduced in the empirical chapters of this thesis, the Ordinary Least Squares (OLS) estimator is not used because the models suffer from unobservable individual heterogeneity (Pindado and Requejo, 2014). This heterogeneity is attributed to time-invariant firm and/or managerial effects (Graham et al., 2012; Demerjian et al., 2013b). The selection of managers by the boards of directors in line with their firms'

¹² The principal component analysis step turns the set of correlated variables to be linearly uncorrelated according to the weights of their variances, thus reducing the number of variables to their principal components.

¹³ The same inferences are obtained using the two factors in this thesis.

¹⁴ The calculations and results of the Roychowdhury's Model are presented in the appendices of this thesis. The inferences are the same as those made from the Gunny's Model.

strategies or corporate cultures results in managers with specific abilities to end up in firms with specific earnings' qualities. Similarly, hiring managers with specific inborn capabilities, personalities, or tendencies to take risk leads managers with certain skills to arrive at firms with specific earnings' qualities. Consequently, the samples are expected to be heterogeneous, as some companies are more predisposed to earnings management than others because of their own specificities. The models introduced later in the empirical chapters can be expressed in general terms according to the following equation for the *ith* observation at time *t*:

$$Y_{it} = \alpha_0 + \alpha_1 X_{i,t} + \alpha_2 Y_{i,t-1} + \varepsilon_{it}$$
5.8

Whrere, (Y_{it}) is the dependent variable, (X_{it}) is the explanatory variables, and (ε_{it}) is an error term. However, to avoid bias in the error term (ε_{it}) it is split it into three components. First, I introduce (η_i) to control for the impact of the unobserved effects in any model.¹⁵ Second, I add a time specific effect (d_t) to control for the macroeconomic variables that may interfere with the results over the period of the study. Finally, I consider the remaining part of the error term (ε_{it}) to be a random disturbance (v_{it}) .

$$Y_{it} = \alpha_0 + \alpha_1 X_{i,t} + \alpha_2 Y_{i,t-1} + \eta_i + d_t + v_{it}$$
5.9

The correlation between the unobservable heterogeneity (η_i) and the explanatory variable (X_{it}) violates one of the key assumptions of OLS $[E(\varepsilon_{it} X_{it}) = 0]$. As a result, I replace the OLS estimator with a static fixed effects model that takes the impact of the unobservable heterogeneity into consideration by demeaning the variables in the equation.

The empirical model, however, suffers from endogeneity problem because of the mutual causality and/or simultaneity between the explanatory variable and the dependent variable. In addition, some variables might be omitted, and measurement errors could be expected in the proxies of the different variables. The three factors result in correlation between the explanatory variable (X_{it}) and the error term (ε_{it}) expressed as $[E(\varepsilon_{it} X_{it}) \neq 0]$. Therefore, the endogeneity problem leads to violating the OLS assumption mentioned above.

The static fixed effects model disregards the endogeneity problem. It assumes strict exogeneity $[E(\varepsilon_{it} X_{it}) = 0]$ where the error term is always uncorrelated with the

 $^{^{15}\,\}eta_{i}$ controls for both firm specific effects and manager specific effects.

explanatory variables. This assumption never occurs in the micro/macroeconomic field. Therefore the within groups estimator (WG) is also invalid and gives a biased estimation in dynamic models (Nickell, 1981). The common use of the fixed effects estimator by researchers in earnings management literature is an outcome of ignoring its dynamic behaviour. Passing over this problem results in another source of endogeneity and ultimately biased inferences (Wintoki et al., 2012).

To deal with the previous problem, I treat the explanatory variable as a predetermined variable that, over different periods (s), is influenced by present and past, but not future, economic shocks $[E(\varepsilon_{it} X_{it}) = 0, \text{ for all } s \ge t]$. I then introduce an instrumental variable (IV_{it}) to solve for the endogeneity in the conditional expectation function $[E(Y_{it}-\alpha_0 - \alpha_1 X_{i,t}-\alpha_2 Y_{i,t-1} / IV_{it}) = 0]$ (Pindado and Requejo, 2014). Using one instrument, however, is inefficient because it makes the model less informative. Therefore, I use a generalized method of moments (GMM) estimator that uses more than one instrument for each variable.

When selecting an instrument (IV_{it}) , it is essential that it satisfies two conditions. First, it has to be highly correlated with the instrumented variable, so that it explains it. Second, it has to be uncorrelated with the error term $[E(\varepsilon_{it} IV_{it}) = 0]$ (Pindado and Requejo, 2014). Because it is difficult to obtain external instruments that satisfy the two previous conditions, I use internal instruments in the form of lagged values of the instrumented variables (the right-hand side variables of the model). Such instruments obviously satisfy the first condition. Besides, current shocks in performance have no effect on historical characteristics; hence the instruments are uncorrelated with the error term and ultimately satisfy the second condition.

Although the first differenced GMM estimator uses all available instruments, it incorporates weak ones because they are inadequately correlated with the differenced predetermined variables (Alonso-Borrego and Arellano, 1999). In this situation, the autoregressive parameter (α_2) approaches unity and the relative variance of fixed effects (σ_n^2/σ_v^2) becomes high. Therefore, although no correlation exists between the instruments and the error term, this estimator does not properly explain the model. Consequently, I use a system GMM estimator that uses level equations in addition to difference equations (Blundell and Bond, 1998). System GMM solves for errors in levels; thus adds more instruments to the linear dynamic model and makes it more efficient and informative (Pindado and Requejo, 2014).

The equation in levels of the system GMM estimator, however, may still include unobserved heterogeneity. To solve this problem, I presume that the correlation between the variables to be instrumented and the unobserved individual effects (η_i) is constant over time (Pindado and Requejo, 2014). By applying the assumption on the explanatory variable, I find that the increments in (X) are uncorrelated with the unobserved heterogeneity $[E(\Delta X_{i,t-1} \eta_i)=0]$. In this way, I transform the instruments to be exogenous to η_i . Therefore, in addition to using the moment conditions of the first difference equations $[E(\Delta v_{it} X_{i,t-s})=0]$, I invest the previously explained lack of correlation in the level equations of the system GMM estimator according to Blundell and Bond (1998) as:

$$E(\Delta X_{i,t-1} \varepsilon_{it}) = E(\Delta X_{i,t-1} \eta_i) + E(\Delta X_{i,t-1} v_{it}) - E(\Delta X_{i,t-2} v_{it}) = 0 + 0 - 0 = 0$$
5.10

To justify the stationarity assumption, the unobservable individual effects identified earlier in the model have to be considered. The impacts of self-selection, corporate culture, inborn capabilities, personality, and the tendency to take risk are all constant over time. In addition, the change in the explanatory variable over time ($\Delta X_{i,t-1}$) does not depend on those unobserved factors, e.g., managerial ability is the aptitude to generate revenues from firm resources and thus depends on factors like the amount of experience and training managers receive. Therefore, I expect the correlation of the unobservable effects with the explanatory variable to remain constant over time. The same logic is applied to the remaining instrumented variables.

Nevertheless, because the thesis is covering a relatively long time period, remote lags are weak instruments. They are not highly correlated with the current values of the instrumented variables. The poor correlation violates the first condition of IV_{it} mentioned above. To avoid the decrease in statistical efficiency due to the high number of instruments, I reduce the width of the instruments matrix. Therefore, I choose the closest instruments (Pindado and Requejo, 2014), which include the first to third lagged values in the difference equations and only one lag in the level equations (Hillier et al., 2011).

Finally, I present the system GMM results compared to those obtained using the OLS and the within-groups estimators for two of the basic models in the thesis. Panel A

in Table 5.2 shows the results from examining the impact of managerial ability on management compensation that is examined in detail in Chapter 6 (see Table 6.5), while Panel B in the same table presents the effect of managerial ability on accrual earnings management discussed further in Chapter 8 (see Appendix C Table V). In line with the previous studies, the results show that the coefficients estimated by the OLS estimator are biased upwards while those estimated by the within-groups estimator are biased downwards (Hillier et al., 2011; Nickell, 1981). The system GMM estimator provides lowest error and highest significance compared to the other estimators as it solves for the problems of heterogeneity and endogeneity in these models as explained above.

Table 5.2 The se	election of	estimator
------------------	-------------	-----------

Panel A: The effect of managerial ability on management compensation					
Variable	Ordinary least squares estimator	Within-groups estimator	System generalized method of moments estimator		
MgrlAbility _{it}	0.085***	0.013	0.076***		
	(0.0166)	(0.012)	(0.001)		
z _I	598.723	2.854	10,640.780		
m_l			-1.490		
<i>m</i> ₂			-1.520		
Hansen			482.630		

Panel B: The effect of managerial ability on accrual earnings management

Variable	Ordinary least squares estimator	Within-groups estimator	System generalized method of moments estimator
MgrlAbility _{it}	0.988***	0.650**	0.973***
	(0.055)	(0.035)	(0.004)
z ₁ m.	144.293	4.634	627.450 -17.950
m_1 m_2			-0.760
Hansen			1,512.910

Notes: This table presents the key parameter estimates for two basic models examined in the thesis later. Panel A presents the effect of managerial ability on management compensation examined in Chapter 6 (see Table 6.5) and Panel B presents the effect of managerial ability on accrual earnings management examined in Chapter 8 (see Appendix C Table V). The most common estimators are used to solve the models in the following order: (1) ordinary least squares, (2) within-groups and (3) system generalized of moments estimators. Each coefficient represents the change in the dependent variable based on a one unit change in the determinant. Variables are defined in the (Variable Definitions) sections in Chapter 6 and 8. Standard errors are displayed in parentheses. The Wald test (z_1) checks for the joint significance of the reported coefficients in the model. The AB test (Arellano and Bond, 1991) examines the serial correlation in the first difference residual (v_{it}) by using the (m_j) statistic and takes the shape of a normal distribution with N(0,1). m_1 values represent the results of the 1st order serial correlation test, while m_2 values represent the results of the 2nd order serial correlation test. Hansen test for over-identifying restrictions takes the shape of χ^2 distribution and checks for the orthogonality between the instrumental variables (IV_{it}) and the error term (ε_{it}). *, **, *** denotes significance at 0.10, 0.05, and 0.01, respectively.

5.6 Conclusions

This chapter has explained the sample construction, data sources, the models used in estimating accrual and real earnings management, and the panel data methodology used in solving the empirical models of the thesis. Three different samples have been selected and different databases have been used for the three empirical chapters because of the different data requirements of each chapter.

For measuring accrual earnings management, I select Stubben's Model because it emphasizes the receivable accruals that are closely related to discretionary revenues, in addition to the Modified Jones Model which shows strong power in explaining total accruals (Dechow et al., 1995). For measuring real earnings management, I use the Gunny's Model and the Roychowdhury's Model because they measure different aspects of this activity.

Finally, to solve the empirical models of this thesis, panel data methodology is implemented using a system Generalized Methods of Moments (GMM) estimator that solves for the problems of unobservable individual heterogeneity and endogeneity encountered by those models. In this sense, the pooled OLS estimator, the fixed effects estimator, and the differenced Generalized Methods of Moments (GMM) estimators have been ignored.

Data and research methods identified in this chapter will be used in the next three chapters i.e., Chapters 6, 7, and 8, that represent the empirical chapters of the thesis. Each of these chapters will examine a specific question about the variables identified earlier. As will be discussed soon, Chapter 6 examines the impact of earnings management behaviour on executive compensation taking into consideration the variation in managerial abilities of those executives.

Chapter 6 Management Compensation: The Impact of Earnings Management and Managerial Ability

6.1 Introduction

This chapter examines management compensation and managerial ability, which form part of the contractual motives of earnings management, to explain the previous finding in the literature of high quality managers using more accrual and less real earnings management (Demerjian et al., 2013b). Compared to accrual earnings management that only allows discretionary choices of accrual accounting, real earnings management involves undertaking actions that have economic consequences for the firm, such as cutting R&D expenditure, and thus may adversely affect its future performance. However, it is questionable whether accrual and real earnings management influence managers' welfare by tarnishing their reputations and future compensation or allowing them to achieve more benefits. Therefore, this chapter examines the impact of earnings management on management compensation conditioned on managerial abilities.

If managerial quality was to be viewed as homogenous, then managers may use earnings management to signal better performance as a result of information asymmetry and incomplete contracting between managers and shareholders (Strong and Walker, 1987; Carter et al., 2009). However, shareholders monitor managerial activities by evaluating the wealth consequences of managerial actions and pay incentives to reinforce those that are likely to improve firm value in the future (Adut et al., 2013). In this sense, firms are likely to reward or punish earnings management activities based on the pay-performance sensitivity (PPS) in their compensation contracts with management (Strong and Walker, 1987; Hart, 1995; Dutta and Fan, 2014).

Managerial ability is, however, heterogeneous and allowing it to vary adds considerable complexity to the situation. Superior managers, therefore, may manage earnings to draw attention to their own performance, and shareholders may adjust their rewards by reducing their compensation. However, the PPS in firm contracts with more able managers is influenced by the higher costs of contracting that involve a lot more negotiation and rewriting, and the better skills of more able managers that result in a greater knowledge gap between shareholders and managers (Oberholzer-Gee and Wulf, 2012; Demerjian et al., 2013b). In support of this view, superior managers are likely to be aware of the consequences of their behaviour on their firms and personal benefits in the short- and long-run, and thus less monitoring might be required from shareholders to evaluate the performance of management (Oberholzer-Gee and Wulf, 2012; Walker, 2013). Therefore, it can be more difficult for shareholders to detect or completely understand the sophisticated behaviour of superior managers and they may be less likely to punish their value-reducing activities (Demerjian et al., 2013b).

Based on the documented individual effects of managerial ability and earnings management and the expectation that both influence firms' managerial compensation decisions, I examine how managerial ability influences the relation between earnings management and management compensation. I use total compensation that includes salaries, bonuses, option awards, stock awards, and long term incentive plans (LTIPs) (Adut et al., 2013). In measuring managerial ability, I follow the model developed by Demerjian et al. (2012) who define it as the leverage to generate revenues from firm resources. Finally, to measure earnings management I use both the Stubben (2010) and the Modified Jones (1995) models to measure accrual earnings management.

I contribute to the literature by documenting that firms punish their managers for real earnings management more than accrual earnings management. Furthermore, I find that firms respond more carefully when the compensation decision is related to top quality managers; and thus they punish/reward them less extremely than they do other managers. The results further explain those of Adut et al. (2013) by distinguishing between the effects of accrual and real earnings management on management compensation.

The remainder of this chapter proceeds as follows. In the next section, I review the related studies from the literature and build the hypotheses of the chapter. In section 6.3 I explain the process of data collection, variable definitions, and the descriptive statistics. In section 6.4 I explain the methodology used in this chapter. In section 6.5 I

135

present and explain the results. In section 6.6 I perform some additional analysis. Finally, in section 6.7 I conclude the chapter.

6.2 Literature and Hypotheses Development

The previous literature has documented that superior managers use more accrual and less real earnings management compared to other managers in the market (Demerjian et al., 2013b). To explain this behaviour, it is essential to understand how high quality managers trade-off between the benefits and costs of earnings management activities. In particular, I focus on management compensation which represents one of the major rents managers extract from their firms. To achieve the goal, I first examine how managerial ability and earnings management individually influence future management compensation. Then, I test the simultaneous effect of managerial ability and earnings management compensation.

6.2.1 Managerial Ability and Management Compensation

The principal-agent theory (Jensen and Meckling, 1976; Holmström and Milgrom, 1987) assumes that there are two parties (shareholders and managers) who attempt to maximize their expected payoffs (Stathopoulos et al., 2007). The separation of ownership and control between these parties leads to specialized risk bearing and specialized decision skills in the organization. While shareholders are assumed to carry the risk, managers are expected to possess the decision making capabilities regardless of their personal wealth (Strong and Walker, 1987).

The conflict in interests between the risk bearing shareholders and the decision making managers raises uncertainty due to the information asymmetry and potential moral hazard between the two parties. Therefore, an optimal structure of contracts is required that links managerial compensation to firm earnings in an optimal pay-performance structure (PPS) and hence motivates managers to exert effort and enhance earnings (Strong and Walker, 1987; Hart, 1995; Dutta and Fan, 2014). Furthermore, because of the strong relation between PPS and firm risk, an efficient contract ideally combines incentives and risk sharing (Stathopoulos et al., 2007). As a result, it helps in

resolving the previous conflict by minimizing the monitoring costs incurred by the principals and the bonding costs incurred by the agents (Jensen and Meckling, 1976; Harris and Raviv, 1979; Bergstresser and Philippon, 2006).

Managerial ability is likely to influence the PPS as able managers possess the skills to improve firm performance (Strong and Walker, 1987). In essence, shareholders tend to offer greater incentives to managers with better abilities to compensate them for their superior efforts and minimize the information gap - especially under high levels of uncertainty (Hart, 1995; Stathopoulos et al., 2007; Baranchuk et al., 2011). While the optimal incentive scheme allows the trade-off between optimal incentives and optimal risk sharing, it enables the firm to attract new superior managers and prevents the mobility of the current ones to competing firms - thus avoiding productivity crises (Eaton and Rosen, 1983; Hart, 1995; Hayes and Schaefer, 1999).

As high quality managers are able to implement more discretion in their decisions, there is higher uncertainty about the outcomes of such decisions in the short-run (Eaton and Rosen, 1983). Therefore, the current firm performance measures will be associated with more noise; and hence management compensation is likely to be determined by future results (Lambert and Larcker, 1987). In a multi period agency model, a firm can assess the labour marginal product in the long-run and thus reveal more information about the quality of managerial performance (Hayes and Schaefer, 1999). Consequently, I expect more able managers to be associated with higher compensation in the long-run and, therefore, I develop the following hypothesis.

H1: Managerial ability has a positive effect on future management compensation.

6.2.2 Earnings Management and Management Compensation

Prior research has focused on management compensation, particularly performancebased payments in the form of bonuses, stock grants, and stock options, as an incentive for earnings management behaviour (Healy, 1985; Gao and Shrieves, 2002; Cheng and Warfield, 2005; Graham et al., 2005; Bergstresser and Philippon, 2006; Burns and Kedia, 2006; Cohen et al., 2008; Carter et al., 2009; Feng et al., 2011; Oberholzer-Gee and Wulf, 2012). As compensation relies on reported earnings, managers attempt to reflect better performance and meet earnings' targets to extract more benefits. Goldman and Slezak (2006) document that managers are inclined to manage earnings upwards in the periods when they are expecting low rewards from their firms. In this sense, the association between earnings management and executive compensation in firms with low-compensation schemes implies a lack of efficiency in contracting (Sun, 2014). Dutta and Fan (2014) examine the same behaviour over a two-year period and document that managers shift earnings from periods with low pay-performance sensitivity to periods with high pay-performance sensitivity. Managers' opportunistic behaviour is supported by their limited liabilities, intentions to sell their stocks and options in the short-run, and their ability to influence the compensation decision that is made by the board of directors and not directly by the shareholders of the firm (Ronen and Yaari, 2008).

In contrast, some studies consider compensation as a governance mechanism that mitigates managerial discretion over financial reporting, especially under the long-term compensation plans where managers aim to reduce analysts' future expectations or decrease the exercise price of their option rewards (Richardson and Waegelein, 2002; O'Connell, 2004; Bauman et al., 2005; Coles et al., 2006; Carter et al., 2009). Therefore, while a compensation contract may fail to improve a firm's intrinsic value and eliminate earnings management when viewed as an incentive, it may be considered as a device that mitigates the conflict of interests between managers and shareholders when viewed as a governance mechanism according to the positive accounting theory (Sun, 2012; Dutta and Fan, 2014).

However, the empirical literature that examines the effect of earnings management on compensation has basically focused on the impact during the same period and documented a negative relation. Adut et al. (2013) point out that bad earnings quality, measured as the poor accruals mapping into cash flows, is associated with lower management compensation as firms tend to punish managers for providing less informative earnings. In other words, firms only offer compensation to managers when the net present value (NPV) of the additional incentive is positive i.e. the expected marginal outcome is more than the compensation paid; which is likely to occur when earnings quality is high. In essence, the better the earnings quality, the higher the tendency of firms to rely on earnings in determining management compensation (Peng, 2011). However, Adut et al. (2013) focus on the mapping of discretionary accruals into future cash flows according to the Dechow and Dichev (2002) model; which creates a link between accrual and real earnings management but does not distinguish between the individual effects of each activity per se.

Theoretically, the study of Sun (2014) develops a model to study the relation between earnings management and executive compensation. The pay-performance sensitivity increases when managers have more opportunities to manage earnings and shareholders are less likely to detect earnings management. Put differently, the additional incentives are required to eliminate the desire of management to manipulate earnings; and thus the positive association between earnings management and executive compensation reflects optimal contracting. As compensation incentives motivate managerial effort as well as the desire to manipulate earnings, managers are always expected to manage earnings as part of their optimal contracts with their firms. In a similar way, the study of Dutta and Fan (2014) develops a model to examine the effect of earnings management cost on managerial compensation through its impact on the pay-performance sensitivity. As the cost of earnings management decreases (e.g., due to the lack of effective governance mechanisms), managers are more inclined to use earnings management to extract rents and thus firms tend to pay lower compensation. In contrast, when the cost of earnings management increases (e.g., under strong governance mechanisms), managers are motivated to exert more effort which requires firms to pay higher compensation. Thus the study documents a positive relation between the cost of earnings management and managerial compensation.

Although earnings reflect the effort made by management during the current accounting period, such effort may influence firms' future cash flows (Basu, 1997; Barclay et al., 2005). Therefore, this chapter examines the effect of earnings management on future compensation to see whether shareholders reward or punish managers for this behaviour when detected. Because of the information asymmetry between insiders and outsiders, shareholders are imperfectly informed about firm performance. As shareholders might not distinguish the quality of earnings because of their bounded rationality, managers send signals in their reported earnings to improve the performance measures that are tied to their compensation plans (Strong and Walker, 1987; Carter et al., 2009). Earnings management assists in improving the security

139

market performance measures and hence results in better market valuation that allows managers to exercise larger amounts of their stock options and/or sell their stock compensation before the higher anomalous returns disappear (Bergstresser and Philippon, 2006). Furthermore, earnings management allows the improvement of some of the accounting measures that reflect a high quality of performance and allow managers to extract better salaries and bonuses (Lambert and Larcker, 1987; Strong and Walker, 1987).

On the other side of the agency problem, shareholders deal with the information asymmetry by screening the information provided by managers and sorting firms according to the quality of their earnings (Strong and Walker, 1987). Shareholders essentially take into account the consequences of the different earnings management activities before deciding on management compensation. As accrual earnings management unwinds in the next accounting period, it does not severely influence future cash flows and hence it would not cause harm to management compensation in the future as far as the net present value of the additional incentive is positive (Adut et al., 2013). In contrast, real earnings management is found to have severe negative consequences for subsequent operating and stock return performance (e.g., Cohen and Zarowin, 2010; Kothari et al., 2016) and thus is likely to have a negative effect on managers' future compensation. Accordingly, I develop the following two hypotheses.

H2a: Accrual earnings management has a positive effect on future management compensation;

H2b: Real earnings management has a negative effect on future management compensation.

6.2.3 Managerial Ability, Earnings Management, and Management Compensation

As shareholders might not be able to distinguish the quality of earnings because of their bounded rationalities, high quality managers are also motivated to use earnings management to signal a better performance quality. While shareholders still screen management signals based on the consequences of earnings management activities as mentioned above, they are more likely to detect earnings management when the conditions of the revelation principle are satisfied. These conditions include the low contracting costs that allow for perfect contracts between shareholders and management, shareholders' rationalities that enable them to maximize their wealth, the presence of common knowledge that assists shareholders in understanding management behaviour, and the costless communication due to the absence of third parties that might interfere with the transfer of information to shareholders (Walker, 2013).

The high contracting costs between firms and superior managers that include the need to negotiate and rewrite the contracts may render their contracts inadequate to monitor performance. In addition, management activities are unlikely to be detected because of the knowledge gap between the highly able managers and less informed shareholders and the tendency of the shareholders to trust the decisions of superior managers who possess the skills to successfully manage their firms (Oberholzer-Gee and Wulf, 2012; Demerjian et al., 2013b). Consequently, I develop the following two hypotheses.

H3a: Higher managerial ability lessens the positive effect of accrual earnings management on future management compensation;

H3b: Higher managerial ability lessens the negative effect of real earnings management on future management compensation.

6.3 Data, Variable Definitions and Descriptive Statistics

6.3.1 Data

I include all firms in the United States from the Annual Compustat, Quarterly Compustat, Historical Segments Compustat, and IBES databases. I also make use of the dataset made available by Demerjian et al. (2012) for the managerial ability variable during the same time period. ¹⁶ For management compensation data, I use the Execucomp database which only provides data from 1992. Among the controls for corporate governance, I use the number of board meetings, which is only available until 2006. As a result, I only cover the period from 1992 to 2006 in the analysis. Finally, I

¹⁶ Data are obtained from the following link:

https://community.bus.emory.edu/personal/PDEMERJ/Pages/Home.aspx.

exclude firm-years experiencing accounting changes, merger or acquisition activities, or discontinued operations.¹⁷

I start with all firms that have the required data for calculating the measures of earnings management, managerial ability, and management compensation. Following Cheng et al. (2011), I exclude regulated industries that contain banks, credit institutions, brokers, insurance, real estate, holding companies, and investment firms. These industries have their unique accounting and financial practices and are subject to distinct regulations. Therefore, managers in these industries have different motivations to manipulate earnings than those of managers in other industry sectors.¹⁸

Following the previous literature I exclude any group of firms with fewer than 6 observations for each SIC code to make sure that sufficient data exists to calculate earnings management measures and the Ordinary Least Square (OLS) assumption regarding the normality of the error term holds (e.g., Rosner, 2003; García Lara et al., 2005; Kothari et al., 2005; Iqbal et al., 2009; Athanasakou et al., 2011). For that purpose, I follow the SIC classification of Fama-French (1997). As the model is dynamic, I make sure that information is available for at least five consecutive years for each firm over the study period (Miguel et al., 2004). When combining between the cross-sectional and time-series dimensions of the data, I obtain total observations of 6,974 in an unbalanced panel dataset. I use an unbalanced panel to avoid survivorship bias.

6.3.2 Earnings Management Measures

Based on the discussion in Chapter 5, I use Stubben's Model (2010) for measuring accrual earnings management because it focuses on discretionary revenues which represent the largest component of earnings in most firms (Stubben, 2010). I also use Gunny's Model (2010) for real earnings management because it covers more types of this activity compared to other models (Gunny, 2010). In addition, I use two other widespread models as corroborating measures that include the Modified Jones' Model (1995) for measuring accrual earnings management and Roychowdhury's Model (2006)

¹⁷ According to McNichols (2002), I specifically exclude firm quarters or years with non-blank values for accounting changes cumulative effects (ACCCHGQ_FN), or merger and acquisition activities (ACQMETH_FN), or discontinued operations (DOQ_FN) in the Compustat database.

¹⁸ I exclude firms with the following SIC codes: $4000 \le SIC \ge 4900$ and $6000 \le SIC \ge 6300$.

for measuring real earnings management. Again, for each of these measures I use a cross sectional model in the calculations for each year and each industry classified by its four-digit SIC code with at least 6 observations. I also take the decile ranks for better comparability and to avoid the effect of the outliers. For the purpose of calculating an overall measure of real earnings management using Gunny's Model (2010) and Roychowdhury's Model (2006), a principal component analysis with varimax rotation is performed to avoid the inclusion of highly correlated variables in the overall measure (Demerjian et al., 2013b).¹⁹

6.3.3 Managerial Ability Measure

To calculate the managerial ability variable, I adopt the approach of Demerjian et al. (2012) as discussed in section 4.3.3. The measure represents managerial potential of generating revenues from the available firm resources. It uses Data Envelopment Analysis (DEA) to calculate firm efficiency then excludes some firm specific characteristics that may influence the performance of managers. The remaining residual (ε_i) is the ultimate measure of managerial ability (*MgrlAbility*). Finally, I take the decile ranks of these measures by industry-year to obtain better comparability and to avoid the effects of outliers.

6.3.4 Compensation Variables

I specifically focus on management benefits in the future because earnings management, and particularly real activities, have a long-term influence on firm performance and the consequences of the sophisticated decisions of high quality managers are more likely to be detected and understood over time. As I aim to examine future compensation as an outcome of the current earnings management behaviour, I emphasize total management benefits as the main measure of compensation. Although the type of compensation may influence earnings management behaviour when compensation is regarded as an incentive (Oberholzer-Gee and Wulf, 2012), I do not expect it to make any significant

¹⁹ The principal component analysis step turns the set of correlated variables to be linearly uncorrelated according to the weights of their variances, thus reducing the number of variables to their principal components.

difference when compensation is viewed as an outcome of the manipulation. I take the natural logarithm of the average over the next three years for the measure of management compensation I have identified (Adut et al., 2013).

I include all the executives' team of each firm in the measurement of compensation variables including the Chief Executive Officer (CEO) and the Chief Financial Officer (CFO) because those senior officers make the firms' major operating, financing, and investment decisions and are responsible for financial reporting under the approval of the board of directors. Therefore, they possess enough knowledge that can be used to achieve compensation benefits via earnings management (Ronen and Yaari, 2008). This point of view is based on studies that have attempted to distinguish between the individual members of the management team and provided evidence that both of the Chief Executive Officer (CEO) and the Chief Financial Officer (CFO) manage earnings in order to meet earnings benchmarks (Dechow and Sloan, 1991; Cheng and Warfield, 2005; Jiang et al., 2011).

6.3.5 Control Variables

In the control variables I include some firm characteristics like firm size (*TotalAssets*), sales volatility (*SalesVolatility*), cash flow volatility (*CashFlowVolatility*), operating cycle (*OperCycle*), and loss history (*Losses*) (Dechow and Dichev, 2002; Adut et al., 2013). I use a national auditor indicator (*NationalAuditor*) to control for an auditor effect (Becker et al., 1998), and another indicator variable to control for litigious industries (*LitigationInd*) (Francis et al., 1994). I add the market to book ratio (*MB*) and the one year sales growth (*SalesGrowth*) (Hribar and Nichols, 2007). I also control for analysts' coverage (*NumAnalyst*), industry revenue leadership (*IndRev%*), and returns momentum (*Momentum*) (Dechow et al., 2011; Zang, 2012). I use the number of segments (*Segments*) and the frequency of foreign transactions (*Foreign*) to control for the complexity of businesses (Karuna et al., 2012). Finally, I use a dummy for executives who serve as directors (*ExecDir*), the executive tenure (*Tenure*), and the number of board meeting (*BoardMtgs*) to control for the impact of corporate governance (Adut et al., 2013). I present the detailed calculations for each of the previous variables in Table 6.1.

Variable	Calculation
AccrualEM	The residual form using Stubben's Model (2010) as explained in the 'Variable Definitions' section above.
RealEM	The first component generated by using Gunny's Model (2010) that represents the discretionary reduction in SG&A expenses and the overproduction to cut prices or to decrease the cost of goods sold as explained in the ' <i>Variable Definitions</i> ' section above.
MgrlAbility	The potential of generating revenues from firm resources measured by the model of Demerjian et al. (2012) as explained in the ' <i>Variable Definitions</i> ' section above.
<i>TotalCompensation</i>	The natural log of the sum of salaries, bonuses, option awards, stock awards, and long term incentive plans (LTIPs) averaged over the next three years as explained in the ' <i>Variable Definitions</i> ' section above.
FirmSize	The natural log of the firm's assets as of the end of year <i>t</i> .
SalesVolatility	The standard deviation of (sales / average assets) over at least three of the last five years $(t-4, t)$.
CashFlowVolatility	The standard deviation of (cash from operations / average assets) over at least three of the last five years (t -4, t).
OperCycle	The operating cycle is the natural log of average sales turnover plus days in inventory over at least three of the last five years $(t-4, t)$.
Losses	Loss history is the percentage of years reporting losses in net income before extraordinary items over at least three of the last five years $(t-4, t)$.
NationalAuditor	An indicator variable equals one for firms audited by national audit firms in year t ; zero otherwise.
LitigationInd	Litigious industry indicator that equals one for firms in industries with SIC Codes: 2833-2836 (biotechnology), 3570-3577 and 7370-7374 (computers), 3600-3674 (electronics), and 52(X)-5961 (retailing).
MB	The market to book ratio that equals the firm's market capitalization divided by book value for year t.
SalesGrowth	Current year's sales less prior year's sales less the increase in receivables all scaled by prior year's sales and decile ranked by industry and year.
NumAnalyst	The log of 1+ the number of analysts covering the firm in year t.
IndRev%	Industry revenue leadership measured by the firm's sales in year t-1 divided by the total sales for the firm's industry in year t-1.
Momentum	Returns momentum calculated by the decile rank (by industry and year) of asset returns during the two years preceding the start of year t.
Segments	The natural log of 1+ the number of firm's business segments in year t.
Foreign	The frequency with which the firm has a non-zero foreign currency transactions during the sample period.
ExecDir	A dummy variable that equals 1 if the executive served as director during the year.
Tenure	The log of the executive tenure measured in days.
BoardMtgs	The number of board meetings held during the year.

Table 6.1 Calculation of all variables

Notes: This table presents the detailed calculations for each of the variables identified in the model as discussed in the (*Variable Definitions*) section above.

Larger firms have huge and more diversified operations and thus are likely to pay higher compensation to keep managers motivated to successfully manage the massive business (Dechow and Dichev, 2002; Adut et al., 2013; Demerjian et al., 2013a, 2013b). Although sales volatility, cash flow volatility, and the operating cycle reflect higher uncertainty in operations which is expected to negatively influence management compensation, it may result in higher payments especially in optimistic firms or those that aim to motivate managers to improve their business stability (Dechow and Dichev, 2002). Prior losses are likely to result in a lower compensation to management because of the lack of resources to make such payments, but they may also drive firms to pay more rewards so that managers are motivated to improve firm performance to meet investors' expectations (Dechow and Dichev, 2002).

Firms audited by national auditors are expected to pay lower compensation compared to those audited by international audit firms (Becker et al., 1998). I also expect firms in litigious industries to pay less compensation than those in other industry sectors to avoid any subsequent punishments (Francis et al., 1994). Market to book ratio, sales growth, and returns momentum reflect the growth prospects of a firm and hence may lead to a higher management compensation. Meanwhile, the high growth might reduce the need to make high payments to management (Hribar and Nichols, 2007; Dechow et al., 2011).

The presence of more analysts suggests higher management compensation to keep the motivation of meeting the analysts' expectations. I also expect industry leaders to make higher payments to their managers to be able to stay in their leadership positions. The number of segments and the foreign currency transactions reflect the complexity of the business and imply higher payments to compensate management for the greater efforts (Karuna et al., 2012). Finally, I introduce a dummy for executives who serve as directors, executive tenure, and the number of board meetings as controls for corporate governance. I expect these governance factors to significantly influence management compensation based on management performance (Adut et al., 2013). I summarize the expected and actual signs of all control variables in Table 6.2.

Variables	Compensation		
	Predicted sign	Actual sign	
FirmSize	+	+	
SalesVolatility	+/-	-	
CashFlowVolatility	+/-	+	
OperCycle	+/-	+	
Losses	+/-	+/-	
NationalAuditor	-	-	
LitigationInd	-	-	
MB	+/-	+/-	
SalesGrowth	+/-	+	
LnNumAnalyst	+	+	
IndRev%	+	+	
Momentum	+/-	+	
Segments	+	+	
Foreign	+	+	
<i>ExecDir</i> _{it}	+/-	+	
Tenure _{it}	+/-	-	
BoardMtgs _{it}	+/-	+/-	

Table 6.2 Expected and actual signs of the control variables

Notes: This table presents the signs expected and actually obtained for the effect of the control variables identified in the models and discussed in the (*Variable Definitions*) section above on management compensation.

6.3.6 Descriptive Statistics

I present the descriptive statistics in Table 6.3. Accrual earnings management, real earnings management, and managerial ability have means of zero. This is because they are calculated as the residuals from the selected models above. The measures of compensation are comparable to those of Adut et al. (2013). However, the differences are due to presenting the untransformed variables in the descriptive statistics. Similarly, the managerial ability measure is in line with that calculated by Demerjian et al. (2013b). The high standard deviations associated with the compensation measures, firm size, and tenure are due to presenting the untransformed variables.

Variables	Mean	Median	Std Dev	P25	P75
AccrualEM	0.00	0.00	0.05	-0.01	0.01
RealEM	0.00	-0.00	1.00	-0.72	0.70
MgrlAbility	-0.00	-0.01	0.15	-0.09	0.07
TotalCompensation	654.53	493.62	682.93	343.73	731.95
Salary	388.02	348.20	203.08	254.90	475.96
Bonus	266.48	122.21	591.56	32.60	289.36
OptionAwards	483.67	222.68	997.01	24.63	595.90
StockAwards	825.78	462.61	1,295.90	151.55	1,067.88
LTIP	69.07	0.00	261.87	0.00	0.00
TotalAssets	2,835.91	719.07	9,333.30	275.55	2,054.50
SalesVolatility	0.22	0.14	0.31	0.08	0.25
CashFlowVolatility	0.07	0.05	0.15	0.03	0.09
OperCycle	127.19	117.05	69.68	80.58	163.04
Losses	0.17	0.00	0.26	0.00	0.25
NationalAuditor	0.03	0.00	0.16	0.00	0.00
LitigationInd	0.14	2.69	0.34	0.00	0.00
MB	3.94	0.10	14.22	1.69	4.44
SalesGrowth	0.24	4.00	4.06	0.02	0.22
NumAnalyst	3.76	0.04	1.66	3.00	5.00
IndRev%	0.12	0.14	0.18	0.01	0.15
Momentum	0.10	1.00	0.32	0.01	0.22
Segments	2.18	0.00	1.77	1.00	3.00
Foreign	0.25	481.33	0.35	0.00	0.44
ExecDir	0.25	1.00	0.33	1.00	1.00
Tenure	2,833.87	2,010.00	2,684.47	974.00	3,744.00
BoardMtgs	7.32	7.00	3.28	5.00	9.00

Table 6.3 Descriptive statistics

Notes: The table presents the descriptive statistics of all the variables identified in the model for 6,974 firm-year observations obtained from Compustat from 1992 to 2006. Variables are defined in the (*Variable Definitions*) section above. I present the untransformed variables for ease of interpretation.

Table 6.4 represents the correlation matrix between the main variables in this chapter: earnings management, managerial ability, and compensation. The correlations between the control variables have not been displayed here for easier demonstration. To control for the multicollinearity between the different variables, however, I make sure that all Variance Inflation Factors (VIF) are less than 10. Managerial ability is positively correlated with accrual earnings management and negatively correlated with real

Table 6.4 Correlation tab	ole								
	Accrual EM	Real EM	Mgrl Ability	Total Compensation	Salary	Bonus	Option Awards	Stock Awards	LTIP
Accrual EM									
Real EM	0.00								
Mgrl Ability	0.05***	-0.05***							
Total Compensation	-0.01	-0.01*	0.08***						
Salary	-0.00	0.01**	0.04***	0.56***					
Bonus	-0.01	-0.02***	0.07***	0.95***	0.31***				
Option Awards	0.01	-0.01	0.05***	0.23***	0.31***	0.13***			
Stock Awards	0.00	-0.02	0.00	0.35***	0.44***	0.21***	0.16***		
LTIP	0.00	-0.01	0.00	0.25***	0.31***	0.20***	0.00	0.00	

Notes: This table presents the correlation matrix for 6,974 firm-year observations obtained from Compustat from 1992 to 2006 between the measures of accrual earnings management measured by Stubben's Model (2010), real earnings management measured by Gunny's Model (2010), managerial ability measured by the model of Demerjian et al. (2012), and total management compensation and its individual components that include salaries, bonuses, option awards, stock awards, and long term incentive plans (LTIPs) averaged over the next three years. Variables are defined in the (*Variable Definitions*) section above. Pearson correlation coefficients are presented. To control for the multicollinearity between the different variables I make sure that all VIF factors are less than 10. VIF factors, however, are not tabulated. *, **, *** denotes a statistical coefficient at the 10, 5 and 1 percent alpha level, respectively.

earnings management; which suggests that high quality managers are more inclined to use accrual rather than real earnings management. Managerial ability is also positively correlated with all compensation measures; which implies that more able managers receive higher compensation. The significant correlations between earnings management and compensation measures show mainly a negative correlation between real earnings management and total management compensation; which suggests that firms are inclined to punish this kind of behaviour but not accrual earnings management.

6.4 Methodology

To test the first hypothesis (*H1*), I derive a model that identifies management compensation as the dependent variable ($Y_{i,t+1,t+3}$). I introduce managerial ability (*MgrlAbility*_{it}) as an explanatory variable (X_{it}).²⁰ To complete the model, I add the previously identified control variables (*Controls*_{it}) and an error term (ε_{it}).

The model suffers from the problem of unobservable individual heterogeneity attributed to time-invariant firm and/or managerial effects. For example, the selection of managers by the boards of directors in line with their firms' strategies or corporate cultures results in managers with specific abilities to end up in firms with specific compensation schemes. Similarly, hiring managers with specific inborn capabilities, personalities, or tendencies to take risk leads managers with certain skills to arrive at firms with specific compensation schemes. Consequently, the sample is expected to be heterogeneous, as some companies would pay higher compensations to their managers than others because of the previous specificities (Graham et al., 2012; Demerjian et al., 2013b).

In addition, the model suffers from an endogeneity problem because of the mutual causality between managerial ability and compensation. While more able managers are expected to be rewarded as they contribute to better performance over time, they are attracted to stay or join firms that are expected to pay higher compensation. This effect would be captured in the error term and ultimately results in a

 $^{^{20}}$ MgrlAbilityDum_{it} is a dummy variable that equals 1 if the ability of a manager in an industry is high (greater than the median) and zero otherwise.

correlation between the explanatory variable and the error term (Pindado and Requejo, 2014).

Therefore, the OLS estimator will not be able to solve the model as it ignores the impact of the unobservable individual heterogeneity or endogeneity problems. Furthermore, while the fixed effects estimator tackles the unobservable heterogeneity by demeaning the variables in the model it does not solve for the endogeneity problem as it assumes strict exogeneity. Therefore, I solve the model using a system generalized method of moments (GMM) estimator that demeans the variables in the model to solve for the heterogeneity and introduces instrumental variables to solve for the endogeneity problem as explained earlier in the methodology section of Chapter 5.²¹

As a result, I split the error term (ε_{it}) it into three components to avoid the consequent bias. First, I introduce (η_i) to control for the impact of the unobserved effects in the model.²² Second, I add a time specific effect (d_t) to control for the macroeconomic variables that also interfere with the results over the time period of the study.²³ Finally, I consider the remaining part of the error term (ε_{it}) a random disturbance (v_{it}) . Consequently, I express the model in general terms in equation (6.1) and in more specific terms in equation (6.2).

$$Y_{i,t+1,t+3} = \alpha_0 + \alpha_1 X_{it} + \alpha_{2-18} Controls_{it} + \varepsilon_{it}$$
(6.1)

$$Compensation_{i,t+1,t+3} = \alpha_0 + \alpha_1 MgrlAbility_{it} + \alpha_2 FirmSize_{it} + \alpha_3 SalesVolatility_{i,t} + \alpha_4 CashFlowVolatility_{i,t-4,t} + \alpha_5 OperCycle_{i,t-4,t} + \alpha_6 Losses_{i,t-4,t} + \alpha_7 NationalAuditor_{it} + \alpha_8 SalesGrowth_{rk_{it}} + \alpha_9 MB_{it} + \alpha_{10} LitigationInd_{it} + \alpha_{11} LnNumAnalyst_{it} + \alpha_{12} IndRev\% + \alpha_{13} Momentum_{it} + \alpha_{14} Segments_{it} + \alpha_{15} Foreign_{it} + \alpha_{16} ExecDir_{it} + \alpha_{17} Tenure_{it} + \alpha_{18} BoardMtgs_{it} + \eta_i + d_t + v_{it}$$
(6.2)

To test the second hypothesis (H2), I again derive a model that identifies management compensation as the dependent variable $(Y_{i,t+1,t+3})$. I introduce earnings management

²¹ Solving for these two problems using a system GMM estimator allows us to obtain less biased and more significant results compared to the previous studies.

 $^{^{22}}$ η_i controls for both firm specific effects and manager specific effects.

²³ I do not tabulate the coefficients of time periods later in the results.

(*EarningsManagement*_{*it*}) as an explanatory variable (Z_{it}). To complete the model, I add the previously identified control variables (*Controls*_{*it*}) and an error term (ε_{it}).

This model also suffers from the problem of unobservable individual heterogeneity attributed to time-invariant firm effects. The selection of earnings management activities in line with firms' strategies results in firms rewarding/punishing earnings management behaviour through specific compensation schemes in the future. Consequently, the sample is expected to be heterogeneous, as some companies are more likely to reward earnings management behaviour than others (Graham et al., 2012; Demerjian et al., 2013b).

In addition, the model suffers from an endogeneity problem because of the mutual causality between earnings management and future compensation. While earnings management behaviour would be reflected on firm performance in the future, it is likely to influence management compensation. On the other hand, the expected compensation would set the current motivation to manage earnings. This effect is captured in the error term and ultimately results in a correlation between the explanatory variable and the error term (Pindado and Requejo, 2014).

Because of the previously explained shortcomings of the OLS and the fixed effects regressions, I solve this model using a system generalized method of moments (GMM) estimator and split the error term (ε_{it}) into three components (η_i , d_t , and v_{it}). I express the model in general terms in equation (6.3) and in more specific terms in equation (6.4).

$$Y_{i,t+1,t+3} = \alpha_0 + \alpha_1 Z_{it} + \alpha_{2-18} Controls_{it} + \varepsilon_{it}$$
(6.3)

$$Compensation_{i,t+1,t+3} = \alpha_{0} + \alpha_{1}EarningsManagement_{it} \\ + \alpha_{2}FirmSize_{it} + \alpha_{3}SalesVolatility_{i,t-4,t} \\ + \alpha_{4}CashFlowVolatility_{i,t-4,t} + \alpha_{5}OperCycle_{i,t-4,t} \\ + \alpha_{6}Losses_{i,t-4,t} + \alpha_{7}NationalAuditor_{it} \\ + \alpha_{8}SalesGrowth_{rk_{it}} + \alpha_{9}MB_{it} + \alpha_{10}LitigationInd_{it} \\ + \alpha_{11}LnNumAnalyst_{it} + \alpha_{12}IndRev\% + \alpha_{13}Momentum_{it} \\ + \alpha_{14}Segments_{it} + \alpha_{15}Foreign_{it} + \alpha_{16}ExecDir_{it} \\ + \alpha_{17}Tenure_{it} + \alpha_{18}BoardMtgs_{it} + \eta_{i} + d_{t} + v_{it}$$

$$(6.4)$$

Finally, to test the third hypotheses (H3), I again derive a model that identifies management compensation as the dependent variable $(Y_{i,t+1,t+3})$. I introduce managerial ability (*MgrlAbility_{it}*) and earnings management (*EarningsManagement_{it}*) as explanatory variables (X_{it} and Z_{it} respectively). I also add the interaction term of earnings management (*EarningsManagement_{it}*) and managerial ability (*MgrlAbility_{it}*).²⁴ To complete the model, I add the previously identified control variables (*Controls_{it}*) and an error term (ε_{it}). This model also suffers from the problems of unobservable individual heterogeneity and endogeneity as explained above (Pindado and Requejo, 2014), and, therefore, I solve it using a system generalized method of moments (GMM) estimator. However, I also present the results of the OLS regression as a robustness test in Appendix A Table II. I express the model in general terms in equation (6.5) and in more specific terms in equation (6.6).

$$Y_{i,t+1,t+3} = \alpha_0 + \alpha_1 Z_{it} + \alpha_2 X_{it} + \beta_1 X_{it} Z_{it} + \alpha_{3-19} Controls_{it} + \varepsilon_{it}$$
(6.5)

$$Compensation_{i,t+1,t+3} = \alpha_{0} + \alpha_{1}EarningsManagement_{it} + \alpha_{2}MgrlAbility_{it} + \beta_{1}MgrlAbility_{it} * EarningsManagement_{it} + \alpha_{3}FirmSize_{it} + \alpha_{4}SalesVolatility_{i,t-4,t} + \alpha_{5}CashFlowVolatility_{i,t-4,t} + \alpha_{6}OperCycle_{i,t-4,t} + \alpha_{7}Losses_{i,t-4,t} + \alpha_{8}NationalAuditor_{it} + \alpha_{9}SalesGrowth_{rk_{it}} + \alpha_{10}MB_{it} + \alpha_{11}LitigationInd_{it} + \alpha_{12}LnNumAnalyst_{it} + \alpha_{13}IndRev\% + \alpha_{14}Momentum_{it} + \alpha_{15}Segments_{it} + \alpha_{16}Foreign_{it} + \alpha_{17}ExecDir_{it} + \alpha_{18}Tenure_{it} + \alpha_{19}BoardMtgs_{it} + \eta_{i} + d_{t} + v_{it}$$
(6.6)

After examining the individual effect of the moderating variable ($MgrlAbilityDum_{it}$) and the independent variable ($EarningsManagement_{it}$), I test for their combined influence. For this purpose, I use a linear restriction test (LRT) that examines the significance of ($\alpha_I + \beta_I$). The results of this test are presented in Table 6.7 and show that the combined effect of managerial ability and earnings management is statistically significant.

²⁴ MgrlAbilityDum_{it} is a dummy variable that equals 1 if the ability of a manager in an industry is high (greater than the median) and zero otherwise. Therefore, α_1 is the coefficient of (*EarningsManagement_{it}*) when managerial ability is low; ($\alpha_1 + \beta_1$) is the coefficient when managerial ability is high.

To ensure that the assumptions of the estimator hold and the model is valid, I initially test whether the GMM estimator properly addresses the problem of endogeneity. For this purpose, I use the Hansen test for over-identifying restrictions. It takes a χ^2 distribution and checks for the orthogonality between the instrumental variables (IV_{it}) and the error term (ε_{it}). The models use multiple lags of the right-hand side variables as instruments, which make them over-identified. Consequently, if I accept Hansen's null hypothesis that the instrumental variables (IV_{it}) and the error term (ε_{it}) are uncorrelated, I guarantee that the instruments are valid and that the estimator is appropriate. I present the results of Hansen test in tables 6.5, 6.6 and 6.7, which show that the instruments are valid.

Additionally, I implement Arellano and Bond (1991) to check for the validity of the model. It takes the shape of a normal distribution with N(0,1). It mainly examines the serial correlation in the first difference residual (v_{it}) over different periods (s) by using the (m_j) statistic $[E(v_{it} v_{is})=0)$, while $t\neq s$]. I accept first order serial correlation in the model because the estimator takes the first difference to eliminate the individual specific effects (η_i) . However, I reject second order serial correlation (m_2) in the residual because it indicates a problem in the model. I present the results of the AB test in tables 6.5, 6.6 and 6.7, which confirm that no second order serial correlations exist in the model.

Finally, I use the Wald test (z_1) to check for the joint significance of the reported coefficients in the model. If I reject the null hypothesis that states no relationship between the dependent and explanatory variables, I make sure that the model is jointly significant. I present the results of the Wald test in tables 6.5, 6.6 and 6.7. They all support the joint significance of the reported coefficients.

6.5 Results

6.5.1 Managerial Ability and Management Compensation

I start by examining the effect of managerial ability on future management compensation. The results appear in the first row of Table 6.5 and show a significant positive coefficient of managerial ability variable $(+0.076^{***})$; which supports

hypothesis H1 of this chapter. The result suggests that high quality managers receive higher compensation in the future.

Variables	Total management
variables	compensation
	-
MgrlAbility _{it}	0.076***
FirmSize _{it}	0.073***
SalesVolatility _{i,t-4,t}	-0.026***
CashFlowVolatility _{it-4,t}	0.229***
<i>OperCycle</i> _{<i>i</i>,<i>t</i>-4,<i>t</i>}	0.005***
Losses _{i,t-4,t}	0.006***
NationalAuditor _{it}	-0.009**
LitigationInd _{it}	-0.025***
["] MB _{it}	-0.000***
Sales Growth _{rkit}	0.011***
LnNumAnalyst _{it}	0.030***
IndRev%	0.111***
Momentum _{it}	0.057***
Segments _{it}	0.004***
Foreign _{it}	0.077***
ExecDir _{it}	0.069***
Tenure _{it}	-0.013***
BoardMtgs _{it}	-0.000
Hansen	482.630
m_1	-1.490
m_2	-1.520
71	10.640.780

Table 6.5 The effect of managerial ability on management compensation

Notes: This table presents the results from the system generalized method of moments regressions for the effect of managerial ability measured by the model of Demerjian et al. (2012) on total management compensation measured as the natural logarithm of the sum of salaries, bonuses, option awards, stock awards, and long term incentive plans (LTIPs) averaged over the next three years. The sample includes 6,974 firm-year observations obtained from Compustat from 1992 to 2006. Each coefficient represents the change in management compensation based on a one unit change in the determinant. Variables are defined in the (Variable Definitions) section above. Hansen test for over-identifying restrictions takes the shape of χ^2 distribution and checks for the orthogonality between the instrumental variables (IV_{it}) and the error term (ε_{it}). The AB test (Arellano and Bond, 1991) examines the serial correlation in the first difference residual (v_{it}) by using the (m_j) statistic and takes the shape of a normal distribution with N(0,1). m_1 values represent the results of the 1st order serial correlation test, while m_2 values represent the results of the 2nd order serial correlation test. The Wald test (z_1) checks for the joint significance at 0.10, 0.05, and 0.01, respectively.

Firms always try to resolve the conflict of interests arising from the separation of ownership from control and thus create a balance between risk bearing and decision making between shareholders and managers. They can achieve this target by introducing a suitable pay-performance structure into their contracts with managers to encourage them to share some risk under the condition of paying higher rewards for the better performance. As high quality managers possess the abilities to take more risk and improve performance, firms compete in attracting them to join or stay by rewarding their superior efforts. Such rewards tend to increase over time as firms gradually realize the consequences of their sophisticated decisions.

6.5.2 Accrual Earnings Management and Management Compensation

The results appear in the first line of the first column of Table 6.6 and represent the average effect of accrual earnings management on management compensation. They show a significant positive coefficient of accrual earnings management variable $(+0.004^{***})$; which suggests that accrual earnings management results in higher management compensation in the future. The finding supports hypothesis *H2a* of this chapter.

Accrual earnings management contributes to improving the measures used in evaluating management performance. However, it does not influence future cash flows as it reverses in the subsequent period. Therefore, accruals' manipulation is perceived as a harmless activity to the firms and thus managers use accrual earnings management as a tool to signal better performance and extract more rewards in the future. In this sense, it can be considered as a less costly activity from the managers' point of view.

The results contribute to the previous literature by specifically explaining the relation between accrual earnings management and management future compensation. I support the argument of Gao and Shrieves (2002) who document that accruals' manipulation merely gives managers a timing option to maximize their compensation. I agree with Dutta and Fan (2014) who document that firms generally pay higher compensation when they do not expect management to extract high benefits from earnings manipulation. I also agree with Sun (2014) who documents that firms highly reward their managers whenever there is a potential of earnings management. However, while Adut et al. (2013) document that shareholders generally punish their managers for low earnings quality, I find that they do not particularly penalize for accrual earnings management because it is not perceived as costly to their firms.

Chapter 6				
Management Compensation:	The Impact of Earnings	Management and	Managerial .	Ability

	(1)	(2)
Variables	Total management	Total management
	compensation	compensation
AccrualEarningsManagement _{it}	0.004***	-
RealEarningsManagement _{it}	-	-0.005***
<i>FirmSize_{it}</i>	0.074***	0.067***
Sales Volatility _{i,t-4,t}	-0.043***	-0.047***
CashFlowVolatility _{i1-4}	0.269***	0.194***
<i>OperCycle</i> _{<i>i</i>,<i>t</i>,4}	0.007***	0.009***
Losses _{it-4,t}	-0.007**	-0.020***
NationalAuditor _{it}	-0.023***	-0.044***
LitigationInd _{it}	-0.014***	-0.024***
 MB _{it}	0.000***	-0.000***
Sales Growth _{rk_{it}	0.021***	0.028***
LnNumAnalyst _{ir}	0.027***	0.035***
IndRev%	0.136***	0.141***
Momentum _{it}	0.060***	0.069***
Segments _{it}	0.011***	0.002***
 Foreign _{ir}	0.076***	0.064***
ExecDir _{it}	0.070***	0.088***
Tenure _{it}	-0.012***	-0.006***
BoardMtgs _{it}	0.001***	0.001***
Hansen	469.660	478.940
m_1	-1.500	-1.440
m_2	-1.440	-1.380
z_l	14,716.530	7,718.130

Table 6.6 The effect of earnings management on management compensation

Notes: This table presents the results from the system generalized method of moments regressions for the effect of earnings management (in the following order: (1) accrual earnings management measured by Stubben's Model (2010), and (2) real earnings management measured by Gunny's Model (2010)) on total management compensation measured as the natural logarithm of the sum of salaries, bonuses, option awards, stock awards, and long term incentive plans (LTIPs) averaged over the next three years. The sample includes 6,974 firm-year observations obtained from Compustat from 1992 to 2006. Each coefficient represents the change in total management compensation based on a one unit change in the determinant. Variables are defined in the (*Variable Definitions*) section above. Hansen test for over-identifying restrictions takes the shape of χ^2 distribution and checks for the orthogonality between the instrumental variables (IV_{it}) and the error term (ϵ_{it}). The AB test (Arellano and Bond, 1991) examines the serial correlation in the first difference residual (v_{it}) by using the (m_j) statistic and takes the shape of a normal distribution with N(0,1). m_1 values represent the results of the 1st order serial correlation test, while m_2 values represent the results of the 2nd order serial correlation test. The Wald test (z_1) checks for the joint significance of the reported coefficients in the model. Intercept is included, but not tabulated. *, **, *** denotes significance at 0.10, 0.05, and 0.01, respectively.

6.5.3 Real Earnings Management and Management Compensation

The results appear in the first line of the second column of Table 6.6 and represent the average effect of real earnings management on management compensation. They show a significant negative coefficient of real earnings management variable (-0.005^{***}); which suggests that real earnings management results in a lower management compensation in the future. The finding supports hypothesis *H2b* of this chapter.

Although real earnings management can be used by managers as a tool to send signals that reflect better current performance, it seriously affects firm future performance as it involves economic actions that may result in sacrificing future projects and losing cash flows. Therefore, real earnings management is perceived as a harmful activity and firms punish their managers for using it. In this sense, shareholders screen real earnings management as a sign of bad performance and poor earnings quality and pay less rewards to managers in the future; hence it can be considered as a more costly activity from management perspective.

Taking the previous studies into consideration, the results agree with those of Adut et al. (2013) in terms of punishing managers for manipulating earnings. However, I document that the penalty occurs when real earnings management specifically takes place as it is a costly activity from the firm's perspective. This part of the results also supports the argument of Dutta and Fan (2014) that firms generally pay lower rewards when they believe that management is already extracting high benefits from earnings management. Finally, the generalization in the study of Sun (2014) regarding how expected earnings management behaviour drives firms to make higher compensation payments does not apply on detrimental activities like real earnings management.

6.5.4 Managerial Ability, Accrual Earnings Management and Management Compensation

The results appear in the second line of the first column of Table 6.7. They show a significant negative coefficient of the interaction term; which suggests that managerial ability significantly influences the relation between accrual earnings management and management compensation in the future. The finding supports hypothesis H3a of this chapter. The coefficient of accrual earnings management that drives superior managers'

compensation in the future ($\propto_1+\beta_1=0.006-0.003=0.003$) is less than that for other managers ($\propto_1=0.006$). In addition, the *t* value of the linear restriction test (LRT) in Table 6.7 shows that the combined effect of managerial ability and accrual earnings management on future management compensation is significantly different from zero. Similar results are obtained by using the Modified Jones' Model (1995) as shown in Appendix A Table I and the OLS estimator as shown in Appendix A Table II.

Shareholders tend to believe in superior managers because of their high skills and thus expect them to demonstrate better performance without the need of high monitoring efforts. However, superior managers keep sending signals to show their distinctive efforts in improving the business under the fear that the less sophisticated shareholders will not be able to directly realize their sophisticated skills compared to those of other managers in the market. Therefore, high quality managers are also inclined to manage earnings. Because accrual earnings management does not require high skills to detect and understand, it can be easily screened by the shareholders; and thus firms tend to punish high quality managers for such activity more severely than other managers. However, the overall effect of accrual earnings management on future compensation is still positive which justifies why superior managers continue to use more accrual earnings management. Therefore, although accrual earnings management is generally considered less costly than real earnings management from managers' perspective, it is relatively more costly to superior managers than to other managers.

The results contribute to the previous literature by showing how accrual earnings management behaviour particularly influences the future compensation of high quality managers. The results are still in line with those of Dutta and Fan (2014) as firms continue to pay higher compensation to superior managers when they do not expect them to achieve high benefits from their opportunistic behaviour. Similarly, I continue to support the results of Sun (2014) as firms still reward the high quality managers when they expect a chance of earnings manipulation. Finally, the results here extend the work of Adut et al. (2013) by demonstrating how shareholders punish superior managers for low earnings quality compared to other managers.

Standbernent Compensations The Impact of Zam		Bernar i te mity
Table 6.7 The effect of earnings management and manage	verial ability on management con	nnensation
Tuble of The effect of curnings manufement and manuf	(1)	(2)
Variables	Total management	Total management
	compensation	compensation
AccrualEarningsManagement _{it}	0.006***	-
MgrlAbility _{it} *Accrual EarningsManagement _{it}	-0.003***	-
RealEarningsManagement	-	-0.003***

MgrlAbility _{it} "Accrual EarningsManagement it	-0.003***	-
<i>RealEarningsManagement</i> _{it}	-	-0.003***
MgrlAbility _{it} *Real EarningsManagement _{it}	-	0.001***
MgrlAbility _{it}	0.013***	0.017***
<i>FirmSize_{it}</i>	0.079***	0.070***
SalesVolatility _{i.t-4.t}	-0.055***	-0.054***
CashFlowVolatility _{i,t-4,t}	0.320***	0.281***
<i>OperCycle</i> _{<i>i</i>,<i>t</i>-4,<i>t</i>}	0.006***	0.002**
$Losses_{i,t-4,t}$	0.002	-0.019***
NationalAuditor _{it}	-0.013**	-0.043***
LitigationInd _{it}	-0.017***	-0.024***
··· ··· ··· ··· ··· ··· ··· ··· ··· ··	0.000***	-0.000***
Sales Growth _{rkit}	0.010***	0.013***
LnNumAnalyst _{it}	0.034***	0.034***
IndRev%	0.135***	0.134***
<i>Momentum_{it}</i>	0.066***	0.067***
Segments _{it}	0.016***	0.004***
Foreign _{ir}	0.083***	0.064***
ExecDir _{it}	0.077***	0.089***
Tenure _{it}	-0.013***	-0.010***
BoardMtgs _{it}	0.001***	0.000***
t	-23.577	19.229
Hansen	467.370	464.170
m_1	-1.570	-1.460
m_2	-1.420	-1.430
z_1	6,902.840	3,350.260

Table 6.7 The effect of earnings management and managerial ability on managemen	it compensation
---	-----------------

Notes: This table presents the results from the system generalized method of moments regressions for the effect of managerial ability measured by the model of Demerjian et al. (2012) and earnings management (in the following order: (1) accrual earnings management measured by Stubben's Model (2010), and (2) real earnings management measured by Gunny's Model (2010)) on total management compensation measured as the natural logarithm of the sum of salaries, bonuses, option awards, stock awards, and long term incentive plans (LTIPs) averaged over the next three years. The sample includes 6,974 firm-year observations obtained from Compustat from 1992 to 2006. Each coefficient represents the change in total management compensation based on a one unit change in the determinant. Variables are defined in the (Variable Definitions) section above. The t values of the linear restriction tests (LRT) reflect the joint significance of the explanatory and moderating variables. The test is performed under the null hypothesis $H_0: \alpha_1 + \beta_1 = 0$. Hansen test for over-identifying restrictions takes the shape of χ^2 distribution and checks for the orthogonality between the instrumental variables (IV_{it}) and the error term (ϵ_{it}) . The AB test (Arellano and Bond, 1991) examines the serial correlation in the first difference residual (v_{it}) by using the (m_j) statistic and takes the shape of a normal distribution with N(0,1). m_1 values represent the results of the 1st order serial correlation test, while m_2 values represent the results of the 2nd order serial correlation test. The Wald test (z_1) checks for the joint significance of the reported coefficients in the model. Intercept is included, but not tabulated. *, **, *** denotes significance at 0.10, 0.05, and 0.01, respectively.

6.5.5 Managerial Ability, Real Earnings Management and Management Compensation

The results appear in the second line of the second column of Table 6.7. They show a significant positive coefficient of the interaction term; which suggests that managerial ability significantly influences the relation between real earnings management and management compensation in the future. The finding supports hypothesis *H3b* of this chapter. The coefficient of real earnings management that drives superior managers' compensation in the future ($\alpha_2+\beta_1=-0.003+0.001=-0.002$) is higher than that for other managers ($\alpha_2=-0.003$). The *t* value of the linear restriction test (LRT) in Table 6.7 shows that the combined effect of managerial ability and real earnings management on future management compensation is significantly different from zero. Similar results are obtained by using the Roychowdhury's Model (2006) as shown in Appendix A Table I and the OLS estimator as shown in Appendix A Table II.

High quality managers are also inclined to send signals through real earnings management. However, real earnings management is more complicated for shareholders to detect or understand compared to accrual earnings management; and thus it cannot be easily screened. Therefore, firms tend to reward superior managers for their sophisticated behaviour that demonstrates their high skills and specialized knowledge. Nevertheless, the overall effect of real earnings management on future compensation is still negative which again justifies why superior managers continue to avoid real earnings management. In this sense, although real earnings management is generally considered more costly than accrual earnings management from managers' perspective, it is not as costly to superior managers as to other managers.

The results build on those of Adut et al. (2013) by showing that firms ultimately punish superior managers for real earnings management behaviour which is still considered as a costly activity to the firm. I also confirm the work of Dutta and Fan (2014) and demonstrate that firms pay overall lower rewards when they realize that high quality managers can extract benefits from real earnings management. Finally, I extend the work of Sun (2014) and show how expected earnings management behaviour relatively drives firms to make higher compensation payments but only to superior managers when using real earnings management.

161

6.6 Additional Analysis: Managerial Ability, Earnings Management and Operating Performance

To further explain the influence of earnings management on management future compensation, I try to examine how earnings management influences firm future performance that will subsequently determine management compensation. I particularly focus on earnings management behaviour of high quality managers and its impact on operating performance (*OpPerformance*) measured as the average return on assets (ROA) over the next three-year-period. Therefore, I develop the following model that identifies operating performance as the dependent variable (*OpPerformance*_{*i*,*t*+1,*t*+3}), earnings management (*EarningsManagement*_{*it*}) and managerial ability(*MgrlAbility*_{*it*}) as explanatory variables, in addition to the interaction term between them. ²⁵ I add the control variables and an error term that is split into $(n_i)^{26}$ to control for the impact of the unobserved effects, $(d_i)^{27}$ to control for the macroeconomic variables, and (v_{it}) as a random disturbance.

$$OpPerformance_{I,t+1,t+3} = \alpha_{0} + \alpha_{1}EarningsManagement_{it} + \alpha_{2}MgrlAbility_{it} + \beta_{1}MgrlAbility_{it} * EarningsManagement_{it} + \alpha_{3}FirmSize_{it} + \alpha_{4}SalesVolatility_{i,t-4,t} + \alpha_{5}CashFlowVolatility_{i,t-4,t} + \alpha_{6}OperCycle_{i,t-4,t} + \alpha_{7}Losses_{i,t-4,t} + \alpha_{8}NationalAuditor_{it} + \alpha_{9}SalesGrowth_{rk_{it}} + \alpha_{10}MB_{it} + \alpha_{11}LnNumAnalyst_{it} + \alpha_{12}IndRev\% + \alpha_{13}Momentum_{it} + \alpha_{14}Segments_{it} + \alpha_{15}ExecDir_{it} + \alpha_{16}Tenure_{it} + \alpha_{17}BoardMtgs_{it} + \eta_{i} + d_{t} + v_{it}$$

$$(6.7)$$

The model suffers from the previously explained problem of unobservable individual heterogeneity that may result in managers with specific abilities and behaviours to end up in firms at certain levels of performance (Pindado and Requejo, 2014). However, no

²⁵ MgrlAbilityDum_{it} is a dummy variable that equals 1 if the ability of a manager in an industry is high (greater than the median) and zero otherwise. Therefore, α_1 is the coefficient of (*EarningsManagement_{it}*) when managerial ability is low; ($\alpha_1 + \beta_1$) is the coefficient when managerial ability is high.

 $^{^{26}\}eta_i$ controls for both firm specific effects and manager specific effects.

²⁷ I do not tabulate the coefficients of time periods later in the results.
endogeneity can be claimed in this case because it is difficult to predict firm future performance at the moment due to different factors that might interfere in determining such variable. As a result, I use a fixed-effects (FE) estimator to solve the model.

Table 0.0 The effect of carmings management and managema	rability on ruture operation	ng periormanee	
Variables	(1)	(2)	
	Operating performance	Operating performance	
AccrualEarningsManagement _{it}	0.001	-	
MgrlAbility, *Accrual EarningsManagement	0.001	-	
RealEarningsManagement _{it}	-	-0.010*	
MgrlAbility*Real EarningsManagement	-	0.008*	
MgrlAbility.	0.004	0.005	
FirmSize _{it}	-0.057***	-0.031***	
SalesVolatility _{it-4}	0.025	0.050	
CashFlowVolatility _{i1-4t}	-0.164***	0.697***	
OperCycle _{i 1-4 t}	-0.032***	-0.017	
Losses _{i,t-4,t}	0.016	0.085***	
NationalAuditor _{it}	0.014	0.003	
MB_{it}	-0.000	-0.001*	
$SalesGrowth_{rk_{it}}$	0.012	-0.000	
LnNumAnalyst _{it}	-0.001	-0.004	
IndRev%	0.017	0.004	
Momentum _{it}	-0.042***	-0.040**	
Segments _{it}	0.010	0.019*	
<i>ExecDir_{it}</i>	-0.008	-0.007	
Tenure _{it}	-0.008***	-0.007*	
Board Mtgs _{it}	-0.003***	-0.002**	
R^2	0.208	0.177	
<i>F-statistic</i>	9.520	7.820	
Prob (F-statistic)	0.000	0.000	

Table 6.8 The effect of earnings management and managerial ability on future operating performance

Notes: This table presents the results from the fixed effect regressions for the influence of managerial ability measured by the model of Demerjian et al. (2012) and earnings management (in the following order: (1) accrual earnings management measured by Stubben's Model (2010) and (2) real earnings management measured by Gunny's Model (2010)) on future operating performance measured as the average return on assets (ROA) over the next three-year-period. The sample includes 6,974 firm-year observations obtained from Compustat from 1992 to 2006. Each coefficient represents the change in future operating performance based on a one unit change in the determinant. Variables are defined in the (*Variable Definitions*) section above. Intercept is included, but not tabulated. *, **, *** denotes significance at 0.10, 0.05, and 0.01, respectively.

The results appear in Table 6.8 and show that managerial ability does not play any significant role in the relationship between accrual earnings management and firm future performance. On the other hand, and as expected, real earnings managed by high quality managers negatively influence firm future operating performance as the

coefficient of real earnings management that drives lower operating performance in the future is still negative when high quality managers are involved in such behaviour ($\alpha_1+\beta_1=-0.010+0.008=-0.002$). The results strongly support the argument that real earnings management is more costly to superior managers than accrual earnings management because it results in a deteriorated firm future performance and subsequently a decrease in management future compensation.

6.7 Conclusions

In this chapter I provide an explanation of the earnings management behaviour of high quality managers that has shown a tendency towards more accrual and less real earnings management according to the previous literature. To do so, I examine the cost and benefit of each of these activities by particularly focusing on their effects on future compensation as a major rent that superior managers might partially lose when manipulating earnings. In other words, I examine the individual and joint effects of managerial ability and earnings management on future compensation.

First, I document that firms reward their managers for their high abilities as the outcomes of their sophisticated decisions are realized in the future. High quality managers contribute towards a better performance over time; which results in higher payments in order to keep the current managers motivated to perform at a high-quality level or attract new ones to move to these firms. The findings suggest that contracts have to take the quality of managers into consideration to establish the optimal pay-performance that keeps a balance between optimal effort and optimal incentives.

Second, I document that firms do not punish their managers for accrual earnings management as much as they do for real earnings management through decreasing their compensation in the future. The findings suggest that firms expect accrual earnings management to reverse in the next accounting period but consider real earnings management as a harmful activity because it severely influences future cash flows. In this sense, I extend the previous literature by documenting that real earnings management is more costly not only from the firm's perspective but also from the management's point of view.

164

When examining the earnings management behaviour of high quality managers in particular, I find that they extract less rents from accrual earnings management but more from real earnings management compared to other managers. However, real earnings management continues to be more costly from the perspective of high quality managers as they are overall punished for real earnings management compared to accrual earnings management that accounts for the main source of their manipulation.

The results show that managerial ability contributes to strong governance by enhancing better performance in the future. While accrual earnings management is not seen as harmful to the firm, real earnings management seems detrimental whether viewed by firms, managers, or even high quality managers. This overall consent should make any regulation that attempts to mitigate real earnings management easier to implement but the challenge will remain in detecting such sophisticated activities and communicating them to the public.

Appendix A

Annendix A Table I The effect of earnin	os management and manageri	ial ahility on management	compensation
Appendix A rable r rie effect of ear min	gs management and manager	fai admity on management	compensation

	(1)	(2)
Variables	Total management	Total management
	compensation	compensation
AccrualEarningsManagement _{it}	0.002***	-
MgrlAbility _{it} *Accrual EarningsManagement _{it}	-0.001***	-
". RealEarningsManagement _{it}	-	-0.016***
MgrlAbility _{it} *Real EarningsManagement _{it}	-	0.008***
	0.016***	0.017***
<i>FirmSize</i> _{it}	0.084***	0.078***
SalesVolatility _{i,t-4,t}	-0.033***	-0.018***
CashFlowVolatility _{it-4,t}	0.253***	0.221***
<i>OperCycle</i> _{<i>i,t-4,t</i>}	-0.001	0.006***
$Losses_{i,t-4,t}$	-0.010***	-0.028***
<i>NationalAuditor_{it}</i>	0.014	-0.011*
LitigationInd _{it}	-0.026***	0.000
MB_{it}	-0.000***	-0.000***
$SalesGrowth_{rk_{it}}$	0.010***	0.010***
LnNumAnalyst _{it}	0.034***	0.029***
IndRev%	0.105***	0.046***
Momentum _{it}	0.056***	0.054***
Segments _{it}	0.003**	0.010***
Foreign _{it}	0.047***	0.073***
ExecDir _{it}	0.081***	0.075***
<i>Tenure</i> _{it}	-0.015***	-0.011***
BoardMtgs _{it}	0.000**	0.000
t	-22.944	5.021
Hansen	477.850	477.880
m_l	-1.450	-1.460
m_2	-1.480	-1.430
z_{I}	8,780.850	11,487.290

Notes: This table presents the results from the system generalized method of moments regressions for the effect of managerial ability measured by the model of Demerjian et al. (2012) and earnings management (in the following order: (1) accrual earnings management measured by Modified Jones' Model (1995), and (2) real earnings management measured by Roychowdhury's Model (2006)) on total management compensation measured as the natural logarithm of the sum of salaries, bonuses, option awards, stock awards, and long term incentive plans (LTIPs) averaged over the next three years. The sample includes 6,974 firm-year observations obtained from Compustat from 1992 to 2006. Each coefficient represents the change in total management compensation based on a one unit change in the determinant. Variables are defined in the (Variable Definitions) section above. The t values of the linear restriction tests (LRT) reflect the joint significance of the explanatory and moderating variables. The test is performed under the null hypothesis $H_0: \alpha_1+\beta_1=0$. Hansen test for over-identifying restrictions takes the shape of χ^2 distribution and checks for the orthogonality between the instrumental variables (IV_{it}) and the error term (ϵ_{it}). The AB test (Arellano and Bond, 1991) examines the serial correlation in the first difference residual (v_{it}) by using the (m_i) statistic and takes the shape of a normal distribution with N(0,1). m_1 values represent the results of the 1st order serial correlation test, while m_2 values represent the results of the 2nd order serial correlation test. The Wald test (z_1) checks for the joint significance of the reported coefficients in the model. Intercept is included, but not tabulated. *, **, *** denotes significance at 0.10, 0.05, and 0.01, respectively.

37 11	(1)	(2)
Variables	Total management	Total management
	compensation	compensation
AccrualEarningsManagement _{it}	0.004*	-
MgrlAbility _{it} *Accrual EarningsManagement _{it}	-8.71e-05*	-
RealEarningsManagement _{it}	-	-0.014*
$MgrlAbility_{it}$ *Real EarningsManagement_{it}	-	0.009*
MgrlAbility _{it}	0.022**	0.019**
<i>FirmSize_{it}</i>	0.092***	0.092***
SalesVolatility _{i.t-4.t}	0.003	0.008
$CashFlowVolatility_{i,t-4,t}$	0.379***	0.362***
<i>OperCycle</i> _{<i>i</i>,<i>t</i>-4,<i>t</i>}	-0.003	-0.002
$Losses_{i,t-4,t}$	-0.016	-0.023
NationalAuditor _{it}	-0.009	-0.005
LitigationInd _{it}	-0.013	-0.015
MB_{it}	0.001	-9.95e-05
$SalesGrowth_{rk_{it}}$	-0.002	0.010
LnNumAnalyst _{it}	0.051***	0.055***
IndRev%	0.089***	0.089***
Momentum _{it}	0.038	0.042*
Segments _{it}	0.008	0.004
Foreign _{it}	0.064***	0.064***
<i>ExecDir_{it}</i>	0.087*	0.087*
<i>Tenure_{it}</i>	0.002	0.001
BoardMtgs _{it}	-0.001	-0.001
t	-3.248	4.009
R^2	0.379	0.380
F-statistic	40.900	42.150
Prob (F-statistic)	0.000	0.000

Notes: This table presents the results from the ordinary least square regressions for the effect of managerial ability measured by the model of Demerjian et al. (2012) and earnings management (in the following order: (1) accrual earnings management measured by Stubben's Model (2010), and (2) real earnings management measured by Gunny's Model (2010)) on total management compensation measured as the natural logarithm of the sum of salaries, bonuses, option awards, stock awards, and long term incentive plans (LTIPs) averaged over the next three years. The sample includes 6,974 firm-year observations obtained from Compustat from 1992 to 2006. Each coefficient represents the change in total management compensation based on a one unit change in the determinant. Variables are defined in the (*Variable Definitions*) section above. The t values of the linear restriction tests (LRT) reflect the joint significance of the explanatory and moderating variables. The test is performed under the null hypothesis $H_0: \propto_1+\beta_1=0$. Intercept is included, but not tabulated. *, **, *** denotes significance at 0.10, 0.05, and 0.01, respectively.

Chapter 7 Market Concentration and Earnings Management: A Non-Linear Relationship

7.1 Intoduction

While the previous chapter has examined the relationship between some of the contractual motives and earnings management, this chapter focuses on market concentration as one of the external drivers of earnings management. Market concentration plays a divergent role in shaping management behaviour. While it makes monitoring more difficult because of the higher information asymmetry according to the agency theory, it renders communicating information less costly due to the lower pressure from the few existing competitors based on the revelation principle (Strong and Walker, 1987; Milgrom and Roberts, 1992; Walker, 2013). The earnings management literature has emphasized the previous roles and found market concentration stimulates earnings management and brings discipline at the same time. While some studies document a positive effect of market concentration on earnings management based on an intensified agency problem (Dalia and Park, 2009; Markarian and Santalo, 2010; Karuna et al., 2012), others evidence a negative relation due to the lower competitive incentives in concentrated markets (Datta et al., 2013). The different points of view and findings in the previous studies suggest that a non-linear relationship can better explain the effect of market concentration on earnings management than a linear approach.

Under both of the previous points of view, earnings management in concentrated markets tends to be determined by information asymmetry between management and shareholders (Dalia and Park, 2009). The quantity of information decreases in concentrated markets because of the lower analysts' coverage that allows firms to obfuscate information; and thus aggravates adverse selection problem (Harris and Raviv, 1979; Healy and Palepu, 2001; Ali et al., 2014; Elbadry et al., 2013; Walker, 2013; Salop, 2015). However, the less competitive environment motivates analysts' discretion

over their forecasts and, therefore, may influence the quality of information up or down (Strong and Walker, 1987; Walker, 2013).

Based on the previous contrasting effects of market concentration on earnings management, I examine a non-linear relation between market concentration and accrual and real earnings management activities. As the previous relations is based on the asymmetry of information, I explain them according to the changes in information quantity and quality in concentrated markets.

For measuring market concentration, I use the Hall Tideman Index (Dalia and Park, 2009; Markarian and Santalo, 2010; Karuna et al., 2012; Datta et al., 2013). For measuring information asymmetry I use three measures that include earnings forecast error, earnings forecast dispersion, and analysts' coverage (Ali et al., 2014; Dalia and Park, 2009). Finally, for measuring earnings management variables, I implement four different models which include Stubben's model and the Modified Jones' model for measuring accruals' manipulation and Gunny's model and Roychowdhury's model for measuring real earnings management.

This chapter shows a significant non-linear effect of market concentration on earnings management. The decrease in information quantity in concentrated markets allows firms to manage accruals until the quality of information starts to decline and thus enables managers to use more of the sophisticated real earnings management. Nevertheless, I find evidence of an optimal level of market concentration with a mutual decrease in accrual and real earnings management at concentration levels between 55% and 60%. The results extend the work of Guo et al. (2015) who examine a non-linear relation between market competition and earnings quality, but this chapter focuses on the effect of market concentration on accrual and real earnings management.

The remainder of the chapter proceeds as follows. In the next section, I review the related studies from the literature and build the hypotheses of the chapter. In section 7.3, I explain the process of data collection, variable definitions, and the descriptive statistics. In section 7.4, I explain the methodology I use in this chapter. In section 7.5, I present and discuss the results. In section 7.6, I add some empirical analysis. Finally, in section 7.7, I conclude the chapter.

7.2 Literature and Hypotheses Development

7.2.1 Market Concentration and Earnings Management

According to the revelation principle, privately informed managers achieve more benefits from revealing the truth as it allows them to avoid any subsequent penalties of misreporting firm results (Ronen and Yaari, 2008). At the same time, shareholders can make better decisions and thus maximize their expected utilities (Milgrom and Roberts, 1992). In this sense, the revelation principle helps in solving the conflict between principals and agents and emphasizes truth telling equilibrium where the truthful revelation of private information leads to maximizing the utility functions of all players in the game.

However, Walker (2013) identifies four conditions of the revelation principle that need to be violated for earnings management to occur. The first condition is violated when contracting costs are high and hence renders contracts between shareholders and management imperfect and difficult to enforce. Second, earnings management may occur because of shareholders' imperfect rationalities, especially under conditions of market uncertainty, which makes them unable to take wealth maximizing decisions all the time. The third condition of the revelation principle is violated when there is no common knowledge between management and shareholders to enable the later to understand managers' actions. Finally, managers may manipulate earnings when they find that communicating with shareholders is costly due to the previous violations is sufficient to give managers the chance to manage earnings (Walker, 2013).

Based on the previous conditions, the effect of market concentration on earnings management can be viewed from two different perspectives of the revelation principle. In more concentrated markets, communicating with shareholders is less costly due to the lower pressure resulting from weak competition that creates fewer chances for comparisons between firms (Strong and Walker, 1987). Therefore, managers feel less need to take risk in order to keep pace with the behaviour of other aggressive competitors and adopt profit maximizing actions to improve their firms' efficiencies (Milgrom and Roberts, 1992). In this sense, I perceive a disciplinary influence of

market concentration on management behaviour, and according to the fourth condition of the revelation principle mentioned above I expect that earnings management is less likely to occur in concentrated markets. In contrast, concentrated markets are less informative causing the lack of common knowledge between management and shareholders. The high information asymmetry does not allow for monitoring management behaviour and thus results in poor follow-up by outsiders; which allows managers to take more non-value-maximizing actions (Milgrom and Roberts, 1992). In this sense, market concentration scatters the interests of principals and agents and, thus, aggravates the agency problem. Consequently, managers may manage earnings as the shareholders are unable to understand their decisions under the uncertainty discussed in the third condition of the revelation principle mentioned earlier.

The literature supports the previous points of view in the theory regarding the effect of market concentration on earnings management and documents contrasting results. Starting with accruals' manipulation, Datta et al. (2013) find that market concentration leads to less accrual earnings management as it plays a disciplinary role and, therefore, mitigates the agency problem between managers and shareholders. On the other hand, Karuna et al. (2012) find that managers in concentrated markets use more accrual earnings management due to the fewer restrictions in such markets. Similarly, Dalia and Park (2009) find that firms in concentrated industries in the manufacturing sector in the United States engage in a higher level of discretionary accruals. They also document that market concentration changes the disciplining function of the market by aggravating the agency problem. Markarian and Santalo (2010) also confirm that the separation of ownership from control does not work efficiently in more concentrated markets.

To the best of my knowledge, the study of Karuna et al. (2012) is the only one that examines the effect of market concentration on real earnings management. It finds that market concentration results in more real earnings management because managers have better opportunities to make benefit of such activities, especially that they can easily understand other managers' incentives to manipulate earnings in such markets (Kallunki and Martikainen, 1999; Bagnoli and Watts, 2000).

Although the previous studies only emphasize the linear relation between market concentration and earnings management, other studies show a non-linear effect of the market structure on firm behaviour. Hermalin (1992) documents that the effect of competition on the agency problem and thus on firm efficiency and management behaviour depends on four different factors. First, the income effect as firms' profitability is expected to decrease under higher competition and thus managers receive lower income. Second, the risk attitude effect as the willingness of managers to take risk changes under competition. Third, the information effect that is related to the influence of competition on information in the market. Finally, actions' value effect because the outcome of different managerial actions may change under the effect of competition. The previous effects do not necessarily have similar signs and thus suggest a non-linear relationship between market competition and management behaviour.

Schmidt (1997) documents a non-linear relationship between market competition and managerial incentives due to the two-sided effects of competition. On the one hand, market competition increases the pressure on management to improve performance in order to stay in the market and thus creates an incentive for more managerial efforts. On the other hand, competition results in lower profitability that is reflected in lower managerial rewards which may demotivate managerial efforts. The overall effect of market competition on managerial incentives is quadratic with managerial incentives increasing at lower levels of market competition and decreasing at higher levels of market competition.

Scherer (1967) documents a non-linear relationship between market concentration and management innovative efforts because of the endogeneity between the two factors. To further explain the results, Aghion et al. (2005) find a non-linear quadratic relationship between market competition and innovation as innovation is endogenous in competitive markets. Therefore, competition may encourage or discourage innovation depending on the level of innovation in the market. At lower levels of competition, firms tend to be more equal in their technologies and thus exert more innovative efforts to escape the competition as the incremental benefits from innovation are increasing. However, at higher levels of market competition where there are big technological gaps between firms in the industry, innovative efforts tend to decrease as firms have already reached to an equilibrium position where the incremental benefit from innovation is decreasing. To my knowledge, the only study that examines a non-linear effect of market competition on earnings quality is that of Guo et al. (2015). Specifically, Guo et al. (2015) document a non-linear relationship between market competition and earnings quality that takes a quadratic shape. They find that earnings quality increases at lower levels of market competition but decreases at higher levels of market competition. They explain their results through the marginal benefit of earnings quality that increases at lower levels of market competition and decreases at higher levels as competitors can make advantage of the high quality information disclosed by a firm. The study uses different proxies for market concentration.

Given the two contrasting views in the agency theory and the revelation principle in addition to the different findings in the previous studies, I expect managers to use different levels of earnings management under different degrees of market concentration. Therefore, I expect a non-linear relationship between market concentration and earnings management. For this purpose, I examine the influence of market concentration (MC_{it}) at two different levels and expect its relationship with earnings management (EM_{it}) to take one of the two shapes that appear in Figure 7.1 and Figure 7.2 below.

Figure 7.1 First assumption for the effect of market concentration on earnings management







Both of the previous assumptions suppose that low and high levels of market concentration have different effects on earnings management. Whether I accept the assumption in Figure 7.1 or that in Figure 7.2, the hypothesis predicts a quadratic relationship between market concentration and earnings management. As I cannot decide which one prevails for now, I develop the following hypothesis.

H1: The effect of market concentration on earnings management is different at low and high levels of market concentration.

7.2.2 Market Concentration and Information Asymmetry

To explain the previous double-edged effect of market concentration on earnings management, information asymmetry has to be considered as it represents the main problem between the principal and the agent. Concentrated markets are inefficient as managers may obfuscate information in order to protect their competitive advantages (Ali et al., 2014). Taken with the decreased coverage by analysts, a condition of imperfect information can be generally assumed in concentrated markets where information is not absent but still incomplete. The result will be a situation of information asymmetry between shareholders and management that can take one of the two forms: moral hazard and adverse selection (Milgrom and Roberts, 1992; Boujelbene and Besbes, 2012). Moral hazard happens because shareholders do not possess information to monitor management and assess whether it works for maximizing the firms' values; resulting in different attitudes and motivations for each

party. On the other hand, adverse selection happens because managers have access to private information that allows them to make decisions for their own benefits and ignore shareholders' value (Walker, 2013). The resulting uncertainty changes the costs and benefits of information for decision makers and gives rise to a problem of coordination in making decisions regarding what information is reported, how it is communicated, and who makes the decision (Harris and Raviv, 1979; Healy and Palepu, 2001; Elbadry et al., 2013; Walker, 2013; Salop, 2015). Overall, the decrease in information influences the agency problem by driving contracting costs like observing and monitoring (Hart, 1995; Arroyo, 2007).

From another perspective, firms have a chance to signal information under the lower competitive pressure in concentrated markets (Strong and Walker, 1987). Although management discretion over financial reporting may involve signaling low quality information, the signals may convey information about future performance and ultimately contribute to improving the quality of the reported information (Gunny, 2010; Walker, 2013). Similarly, the less competitive environment drives analysts' judgements and further contributes to the variation in the quality of their forecasts.

The literature provides evidence on the interaction between market competition and information asymmetry to determine the response of the market. Balakrishnan et al. (2012) examine the effect of market competition on the relation between information asymmetry and the cost of capital. They document that investors discount firms' prices according to the level of information asymmetry and competition in the market. This implies a higher cost of capital and lower future returns to the firms under more intense competition and higher levels of information asymmetry. In this sense, both market competition and information asymmetry contribute to adverse selection problem.

More specifically, other studies show the effect of market concentration on information asymmetry. Information asymmetry is likely to occur in markets where a specific group possesses information without allowing the others to access it (Boujelbene and Besbes, 2012). Dalia and Park (2009) find higher levels of analyst forecasts' errors and dispersions at higher levels of market concentration and, therefore, lower financial reporting quality in more concentrated markets. On the other hand, they find less information asymmetry in competitive markets even when firms suffer from low profitability that is expected to create higher incentives to manage earnings.

175

Markarian and Santalo (2010) document that when shareholders do not have access to information about real firms' outputs and market prices, it becomes easier for managers to justify the manipulation and, therefore, the cost of earnings management to the firms gets lower. Ali et al. (2014) find lower analysts' coverage, higher forecasts' errors and dispersions, and bigger bid-ask spreads in more concentrated markets that reflect a more serious adverse selection problem. They use market concentration as a proxy for firm disclosures because they expect firms in more concentrated markets to be less likely to disclose information to avoid losing their competitive advantages. However, the study uses the industry concentration measures available from the US Census Bureau, rather than the HHI, and for the manufacturing sector only. Finally, Bhattacharya et al. (2013) find a non-linear quadratic relationship between earnings quality and information asymmetry and document that information can be considered as a measure of the quality of disclosure (Bhattacharya et al., 2013).

Although I can expect less information due to the lower analysts' coverage in more concentrated markets according to the obfuscation hypothesis, the contrasting theoretical points of view in the signalling hypothesis and the different literature findings explained earlier suggest a changing information quality at different levels of market concentration. Accordingly, I develop the following two hypotheses.

H2a: There is a negative effect of market concentration on the quantity of information in the market.

H2b: The effect of market concentration on information quality is different at low and high levels of market concentration.

7.3 Data, Variable Definitions and Descriptive Statistics

7.3.1 Data

I include all firms in the United States from the Annual Compustat, Quarterly Compustat, Historical Segments Compustat, and IBES databases, for the period from 1989 to 2011. I also make use of the dataset made available by Demerjian et al. (2012) for managerial ability control variable during the same time period. I start with all firms that have the required data for calculating the measures of earnings management,

market concentration, and information asymmetry after excluding firm-years that experienced accounting changes, merger and acquisition activities, or discontinued operations.²⁸

Following Cheng et al. (2011), I exclude regulated industries that contain banks, credit institutions, brokers, insurance, real estate, holding companies, and investment firms because they have their unique accounting and financial practices and are subject to distinct regulations. Therefore, managers in these industries have different motivations to manipulate earnings than those of managers in other industry sectors.²⁹

Following prior literature, I exclude any industry with fewer than six observations for each SIC code in a specific year to ensure sufficient data exists to calculate earnings management measures and make sure that OLS assumption regarding the normality of the error term holds (e.g., Rosner, 2003; García Lara et al., 2005; Kothari et al., 2005; Athanasakou et al., 2009; Iqbal et al., 2009). For that purpose, I follow the SIC classification of Fama-French (1997).

As the model is dynamic, I make sure that information is available for at least five consecutive years for each firm over the study period (Miguel et al., 2004). When combining between the cross-sectional and time-series dimensions of the data, I obtain total observations of 25,119 in an unbalanced panel dataset. I do not use balanced panels to avoid survivorship bias problem.

7.3.2 Earnings Management Measures

Based on the discussion in Chapter 5, I use Stubben's Model (2010) for measuring accrual earnings management because it focuses on discretionary revenues which represent the largest component of earnings in most firms (Stubben, 2010). I also use Gunny's Model (2010) for real earnings management because it covers more types of this activity compared to other models (Gunny, 2010). In addition, I use two other widespread models as corroborating measures that include the Modified Jones' Model

²⁸ According to McNichols (2002), I specifically exclude firm quarters or years with non-blank values for accounting changes cumulative effects (ACCCHGQ_FN), or merger and acquisition activities (ACQMETH_FN), or discontinued operations (DOQ_FN) in the Compustat database.

²⁹ I exclude firms with the following SIC codes: $4000 \le SIC \ge 4900$ and $6000 \le SIC \ge 6300$.

(1995) for measuring accrual earnings management and Roychowdhury's Model (2006) for measuring real earnings management. Again, for each of these measures I use a cross sectional model in the calculations for each year and each industry classified by its four-digit SIC code with at least 6 observations. I also take the decile ranks for better comparability and to avoid the effect of the outliers. For the purpose of calculating an overall measure of real earnings management using Gunny's Model (2010) and Roychowdhury's Model (2006), a principal component analysis with varimax rotation is performed to avoid the inclusion of highly correlated variables in the overall measure (Demerjian et al., 2013b).³⁰

7.3.3 Market Concentration Measures

To measure market concentration, I use the Hall Tideman Index (*HTI*) because it takes into consideration the rank of each firm based on its market share in the industry. In this sense, the *HTI* accounts for the absolute number of firms which reflects the entry barriers to the industry, in addition to emphasizing the relative sizes of those firms (Hall and Tideman, 1967). Finally, I decile rank the measure for better comparability and to mitigate for outliers. I present the calculation of the previous measure in Table 7.1.

7.3.4 Information Asymmetry Measures

As there is no "best" measure of information asymmetry (Elbadry et al., 2013), I use several proxies to calculate this variable. Analysts' coverage is negatively associated with information asymmetry and, thus, used as a measure of this variable in many studies (Houston at al., 2008). Therefore, I particularly focus on analysts-related measures because financial analysts act as information intermediaries who generate information through their forecasts (Healy and Palepu, 2001). Information asymmetry is associated with a decrease in analysts' coverage and an increase in the dispersion and error of their earnings forecasts (Ali et al., 2014). Thus, I identify earnings forecast error, earnings forecast dispersion, and analysts' coverage as the proxies of information

³⁰ The principal component analysis step turns the set of correlated variables to be linearly uncorrelated according to the weights of their variances, thus reducing the number of variables to their principal components.

asymmetry (Dalia and Park, 2009). The calculation of each of these measures is presented in table 7.1.

7.3.5 Control Variables

Following Dechow and Dichev (2002), the first set of controls consider firm specific determinants and account for firm size (*FirmSize*), sales volatility (*SalesVolatility*), cash flow volatility (CashFlowVolatility), operating cycle (OperCycle), and historical losses (Losses). Larger firms have operations that are more predictable and more diversified businesses and thus earnings should be of higher quality and better communicated to the stakeholders (Dechow and Dichev, 2002). Managers in these firms, however, have more opportunities to manipulate earnings because of the large number of transactions they undertake and their complicated operations (Demerjian et al., 2013b). Sales volatility reflects uncertainty in operations and, therefore, implies a higher likelihood of earnings management (Dechow and Dichev, 2002). Cash flow volatility also reflects increased volatility in the operations of the firm creating more opportunities for earnings management. Similarly, longer operating cycles increase uncertainty and therefore, the potential for earnings management (Dechow and Dichev, 2002). However, the higher uncertainty may draw the attention of the analysts to such firms. Prior losses are likely to result in more earnings management to meet investors' expectations in making profit and thus more information asymmetry (Dechow and Dichev, 2002). Equally, they may drive managers to avoid earnings management as they will be facing more scrutiny by the market.

I also control for managerial ability (*MgrlAbility*) that drives more accrual but less real earnings management (Demerjian et al., 2013b). While superior managers make sophisticated decisions that contribute to more information asymmetry, they possess the skills to communicate information to the stakeholders. I use a national auditor indicator (*NationalAuditor*) to control for any auditor effects as firms audited by national auditors use more earnings management, particularly real activities, than those audited by international audit firms (Becker et al., 1998). Therefore, the overall result on information asymmetry can be positive or negative. I add another indicator variable to control for litigious industries (*LitigationInd*) following Francis et al. (1994) because firms in such industries are more likely to be involved in earnings management. However, the opposite effect may appear because of the probable regulatory intervention.

The market to book ratio (*MB*), the one year sales growth (*SalesGrowth*), and returns momentum (*Momentum*) are added as they have been found to influence earnings management (Hribar and Nichols, 2007; Dechow et al., 2011). As firms with higher market to book ratios have more growth prospects, they may be involved in more earnings management to meet market expectations (Hribar and Nichols, 2007). Firms with growing sales and high returns momentum also have high growth prospects but the increase in sales and returns reduces the pressure on management to manipulate earnings (Dechow et al., 2011). Therefore, the influence of the previous three factors can take two different directions.

In addition, this chapter controls for analyst coverage (*NumAnalyst*). The presence of more analysts increases the burden on management to present better earnings and, therefore, may lead to more earnings management (Dechow et al., 2011; Zang, 2012). I also include an industry revenue leadership variable (*IndRev%*) (Dechow et al., 2011; Zang, 2012). As Zang (2012) could not specify the influence of industry revenue leadership, I expect it to be bi-directional due to the simultaneous high power of the industry leaders and the more scrutiny they face in the market.

The number of segments (*Segments*) and the frequency of foreign transactions (*Foreign*) are used to control for businesses complexity (Karuna et al., 2012). With more complex transactions managers use more real earnings management because they are less likely to be detected than discretionary revenue manipulation. Meanwhile, business complexity may drive more scrutiny to such firms (Karuna et al., 2012). As a result, it may have a double-edged effect on earnings management and information asymmetry.

Finally, I use dummies for the global financial crisis (*GFC*) (Badertscher et al., 2014; Kanagaretnam et al., 2014; Magnan et al., 2015), the dot-com bubble (*DotCom*) (Lieberman and Asaba, 2006), and the post-Sarbanes Oxley period (SOX) (Cohen et al., 2008) to control for some macro-economic events over the study period. I present the

detailed calculations for each of the previous variables in Table 7.1 and summarize the expected and actual signs of all control variables in Table 7.2.

Variable	Calculation
AccrualEM	The residual from using Stubben's Model (2010) as explained in the 'Variable Definitions' section above.
RealEM	The first component generated by using Gunny's Model (2010) that represents the discretionary reduction in SG&A expenses and the overproduction to cut prices or to decrease the cost of goods sold as explained in the 'Variable Definitions' section above.
Concentration	Market concentration measured by the Hall Tideman index; calculated as $HTI=1/[(2\sum_{i=1}^{N}(k*s))-1]$ where <i>s</i> represents market share, N is the number of firms per year-industry, and <i>k</i> represents firm rank according to market share.
ForecastError	The log of $1 +$ the difference between the median next-year forecast earnings per share (EPS) and the realized EPS normalized by the median EPS forecast.
ForecastDispersion	The log of $1 +$ the standard deviation of EPS forecasts for the next year normalized by the absolute value of the median EPS forecast.
InvAnalystsCoverage	The log of 1 + the number of analysts covering the firm in year t, multiplied by -1 to reflect information asymmetry.
FirmSize	The natural log of the firm's assets as of the end of year <i>t</i> .
SalesVolatility	The standard deviation of (sales / average assets) over at least three of the last five years $(t-4, t)$.
CashFlowVolatility	The standard deviation of (cash from operations / average assets) over at least three of the last five years (t –4, t).
OperCycle	The operating cycle is the natural log of average sales turnover plus days in inventory over at least three of the last five years $(t-4, t)$.
Losses	Loss history is the percentage of years reporting losses in net income before extraordinary items over at least three of the last five years $(t-4, t)$.
MgrlAbility	This measure represents how efficiently managers generate revenues from the available firm resources according to the approach of Demerjian et al. (2012). The variable was directly obtained from <i>https://community.bus.emory.edu/personal/PDEMERJ/Pages/Home.aspx.</i>
NationalAuditor	An indicator variable equals one for firms audited by national audit firms in year <i>t</i> ; zero otherwise.
LitigationInd	Litigious industry indicator that equals one for firms in industries with SIC Codes: 2833-2836 (biotechnology), 3570-3577 and 7370-7374 (computers), 3600-3674 (electronics), and 52(X)-5961 (retailing).

Table 7.1 Calculation of the variables

(The table is continued on the next page)

Variable	Calculation
MB	The market to book ratio that equals the firm's market capitalization divided by book value for year t.
SalesGrowth	Current year's sales less prior year's sales less the increase in receivables all scalded by prior year's sales and decile ranked by industry and year.
Momentum	Returns momentum calculated by the decile rank (by industry and year) of asset returns during the two years preceding the start of year t.
NumAnalyst	The log of 1+ the number of analysts covering the firm in year t.
IndRev%	Industry revenue leadership measured by the firm's sales in year t-1 divided by the total sales for the firm's industry in year t-1.
Segments	The natural log of 1+ the number of firm's business segments in year t.
Foreign	The frequency with which the firm has a non-zero foreign currency transactions during the sample period.
GFC	An indicator variable equals one for the years of the global financial crisis (2007-2009); zero otherwise.
Dot – Com	An indicator variable equals one for the years of the Dot-Com bubble (1997-2000); zero otherwise.
SOX	An indicator variable equals one for the years after the release of the Sarbanes Oxley act (2002, 2005); zero otherwise

Table 7.1 (continued)

Notes: This table presents the detailed calculations for each of the control variables identified in the model as discussed in the (Variable Definitions) section above.

Variables	Accrua	lEM	RealEM		
	Predicted sign Actual sign		Predicted sign	Actual sign	
<i>FirmSize_{it}</i>	+/-	+	+/-	+	
Sales Volatility _{i 1-4 t}	+	+	+	+	
$CashFlowVolatility_{i.t-4.t}$	+	+	+	+	
OperCycle _{i 1-4 t}	+	+	+	+	
$Losses_{i,t-4,t}$	+/-	+	+/-	-	
MgrlAbility _{it}	+	+	-	-	
NationalAuditor _{it}	+/-	-	+	+	
LitigationInd _{it}	+/-	+	+/-	+	
MB_{it}	+/-	+	+/-	+	
$SalesGrowth_{rk_{it}}$	+/-	-	+/-	+	
<i>Momentum_{it}</i>	+/-	+	+/-	-	
LnNumAnalyst _{it}	+	+	+	+	
IndRev%	+/-	-	+/-	+	
Segments _{it}	+/-	-	+/-	-	
Foreign _{it}	+/-	-	+/-	+	
GFC_{it}	+/-	+	+/-	+	
Dot-Com;,	+/-	-	+/-	+	
SOX _{it}	+/-	-	+/-	+	

Table 7.2 Panel A Expected and actual signs of the control variables - The effect on earnings management

Panel B Expected and actual sig	ns of the control variables -	- The effect on information asymmetry

Variables	InfoAssymetry _{it}			
	Predicted sign	Actual sign		
FirmSize _{it}	+/-	-		
SalesVolatility _{i.t-4.t}	+/-	+/-		
CashFlowVolatility _{i.t-4.t}	+/-	-		
<i>OperCycle</i> _{<i>i t</i>-4 <i>t</i>}	+/-	+/-		
Losses _{i,t-4,t}	+/-	+/-		
MgrlAbility _{it}	+/-	-		
NationalAuditor _{it}	+/-	+		
LitigationInd _{it}	+/-	+/-		
MB _{it}	+/-	+/-		
$SalesGrowth_{rk_{it}}$	+/-	-		
Momentum _{it}	+/-	-		
IndRev%	+/-	+/-		
Segments _{it}	+/-	+		
Foreign _{it}	+/-	+		

Notes: This table presents the signs I expect and actually get for all control variables identified in the models and discussed in the (Variable Definitions) section above.

7.3.6 Descriptive Statistics

I present the descriptive statistics in Table 7.3. Consistent with Demerjian et al. (2013b), both accrual and real earnings management measures have means and medians of zero because they are calculated as the residuals from the models selected above. Meanwhile, the measures of market concentration and information asymmetry are comparable to those of Karuna et al. (2012) and Dalia and Park (2009), taking into consideration the difference of the sample from those of the previous studies.

Variables	Mean	Median	Std Dev	P25	P75
AccrualEM	0.00	0.00	0.05	-0.01	0.01
RealEM	0.00	-0.00	1.00	-0.72	0.70
Concentration	0.22	0.17	0.16	0.10	0.29
ForecastError	0.28	0.10	0.83	0.03	0.26
ForecastDispersion	0.15	0.05	0.71	0.02	0.13
AnalystsCoverage	1.22	1.09	0.41	0.69	1.60
TotalAssets	2,622.85	147.34	14,666.88	30.75	789.46
SalesVolatility	0.26	0.15	0.40	0.08	0.29
CashFlowVolatility	0.12	0.06	0.34	0.04	0.12
OperCycle	133.22	119.26	81.27	76.72	172.87
Losses	0.37	0.25	0.36	0.00	0.67
MgrlAbility	0.00	-0.01	0.15	-0.10	0.08
NationalAuditor	0.22	0.00	0.41	0.00	0.00
LitigationInd	0.09	0.00	0.28	0.00	0.00
MB	3.15	1.89	50.61	1.04	3.46
SalesGrowth	0.90	0.08	66.38	-0.03	0.22
Momentum	-0.17	0.06	2.09	-0.13	0.15
NumAnalyst	3.05	3.00	1.71	2.00	4.00
IndRev%	0.07	0.01	0.15	0.00	0.06
Segments	2.04	1.00	1.72	1.00	3.00
Foreign	0.21	0.00	0.34	0.00	0.33
GFC	0.13	0.00	0.34	0.00	0.00
Dot-Com	0.24	0.00	0.43	0.00	0.00
SOX	0.20	0.00	0.40	0.00	0.00

Table 7.3 Descriptive statistics

Notes: The table presents the descriptive statistics of all the variables identified in the model for 25,119 firm-year observations obtained from Compustat from 1989 to 2011. Variables are defined in the (Variable Definitions) section above. I present the untransformed variables for ease of interpretation.

The control variables are also comparable to Karuna et al. (2012) and Demerjian et al. (2013b) with some differences appearing because of the variations between the samples. Cash flow volatility is lower than sales volatility and thus implies that firms consider it more costly as it is negatively valued by the market (Allayannis and Weston, 2007). A huge portion of the firms show losses during the recent years which ultimately resulted in negative returns momentum. Because of presenting the untransformed variables, firm size and operating cycle are associated with high standard deviations.

Table 7.4 represents the correlation matrix between the major variables in the chapter: earnings management, market concentration, and information asymmetry. The correlations between the control variables have not been displayed here for easier demonstration. There is a negative correlation between accrual and real earnings management which suggests that they might be used as substitutes consistent with the subsequent findings in the results section. I also observe a strong correlation between the proxy of market concentration and both accrual and real earnings management, suggesting a potential impact of market concentration on earnings management. While for the proxies of information asymmetry, Table 7.4 shows a positive correlation between the three measures of information asymmetry, which implies that they all represent strong measures of the same variable. Further, I evidence a strong correlation between information asymmetry measures and market concentration asymmetry. Finally, I check the VIF between the previous variables and make sure that all VIF factors are less than 10 in order to control for multicollinearity.

Table 7.4 Correlation table						
	Accrual EM	Real EM	Concentration	Forecast Error	Forecast Dispersion	Analysts Coverage
Accrual EM						
Real EM	-0.01**					
Concentration	0.04***	0.03***				
Forecast Error	-0.01*	0.00	-0.04***			
Forecast Dispersion	-0.01*	-0.01**	-0.04***	0.92***		
Analysts Coverage	-0.01	-0.02***	0.05***	0.04***	0.02***	

Notes: This table presents the correlation matrix for 25,119 firm-year observations obtained from Compustat from 1989 to 2011 between the measures of accrual earnings management measured by Stubben's Model (2010), real earnings management measured by Gunny's Model (2010), market concentration measured by the Hall Tideman Index (*HTI*), and information asymmetry measured by analysts' coverage, and the dispersion and error of their earnings forecasts. Variables are defined in the (Variable Definitions) section above. Pearson correlation coefficients are presented. To control for the multicollinearity between the different variables I make sure that all VIF factors are less than 10. VIF factors, however, are not tabulated. *, **, **** denotes a statistical coefficient at the 10, 5 and 1 percent alpha level, respectively.

7.4 Methodology

To test the first hypothesis (*H1*), I present a quadratic relationship in the model that identifies earnings management (*EarningsManagement*_t) as the dependent variable (Y_{it}) with its two proxies of real (*RealEM*) and accrual (*AccrualEM*) earnings management.³¹ I introduce market concentration (*Concentration*_{it}) and its square (*Concentration*²_{it}) as the explanatory variables (X_{it}) and (X^2_{it}) respectively. Meanwhile, I take the dynamic effect of earnings management into the model. Earnings performance in the past years determines earnings management behaviour during the current year (Kim et al., 2003). In addition, earnings management levels are associated with meeting prior earnings' benchmarks (Graham et al., 2005; Cohen et al., 2008; Gunny, 2010). Consequently, managers look back at the previously manipulated earnings when contemplating to manage earnings management measures (*EarningsManagement*_{i,t-1}) to explain its current levels. To complete the model, I add the previously identified control variables (*Controls_{it}*) and an error term (ε_{it}).

The model suffers from the problem of unobservable individual heterogeneity attributed to time-invariant industry effects. The industry culture (e.g., the nature of relations and knowledge sharing between competitors in the same sector) may contribute to the information asymmetry in the market; hence using higher levels of earnings management activities by firms in specific industries compared to other sectors. Consequently, the sample is expected to be heterogeneous (Graham et al., 2012).

In addition, the model suffers from an endogeneity problem because of the mutual causality between the dependent and explanatory variables. While market concentration influences information asymmetry and thus the levels of earnings management, earnings management and information asymmetry are determinants of market concentration as they contribute to misleading investors and ultimately driving some firms to exit the market. This effect would be captured in the error term and

³¹ The dependent variables have been tested for normality using the Jarque and Bera test which follows a chi-square distribution. The test statistic is calculated as $\left[\frac{n-k}{6}\left(S^2 + \frac{(K-3)^2}{4}\right)\right]$; where *n* is the number of observations, *k* is the number of regressors, *S* represents skewness and *K* represents kurtosis. The results provide statistics of 2.37 and 2.66 for accrual and real earnings management respectively leading to accepting the null hypotheses of normality with 95% confidence.

ultimately results in a correlation between the explanatory variable and the error term (Pindado and Requejo, 2014).

Therefore, the OLS estimator will not be able to solve the model as it ignores the impact of the unobservable individual heterogeneity or endogeneity problems. Furthermore, while the fixed effects estimator tackles the unobservable heterogeneity by demeaning the variables in the model it does not solve for the endogeneity problem as it assumes strict exogeneity. Therefore, I solve the model using a system generalized method of moments (GMM) estimator that demeans the variables in the model to solve for the heterogeneity and introduces instrumental variables to solve for the endogeneity problem as explained earlier in the methodology section of Chapter 5.³² In addition, I present the results of the OLS regression as a robustness test in Appendix B Table II.

To avoid the consequent bias in the error term (ε_{it}) I split it into three components. First, I introduce (η_i) to control for the impact of the unobserved effects in the model.³³ Second, I add a time specific effect (d_t) to control for the macroeconomic variables that also interfere with the results over the time period of the study.³⁴ Finally, I consider the remaining part of the error term (ε_{it}) a random disturbance (v_{it}) . Consequently, I express the model in general terms in equation (7.1) and in more specific terms in equation (7.2).

$$Y_{it} = \alpha_0 + \alpha_1 Y_{i,t-1} + \alpha_2 X_{it} + \alpha_3 X_{it}^2 + \alpha_{4-21} Controls_{it} + \varepsilon_{it}$$

$$(7.1)$$

³² Solving for these two problems using a system GMM estimator allows us to obtain less biased and more significant results compared to the previous studies.

 $^{^{33}\}eta_i$ controls for industry specific effects.

³⁴I do not tabulate the coefficients of time periods later in the results.

$$EarningsManagement_{it} = \alpha_0 + \alpha_1 EarningsManagement_{i,t-1} + \alpha_2 Concentration_{it} + \alpha_3 Concentration^2_{it} + \alpha_4 FirmSize_{it} + \alpha_5 SalesVolatility_{i,t-4,t} + \alpha_6 CashFlowVolatility_{i,t-4,t} + \alpha_7 OperCycle_{i,t-4,t} + \alpha_8 Losses_{i,t-4,t} + \alpha_9 MgrlAbility_{it} + \alpha_{10} NationalAuditor_{it} + \alpha_{11} LitigationInd_{it} + \alpha_{12} MB_{it} + \alpha_{13} SalesGrowth_{rk_{it}} + \alpha_{14} Momentum_{it} + \alpha_{15} LnNumAnalyst_{it} + \alpha_{16} IndRev\% + \alpha_{17} Segments_{it} + \alpha_{18} Foreign_{it} + \alpha_{19} GFC_{it} + \alpha_{20} Dot-Com_{it} + \alpha_{21} SOX_{it} + \eta_i + d_t + v_{it}$$
(7.2)

After examining the relation between market concentration and earnings management, I find the breakpoints of the regression. As I study a quadratic relation, I calculate one breakpoint (MC_1) derived from differentiating value with respect to market concentration which equal – ($\alpha_2/2\alpha_3$). According to the hypothesis, I expect α_2 and α_3 to have opposite signs. Accordingly, if I accept the assumption in Figure 7.1, MC_1 will be a maximum but the opposite applies if I accept the assumption in Figure 7.2.

Similarly, to test the second hypothesis (*H2*), I present a quadratic relationship in a model that identifies information asymmetry (*InfoAssymetry*_{ii}) as the dependent variable (Z_{it}), where (*InfoAssymetry*_{it}) is the proxy of (*ForecastError*), (*ForecastDispersion*), and (*InvAnalystsCoverage*). I again introduce market concentration (*Concentration*_{it}) and its square (*Concentration*²_{it}) as the explanatory variables (X_{it}) and (X^{2}_{it}) respectively. To complete the model, I add the previously identified control variables (*Controls*_{it}) and an error term (ε_{it}). This model also suffers from the problems of unobservable individual heterogeneity and endogeneity as explained above (Pindado and Requejo, 2014), and, therefore, I solve it using a system generalized method of moments (GMM) estimator and split the error term (ε_{it}) into three components ($\eta_{it}d_{it}$, and v_{it}). However, I also present the results of the OLS regression as a robustness test in Appendix B Table III. I express the model in general terms in equation (7.3) and in more specific terms in equation (7.4).

$$Z_{it} = \alpha_0 + \alpha_1 X_{it} + \alpha_2 X_{it}^2 + \alpha_{3-20} Controls_{it} + \varepsilon_{it}$$
(7.3)

$$InfoAssymetry_{it} = \alpha_0 + \alpha_1 Concentration_{it} + \alpha_2 Concentration^2_{it} + \alpha_3 FirmSize_{it} + \alpha_4 Sales Volatility_{i,t-4,t} + \alpha_5 CashFlowVolatility_{i,t-4,t} + \alpha_6 OperCycle_{i,t-4,t} + \alpha_7 Losses_{i,t-4,t} + \alpha_8 MgrlAbility_{it} + \alpha_9 NationalAuditor_{it} + \alpha_{10} LitigationInd_{it} + \alpha_{11} MB_{it} + \alpha_{12} Sales Growth_{rk_{it}} + \alpha_{13} Momentum_{it} + \alpha_{14} IndRev\% + \alpha_{15} Segments_{it} + \alpha_{16} Foreign_{it} + \eta_i + d_t + v_{it}$$
(7.4)

After the completion of the empirical work, I ensure that the assumptions of the estimator hold and that the model is valid. I initially test whether the GMM estimator properly addresses the problem of endogeneity. For this purpose, I use the Hansen test for over-identifying restrictions. It takes a χ^2 distribution and checks for the orthogonality between the instrumental variables (IV_{it}) and the error term (ε_{it}). The model uses multiple lags of the right-hand side variables as instruments, which make it over-identified. Consequently, if I accept Hansen's null hypothesis that the instrumental variables (IV_{it}) and the error term (ε_{it}) are uncorrelated, I guarantee that the instruments are valid and the estimator is appropriate. I present the results of Hansen test in tables 7.5 and 7.6, which show that the instruments are valid.

Additionally, I implement Arellano and Bond (1991) to check for the validity of the model. It takes the shape of a normal distribution with N(0,1). It mainly examines the serial correlation in the first difference residual (v_{it}) over different periods (s) by using the (m_j) statistic $[E(v_{it} v_{is})=0$, while $t\neq s$]. I accept first order serial correlation in the model because the estimator takes the first difference to eliminate the individual specific effects (η_i) . However, I reject second order serial correlation (m_2) in the residual because it indicates a problem in the model. I present the results of the AB test in tables 7.5 and 7.6, which confirm that no second order serial correlations exist in the model.

Finally, I use the Wald test (z_1) to check for the joint significance of the reported coefficients in the model. If I reject the null hypothesis that states no relationship between the dependent and explanatory variables, I make sure that the model is jointly

significant. I present the results of the Wald test in tables 7.5 and 7.6. They all support the joint significance of the reported coefficients.

7.5 Results

7.5.1 Market Concentration and Earnings Management

I start by examining the effect of market concentration, represented by the Hall Tideman index (*HTI*), on accrual earnings management. Column (1) of Table 7.5 presents the results of the system GMM estimation of the model. The results show that α_I is significantly positive while α_2 is significantly negative; which support hypothesis *H1* and confirm the quadratic relationship between market concentration and accrual earnings management according to Figure 7.1. By calculating the breakpoint (*MC*₁), I find that accrual earnings management increases at market concentration levels between 0 and 55% but starts decreasing after that level. Similar results are obtained by using the Modified Jones' Model (1995) as shown in Appendix B Table I and the OLS estimator as shown in Appendix B Table II.

I also show the results of the system GMM estimation for the effect of market concentration on real earnings management in column (2) of Table 7.5. They show that α_1 is significantly negative while α_2 is significantly positive; which also support hypothesis *H1* and confirm the quadratic relationship presented in Figure 7.2. By calculating the breakpoint, I find that real earnings management decreases at market concentration levels between 0 and 60% but starts increasing afterwards. Similar results are obtained by using the Roychowdhury's Model (2006) as shown in Appendix B Table I and the OLS estimator as shown in Appendix B Table II.

	(1)	(2)
Variables	AccrualEM _{it}	<i>RealEM_{it}</i>
Concentration _{it}	0.306***	-0.827***
$Concentration_{it}^2$	-0.276***	0.694***
AccrualEM _{i,t-1}	-0.092***	-
RealEM _{i,t-1}	-	0.323***
<i>FirmSize_{it}</i>	0.070***	0.038***
$SalesVolatility_{i,t-4,t}$	0.295***	0.346***
CashFlowVolatility _{i,t-4,t}	0.093***	0.034***
<i>OperCycle</i> _{<i>i</i>,<i>t</i>-4,<i>t</i>}	0.035***	0.016***
Losses _{i,t-4,t}	0.073***	-0.032***
MgrlAbility _{it}	0.930***	-0.084***
NationalAuditor _{it}	-0.081***	0.222***
LitigationInd _{it}	0.109***	0.072***
MB_{it}	0.000***	0.000***
$SalesGrowth_{rk_{it}}$	-0.721***	0.040***
Momentum _{it}	0.472***	-0.150***
LnNumAnalyst _{it}	0.065***	0.016***
IndRev%	-0.484***	0.080***
Segments _{it}	-0.100***	-0.056***
Foreign _{it}	-0.049***	0.165***
GFC_{it}	0.021***	0.142***
Dot-Com _{it}	-0.083***	0.120***
SOX _{it}	-0.025***	0.124***
Hansen	1,774.690	1,782.030
m_1	-19.450	-9.560
<i>m</i> ₂	-1.110	1.600
	31,069.380	2.3e+05

Table 7.5 The effect of marke	t concentration on	i earnings management
-------------------------------	--------------------	-----------------------

Notes: This table presents the results from the system generalized method of moments regressions for the effect of market concentration on earnings management. The sample includes 25,119 firm-year observations obtained from Compustat from 1989 to 2011. Market concentration (*Concentration*) is measured using the Hall Tideman Index (*HTI*) and each column represents the estimate from examining its effect on earnings management activities in the following order: (1) accrual earnings management measured by Stubben's Model (2010), and (2) real earnings management measured by Gunny's Model (2010). Each coefficient represents the change in earnings management based on a one unit change in the determinant. Variables are defined in the (Variable Definitions) section above. *Concentration* is the Hall Tideman Index and *Concentration*² is its square. Hansen test for over-identifying restrictions takes the shape of χ^2 distribution and checks for the orthogonality between the instrumental variables (IV_{ii}) and the error term (ε_{ii}). The AB test (Arellano and Bond, 1991) examines the serial correlation in the first difference residual (v_{ii}) by using the (m_j) statistic and takes the shape of a normal distribution with N(0,1). m_1 values represent the results of the 1st order serial correlation test, while m_2 values represent the results of the 2nd order serial correlation test. The Wald test (z_1) checks for the joint significance of the reported coefficients in the model. The table also displays evidence about the dynamic behaviour of earnings management in the third and fourth rows. Intercept is included, but not tabulated. *, **, **** denotes significance at 0.10, 0.05, and 0.01, respectively.

The previous results emphasize the double-edged effect of market concentration on earnings management through decreasing the costs of communication while increasing the asymmetry of information between management and shareholders. Therefore, while Dalia and Park (2009), Markarian and Santalo (2010), Karuna et al. (2012), and Datta et al. (2013) find linear relations, I document a non-linear effect of market concentration on both accrual and real earnings management. On the other hand, the results support those of Scherer (1967), Hermalin (1992), Schmidt (1997), and Aghion et al. (2005) that document a non-linear effect of the market structure on firm behaviour. Particularly, I am in line with Guo et al. (2015) who find a non-linear relationship between market concentration on earnings management based on information asymmetry theory in the next section.

7.5.2 Market Concentration and Information Asymmetry

I start by examining the effect of market concentration, represented by the Hall Tideman index (*HTI*), on information asymmetry measured by the inverse of analysts' coverage.³⁵ Column (1) of Table 7.6 presents the results of the system GMM estimation of the related model. I find that α_I is significantly positive; which supports hypothesis *H2a* and shows that there is more information asymmetry at higher levels of market concentration due to the lower analysts' coverage.³⁶ In this sense, the previous result shows a decline in the quantity of information with the increase in market concentration. Similar results are obtained by using the OLS estimator as shown in Appendix B Table III.

³⁵ The analysts' coverage variable is multiplied by -1 to reflect information asymmetry as explained in Table 7.1.

³⁶ As I am already expecting a linear relationship in hypothesis H2a, I do not test for the quadratic effect of market concentration on the inverse of analysts' coverage; and thus I leave a blank value for *Concentration*²_{it} in table 7.6.

	(1)	(2)	(3)
Variables	InvAnalystsCoverage _{it}	<i>ForecastError_{it}</i>	<i>ForecastDispersion</i> _{it}
<i>Concentration</i> _{it}	0.222***	-0.395***	-0.418***
$Concentration_{it}^2$	-	0.339***	0.393***
<i>FirmSize</i> _{it}	-0.134***	-0.039***	-0.019***
SalesVolatility _{i,t-4,t}	-0.102***	0.097***	0.054***
CashFlowVolatility _{i,t-4,t}	-0.156***	-0.161***	-0.191***
<i>OperCycle</i> _{<i>i</i>,<i>t</i>-4,<i>t</i>}	0.028***	0.006***	-0.002*
Losses _{i,t-4,t}	-0.032***	0.290***	0.213***
MgrlAbility _{it}	0.028***	-0.113***	-0.065***
NationalAuditor _{it}	0.043***	0.182***	0.077***
LitigationInd _{it}	0.032**	-0.003	-0.078***
MB _{it}	-0.000***	6.81e-05***	3.14e-05***
$SalesGrowth_{rk_{it}}$	-0.020***	-0.164***	-0.074***
<i>Momentum_{it}</i>	-0.102***	-0.169***	-0.116***
IndRev%	0.441***	-0.264***	-0.207***
Segments _{it}	0.126***	0.026***	0.006***
Foreign _{it}	0.032**	0.137***	0.069***
Hansen	1,790.650	1,823.270	1,879.210
m_l	-21.440	-10.550	-8.510
<i>m</i> ₂	-2.170	-0.950	-1.360
z_1	990.610	3,791.370	11,269.540

Table 7.6 The effect of market concentration on infor	mation asymmetry
---	------------------

Notes: This table presents the results from the system generalized method of moments regressions for the effect of market concentration on information asymmetry. The sample includes 25,119 firm-year observations obtained from Compustat from 1989 to 2011. Market concentration is measured using the Hall Tideman Index (*HTI*) and each column represents an estimate from examining its effect on information asymmetry measures in the following order: (1) The inverse of analysts' coverage, (2) earnings forecast error, and (3) earnings forecast dispersion. Each coefficient represents the change in information asymmetry based on a one unit change in the determinant. Variables are defined in the (Variable Definitions) section above. *Concentration* is the Hall Tideman Index and *Concentration*² is its square. Hansen test for over-identifying restrictions takes the shape of χ^2 distribution and checks for the orthogonality between the instrumental variables (IV_{it}) and the error term (ε_{it}). The AB test (Arellano and Bond, 1991) examines the serial correlation in the first difference residual (v_{it}) by using the (m_j) statistic and takes the shape of a normal distribution with N(0,1). m_1 values represent the results of the 1st order serial correlation test, while m_2 values represent the results of the 2nd order serial correlation test. The Wald test (z_1) checks for the joint significance of the reported coefficients in the model. Intercept is included, but not tabulated. *, **, *** denotes significance at 0.10, 0.05, and 0.01, respectively.

I then examine the effect of market concentration, represented by the Hall Tideman index (*HTI*), on information asymmetry measured by earnings forecast error and earnings forecast dispersion. Column (2) and (3) of Table 7.6 present the results of the system GMM estimation of the related model. The results show that α_1 is significantly

negative while α_2 is significantly positive; which support hypothesis *H2b* and confirm the quadratic relationship between market concentration and earnings forecast error and dispersion. By calculating the breakpoints, I find that earnings forecast error and dispersion decrease up to market concentration levels of 58% and 53% respectively but start increasing afterwards. Similar results are obtained by using the OLS estimator as shown in Appendix B Table III. In this sense, I evidence an increase in the quality of information provided by analysts at lower levels of market concentration because of the lower errors and dispersion in their forecasts, followed by a decrease in that quality at higher levels of market concentration. These results suggest that with the increase in market concentration, analysts are able to produce better forecasts based on firms' higher motivation to signal quality information under the lower probability of competitors to threaten their competitive advantages. However, after a certain point, the few remaining competitors are more likely to collude and ultimately provide low quality information and, therefore, make it more difficult for the analysts to compare the performance between the few existing firms.

The results provide another piece of evidence on the effect of market concentration on information asymmetry. While Ali et al. (2014) and Dalia and Park (2009) emphasize a linear positive effect, I document a non-linear relationship. However, the results are more supportive to those of Bhattacharya et al. (2013) who document a non-linear relationship between earnings quality and information asymmetry.

7.6 Additional Analysis: Governance Controls

Many studies have documented that corporate governance influences earnings management (García Lara et al., 2007; Duh et al., 2009; Feng et al., 2009; Jaggi et al., 2009; Prencipe and Bar-Yosef, 2011; Demerjian et al., 2013a). Better governance quality contributes in aligning information between management and shareholders and, therefore, mitigates the information asymmetry and contributes to decreasing the uncertainty in the market (Elbadry et al., 2013). As a result, I perform all the previous regressions once more using additional controls for corporate governance. I add three

control variables which I obtain form Execuocomp database.³⁷ I use a dummy for executives who serve as directors (*ExecDir*) which equals 1 if the executive served as director during the year. I also use the executive tenure (*Tenure*) calculated as the log of the executive tenure measured in days. Finally, I add the number of board meeting (*BoardMtgs*) held during the year (Adut et al., 2013). The previous variables, however, are only available from 1992 to 2006 and, therefore, I cover only this period in the additional analysis. As a result, I end up with 18,505 observations. The results emphasize the previous findings related to the relationship between market concentration, information asymmetry and earnings management. The results, however, are not tabulated here.

7.7 Conclusion

In this chapter, I examine the effect of market concentration on earnings management that has been widely debated in the literature. For this purpose, I explain the relationship based on the asymmetry of information in concentrated markets and provide some new insights. First, while the previous studies have examined the linear effect of market concentration on earnings management, I document a significant non-linear effect where accrual earnings management increases up to a certain level of market concentration then real earnings management starts to dominate.

Second, I explain the previous trend of earnings management based on the quantity and quality of information in the market. As the quantity of information decreases in concentrated markets, managers get motivated to use discretionary accruals due to the weak monitoring. However, when the quality of information starts to decline, managers tend to use more of the sophisticated real earnings management as it is more difficult to detect and understand.

Third, the results emphasize the substitutability of earnings management activities based on their relative costs to the firms. While, firms consider accrual earnings management less costly at lower levels of market concentration, they find real earnings management a better choice at higher levels of market concentration. However,

³⁷ Data generously provided by Francesco Vallascas and Paula Castro.

an optimal level of market concentration exists between the levels of 55% and 60% where both accrual and real earnings management are falling and, therefore, regulatory intervention is required to emphasize that level. For future research, I recommend examining the non-linear relationship between the other measures of market competition and earnings management as it may provide a new understanding of this area of research.

Appendix B

Appendix D Table I The check of market concentration on carmings managemen
--

Variables	(1) AccrualEM _{it}	(2) $RealEM_{it}$
<i>Concentration</i> _{it}	0.270***	-0.039***
<i>Concentration</i> ² _{<i>it</i>}	-0.240***	0.032***
AccrualEM _{i,t-1}	0.055***	-
RealEM _{i,t-1}	-	0.597***
<i>FirmSize_{it}</i>	-0.015***	0.163***
$SalesVolatility_{i.t-4.t}$	0.044***	0.318***
CashFlowVolatility _{i 1-4 1}	0.061***	0.104***
<i>OperCycle</i> _{<i>i</i>,<i>t</i>-4,<i>t</i>}	0.086***	0.091***
Losses _{i,t-4,t}	-0.269***	0.195***
MgrlAbility _{it}	0.935***	-0.235***
NationalAuditor _{it}	-0.010***	0.014***
LitigationInd _{it}	0.299***	0.040***
MB _{it}	0.000***	0.000***
$SalesGrowth_{rk_{it}}$	-0.336***	0.217***
Momentum _{it}	0.191***	0.327***
LnNumAnalyst _{ir}	0.030***	0.053***
IndRev%	-0.158***	-0.902***
Segments _{it}	-0.033***	-0.104***
Foreign _{it}	-0.056***	-0.148***
<i>GFC</i> _{it}	0.073***	-0.422***
Dot-Com,,	0.065***	-0.354***
SOX _{it}	0.078***	-0.387***
Hansen	1,788.330	1,757.050
m_1	-17.000	-6.760
<i>m</i> ₂	0.190	1.220
<i>z</i> ₁	20,035.810	4.5e+05

Notes: This table presents the results from the system generalized method of moments regressions for the effect of market concentration on earnings management. The sample includes 25,119 firm-year observations obtained from Compustat from 1989 to 2011. Market concentration (*Concentration*) is measured using the Hall Tideman Index (*HTI*) and each column represents the estimate from examining its effect on earnings management activities in the following order: (1) accrual earnings management measured by Modified Jones' Model (1995), and (2) real earnings management measured by Roychowdhury's Model (2006). Each coefficient represents the change in earnings management based on a one unit change in the determinant. Variables are defined in the (Variable Definitions) section above. *Concentration* is the Hall Tideman Index and *Concentration*² is its square. Hansen test for overidentifying restrictions takes the shape of χ^2 distribution and checks for the orthogonality between the instrumental variables (IV_{it}) and the error term (ε_{it}). The AB test (Arellano and Bond, 1991) examines the serial correlation in the first difference residual (v_{it}) by using the (m_j) statistic and takes the shape of a normal distribution with N(0,1). m_1 values represent the results of the 1st order serial correlation test, while m_2 values represent the results of the 2nd order serial correlation test. The Wald test (z_1) checks for the joint significance of the reported coefficients in the model. The table also displays evidence about the dynamic behaviour of earnings management in the third and fourth rows. Intercept is included, but not tabulated. *, ***, *** denotes significance at 0.10, 0.05, and 0.01, respectively.
	(1)	(2)
Variables	AccrualEM _{it}	<i>RealEM_{it}</i>
Concentration _{it}	0.255*	-0.286***
Concentration ² _{it}	-0.229*	0.242**
AccrualEM _{i,t-1}	-0.104**	-
RealEM _{i,t-1}	-	0.652***
<i>FirmSize_{it}</i>	0.011	0.005
$SalesVolatility_{i,t-4,t}$	0.011	0.076**
$CashFlowVolatility_{i,t-4,t}$	0.026	-0.029
<i>OperCycle</i> _{<i>i</i>,<i>t</i>-4,<i>t</i>}	0.015	0.008
$Losses_{i,t-4,t}$	-0.003	-0.000
MgrlAbility _{it}	0.385***	-0.102***
<i>NationalAuditor_{it}</i>	0.000	0.062***
LitigationInd _{it}	0.041	0.092***
MB_{it}	0.000	5.77e-05
$SalesGrowth_{rk_{it}}$	-0.730***	0.047*
Momentum _{it}	0.129**	-0.013
LnNumAnalyst _{it}	0.055**	-0.004
IndRev%	-0.223***	0.042
Segments _{it}	0.003	-0.005
Foreign _{it}	0.046	0.032
<i>GFC</i> _{it}	0.012	0.066
Dot-Com _{it}	0.000	0.121
SOX _{it}	-0.011	0.166
R^2	0.255	0.452
<i>F-statistic</i>	15.300	22.580
Prob (F-statistic)	0.000	0.000

Appendix B	Table II	The effect of	' market	concentration	on earnings	management
------------	----------	---------------	----------	---------------	-------------	------------

Notes: This table presents the results from the ordinary least square regressions for the effect of market concentration on earnings management. The sample includes 25,119 firm-year observations obtained from Compustat from 1989 to 2011. Market concentration *(Concentration)* is measured using the Hall Tideman Index *(HTI)* and each column represents the estimate from examining its effect on earnings management activities in the following order: (1) accrual earnings management measured by Stubben's Model (2010), and (2) real earnings management measured by Gunny's Model (2010). Each coefficient represents the change in earnings management based on a one unit change in the determinant. Variables are defined in the (Variable Definitions) section above. *Concentration* is the Hall Tideman Index and *Concentration*² is its square. Intercept is included, but not tabulated. *, **, *** denotes significance at 0.10, 0.05, and 0.01, respectively.

	(1)	(2)	(3)
Variables	InvAnalystsCoverage _{it}	<i>ForecastError</i> _{it}	<i>ForecastDispersion</i> _{it}
Concentration _{it}	0.156***	-0.181*	-0.056
Concentration ² _{it}	-	0.169*	0.045
<i>FirmSize_{it}</i>	-0.108***	-0.019***	-0.003
$Sales Volatility_{i,t-4,t}$	-0.017	0.161***	0.078***
CashFlowVolatility _{i,t-4,t}	-0.166***	-0.009	0.011
<i>OperCycle</i> _{<i>i</i>,<i>t</i>-4,<i>t</i>}	0.007*	0.010	-0.005
$Losses_{i,t-4,t}$	-0.110***	0.196***	0.156***
MgrlAbility _{it}	-0.002	-0.075***	-0.046**
NationalAuditor _{it}	0.093***	0.046**	0.006
LitigationInd _{it}	-0.034***	0.038*	-0.006
MB _{it}	-0.000**	-3.50e-05	-1.60e-05
$SalesGrowth_{rk_{it}}$	-0.083***	-0.087***	-0.058***
<i>Momentum_{it}</i>	-0.155***	-0.183***	-0.104***
IndRev%	0.041	-0.175***	-0.111***
Segments _{it}	0.092***	0.078***	0.042***
Foreign _{it}	0.052***	0.027	0.017
R^2	0.254	0.137	0.122
<i>F-statistic</i>	137.540	14.80	8.49
Prob (F-statistic)	0.000	0.000	0.000

Notes: This table presents the results from the ordinary least square regressions for the effect of market concentration on information asymmetry. The sample includes 25,119 firm-year observations obtained from Compustat from 1989 to 2011. Market concentration is measured using the Hall Tideman Index (*HTI*) and each column represents an estimate from examining its effect on information asymmetry measures in the following order: (1) The inverse of analysts' coverage, (2) earnings forecast error, and (3) earnings forecast dispersion. Each coefficient represents the change in information asymmetry based on a one unit change in the determinant. Variables are defined in the (Variable Definitions) section above. *Concentration* is the Hall Tideman Index and *Concentration*² is its square. Intercept is included, but not tabulated. *, **, *** denotes significance at 0.10, 0.05, and 0.01, respectively.

Chapter 8 Earnings Management, Managerial Ability and the Role of Competition

8.1 Introduction

While chapter 6 and 7 have focused on examining factors from the contractual or thirdparty motives of earnings management individually, this chapter examines how both groups interact in determining earnings management behaviour. High quality managers possess the abilities to manage earnings and achieve private benefits, but they may avoid earnings management because they are aware of its unfavourable consequences on their firms (Francis et al., 2008; Demerjian et al., 2013b). Meanwhile, the behaviour of superior managers is unlikely to be independent of the external factors that influence the firm. Some managers with high levels of ability, for example, will operate in business environments that are highly competitive, while others will operate in environments with significantly lower levels of competition.³⁸ Therefore, in this chapter I examine whether the influence of a third-party like competitors disciplines more able managers or drives their opportunistic behaviour.

Earnings management behaviour within a firm is determined by different sets of motives that can be related to contracting, capital market structure, or other external factors (Walker, 2013). Some of the contractual drivers of earnings management are the abilities and competencies of the managers of the firm, which affect the cost of managerial contracts with shareholders (Francis et al., 2008; Demerjian et al., 2013b). In the presence of high quality managers, shareholders may have to continuously consider, write, negotiate, and renegotiate the employment contracts of skilled managerial labour (Hart, 1983; Walker, 2013). Therefore, contracts with highly skilled managers may be insufficient to monitor their performance effectively, thereby giving them more space to manage their firms successfully. As such, highly skilled managers

³⁸ The converse is also true in the case of managers with lower levels of ability.

may be more able to realize the long-term gains associated with creating sustainable value, both through current and future remuneration and via increased labour market capital. In this situation, shareholders are more likely to trust the signals produced by these managers (e.g. earnings); and these managers' incentives are better aligned with shareholders who consequently feel less need to monitor their behaviour (Hart, 1983; Walker, 2013).

A key external driver of earnings management that has been widely examined in the literature is the level of market competition present in an industry (Dalia and Park, 2009; Markarian and Santalo, 2010; Karuna et al., 2012; Datta et al., 2013). Competition increases the level of transparency in an industry, and as a result, it decreases the information asymmetry between managers and shareholders (Dalia and Park, 2009). However, while the presence of increased information in highly competitive environments allows shareholders to more effectively compare firm performance, it may motivate managers to undertake increased levels of earnings management.

Based on the previous arguments concerning managerial ability and market competition, and the expectation that these two factors interact in explaining earnings management, I examine the influence of managerial ability on the earnings management behaviour in the presence of differential levels of market competition. In measuring managerial ability, I follow the model developed by Demerjian et al. (2012) that measures ability as the capacity of management to generate revenues from firm resources. For market competition I use three of the most commonly used measures, namely, market concentration, product substitutability, and market pricing power (Dalia and Park, 2009; Markarian and Santalo, 2010; Karuna et al., 2012; Datta et al., 2013). Finally, for the measures of earnings management I apply the Stubben (2010) and the Modified Jones (1995) models to measure accrual earnings management.

I find that high quality managers respond to market competition by using more discretionary accruals while avoiding real earnings management. The results extend the work of Karuna et al. (2012) who document an overall positive effect of market competition on both accrual and real earnings management. However, I particularly

202

show that superior managers are not likely to use the later activity because of its higher subsequent costs to their firms.

The remainder of the chapter proceeds as follows. In the next section I review the related studies from the literature and build the hypotheses of the chapter. In section 8.3 I explain the process of data collection, variable definitions, and the descriptive statistics. In section 8.4 I explain the methodology I use in this chapter. In section 8.5 I present and explain the results. In section 8.6 I add some empirical analysis. Finally, in section 8.7 I offer conclusions.

8.2 Literature and Hypotheses Development

8.2.1 Earnings Management Behaviour

According to the theory of the firm, the separation of ownership from control in public companies creates an agency problem (Ronen and Yaari, 2008). In this conflict of interest, shareholders and managers act rationally to maximize their own welfare. However, earnings serve as a performance measure that enables shareholders to monitor managers and hence alleviate the previous tension. That is, earnings have potential value to the shareholders in making their decisions – namely providing them with relevant information to establish or adjust their contracts with the management (Walker, 2013).

However, earnings need not always deliver their expected value as management prepares the financial statements that shareholders rely on. The information asymmetry gives management the opportunity to obfuscate its failures or signal specific messages to outsiders. Under such uncertainty, shareholders will not be able to completely understand management actions (Merkl-Davies and Brennan, 2007). In response to this weak form of market efficiency, managers may take advantage of the imperfect contracts with shareholders and use their judgement to alter the financial statements and report earnings in a different way from what is already known to them (Ronen and Yaari, 2008).

In light of the previous discussion, the motives of earnings management behaviour can be explained using the revelation principle and the conditions identified by Walker (2013). According to Walker (2013), at least one of the following four conditions needs to be satisfied for earnings management to occur. First, there are high contracting costs, especially under the impossibility of establishing perfect contracts to cover all potential terms required to co-ordinate between management and the large number of shareholders. Second, shareholders are 'imperfect rational' and are unable to make wealth maximizing decisions all of the time. Third, there is a lack of common shared knowledge (in the form of a management information systems), which precludes shareholders from fully understanding management actions. Finally, there are high relative communication costs due to the presence of third parties such as regulators, competitors, and tax authorities – essentially, the performance of a firm that is relevant to shareholders is conditioned by the performance of immediate competitors, how far the regulatory environment affects the specific behaviour of the firm and how able the firm is to manage the tax regime. However, as Walker (2013) has not distinguished between accrual and real earnings management, I incorporate the costs of both activities into the previous model when explaining the selected motives of earnings management, as discussed in the next sections.

8.2.2 Managerial Ability, Market Competition and Accrual Earnings Management

To the extent that perfect contracts do not exist in terms of accommodating all current and potential terms, more simplified general formats are likely to be used. Within this context, more able managers are more likely to emphasize aspects of their firms' performance that allow them to achieve better personal benefits. This conjecture is in line with the rent extraction hypothesis and leads to aggravating moral hazard problem between shareholders and superior managers. Consistent with the previous argument, Francis et al. (2008) provide evidence that superior managers do not necessarily produce better quality disclosures. They point out that managerial ability leads to a decrease in earnings quality because better managers are more able to manage earnings to obtain higher stock compensation, etc. Likewise, Demerjian et al. (2013b) examine the effect of managerial ability on accrual earnings management and document a positive relationship.

At a more macro-level, I expect earnings management behaviour to depend on market competition. To the extent that third parties like competitors influence managerial communication with shareholders, the effect of market competition can be best explained by violating the fourth condition of the revelation principle mentioned above and hence changing the motivation of managers to manage earnings (Walker, 2013). Communicating more information in competitive markets allows shareholders and managers to make more comparisons between firms in the same industry which potentially drives managers to imitate the aggressive activities of other managers (Lieberman and Asaba, 2006). Put differently, market competition drives more adverse selection by triggering management herding behaviour in order to meet market expectations. Consistent with this argument, Milgrom and Roberts (1992) analyse a case study in the savings and loans industry in the United States in the 1980s where conservative managers imitated the aggressive behaviour in the industry by offering high interest rates to depositors in order to collect more money for their projects and stay in the business. Other studies particularly study the effect of market competition on accrual earnings management. Karuna et al. (2012) and Datta et al. (2013) find a positive impact because of the comparisons that create pressure to meet competitors' performance. Moreover, the herding behaviour is very likely in this setting because managers can easily understand each other's incentives in the same industry (Kallunki and Martikainen, 1999; Bagnoli and Watts, 2000). In contrast, Dalia and Park (2009) provide evidence that competition discourages discretionary accruals in the manufacturing sector in the United States because information spreads more easily and hence the stock market promptly punishes earnings management behaviour. Following a similar line of argument, Markarian and Santalo (2010) document that when shareholders have access to more information about the firm and its competitors, earnings management activities become more costly as they are difficult to be justified by management.

Given the abilities of superior managers to trade-off between the short and long horizons after evaluating the costs of earnings management activities as explained earlier, I expect them to react carefully to the pressure of market competition. To the extent that the high communication costs can drive managers to act more aggressively in order to survive in the market, high quality managers are more likely to use the less

205

costly accrual earnings management to achieve this target. Consistent with this argument, I develop the following hypothesis:

H1a: The positive relationship between managerial ability and accrual earnings management will be enhanced with the extent of market competition in the industry where the firm operates.

8.2.3 Managerial Ability, Market Competition and Real Earnings Management

In contrast to the previous argument, more able managers provide higher quality performance and are obviously aware of the consequences of their behaviour on their reputations and future benefits. Therefore, shareholders expect superior managers to emphasize the long horizons of their firms. The assumption is in line with the efficient contracting hypothesis and implies a decrease in moral hazard problem between more able managers and shareholders in the long-run. The above idea has some foundation in the prior research that examines the effect of managerial ability on earnings management. Demerjian et al. (2013b) find that superior managers are less inclined to use earnings management by particularly avoiding real earnings management activities because of the negative consequences for their firms. To the extent that real earnings management involves activities like delaying expenses, asset sales, increasing production, sales discounts, or R&D cuts, it results in giving up positive net present value (NPV) projects and burning cash flows and, therefore, achieving short-term earnings' targets at the expense of firm value. Because superior managers can anticipate the negative long-term consequences of real earning management, they are less likely to use such costly activities.

At the same time, superior managers are not independent of the external motives and thus market competition may also influence their real earnings management behaviour. Some studies provide evidence on the disciplinary function of market competition. Holmstrom (1982) finds that the performance evaluation of a manager relative to others in the same industry results in more information about the common uncertainties in the market and, thus, helps to mitigate the conflict of interest between management and shareholders. Similarly, Hart (1983) provides evidence that there is less space for discretionary behaviour under competition as firms within the same

206

industry share a business environment that allows performance comparisons and ultimately rules out any supernormal profits. Furthermore, Chhaochharia et al. (2012) specifically find that market competition is associated with better operational efficiency that makes it a potential substitute for internal governance mechanisms. In contrast, Karuna et al. (2012) examines the effect of market competition on real earnings management. Their results show managers use more real earnings management under the intensive comparisons between firms and the tendency to imitate the opportunistic behaviour of other managers in the same industry. Because superior managers possess the abilities to extract private benefits without sacrificing long-term gains (Demerjian et al., 2013b), I expect them to pursue improving shareholder value even under the pressure of the high communication costs and adverse selection driven by market competition. Top managers are therefore more likely to avoid real earnings management because of the associated costs. Drawing on the previous discussion I develop the following hypothesis:

H1b: The negative relationship between managerial ability and real earnings management will be enhanced with the extent of market competition in the industry where the firm operates.

8.3 Data, Variable Definitions, and Descriptive Statistics

8.3.1 Data

To undertake the analysis, I include all firms in the United States from the Annual Compustat, Quarterly Compustat, Historical Segments Compustat, and IBES databases. I start with all firms that have the required data for calculating the measures of earnings management, competition, and managerial ability.³⁹ Following Cheng et al. (2011), I exclude regulated industries such as banks, credit institutions, brokers, insurance, real estate, holding companies, and investment firms. These industries have their unique accounting and financial practices and are subject to distinct regulations. Therefore, managers in these industries have different motivations to manipulate earnings than

³⁹ For estimates of managerial ability, I use the dataset of Demerjian et al. (2012) which is available at <u>https://community.bus.emory.edu/personal/PDEMERJ/Pages/Home.aspx</u>

those of managers in other industry sectors.⁴⁰ In addition, I exclude firm-years where accounting changes, merger and acquisition activities, or discontinued operations occur.⁴¹

Following prior literature, I exclude any industry with fewer than six observations for each SIC code in a specific year to ensure sufficient data exists to calculate earnings management measures and make sure that OLS assumption regarding the normality of the error term holds (e.g., Rosner, 2003; García Lara et al., 2005; Kothari et al., 2005; Athanasakou et al., 2009; Iqbal et al., 2009). For that purpose, I follow the SIC classification of Fama-French (1997). As the model is dynamic, I ensure that information is available for at least five consecutive years for each firm over the study period (Miguel et al., 2004). When I combine the cross-sectional and time-series dimensions of the data, I obtain 66,695 observations in an unbalanced panel.⁴²

8.3.2 Earnings Management Measures

Based on the discussion in Chapter 5, I use Stubben's Model (2010) for measuring accrual earnings management because it focuses on discretionary revenues which represent the largest component of earnings in most firms (Stubben, 2010). I also use Gunny's Model (2010) for real earnings management because it covers more types of this activity compared to other models (Gunny, 2010). In addition, I use two other widespread models as corroborating measures that include the Modified Jones' Model (1995) for measuring accrual earnings management and Roychowdhury's Model (2006) for measuring real earnings management. Again, for each of these measures I use a cross sectional model in the calculations for each year and each industry classified by its four-digit SIC code with at least 6 observations. I also take the decile ranks for better comparability and to avoid the effect of the outliers. For the purpose of calculating an overall measure of real earnings management using Gunny's Model (2010) and Roychowdhury's Model (2006), a principal component analysis with varimax rotation is

⁴⁰ I exclude firms with the following SIC codes: $4000 \le SIC \ge 4900$ and $6000 \le SIC \ge 6300$.

⁴¹ According to McNichols (2002), I specifically exclude firm quarters or years with non-blank values for accounting changes cumulative effects (ACCCHGQ_FN), or merger and acquisition activities (ACQMETH_FN), or discontinued operations (DOQ_FN) in the Compustat database.

⁴² By having an unbalanced panel I avoid any issues regarding survivorship bias.

performed to avoid the inclusion of highly correlated variables in the overall measure (Demerjian et al., 2013b).⁴³

8.3.3 Managerial Ability Measure

To calculate the managerial ability variable, I adopt the approach of Demerjian et al. (2012) as discussed in section 4.3.3. The measure represents managerial potential of generating revenues from the available firm resources. It uses Data Envelopment Analysis (DEA) to calculate firm efficiency then excludes some firm specific characteristics that may influence the performance of managers. The remaining residual (ε_i) is the ultimate measure of managerial ability (*MgrlAbility*). Finally, I take the decile ranks of these measures by industry-year to obtain better comparability and to avoid the effects of outliers.

8.3.4 Competition Measures

As market competition has various dimensions, I use three of the most commonly used measures in the literature for this variable. These measures include market concentration, product substitutability, and market pricing power. The Herfindahl-Hirschman index (*HHI*) is the most commonly expressed measure in the literature to proxy for market concentration. I calculate it using the following equation:

$$HHI = \sum_{i=1}^{N} \omega_i^2 \tag{8.1}$$

In equation (8.1), ω is market share; and N is the number of firms per year-industry. Following Karuna et al. (2012) I use product substitutability (*ProdSubstitutability*) to represent the gross margin in the industry; and I measure it by the average operating profit divided by the sales in the industry. Finally, I measure firm-specific product market pricing power (*PricPower*) by the industry adjusted Lerner Index (*LI*). As in

⁴³ The principal component analysis step turns the set of correlated variables to be linearly uncorrelated according to the weights of their variances, thus reducing the number of variables to their principal components.

Datta et al. (2013) I adjust the Lerner Index by excluding industry specific characteristics that interfere with market pricing power as follows:

$$LI_i = \frac{Sales - COGS - SG \& A}{Sales}$$
(8.2)

$$PricPower = LI_i - \sum_{i=1}^{N} \omega_i LI_i$$
(8.3)

In equations (8.2) and (8.3), LI_i is the Lerner index for firm i, also referred to as the price cost margin (PCM); and ω_i is market share for firm i measured by its sales divided by total industry sales.

8.3.5 Control Variables

Following Dechow and Dichev (2002), the first set of controls consider firm specific determinants and account for firm size (*TotalAssets*), sales volatility (*SalesVolatility*), cash flow volatility (*CashFlowVolatility*), operating cycle (*OperCycle*), and historical losses (*Losses*). Larger firms have operations that are more predictable and more diversified businesses and thus earnings should be of higher quality. Managers in these firms, however, have more opportunities to manipulate earnings because of the large number of transactions they undertake and their complicated operations (Demerjian et al., 2013b). Sales volatility reflects uncertainty in operations and, therefore, implies a higher likelihood of earnings management. Cash flow volatility also reflects increased volatility in the operations of the firm creating more opportunities for earnings management. Similarly, longer operating cycles increase uncertainty and therefore, the potential for earnings management. Prior losses are likely to result in more earnings management to meet investors' expectations in making profit but equally, they may drive managers to avoid earnings management as they will be facing more scrutiny by the market.

I use a national auditor indicator (*NationalAuditor*) to control for any auditor effects as firms audited by national auditors use more earnings management than those audited by international audit firms (Becker et al., 1998). I add another indicator variable to control for litigious industries (*LitigationInd*) following Francis et al. (1994).

Firms in such industries are expected to use more real activities as accrual earnings management is more likely to be detected. However, the opposite may occur as real earnings management has serious economic consequences to the firm. I also add the market to book ratio (*MB*), the one year sales growth (*SalesGrowth*), and returns momentum (*Momentum*) as they have been found to influence earnings management (Hribar and Nichols, 2007). As firms with higher market to book ratios have more growth prospects, they may be involved in more earnings management to meet market expectations (Hribar and Nichols, 2007). Firms with growing sales and high returns momentum also have high growth prospects but the increase in sales and returns reduces the pressure on management to manipulate earnings (Dechow et al., 2011).

I also control for analyst coverage (*NumAnalyst*). Although the presence of more analysts may discipline managers, it can increase the burden to present better earnings and, therefore, may lead to more of the sophisticated earnings management activities. I also include an industry revenue leadership variable (*IndRev%*) (Dechow et al., 2011; Zang, 2012). Although Zang (2012) could not specify the influence of industry revenue leadership, I expect a negative effect because industry leaders are under more scrutiny from investors and the SEC. I use the number of segments (*Segments*) and the frequency of foreign transactions (*Foreign*) to control for businesses complexity (Karuna et al., 2012). With more complex transactions managers use more real earnings management because they are less likely to be detected than discretionary revenue manipulation (Karuna et al., 2012).

Finally, I use dummies for the global financial crisis (*GFC*) (Badertscher et al., 2014; Kanagaretnam et al., 2014; Magnan et al., 2015), the dot-com bubble (*DotCom*) (Lieberman and Asaba, 2006), and the post-Sarbanes Oxley period (*SOX*) (Cohen et al., 2008) to control for some macro-economic events over the study period. The possibility of trade-off between accrual and real earnings management has to be always taken into consideration when examining the previous factors. I present the detailed calculations for each of the previous variables in Table 8.1 and summarize the expected and actual signs of all control variables in Table 8.2.

Variable	Calculation
FirmSize	The natural log of the firm's assets as of the end of year <i>t</i> .
SalesVolatility	The standard deviation of (sales / average assets) over at least three of the last five years $(t-4, t)$.
CashFlowVolatility	The standard deviation of (cash from operations / average assets) over at least three of the last five years (t -4, t).
OperCycle	The operating cycle is the natural log of average sales turnover plus days in inventory over at least three of the last five years $(t-4, t)$.
Losses	Loss history is the percentage of years reporting losses in net income before extraordinary items over at least three of the last five years $(t-4, t)$.
NationalAuditor	An indicator variable equals one for firms audited by national audit firms in year t ; zero otherwise.
LitigationInd	Litigious industry indicator that equals one for firms in industries with SIC Codes: 2833-2836 (biotechnology), 3570-3577 and 7370-7374 (computers), 3600-3674 (electronics), and 52(X)-5961 (retailing).
МВ	The market to book ratio that equals the firm's market capitalization divided by book value for year t.
SalesGrowth	Current year's sales less prior year's sales less the increase in receivables all scaled by prior year's sales and decile ranked by industry and year.
NumAnalyst	The log of 1+ the number of analysts covering the firm in year t.
IndRev%	Industry revenue leadership measured by the firm's sales in year t-1 divided by the total sales for the firm's industry in year t-1.
Momentum	Returns momentum calculated by the decile rank (by industry and year) of asset returns during the two years preceding the start of year t.
Segments	The natural log of 1+ the number of firm's business segments in year t.
Foreign	The frequency with which the firm has a non-zero foreign currency transactions during the sample period.
GFC	An indicator variable equals one for the years of the global financial crisis (2007-2009); zero otherwise.
Dot-Com	An indicator variable equals one for the years of the Dot-Com bubble (1997-2000); zero otherwise.
SOX	An indicator variable equals one for the years after the release of the Sarbanes Oxley act (2002-2005); zero otherwise.

Table 8.1 Calculation of control variables

Notes: This table presents the detailed calculations for each of the control variables identified in the model as discussed in the (Variable Definitions) section above.

Variables	Accrual	EM	RealE	M
	Predicted sign	Actual sign	Predicted sign	Actual sign
FirmSize	+/-	+	+/-	+
SalesVolatility	+	+	+	+
CashFlowVolatility	+	+	+	+
OperCycle	+	+	+	+
Losses	+/-	+	+/-	-
NationalAuditor	+	+	+	+
LitigationInd	+/-	-	+/-	+
MB	+	+	+	+
SalesGrowth	+/-	-	+/-	+
NumAnalyst	+/-	+	+/-	+
IndRev%	-	-	-	-
Momentum	+/-	+	+/-	-
Segments	+/-	-	+/-	-
Foreign	+/-	-	+/-	+
GFC	+/-	-	+/-	-
Dot-Com	+/-	-	+/-	-
SOX	+/-	-	+/-	-

Table 8.2 Expected and actual	signs of the control variables
-------------------------------	--------------------------------

Notes: This table presents the signs I expect and actually get for all control variables identified in the model and discussed in the (Variable Definitions) section above.

8.3.6 Descriptive Statistics

I present the descriptive statistics in Table 8.3. Consistent with Demerjian et al. (2013b), accrual earnings management and real earnings management have means and medians of zero as these variables are taken from the residuals of the models selected above. The measures of market competition are comparable to those calculated by prior studies (e.g., Karuna et al., 2012; Datta et al., 2013). Similarly, managerial ability measure is consistent with Demerjian et al. (2013b) and has a mean and median close to zero because it is measured as the residual from the firm efficiency equation.

Variables	Mean	Median	Std Dev	P25	P75
AccrualEM	0.00	0.00	0.06	-0.02	0.02
RealEM	0.00	0.00	1.00	-0.72	0.71
HHI	0.28	0.22	0.18	0.14	0.37
ProdSubstitutability	0.17	0.15	0.10	0.10	0.23
PricPower	-3.60	-0.05	73.04	-0.22	0.02
MgrlAbility	0.00	-0.01	0.15	-0.10	0.08
TotalAssets	1,759.47	86.79	10,847.54	18.43	446.71
SalesVolatility	0.29	0.17	0.46	0.09	0.34
CashFlowVolatility	0.14	0.07	0.41	0.04	0.14
OperCycle	134.37	120.05	81.96	76.79	176.29
Losses	0.40	0.25	0.37	0.00	0.75
NationalAuditor	0.24	0.00	0.43	0.00	0.00
LitigationInd	0.08	0.00	0.27	0.00	0.00
МВ	3.50	1.98	58.04	1.05	3.70
SalesGrowth	1.20	0.08	83.94	-0.04	0.25
NumAnalyst	2.94	3.00	1.65	2.00	4.00
IndRev%	0.06	0.01	0.13	0.00	0.04
Momentum	-0.27	0.05	2.79	-0.22	0.16
Segments	1.74	1.00	1.45	1.00	2.00
Foreign	0.20	0.00	0.33	0.00	0.31
GFC	0.10	0.00	0.30	0.00	0.00
Dot-Com	0.24	0.00	0.42	0.00	0.00
SOX	0.17	0.00	0.38	0.00	0.00

Table 8.3 Descriptive statistics

Notes: The table presents the descriptive statistics of all the variables identified in the model for 66,695 firm-year observations obtained from Compustat from 1989 to 2011. Variables are defined in the (Variable Definitions) section above. I present the untransformed variables for ease of interpretation.

The control variables are comparable to Karuna et al. (2012) and Demerjian et al. (2013b) with some differences appearing because of the variations in the samples. Cash flow volatility is lower than sales volatility and thus implies that firms consider it more costly as it is negatively valued by the market (Allayannis and Weston, 2007). 40% of the firms in the sample reported a history of losses in the last five years which has contributed to an overall negative returns momentum. The high standard deviations associated with firm size and operating cycle are due to presenting the untransformed variables.

Table 8.4 presents the correlation matrix between the measures of earnings management, managerial ability, and competition. From the table it can be seen that managerial ability is positively correlated with accrual earnings management and negatively correlated with real earnings management, which suggests that high quality managers are more inclined to use accrual rather than real earnings management. However, there is no consistent direction for the measures of market competition and earnings management. As well as examining the correlation between the variables of interest, I also check the VIF between all the variables in the model and confirm they are less than 10.

Table 8.4 Correlation table						
	Accrual EM	Real EM	ННІ	Prod Substitutability	Pric Power	Mgrl Ability
Accrual EM						
Real EM	0.00					
ННІ	0.01**	0.02***				
Prod Substitutability	-0.01*	-0.03***	-0.21***			
Pric Power	0.00	0.02***	0.01***	-0.03***		
Mgrl Ability	0.06***	-0.06***	0.01**	-0.01*	0.05***	

Table 8.4 Correlation table

Notes: This table presents the correlation matrix for 66,695 firm-year observations obtained from Compustat from 1989 to 2011 between the measures of accrual earnings management measured by Stubben's Model (2010), real earnings management measured by Gunny's Model (2010), managerial ability measured by the model of Demerjian et al. (2012), market concentration measured by the Herfindahl-Hirschman index (*HHI*), product substitutability measured by the gross margin in the industry, and pricing power measured by the industry adjusted Lerner Index (*LI*). Variables are defined in detail in the (Variable Definitions) section above. Pearson correlation coefficients are presented. To control for the multicollinearity between the different variables I make sure that all VIF factors are less than 10. VIF factors, however, are not tabulated. *, **, *** denotes a statistical coefficient at the 10, 5 and 1 percent alpha level, respectively.

I next perform univariate analysis where I examine earnings management, competition and managerial ability simultaneously. I first identify in Panel A of Table 8.5 earnings management observations under high competition i.e. less than the median of the three measures of market competition calculated by industry and year. Among the resulting observations, I create a group for the most able managers (highest quintile of managerial ability variable by industry and year) and another group for the least able managers (lowest quintile of managerial ability variable by industry and year). Next, I identify in Panel B of Table 8.5 earnings management observations under low competition i.e. more than the median of the three measures of market competition calculated by industry and year, and again I divide the results into a high managerial ability group and a low managerial ability group.

When considering accrual earnings management, I find that under low and high levels of market concentration the means⁴⁴ of accrual earnings managed by high quality managers (0.006 and 0.005 successively) are higher than those evidenced by low quality managers (-0.007 and -0.004 successively). Similar results are achieved for the measures of product substitutability and pricing power. In contrast, when considering real earnings management I find that under low and high levels of market concentration the means of real earnings managed by high quality managers (-0.086 and -0.084 successively) are lower than those evidenced by low quality managers (0.081 and 0.104 successively). Again, similar results are achieved for the measures of product substitutability and pricing power. Overall, the results imply that high quality managers always use more accrual and less real earnings management compared to low quality managers.

⁴⁴ The same applies when the median is used for comparisons.

		Market	Competition belo	w median <i>HHI</i>		
	Lowest Quint	ile of Managerial	Highest Quinti	le of Managerial	Signif	icance of
	A	bility	Ab	oility	differ	ence test
Variable	Mean	Median	Mean	Median	Mean	Median
AccrualEM	-0.007	-0.003	0.006	0.001	***	***
RealEM	0.081	0.088	-0.086	-0.086	***	***
		Market Competi	tion below media	n <i>ProdSubstituta</i>	bility	
	Lowest Quint	ile of Managerial	Highest Quinti	le of Managerial	Signif	icance of
	A	bility	Ab	oility	differ	ence test
Variable	Mean	Median	Mean	Median	Mean	Median
AccrualEM	-0.006	-0.002	0.005	0.001	***	***
RealEM	0.046	0.056	-0.097	-0.089	***	***
		Market Cor	npetition below n	nedian PricPower		
	Lowest Quint	ile of Managerial	Highest Quinti	le of Managerial	Significance of	
	A	bility	Ab	oility	difference test	
Variable	Mean	Median	Mean	Median	Mean	Median
AccrualEM	-0.006	-0.002	0.002	0.002	***	***
RealEM	0.103	0.096	-0.149	-0.159	***	***
Panel B:		I	Low Market Com	petition		
		Market	Competition abov	ve median HHI		
	Lowest Quint	ile of Managerial	Highest Quinti	le of Managerial	Signif	icance of
	A	bility	Ab	oility	differ	ence test
Variable	Mean	Median	Mean	Median	Mean	Median
AccrualEM	-0.004	-0.001	0.005	0.001	***	***
RealEM	0.104	0.101	-0.084	-0.069	***	***
		Market Competi	tion above media	n <i>ProdSubstituta</i>	bility	
	Lowest Quint	ile of Managerial	Highest Quinti	le of Managerial	Signif	icance of
	A	bility	Ab	oility	differ	ence test
Variable	Mean	Median	Mean	Median	Mean	Median
AccrualEM	-0.005	-0.002	0.004	0.001	***	***
RealEM	0.132	0.123	-0.071	-0.055	***	***

Table 8.5 Earnings management variables partitioned by competition and managerial abilityPanel A:High Market Competition

	Market Competition above median PricPower					
	Lowest Quintile of Managerial Highest Quintile of Managerial					icance of
	Ability		Ability		difference test	
Variable	Mean	Median	Mean	Median	Mean	Median
AccrualEM	-0.003	-0.001	0.001	0.001	***	***
RealEM	0.107	0.096	-0.113	-0.112	***	***

Notes: This table presents the univariate analysis of earnings management by competition and managerial ability for 66,695 firm-year observations obtained from Compustat from 1989 to 2011. Accrual earnings management is measured by Stubben's Model (2010), real earnings management is measured by Gunny's Model (2010), managerial ability is measured by the model of Demerjian et al. (2012), market concentration is measured by the Herfindahl-Hirschman index (*HHI*), product substitutability is measured by the gross margin in the industry, and pricing power is measured by the industry adjusted Lerner Index (*LI*). Variables are defined in detail in the (Variable Definitions) section above. Panel A and panel B use the three measures I identified for market competition. I present the untransformed variables for ease of interpretation. The significances of the two-tailed t-test for the differences between means and medians are displayed. *, **, *** denotes a statistical coefficient at the 10, 5 and 1 percent alpha level, respectively.

However, the intensity of such behaviour changes at different levels of market competition. When considering accrual earnings management, I find that under lower levels of market concentration (higher competition) the mean⁴⁵ of accrual earnings managed by high quality managers (0.006) is higher than that observed under higher levels of market concentration (0.005). Similar results are achieved for the measures of product substitutability and pricing power. When shifting the attention to real earnings management, I find that under lower levels of market concentration the mean of real earnings managed by high quality managers (-0.086) is lower than that observed under higher levels of market concentration (-0.084). Similar results are achieved for the measures of product substitutability and pricing power. Overall, the results suggest that high quality managers use more accrual and less real earnings management under higher levels of market competition.

8.4 Methodology

I derive a model that identifies earnings management as the dependent variable (Y_{it}) . I also use the lagged values $(Y_{i,t-1})$ of earnings management (*EarningsManagement*_{i,t-1}) to explain its current levels. I control for the dynamic effect of earnings management because earnings performance in the past year will influence earnings management behaviour during the current year (Kim et al., 2003). In addition, earnings management levels are associated with meeting prior earnings' benchmarks (Graham et al., 2005; Cohen et al., 2008; Gunny, 2010). As a result, managers have to be aware of the prior year's earnings management levels when managing earnings in the current period.

⁴⁵ The same applies when the median is used for comparisons.

I introduce managerial ability ($MgrlAbility_{it}$) and a dummy for market competition dummy ($Competition_{it}$) as an the explanatory variables (X_{it}) and ($Z_{i,t}$). I also use competition dummy as a moderating variable.⁴⁶ To complete the model, I add the previously identified control variables ($Controls_{it}$) and an error term (ε_{it}). I express the model in general terms in equation (8.4) and in more specific terms in equation (8.5).

$$Y_{it} = \alpha_0 + \alpha_1 X_{i,t} + \alpha_2 Z_{i,t} + \beta_1 Z_{it} X_{it} + \alpha_4 Y_{i,t-1} + \alpha_{5-21} Controls_{it} + \varepsilon_{it}$$

$$(8.4)$$

$$\begin{aligned} EarningsManagement_{i} = & \alpha_{0} + \alpha_{I}MgrlAbility_{it} + \alpha_{2}Competition_{it} \\ + & \beta_{I}Competition_{it} * MgrlAbility_{it} \\ + & \alpha_{4}EarningsManagement_{i,t-1} + & \alpha_{5}FirmSize_{it} \\ + & \alpha_{6}SalesVolatility_{i,t-4,t} + & \alpha_{7}CashFlowVolatility_{i,t-4,t} \\ + & \alpha_{8}OperCycle_{i,t-4,t} + & \alpha_{9}Losses_{i,t-4,t} + & \alpha_{10}NationalAuditor_{it} \\ + & \alpha_{11}SalesGrowth_{rk_{it}} + & \alpha_{12}MB_{it} + & \alpha_{13}LitigationInd_{it} \\ + & \alpha_{14}LnNumAnalyst_{it} + & \alpha_{15}IndRev\% + & \alpha_{16}Momentum_{it} \\ + & \alpha_{17}Segments_{it} + & \alpha_{18}Foreign_{it} + & \alpha_{19}GFC_{it} \\ + & \alpha_{20}Dot-Com_{it} + & \alpha_{21}SOX_{it} + & \varepsilon_{it} \end{aligned}$$

$$(8.5)$$

The model suffers from the problem of unobservable individual heterogeneity attributed to time-invariant firm and managerial effects. Firm strategies and corporate cultures may result in firms with specific levels of earnings quality selecting managers with specific abilities. Similarly, managers' characteristics e.g., inborn capabilities, personalities, or tendencies to take risk, may drive them to target firms with specific levels of earnings quality. Consequently, the sample is expected to be heterogeneous, as some companies are more predisposed to earnings management than others (Graham et al., 2012; Demerjian et al., 2013b).

In addition, the model suffers from an endogeneity problem because of the mutual causality between managerial ability and earnings management. While more

⁴⁶ The dummy takes a value of one when competition is high in the market and zero otherwise. Therefore, α_1 is the coefficient of $(MgrlAbility_{it})$ when competition is low; $(\alpha_1 + \beta_1)$ is the coefficient when competition is high. I introduce a dummy for each of the three measures of market competition I identified earlier. *HHI* is a dummy variable that equals 1 if the firm is in an industry where market concentration is lower than the median and zero otherwise. ProdSubstitutability is lower than the median and zero otherwise. Finally, PricPower is a dummy variable that equals 1 if the firm has lower pricing power than the median and zero otherwise.

able managers are expected to be more selective in using the different earnings management activities, higher levels of earnings management may require hiring managers with specific abilities to successfully run the business. This effect would be captured in the error term and ultimately results in a correlation between the explanatory variable and the error term (Pindado and Requejo, 2014).

Therefore, the OLS estimator will not be able to solve the model as it ignores the impact of the unobservable individual heterogeneity or endogeneity problems. Furthermore, while the fixed effects estimator tackles the unobservable heterogeneity by demeaning the variables in the model it does not solve for the endogeneity problem as it assumes strict exogeneity. Therefore, I solve the model using a system generalized method of moments (GMM) estimator that demeans the variables in the model to solve for the heterogeneity and introduces instrumental variables to solve for the endogeneity problem as explained earlier in the methodology section of Chapter 5.⁴⁷ In addition, I present the results of the OLS regressions as a robustness test in Appendix C Table III and IV.

Therefore, I split the error term (ε_{it}) into three components. First, I introduce (η_i) to control for the impact of the unobserved effects in the model.⁴⁸ Second, I add a time specific effect (d_t) to control for the macroeconomic variables that may interfere with the results over the period of the study. Finally, I consider the remaining part of the error term (ε_{it}) to be a random disturbance (v_{it}).

$$\begin{aligned} EarningsManagement_{i} = & \alpha_{0} + \alpha_{1}MgrlAbility_{it} + \alpha_{2}Competition_{it} \\ + & \beta_{1}Competition_{it} * MgrlAbility_{it} \\ + & \alpha_{4}EarningsManagement_{i,t-1} + & \alpha_{5}FirmSize_{it} \\ + & \alpha_{6}SalesVolatility_{i,t-4,t} + & \alpha_{7}CashFlowVolatility_{i,t-4,t} \\ + & \alpha_{8}OperCycle_{i,t-4,t} + & \alpha_{9}Losses_{i,t-4,t} + & \alpha_{10}NationalAuditor_{ii} \\ + & \alpha_{11}SalesGrowth_{rk_{it}} + & \alpha_{12}MB_{it} + & \alpha_{13}LitigationInd_{it} \\ + & \alpha_{14}LnNumAnalyst_{it} + & \alpha_{15}IndRev\% + & \alpha_{16}Momentum_{it} \\ + & \alpha_{17}Segments_{it} + & \alpha_{18}Foreign_{it} + & \alpha_{19}GFC_{it} \\ + & \alpha_{20}Dot-Com_{it} + & \alpha_{21}SOX_{it} + & \eta_{i} + & d_{t} + & v_{it} \end{aligned}$$

⁴⁷ Solving for these two problems using a system GMM estimator allows us to obtain less biased and more significant results compared to the previous studies.

 $^{^{48}}$ η_i controls for both firm specific effects and manager specific effects.

After examining the individual effect of the moderating variable (*Competition_{it}*) and the independent variable (*MgrlAbility_{it}*), I test for their combined influence. For this purpose, I use a linear restriction test (LRT) that examines the significance of $(\alpha_1 + \beta_1)$. I use the different measures identified earlier for market competition (*Competition_{it}*) as moderating factors. Therefore, I repeat the test for each measure separately as shown in the next section.

After the completion of the empirical work, I ensure that the assumptions of the estimator hold and that the model is valid. I initially test whether the GMM estimator properly addresses the problem of endogeneity. For this purpose, I use the Hansen test for over-identifying restrictions. It takes a χ^2 distribution and checks for the orthogonality between the instrumental variables (IV_{it}) and the error term (ε_{it}). The model uses multiple lags of the right-hand side variables as instruments, which make it over-identified. Consequently, if I accept Hansen's null hypothesis that the instrumental variables (IV_{it}) and the error term (ε_{it}) are uncorrelated, I guarantee that the instruments are valid and the estimator is appropriate. I present the results of Hansen test in tables 8.6 and 8.7, which show that the instruments are valid.

Additionally, I implement Arellano and Bond (1991) to check for the validity of the model. It takes the shape of a normal distribution with N(0,1). It mainly examines the serial correlation in the first difference residual (v_{it}) over different periods (s) by using the (m_j) statistic $[E(v_{it} v_{is})=0$, while $t\neq s$]. I accept first order serial correlation in the model because the estimator takes the first difference to eliminate the individual specific effects (η_i) . However, I reject second order serial correlation (m_2) in the residual because it indicates a problem in the model. I present the results of the AB test in tables 8.6 and 8.7, which confirm that no second order serial correlations exist in the model.

Finally, I use the Wald test (z_1) to check for the joint significance of the reported coefficients in the model. If I reject the null hypothesis that states no relationship between the dependent and explanatory variables, I make sure that the model is jointly significant. I present the results of the Wald test in tables 8.6 and 8.7. They all support the joint significance of the reported coefficients.

8.5 Results

8.5.1 Managerial Ability, Market Competition and Accrual Earnings Management

The results show that increased competition has a positive effect on the relation between managerial ability and accrual earnings management - higher levels of competition drive more able managers to use more discretionary accruals. From Table 8.6, column (1) shows that the coefficient of superior managers who use accrual earnings management in less concentrated markets ($\alpha_1+\beta_1=1.474+0.103=1.577$) is higher than that in more concentrated markets ($\alpha_1=1.474$). Column (2) shows that the same coefficient in industries with higher product substitutability ($\alpha_1+\beta_1=1.507+0.202=1.709$) is more than that in industries with fewer substitutes ($\alpha_1=1.507$). Column (3) shows that the coefficient in firms with less pricing power ($\alpha_1+\beta_1=1.223+0.646=1.869$) is higher than that in other firms ($\alpha_1=1.223$). Finally, the *t* values of the linear restriction test (LRT) in Table 8.6 show that the combined effects of market competition and managerial ability on accrual earnings management are significantly different from zero. These results support hypothesis *H1a* of this chapter. Similar results are obtained by using the Modified Jones' Model (1995) as shown in Appendix C Table I and the OLS estimator as shown in Appendix C Table III.

The individual average effects of managerial ability and market competition on accrual earnings management appear in Appendix C Table V and VII as Table 8.6 presents the conditional effects only. The results in Appendix C Table V confirm that high quality managers manage earnings by exploiting their incomplete contracts with the shareholders. The high costs of writing, negotiating, and renegotiating contracts with high quality managers allow them to exercise more discretion in financial reporting. In this case, they use accrual earnings management to improve the perception about their abilities and earn high imminent rewards without influencing the long-term performance of their firms (Narayanan, 1985; Milgrom and Roberts, 1992; Graham et al., 2005; Cohen et al., 2008; Ronen and Yaari, 2008; Demerjian et al., 2013b). Meanwhile, the results in Appendix C Table VII show that market competition plays a role in changing the behaviour of managers in utilising earnings management as evidenced in the coefficients of its three proxies. Information-driven comparisons

Variables	(1)	(2)	(3)
variables	Acc	rual earnings manager	ment
MorlAhilitvia	1 474***	1 507***	1 223***
HHI _{it}	-0.041***	-	-
ProdSubstitutability:	-	0 069***	_
PricPowerit	-	-	-0.134***
EarningsManagement _{i t-1}	-0.111***	-0.111***	-0.108***
FirmSizeit	0.105***	0.094***	0.089***
SalesVolatility _{i.t-4,t}	0.225***	0.256***	0.230***
CashFlowVolatility _{i.t-4,t}	0.090***	0.084***	0.058***
OperCycle _{i,t-4,t}	0.039***	0.012**	0.020***
Losses _{i,t-4,t}	0.093***	0.055***	0.153***
NationalAuditorit	0.056***	0.071***	0.071***
<i>LitigationInd</i> _{it}	-0.169***	-0.094***	-0.095***
MB _{it}	2.65e-05***	5.88e-05***	5.26e-05***
SalesGrowthrk _{it}	-0.610***	-0.621***	-0.672***
LnNumAnalyst _{it}	0.047***	0.057***	0.089***
IndRev%	-0.390***	-0.484***	-0.504***
Momentum _{it}	0.409***	0.396***	0.331***
Segments _{it}	-0.107***	-0.085***	-0.093***
Foreign _{it}	-0.213***	-0.065***	-0.020
<i>GFC</i> _{it}	-0.272***	-0.270***	-0.564***
Dot-Com _{it}	-0.108***	-0.172***	-0.420***
SOX _{it}	-0.153***	-0.061**	-0.460***
(HHI _{it})MgrlAbility _{it}	0.103***	-	-
(ProdSubstitutability _{it})MgrlAbility _{it}	-	0.202***	-
(PricPower _{it})MgrlAbility _{it}	-	-	0.646***
t	52.426	51.892	49.607
Hansen	1,696.560	1,672.400	1,698.560
m_l	-17.840	-18.040	-17.850
<i>m</i> ₂	-1.200	-0.970	-1.050
Z1	3,417.130	13,936.510	8,792.840

Table 8.6 The moderating role of market competition in the relation between managerial ability and accrual
earnings management (AccrualEM)

Notes: This table presents the results from the system generalized method of moments regressions for the effect of market competition on the relation between managerial ability and accrual earnings management. The sample includes 66,695 firm-year observations obtained from Compustat from 1989 to 2011. Accrual earnings management is measured by Stubben's Model (2010). Managerial ability is measured by the model of Demerjian et al. (2012). Each column represents the estimates from examining a dummy measure of market competition in the following order: (1) market concentration measured by the Herfindahl-Hirschman index (HHI), (2) product substitutability measured by the gross margin in the industry, and (3) pricing power measured by the industry adjusted Lerner Index (LI). HHI is a dummy variable that equals 1 if the firm is in an industry where market concentration is lower than the median and zero otherwise. ProdSubstitutability is a dummy variable that equals 1 if the firm is in an industry where product substitutability is lower than the median and zero otherwise. PricPower is a dummy variable that equals 1 if the firm has lower pricing power than the median and zero otherwise. Variables are defined in detail in the (Variable Definitions) section above. Each coefficient represents the change in accrual earnings management based on a one unit change in the determinant. The t values of the linear restriction tests (LRT) reflect the joint significance of the explanatory and moderating variables. The test is performed under the null hypothesis $H_0: \alpha_1 + \beta_1 = 0$. Hansen test for over-identifying restrictions takes the shape of χ^2 distribution and checks for the orthogonality between the instrumental variables (IV_{it}) and the error term (ε_{it}). The AB test (Arellano and Bond, 1991) examines the serial correlation in the first difference residual (v_{it}) by using the (m_i) statistic and takes the shape of a normal distribution with N(0,1). m_1 values represent the results of the 1st order serial correlation test, while m_2 values represent the results of the 2nd order serial correlation test. The Wald test (z_i) checks for the joint significance of the reported coefficients in the model. The table also shows evidence about the dynamic behaviour of accrual earnings management in the fifth row. Intercept is included, but not tabulated. *, **, *** denotes significance at 0.10, 0.05, and 0.01, respectively.

impose higher expectations on managers to beat rivals' earnings and, therefore, make communications more costly between managers and shareholders. In this setting, managers tend to accept more risk to keep pace with the behaviour of other aggressive competitors. The results imply that competition drives adverse selection and encourages opportunistic managerial behaviour.

The fact that superior managers use accrual earnings management in the face of increased competition as presented in Table 8.6 is telling. Accrual earnings management is less costly to shareholders as it does not affect the underlying business and its future cash flows. Moreover, it is less costly for managers as it will allow short-term targets to be met in the current period i.e. bonuses, but also lowers the risk of missing targets in future periods. The results extend the previous work of Dalia and Park (2009), Markarian and Santalo (2010), Karuna et al. (2012), and Datta et al. (2013) by showing that all managers, including high quality ones, tend to manage more accruals under the pressure of competition.

8.5.2 Managerial Ability, Market Competition and Real Earnings Management

The results show that competition plays a moderating role in the relation between managerial ability and real earnings management. Table 8.7, column (1) shows that the coefficient of superior managers who use real earnings management in less concentrated markets ($\alpha_1+\beta_1=-0.123-0.163=-0.286$) is lower than that in more concentrated markets ($\alpha_1=-0.123$). Column (2) shows that the coefficient in industries with higher product substitutability ($\alpha_1+\beta_1=-0.145-0.308=-0.453$) is lower than that in industries with fewer substitutes ($\alpha_1=-0.145$). Column (3) shows that the coefficient in firms with less pricing power ($\alpha_1+\beta_1=-0.136-0.167=-0.303$) is lower than that in other firms ($\alpha_1=-0.136$). Finally, the *t* values of the linear restriction test (LRT) in Table 8.7 show that the combined effects of market competition and managerial ability on real earnings management are significantly different from zero. Once more, the previous results support hypothesis *H1b* of this chapter. Similar results are obtained by using the Roychowdhury's Model (2006) as shown in Appendix C Table II and the OLS estimator as shown in Appendix C Table IV.

Verichles	(1)	(2)	(3)
variables	R	nt	
MgrlAbility _{it}	-0.123***	-0.145***	-0.136***
HHI _{it}	-0.070***	-	-
<i>ProdSubstitutability</i> _{it}	-	-0.031***	-
PricPowerit	-	-	0.082***
EarningsManagement _{i,t-1}	0.295***	0.290***	0.291***
FirmSizeit	0.064***	0.052***	0.076***
SalesVolatility _{i,t-4,t}	0.302***	0.249***	0.257***
CashFlowVolatility _{i,t-4,t}	0.036***	0.052***	0.039***
OperCycle _{i,t-4,t}	0.011***	0.005**	0.003
Losses _{i,t-4,t}	-0.093***	-0.007	-0.077***
<i>NationalAuditor_{it}</i>	0.187***	0.181***	0.201***
LitigationInd _{it}	0.218***	0.188***	0.175***
MB _{it}	2.97e-05***	1.51e-05***	2.33e-05***
SalesGrowthrk _{it}	0.011**	0.029***	0.038***
LnNumAnalyst _{it}	0.056***	0.056***	0.041***
IndRev%	-0.127***	-0.124***	-0.026*
Momentum _{it}	-0.178***	-0.100***	-0.133***
Segments _{it}	-0.027***	-0.018**	-0.028***
Foreign _{it}	0.177***	0.228***	0.152***
<i>GFC</i> _{it}	-0.179***	-0.169***	-0.493***
Dot-Com _{it}	-0.117***	-0.076***	-0.370***
SOX _{it}	-0.111***	-0.093***	-0.395***
(HHI _{it})MgrlAbility _{it}	-0.163***	-	-
(ProdSubstitutability _{it})MgrlAbility _{it}	-	-0.308***	-
(PricPower _{it})MgrlAbility _{it}	-	-	-0.167***
t	-12.205	-10.987	-3.139
Hansen	1,723.230	1,736.200	1,731.640
m_1	-14.250	-14.220	-14.300
<i>m</i> ₂	1.030	0.990	1.080
Z1	8,414.850	10,021.850	9,691.780

Table 8.7 The moderating role of market competition in the relation between managerial a	bility :	and real
earnings management (RealEM)		

Notes: This table presents the results from the system generalized method of moments regressions for the effect of market competition on the relation between managerial ability and real earnings management. The sample includes 66,695 firm-year observations obtained from Compustat from 1989 to 2011. Real earnings management is measured by Gunny's Model (2010). Managerial ability is measured by the model of Demerjian et al. (2012). Each column represents the estimates from examining a dummy measure of market competition in the following order: (1) market concentration measured by the Herfindahl-Hirschman index (HHI), (2) product substitutability measured by the gross margin in the industry, and (3) pricing power measured by the industry adjusted Lerner Index (LI). HHI is a dummy variable that equals 1 if the firm is in an industry where market concentration is lower than the median and zero otherwise. ProdSubstitutability is a dummy variable that equals 1 if the firm is in an industry where product substitutability is lower than the median and zero otherwise. PricPower is a dummy variable that equals 1 if the firm has lower pricing power than the median and zero otherwise. Variables are defined in detail in the (Variable Definitions) section above. Each coefficient represents the change in real earnings management based on a one unit change in the determinant. The t values of the linear restriction tests (LRT) reflect the joint significance of the explanatory and moderating variables. The test is performed under the null hypothesis $H_0: \alpha_1 + \beta_1 = 0$. Hansen test for over-identifying restrictions takes the shape of χ^2 distribution and checks for the orthogonality between the instrumental variables (IV_{it}) and the error term (ϵ_{it}). The AB test (Arellano and Bond, 1991) examines the serial correlation in the first difference residual (v_{it}) by using the (m_i) statistic and takes the shape of a normal distribution with N(0,1). m_1 values represent the results of the 1st order serial correlation test, while m_2 values represent the results of the 2nd order serial correlation test. The Wald test (z_1) checks for the joint significance of the reported coefficients in the model. The table also displays evidence about the dynamic behaviour of real earnings management in the fifth row. Intercept is included, but not tabulated. *, **, *** denotes significance at 0.10, 0.05, and 0.01, respectively.

The individual average effect of managerial ability on real earnings management appears in Appendix C Table VI as Table 8.7 presents the conditional effects only. Lower levels of monitoring by shareholders allow managers greater flexibility. However, real earnings management activities, while harder to detect in the short-run, are detrimental to the sustainability and performance of the firm as they have long-run economic consequences for the business (Graham et al., 2005; Merkl-Davies and Brennan, 2007; Cohen et al., 2008). Therefore, managers with higher levels of ability do not undertake such activities, as they are detrimental to their long-term compensation and labour market value. The results show how high quality managers link their own interests and shareholder wealth in the long-run. Similar to Karuna et al. (2012), Appendix C Table VIII confirm the role market competition plays in imposing higher pressure on managers to beat rivals' earnings and driving adverse selection.

Given the influence of competition as appears in Table 8.6 and Table 8.7, the results provide evidence that while high quality managers respond to increased market competition by relying on accrual earnings management, they avoid costly real earnings management. In this sense, they balance the short-term and long-term benefits of their earnings management decisions based on their insight and superior competencies. The results extend those of Karuna et al. (2012) by excluding high quality managers from using real earning management as a response to competition pressure.

8.5.3 Comparison of Competition Measures

To compare the explanatory power of the competition measures I calculate their individual elasticity indices (EI). As Tables 8.6 and 8.7 show, all the competition measures have significant coefficients. This test, however, helps to identify which measure is the most restrictive with regards to earnings management. I present the results in Table 8.8, which are calculated according to the following equation.

$$EI_{Competition} = \frac{h_{MgrlAbility} + h_{Competition}}{\sum h}$$
(8.7)

Where, $EI_{Competition}$ is the elasticity index of a specific competition measure, $h_{MgrlAbility}$ is the elasticity of managerial ability variable, $h_{Competition}$ is the elasticity of a specific competition measure, and Σ h is the sum of all explanatory variables' elasticities. I

calculate elasticity (h) as $h_k = b_k \frac{\overline{x}_k}{b\overline{x}}$, where k is the variable of interest, b_k is its coefficient, \overline{x}_k is its mean, and $b\overline{x}$ is the expected value of the dependent variable obtained by multiplying the coefficient of each explanatory variable by its mean.

The results show that product substitutability has the highest elasticity index for its moderating effect on both accrual and real earnings management. Among the three competition measures I cover, product substitutability is the strongest driver of superior managers in manipulating accruals [Highest index EI= 0.577 in Table 8.8] and has the strongest mitigating effect on real earnings management by more able managers [Highest index EI= 0.395 in Table 8.8].

Table 8.8 Elasticity Indices			
Variables	(1)	(2)	(3)
AccrualEM	0.490	0.577	0.426
RealEM	0.215	0.395	0.343

Notes: This table presents the individual elasticity indices (EI) of the competition measures identified in this chapter. The sample includes 66,695 firm-year observations obtained from Compustat from 1989 to 2011. EI is calculated as the sum of the elasticities of the main explanatory variable (managerial ability) and the moderating variable (market competition) divided by the total elasticities of all the variables identified in the model, as explained in the (Comparison of competition measures) section above. EIs assist in comparing the explanatory power of competition measures. Each column represents the estimates from examining one measure of market competition in the following order: (1) market concentration measured by the Herfindahl-Hirschman index (*HHI*), (2) product substitutability measured by the gross margin in the industry, and (3) pricing power measured by the industry adjusted Lerner Index (*LI*).

8.6 Additional Analysis

8.6.1 Comparing the Effect of Managerial Ability on Earnings Management between High and Low Competition Settings

Although the GMM estimator used earlier in the methodology section solves for the problem of heterogeneity, in this section I control for the correlation between market competition and managerial ability that may influence the results. As more competitive markets are more likely to attract high quality managers and vice versa, I divide the sample into a subsample of high competition and another of low competition. Based on the first proxy of market competition, observations with "HHI" lower than the median

are considered in the high competition subsample, and in the low competition subsample otherwise. Under the second proxy of market competition, observations with "ProdSubstitutability" lower than the median are considered in the high competition subsample, and in the low competition subsample otherwise. Under the third proxy of market competition, observations with "PricPower" lower than the median are considered in the high competition subsample, and in the low competition subsample otherwise.

To generate two comparable subsamples under each of the previous proxies, I exclude the extreme values of managerial ability that exist under the normal distribution of high or low levels of competition by trimming at 1%. To make sure that the subsamples are comparable, I perform the t-test for the differences between means and confirm that the results are not significant. When combining the high and low competition subsamples, the sample size for each of the previous three proxies of competition include 57,040, 56,886, and 59,516 firm-year observations respectively.

Finally, I perform the system generalized method of moments regressions for the effect of managerial ability on accrual earnings management and real earnings management separately in each subsample following this model:

$$\begin{aligned} EarningsManagement_{i} = & \alpha_{0} + \alpha_{1}MgrlAbility_{it} + \alpha_{2}EarningsManagement_{i,t-1} \\ & + \alpha_{3}FirmSize_{it} + \alpha_{4}SalesVolatility_{i,t-4,t} \\ & + \alpha_{5}CashFlowVolatility_{i,t-4,t} + \alpha_{6}OperCycle_{i,t-4,t} \\ & + \alpha_{7}Losses_{i,t-4,t} + \alpha_{8}NationalAuditor_{it} \\ & + \alpha_{9}SalesGrowth_{rk_{it}} + \alpha_{10}MB_{it} + \alpha_{11}LitigationInd_{it} \\ & + \alpha_{12}LnNumAnalyst_{it} + \alpha_{13}IndRev\% + \alpha_{14}Momentum_{it} \\ & + \alpha_{15}Segments_{it} + \alpha_{16}Foreign_{it} + \alpha_{17}GFC_{it} \\ & + \alpha_{18}Dot-Com_{it} + \alpha_{19}SOX_{it} + \eta_{i} + d_{t} + v_{it} \end{aligned}$$

$$(8.8)$$

The results appear in tables 8.9 and 8.10 and confirm the previous findings. Superior managers use more accrual earnings management in more competitive markets as the coefficients of managerial ability variable are higher than those under less competition in table 8.9. The results also show that while high quality managers avoid real earnings management under high levels of competition, they tend to use more of this costly activity in less competitive markets as appears in the coefficients of managerial ability variable in table 8.10. The results again support the two hypotheses of this chapter.

competition						
	(1)		(2)		(3)	
Variables	High	Low	High	Low	High	Low
variables	Competition	Competition	Competition	Competition	Competition	Competition
<i>MgrlAbility</i> _{it}	0.870***	0.660***	0.714***	0.654***	0.868***	0.565***
AccrualEM _{i,t-1}	-0.170***	-0.142***	-0.140***	-0.162***	-0.191***	-0.094***
<i>FirmSize</i> _{it}	0.098***	0.086***	0.068***	0.116***	0.182***	-0.048***
SalesVolatility _{i,t-4,t}	0.157***	0.282***	0.161***	0.009***	0.112***	0.103***
CFVolatility _{i,t-4,t}	0.096***	0.121***	0.004***	1.522***	1.338***	0.037***
OperCycle _{i,t-4,t}	0.035***	0.140***	0.086***	0.008***	0.021***	0.108***
Losses _{i,t-4,t}	-0.055***	-0.032***	0.001	-0.113***	0.359***	-0.116***
NationalAuditor _{it}	-0.088***	0.167***	-0.005**	0.114^{***}	0.248***	-0.254***
LitigationInd _{it}	0.071***	-0.115***	0.113***	-0.159***	-0.014	0.107***
MB_{it}	0.000***	0.001^{***}	-0.000***	0.000***	-0.000***	-0.002***
SalesGrowthrk _{it}	-0.606***	-0.555***	-0.592***	-0.639***	-0.615***	-0.642***
LnNumAnalyst _{it}	0.094***	0.084^{***}	0.080***	0.048***	-0.002**	0.160***
IndRev%	-1.557***	-0.524***	-0.545***	-0.986***	0.170***	-0.624***
<i>Momentum_{it}</i>	0.186***	0.445***	0.437***	0.108^{***}	0.496***	0.195***
Segmentsit	-0.099***	-0.084***	-0.094***	-0.183***	-0.136***	0.075***
<i>Foreign</i> _{it}	-0.217***	-0.195***	-0.155***	0.045***	-0.011**	-0.192***
<i>GFC</i> _{it}	0.052***	0.177***	0.039***	-0.488***	-0.377***	0.042***
Dot-Com _{it}	0.190***	0.166***	-0.070***	-0.460***	-0.141***	-0.103***
SOX _{it}	0.103***	0.133***	0.017***	-0.406***	-0.266***	-0.021***
Observations	27,538	29,502	28,959	27,927	29,904	29,612
Hansen	1,145.730	1,005.340	958.550	1,099.750	1,026.620	1,118.230
m_l	-12.570	-11.880	-11.960	-11.760	-10.120	-14.200
m_2	-1.690	-0.910	0.210	-1.630	-1.230	-1.140
Z.1	2.2e+06	65,008.940	2.1e+05	3.5e+05	1.1e+06	1.1e+06

Table 8.9 The effect of managerial ability on accrual earnings management (AccrualEM) conditional on competition

Notes: This table presents the results from the system generalized method of moments regressions for the effect of managerial ability on accrual earnings management based on the level of different proxies of market competition in the following order: (1) market concentration measured by the Herfindahl-Hirschman index (HHI), (2) product substitutability measured by the gross margin in the industry, and (3) pricing power measured by the industry adjusted Lerner Index (LI). Under each of the previous three proxies of competition, I divide the sample into a subsample of high competition and another of low competition. Based on the first proxy of market competition, observations with HHI lower than the median are considered in the high competition subsample, and in the low competition subsample otherwise. Under the second proxy of market competition, observations with ProdSubstitutability lower than the median are considered in the high competition subsample, and in the low competition subsample otherwise. Under the third proxy of market competition, observations with PricPower lower than the median are considered in the high competition subsample, and in the low competition subsample otherwise. I trim each subsample at 1% level to exclude the extreme values of managerial ability that might exist under high or low levels of competition. When combining the high and low competition subsamples, the sample size for each of the previous three proxies of competition include 57,040, 56,886, and 59,516 firm-year observations respectively, obtained from Compustat from 1989 to 2011. To make sure that the subsamples are comparable, I perform the t-test for the differences between means and make sure that the results are not significant. Accrual earnings management is measured by Stubben's Model (2010). Managerial ability is measured by the model of Demerjian et al. (2012). Variables are defined in detail in the (Variable Definitions) section above. Each coefficient represents the change in accrual earnings management based on a one unit change in the determinant. Hansen test for over-identifying restrictions takes the shape of χ^2 distribution and checks for the orthogonality between the instrumental variables (IV_{it}) and the error term (ε_{it}). The AB test (Arellano and Bond, 1991) examines the serial correlation in the first difference residual (vit) by using the (mi) statistic and takes the shape of a normal distribution with N(0,1). m1 values represent the results of the 1st order serial correlation test, while m₂ values represent the results of the 2nd order serial correlation test. The Wald test (z_1) checks for the joint significance of the reported coefficients in the model. The table also displays evidence about the dynamic behaviour of accrual earnings management in the second row. Intercept is included, but not tabulated. *, **, *** denotes significance at 0.10, 0.05, and 0.01, respectively.

	((1) (2)		2)	(3)	
Variables	High	Low	High	Low	High	Low
variables	Competition	Competition	Competition	Competition	Competition	Competition
MgrlAbility _{it}	-0.226***	0.048***	-0.263***	0.123***	-0.312***	0.078***
RealEM _{i,t-1}	0.364***	0.160***	0.296***	0.239***	0.304***	0.256***
<i>FirmSize</i> _{it}	0.045***	0.058***	0.058***	-0.009***	0.109***	0.009***
SalesVolatility _{i,t-4,t}	0.124***	0.134***	0.372***	0.079***	0.203***	0.133***
CFVolatility _{i,t-4,t}	0.055***	0.223***	0.074***	0.066***	0.290***	0.033***
OperCycle _{i,t-4,t}	-0.005***	0.088^{***}	-0.030***	0.184***	-0.006***	0.074***
Losses _{i,t-4,t}	-0.182***	-0.062***	-0.051***	-0.115***	-0.091***	-0.096***
NationalAuditor _{it}	0.117***	0.010***	0.224***	0.019***	0.138***	0.072***
LitigationInd _{it}	0.096***	0.239***	0.031***	0.265***	0.041**	0.172***
MB _{it}	0.000***	0.000***	-0.000***	0.000***	0.000***	0.000***
SalesGrowthrk _{it}	0.049***	0.041***	0.071***	0.083***	0.009***	0.118***
LnNumAnalyst _{it}	0.049***	0.027***	0.012***	0.142***	-0.095***	0.074***
IndRev%	-0.431***	-0.026***	0.132***	0.139***	-0.526***	0.193***
<i>Momentum_{it}</i>	-0.046***	0.024***	0.007***	-0.074***	-0.219***	-0.003***
Segments _{it}	-0.010***	0.000	-0.021***	0.021***	0.197***	-0.092***
Foreignit	0.194***	-0.049***	0.103***	0.071***	0.098***	0.031***
GFC_{it}	-0.011***	-0.084***	0.023***	-0.010***	-0.168***	0.053***
Dot-Com _{it}	0.026***	0.007***	0.082***	-0.075***	0.006***	-0.005***
SOX _{it}	0.134***	0.025***	0.205***	-0.005***	0.095***	0.049***
Observations	27,538	29,502	28,959	27,927	29,904	29,612
Hansen	1,168.640	1,027.720	1,120.570	976.190	1,074.200	1,168.160
m_l	-12.300	-9.940	-11.570	-10.670	-10.180	-12.210
m_2	1.150	1.540	1.590	0.040	1.220	0.720
<i>Z1</i>	2.9e+06	1.1e+05	1.4e+06	4.8e+05	5.3e+05	2.4e+06

Table 8.10 The effect of managerial ability on real earnings management (RealEM) conditional on competition

Notes: This table presents the results from the system generalized method of moments regressions for the effect of managerial ability on real earnings management based on the level of different proxies of market competition in the following order: (1) market concentration measured by the Herfindahl-Hirschman index (HHI), (2) product substitutability measured by the gross margin in the industry, and (3) pricing power measured by the industry adjusted Lerner Index (LI). Under each of the previous three proxies of competition, I divide the sample into a subsample of high competition and another of low competition. Based on the first proxy of market competition, observations with HHI lower than the median are considered in the high competition subsample, and in the low competition subsample otherwise. Under the second proxy of market competition, observations with ProdSubstitutability lower than the median are considered in the high competition subsample, and in the low competition subsample otherwise. Under the third proxy of market competition, observations with PricPower lower than the median are considered in the high competition subsample, and in the low competition subsample otherwise. I trim each subsample at 1% level to exclude the extreme values of managerial ability that might exist under high or low levels of competition. When combining the high and low competition subsamples, the sample size for each of the previous three proxies of competition include 57,040, 56,886, and 59,516 firm-year observations respectively, obtained from Compustat from 1989 to 2011. To make sure that the subsamples are comparable, I perform the t-test for the differences between means and make sure that the results are not significant. Real earnings management is measured by Gunny's Model (2010). Managerial ability is measured by the model of Demerjian et al. (2012). Variables are defined in detail in the (Variable Definitions) section above. Each coefficient represents the change in real earnings management based on a one unit change in the determinant. Hansen test for over-identifying restrictions takes the shape of χ^2 distribution and checks for the orthogonality between the instrumental variables (IV_{it}) and the error term (ɛit). The AB test (Arellano and Bond, 1991) examines the serial correlation in the first difference residual (v_{il}) by using the (m_i) statistic and takes the shape of a normal distribution with N(0,1). m_1 values represent the results of the 1st order serial correlation test, while m₂ values represent the results of the 2nd order serial correlation test. The Wald test (z_1) checks for the joint significance of the reported coefficients in the model. The table also displays evidence about the dynamic behaviour of real earnings management in the second row. Intercept is included, but not tabulated. *, **, *** denotes significance at 0.10, 0.05, and 0.01, respectively.

8.6.2 Earnings Smoothing, Managerial Ability, and Market Competition

Earnings smoothing is another way that management may seek to improve the earnings of the firm. I therefore examine the effect of market competition on earnings smoothed by high quality managers. Earnings smoothing may improve the informativeness of earnings when managers try to reduce current high earnings numbers under the expectation of a decrease in future earnings, and vice versa. In this sense, earnings smoothing makes future earnings more predictable to shareholders. Nevertheless, managers may opportunistically smooth earnings in order to hide bad news in the current period, the result of which will be worse earnings numbers in the future when performance does not improve (Dalia and Park, 2009).

Based on Hunt et al. (1997) and Pincus and Rajgopal (2002), I calculate the earnings smoothing ratio as the standard deviation of pre-managed earnings to the standard deviation of reported earnings over at least three of the last five years. Pre-managed earnings are calculated as non-discretionary accruals from the Modified Jones Model (1995) added to the cash flows from operating activities minus extra-ordinary items and discontinued operations.

I start by examining the effect of managerial ability on earnings smoothing under low market competition and find that managerial ability has a significant negative effect on earnings smoothing. The results appear in the first row of Table 8.11 and show managers with higher levels of ability avoid smoothing earnings when competition is low in the market. This is consistent with the lack of earnings management motive in the absence of competition especially that managers with higher abilities are aware of the reputational costs they would incur if the decline in earnings does not reverse in subsequent periods.

Variables	(1)	(2)	(3)		
Variables	Earnings smoothing				
MgrlAbility _{it}	-0. 138***	-0.265***	-0.175***		
HHI _{it}	0.057***	-	-		
<i>ProdSubstitutability</i> _{it}	-	0.566***	-		
PricPowerit	-	-	0.000***		
Smoothing _{i,t-5}	0.389***	0.391***	0.393***		
<i>FirmSize</i> _{it}	-0.072***	-0.083***	-0.061***		
$SalesVolatility_{i,t-4,t}$	-0.762***	-0.832***	-0.831***		
CashFlowVolatility _{i,t-4,t}	1.862***	1.924***	1.948***		
<i>OperCycle</i> _{i,t-4,t}	-0.122***	-0.120***	-0.108***		
Losses _{i,t-4,t}	-1.223***	-1.266***	-1.233***		
<i>NationalAuditor_{it}</i>	-0.133***	-0.090***	-0.111***		
LitigationInd _{it}	0.100***	0.073***	-0.015***		
MB _{it}	6.16e-05***	5.48e-05***	2.93e-05***		
<i>SalesGrowthrk</i> _{it}	0.090***	0.090***	0.102***		
LnNumAnalyst _{it}	-0.210***	-0.210***	-0.208***		
IndRev%	1.413***	1.550***	1.261***		
<i>Momentum_{it}</i>	-0.100***	-0.114***	-0.136***		
Segments _{it}	-0.063***	-0.053***	-0.018***		
Foreign _{it}	-0.187***	-0.218***	-0.300***		
<i>GFC</i> _{it}	2.613***	0.234***	0.506***		
Dot-Com _{it}	2.414***	0.000	0.047***		
SOX _{it}	2.470***	0.114***	0.047***		
(HHI _{it})MgrlAbility _{it}	0.213***	-	-		
(ProdSubstitutability _{it})MgrlAbility _{it}	-	0.434***	-		
(PricPower _{it})MgrlAbility _{it}	-	-	0.380***		
t	85.185	111.261	111.638		
Hansen	1,705.400	1,701.510	1,716.710		
m_l	-4.200	-4.190	-4.200		
<i>m</i> ₂	0.210	0.200	0.230		
<i>Z1</i>	4.0e+07	5.9e+06	6.2e+06		

Table 8.11 The moderating role of market competition in the relation between managerial ability and earnings smoothing

Notes: This table presents the results from the system generalized method of moments regressions for the effect of market competition on the relation between managerial ability and earnings smoothing. The sample includes 66,695 firm-year observations obtained from Compustat from 1989 to 2011. Earnings smoothing is calculated as the standard deviation of pre-managed earnings to the standard deviation of reported earnings over at least three of the last five years, as described in the (Additional analysis) section above. Managerial ability is measured by the model of Demerjian et al. (2012). Each column represents the estimates from examining a dummy measure of market competition in the following order: (1) market concentration measured by the Herfindahl-Hirschman index (HHI), (2) product substitutability measured by the gross margin in the industry, and (3) pricing power measured by the industry adjusted Lerner Index (LI). HHI is a dummy variable that equals 1 if the firm is in an industry where market concentration is lower than the median and zero otherwise. ProdSubstitutability is a dummy variable that equals 1 if the firm is in an industry where product substitutability is lower than the median and zero otherwise. PricPower is a dummy variable that equals 1 if the firm has lower pricing power than the median and zero otherwise. Variables are defined in detail in the (Variable Definitions) section above. Each coefficient represents the change in earnings smoothing ratio based on a one unit change in the determinant. The t values of the linear restriction tests (LRT) reflect the joint significance of the explanatory and moderating variables. The test is performed under the null hypothesis $H_0: \propto_1 + \beta_1 = 0$. Hansen test for over-identifying restrictions takes the shape of χ^2 distribution and checks for the orthogonality between the instrumental variables (IV_{it}) and the error term (ε_{it}). The AB test (Arellano and Bond, 1991) examines the serial correlation in the first difference residual (v_{it}) by using the (m_i) statistic and takes the shape of a normal distribution with N(0,1). m_1 values represent the results of the 1st order serial correlation test, while m_2 values represent the results of the 2nd order serial correlation test. The Wald test (z_1) checks for the joint significance of the reported coefficients in the model. The table also displays evidence about the dynamic behaviour of earnings smoothing in the fifth row. Intercept is included, but not tabulated. *, **, *** denotes significance at 0.10, 0.05, and 0.01, respectively.

Next, I incorporate the effect of high market competition into the relationship between managerial ability and earnings smoothing. The results here show a positive moderating role of market competition on the relation between managerial ability and earnings smoothing. Table 8.11, column (1) shows that the coefficient of superior managers who smooth earnings in less concentrated markets ($\alpha_1+\beta_1=-0.138+0.213=0.075$) is higher than that in more concentrated markets (\propto_1 =-0.138). Column (2) shows that the same with higher product substitutability ($\alpha_1 + \beta_1 =$ coefficient in industries 0.265+0.434=0.169) is more than that in industries with fewer substitutes (\propto_1 =-0.265). Column (3) shows that the coefficient in firms with less pricing power ($\alpha_1 + \beta_1 = -$ 0.175+0.380=0.205) is higher than that in other firms (α_1 =-0.175). Finally, the t values of the linear restriction tests (LRT) in Table 8.11 show that the combined effects of market competition and managerial ability on earnings smoothing are overall significantly different from zero. These results confirm the major finding of this chapter that market competition changes the behaviour of high quality managers when undertaking earnings management by driving earnings smoothing in this case.

8.6.3 Additional Control Variables

As corporate governance influences earnings management (Duh et al., 2009; Feng et al., 2009; Jaggi et al., 2009; Prencipe and Bar-Yosef, 2011; Demerjian et al., 2013a), I perform the previous regressions once more using additional controls for corporate governance. I add three control variables which I obtain form Execuocomp database.⁴⁹ I use a dummy for executives who serve as directors (*ExecDir*) which equals 1 if the executive served as director during the year. I also use the executive tenure (*Tenure*) calculated as the log of the executive tenure measured in days. Finally, I add the number of board meeting (*BoardMtgs*) held during the year (Adut et al., 2013). The previous variables, however, are only available from 1992 to 2006 and, therefore, I cover only this period in this additional test. As a result, I end up with 42,183 observations. The results emphasize the previous findings related to the moderating effect of market

⁴⁹ Data generously provided by Francesco Vallascas and Paula Castro.
competition on earnings management behaviour of superior managers. The results, however, are not tabulated in this thesis.

8.7 Conclusions

In this chapter, I examine the role of market competition in moderating the relation between managerial ability and earnings management. I build the core question of this chapter based on the expectation that the communication of information in more competitive markets contributes to further discipline the behaviour of high quality managers. Firstly, I find that superior managers use more of the less costly accrual earnings management and less of the more costly real earnings management in facing competition. The results provide another evidence on the substitutability of earnings management activities.

Secondly, although market competition can individually drive both accrual and real earnings management activities, it contributes to mitigating real earnings management behaviour of high quality managers. In other words, high quality managers use their superior competencies to manage earnings more wisely under the pressure of competition. Therefore, high managerial skills and strong market competition may serve together as alternative forms of governance mechanisms by preventing the firms from costly activities in the future.

Finally, this chapter emphasizes the importance of understanding the different motives of earnings management at a time to get a better vision of management decisions in reality. I particularly provide evidence on how industry level factors influence management decisions at the firm level and thus show the interaction between the external and contracting motives of earnings management. For future research, further factors might be selected at different levels to examine how they interact in determining earnings management behaviour.

Appendix C

Appendix C Table I The moderating role of market competition in the relation between managerial	ability	and
accrual earnings management (AccrualEM)		

Variables	(1)	(2)	(3)	
variables	Accrual earnings management			
MarlAhility	1 233***	1 072***	1 005***	
HHI:	-0.030***	-	-	
ProdSubstitutability	-	0 039***	-	
PricPower:	-	-	-0 193***	
FarningsManagement _{i t-1}	0.035***	0.036***	0.036***	
FirmSize _{it}	0.026***	0.011***	0.010***	
SalesVolatility; 1-4 t	0.056***	-0.010	0.057***	
CashFlowVolatility _{it-4.t}	0.097***	0.050***	0.032***	
OperCycle _{i,t-4,t}	0.001	0.026***	-0.008	
Losses _{i.t-4,t}	-0.100***	-0.209***	-0.097***	
NationalAuditorit	0.152***	0.170***	0.173***	
<i>LitigationInd</i> _{it}	0.069***	-0.010	-0.000	
MB _{it}	0.000***	0.000***	0.000***	
SalesGrowthrk _{it}	-0.242***	-0.256***	-0.297***	
LnNumAnalyst _{it}	-0.043***	-0.022***	-0.017***	
IndRev%	-0.163***	-0.210***	-0.285***	
Momentum _{it}	0.349***	0.299***	0.274***	
Segments _{it}	-0.077***	-0.076***	-0.051***	
Foreign _{it}	-0.014	0.083***	-0.008	
<i>GFC</i> _{it}	-0.120***	0.082***	0.060***	
Dot-Com _{it}	0.011	0.062***	0.196***	
SOX _{it}	-0.087**	0.190***	0.204***	
(HHI _{it})MgrlAbility _{it}	0.127***	-	-	
(ProdSubstitutability _{it})MgrlAbility _{it}	-	0.855***	-	
(PricPower _{it})MgrlAbility _{it}	-	-	0.457***	
t	50.428	39.908	37.839	
Hansen	1,692.500	1,687.180	1,715.820	
m_1	-20.850	-20.830	-20.920	
<i>m</i> ₂	-0.680	-0.580	-0.540	
Z1	1,967.600	3,354.900	4,930.200	

Notes: This table presents the results from the system generalized method of moments regressions for the effect of market competition on the relation between managerial ability and accrual earnings management. The sample includes 66,695 firm-year observations obtained from Compustat from 1989 to 2011. Accrual earnings management is measured by Modified Jones' Model (1995). Managerial ability is measured by the model of Demerjian et al. (2012). Each column represents the estimates from examining a dummy measure of market competition in the following order: (1) market concentration measured by the Herfindahl-Hirschman index (HHI), (2) product substitutability measured by the gross margin in the industry, and (3) pricing power measured by the industry adjusted Lerner Index (LI). HHI is a dummy variable that equals 1 if the firm is in an industry where market concentration is lower than the median and zero otherwise. ProdSubstitutability is a dummy variable that equals 1 if the firm is in an industry where product substitutability is lower than the median and zero otherwise. PricPower is a dummy variable that equals 1 if the firm has lower pricing power than the median and zero otherwise. Variables are defined in detail in the (Variable Definitions) section above. The t values of the linear restriction tests (LRT) reflect the joint significance of the explanatory and moderating variables. The test is performed under the null hypothesis $H_0: \alpha_1 + \beta_1 = 0$. Hansen test for over-identifying restrictions takes the shape of χ^2 distribution and checks for the orthogonality between the instrumental variables (IV_{it}) and the error term (ε_{it}). The AB test (Arellano and Bond, 1991) examines the serial correlation in the first difference residual (v_{it}) by using the (m_i) statistic and takes the shape of a normal distribution with N(0,1). m₁ values represent the results of the 1st order serial correlation test, while m₂ values represent the results of the 2nd order serial correlation test. The Wald test (z_1) checks for the joint significance of the reported coefficients in the model. The table also displays evidence about the dynamic behaviour of accrual earnings management in the fifth row. Intercept is included, but not tabulated. *, **, *** denotes significance at 0.10, 0.05, and 0.01, respectively.

Variables	(1)	(2)	(3)
variables	R	eal earnings managemei	nt
MgrlAbility _{it}	-0.508***	-0.516***	-0.468***
HHI _{it}	-0.053***	-	-
<i>ProdSubstitutability</i> _{it}	-	-0.007***	-
PricPowerit	-	-	-0.027***
EarningsManagement _{i,t-1}	0.567***	0.566***	0.568***
FirmSizeit	0.178***	0.174***	0.184***
SalesVolatility _{i,t-4,t}	0.185***	0.229***	0.167***
CashFlowVolatility _{i,t-4,t}	0.104***	0.119***	0.101***
<i>OperCycle</i> _{<i>i</i>,<i>t</i>-4,<i>t</i>}	0.046***	0.061***	0.059***
Losses _{i,t-4,t}	0.168***	0.148***	0.177***
NationalAuditor _{it}	0.079***	0.073***	0.073***
<i>LitigationInd_{it}</i>	-0.143***	-0.070***	-0.143***
MB _{it}	0.000***	0.000***	0.000***
<i>SalesGrowthrk</i> _{it}	0.195***	0.197***	0.190***
LnNumAnalyst _{it}	0.065***	0.056***	0.056***
IndRev%	-0.924***	-0.923***	-0.943***
Momentum _{it}	0.275***	0.283***	0.294***
Segments _{it}	-0.084***	-0.068***	-0.082***
Foreign _{it}	0.029**	0.039***	0.110***
GFC _{it}	0.044***	-1.646***	-0.293***
Dot-Com _{it}	0.170***	-1.604***	-0.230***
SOX _{it}	0.157***	-1.549***	-0.189***
(HHI _{it})MgrlAbility _{it}	-0.006	-	-
(ProdSubstitutability _{it})MgrlAbility _{it}	-	-0.089***	-
(PricPower _{it})MgrlAbility _{it}	-	-	-0.125***
t	-49.469	-36.473	-42.602
Hansen	1,731.660	1,755.180	1,727.840
m_l	-10.360	-10.350	-10.270
m_2	0.690	0.850	0.800
Ζ1	22,770.440	33,300.670	14,569.600

Appendix C Table II The moderating role of market competition in the relation between managerial ability and real earnings management (RealEM)

Notes: This table presents the results from the system generalized method of moments regressions for the effect of market competition on the relation between managerial ability and real earnings management. The sample includes 66,695 firm-year observations obtained from Compustat from 1989 to 2011. Real earnings management is measured by Roychowdhury's Model (2006). Managerial ability is measured by the model of Demerjian et al. (2012). Each column represents the estimates from examining a dummy measure of market competition in the following order: (1) market concentration measured by the Herfindahl-Hirschman index (HHI), (2) product substitutability measured by the gross margin in the industry, and (3) pricing power measured by the industry adjusted Lerner Index (LI). HHI is a dummy variable that equals 1 if the firm is in an industry where market concentration is lower than the median and zero otherwise. ProdSubstitutability is a dummy variable that equals 1 if the firm is in an industry where product substitutability is lower than the median and zero otherwise. PricPower is a dummy variable that equals 1 if the firm has lower pricing power than the median and zero otherwise. Variables are defined in detail in the (Variable Definitions) section above. Each coefficient represents the change in real earnings management based on a one unit change in the determinant. The t values of the linear restriction tests (LRT) reflect the joint significance of the explanatory and moderating variables. The test is performed under the null hypothesis $H_0: \alpha_1 + \beta_1 = 0$. Hansen test for over-identifying restrictions takes the shape of χ^2 distribution and checks for the orthogonality between the instrumental variables (IVit) and the error term (Eit). The AB test (Arellano and Bond, 1991) examines the serial correlation in the first difference residual (v_{i}) by using the (m_i) statistic and takes the shape of a normal distribution with N(0,1). m₁ values represent the results of the 1st order serial correlation test, while m₂ values represent the results of the 2nd order serial correlation test. The Wald test (z_1) checks for the joint significance of the reported coefficients in the model. The table also displays evidence about the dynamic behaviour of real earnings management in the fifth row. Intercept is included, but not tabulated. *, **, *** denotes significance at 0.10, 0.05, and 0.01, respectively.

Vorichlas	(1)	(2)	(3)
variables	Acc	rual earnings manager	nent
	0.400++++		
MgrlAbility _{it}	0.498***	0.38/***	0.236***
HHI _{it}	-0.025	-	-
<i>ProdSubstitutability</i> _{it}	-	0.044**	-
PricPower _{it}	-	-	-0.131***
EarningsManagement _{i,t-1}	-0.091***	-0.090***	-0.089***
FirmSize _{it}	0.004	0.005	-0.001
SalesVolatility _{i,t-4,t}	-0.069	-0.081	-0.064
CashFlowVolatility _{i,t-4,t}	0.091	0.093	0.084
OperCycle _{i,t-4,t}	0.019	0.019	0.024*
Losses _{i,t-4,t}	-0.016	-0.007	0.070*
<i>NationalAuditor</i> _{it}	0.041	0.043	0.041
LitigationInd _{it}	0.028	0.004	0.009
MB _{it}	2.13e-05	3.83e-05	2.42e-05
<i>SalesGrowthrk</i> _{it}	-0.502***	-0.503***	-0.537***
LnNumAnalyst _{it}	0.067***	0.069***	0.075***
IndRev%	-0.196**	-0.190**	-0.208**
Momentum _{it}	0.178***	0.183***	0.167***
Segmentsit	0.006	0.000	0.003
Foreign _{it}	0.026	0.025	0.024
GFC _{it}	-0.017	-0.078	-0.074
Dot-Com _{it}	-0.056	-0.053	-0.051
SOX _{it}	-0.003	-0.079	-0.084
(HHI _{it})MgrlAbility _{it}	0.078	-	-
(ProdSubstitutabilityit)MgrlAbilityit	-	0.222*	-
(PricPower _{it})MgrlAbility _{it}	-	-	0.469***
t	4.582	9.788	17.687
R^2	0.131	0.135	0.149
<i>F-statistic</i>	11.060	11.160	12.280
Prob (F-statistic)	0.000	0.000	0.000

Appendix C Table III The moderating role of market	competition in the	relation	between	managerial ab	ility
and accrual earnings management (AccrualEM)					

Notes: This table presents the results from the ordinary least square regressions for the effect of market competition on the relation between managerial ability and accrual earnings management. The sample includes 66,695 firm-year observations obtained from Compustat from 1989 to 2011. Accrual earnings management is measured by Stubben's Model (2010). Managerial ability is measured by the model of Demerjian et al. (2012). Each column represents the estimates from examining a dummy measure of market competition in the following order: (1) market concentration measured by the Herfindahl-Hirschman index (*HHI*), (2) product substitutability measured by the gross margin in the industry, and (3) pricing power measured by the industry adjusted Lerner Index (*LI*). HHI is a dummy variable that equals 1 if the firm is in an industry where market concentration is lower than the median and zero otherwise. ProdSubstitutability is a dummy variable that equals 1 if the firm is in an industry where product substitutability is lower than the median and zero otherwise. PricPower is a dummy variable that equals 1 if the firm has lower pricing power than the median and zero otherwise. Variables are defined in detail in the (Variable Definitions) section above. Each coefficient represents the change in accrual earnings management based on a one unit change in the determinant. The t values of the linear restriction tests (LRT) reflect the joint significance of the explanatory and moderating variables. The test is performed under the null hypothesis $H_0:\alpha_1+\beta_1=0$. Intercept is included, but not tabulated. *, **, *** denotes significance at 0.10, 0.05, and 0.01, respectively.

Variables	(1)	(2)	(3)	
variables	Real earnings management			
MgrlAbility _{it}	-0.310*	-0.305***	-0.172**	
HHI _{it}	-0.017	-	-	
<i>ProdSubstitutability</i> _{it}	-	0.000	-	
PricPowerit	-	-	0.051***	
EarningsManagement _{i,t-1}	0.536***	0.537***	0.536***	
FirmSizeit	0.010*	0.010*	0.012**	
$SalesVolatility_{i,t-4,t}$	0.107**	0.107**	0.107**	
CashFlowVolatility _{i,t-4,t}	-0.082	-0.084	-0.084	
OperCycle _{i,t-4,t}	0.011	0.013	0.011	
Losses _{i,t-4,t}	-0.042	-0.042	-0.080**	
<i>NationalAuditor</i> _{it}	0.079***	0.080***	0.081***	
<i>LitigationInd_{it}</i>	0.097***	0.096***	0.104***	
MB _{it}	3.00e-05	3.07e-05	2.92e-05	
SalesGrowthrkit	0.025	0.027	0.043	
LnNumAnalyst _{it}	0.016	0.014	0.010	
IndRev%	-0.001	0.014	0.034	
Momentum _{it}	-0.039	-0.040	-0.036	
Segments _{it}	-0.003	-0.004	-0.003	
Foreign _{it}	0.022	0.020	0.020	
GFC _{it}	0.102	0.076	0.069	
Dot-Com _{it}	0.063	0.035	0.033	
SOX _{it}	0.110*	0.076	0.075	
(HHI _{it})MgrlAbility _{it}	-0.075*	-	-	
(ProdSubstitutability _{it})MgrlAbility _{it}	-	-0.106	-	
(PricPower _{it})MgrlAbility _{it}	-	-	-0.208*	
t	-4.914	-3.653	-5.232	
R^2	0.312	0.311	0.313	
<i>F-statistic</i>	140.250	139.890	140.310	
Prob (F-statistic)	0.000	0.000	0.000	

Appendix C Table IV The moderating role of market competition in the relation between managerial ability
and real earnings management (RealEM)

Notes: This table presents the results from the ordinary least square regressions for the effect of market competition on the relation between managerial ability and real earnings management. The sample includes 66,695 firm-year observations obtained from Compustat from 1989 to 2011. Real earnings management is measured by Gunny's Model (2010). Managerial ability is measured by the model of Demerjian et al. (2012). Each column represents the estimates from examining a dummy measure of market competition in the following order: (1) market concentration measured by the Herfindahl-Hirschman index (*HHI*), (2) product substitutability measured by the gross margin in the industry, and (3) pricing power measured by the industry adjusted Lerner Index (*LI*). HHI is a dummy variable that equals 1 if the firm is in an industry where market concentration is lower than the median and zero otherwise. ProdSubstitutability is a dummy variable that equals 1 if the firm is in an industry where product substitutability is lower than the median and zero otherwise. PricPower is a dummy variable that equals 1 if the firm has lower pricing power than the median and zero otherwise. Variables are defined in detail in the (Variable Definitions) section above. Each coefficient represents the change in real earnings management based on a one unit change in the determinant. The t values of the linear restriction tests (LRT) reflect the joint significance of the explanatory and moderating variables. The test is performed under the null hypothesis $H_0: \propto_1+\beta_1=0$. Intercept is included, but not tabulated. *, **, **** denotes significance at 0.10, 0.05, and 0.01, respectively.

Variables	Accrual earnings management
14 141 ⁻¹ -	0.072***
MgrlAbilityit	0.9/3***
EarningsManagement _{i,t-1}	-0.109***
FirmSizeit	0.128***
$SalesVolatility_{i,t-4,t}$	0.323***
CashFlowVolatility _{i,t-4,t}	0.033***
<i>OperCycle</i> _{<i>i</i>,<i>t</i>-4,<i>t</i>}	0.013
Losses _{i,t-4,t}	0.024
<i>NationalAuditor</i> _{it}	0.082***
<i>LitigationInd</i> _{it}	-0.176***
MB _{it}	8.65e-05***
<i>SalesGrowthrk</i> _{it}	-0.688***
LnNumAnalyst _{it}	0.044***
IndRev%	-0.572***
<i>Momentum_{it}</i>	0.298***
Segments _{it}	-0.117***
Foreign _{it}	-0.091**
<i>GFC</i> _{it}	-1.122***
Dot-Com _{it}	-1.080***
SOX _{it}	-1.009***
Hansen	1,512.910
<i>m</i> ₁	-17.950
m_2	-0.760
Z1	627.450

Appendix C Table V The individual effect of managerial ability on accrual earnings management (AccrualEM	I)
--	----

Notes: This table presents the results from the system generalized method of moments regressions for the individual effect of managerial ability on accrual earnings management. The sample includes 66,695 firm-year observations obtained from Compustat from 1989 to 2011. Accrual earnings management is measured by Stubben's Model (2010). Managerial ability is measured by the model of Demerjian et al. (2012). Variables are defined in detail in the (Variable Definitions) section above. Each coefficient represents the change in accrual earnings management based on a one unit change in the determinant. Hansen test for over-identifying restrictions takes the shape of χ^2 distribution and checks for the orthogonality between the instrumental variables (IV_{il}) and the error term (ε_{it}). The AB test (Arellano and Bond, 1991) examines the serial correlation in the first difference residual (v_{il}) by using the (m_j) statistic and takes the shape of a normal distribution with N(0,1). m₁ values represent the results of the 1st order serial correlation test, while m₂ values represent the results of the 2nd order serial correlation test. The Wald test (z_I) checks for the joint significance of the reported coefficients in the model. Intercept is included, but not tabulated. *, ***, *** denotes significance at 0.10, 0.05, and 0.01, respectively.

Variables	Real earnings management
MgrlAbility _{it}	-0.133***
EarningsManagement _{i,t-1}	0.289***
<i>FirmSize</i> _{it}	0.074***
$SalesVolatility_{i,t-4,t}$	0.274***
CashFlowVolatility _{i,t-4,t}	0.046***
<i>OperCycle</i> _{<i>i</i>,<i>t</i>-4,<i>t</i>}	0.013*
Losses _{i,t-4,t}	-0.003
<i>NationalAuditor_{it}</i>	0.189***
<i>LitigationInd</i> _{it}	0.227***
MB _{it}	5.88e-06
SalesGrowthrk _{it}	0.013
LnNumAnalyst _{it}	0.043***
IndRev%	-0.112***
<i>Momentum_{it}</i>	-0.108***
Segments _{it}	0.001
Foreign _{it}	0.146***
GFC _{it}	-0.471***
Dot-Com _{it}	-0.373***
SOX _{it}	-0.338***
Hansen	1,536.760
m_1	-14.340
m_2	1.030
Ζ1	1,022.200

Appendix C Table VI The individual effect of man	gerial ability on real earnin	gs management (RealEM)
--	-------------------------------	------------------------

Notes: This table presents the results from the system generalized method of moments regressions for the individual effect of managerial ability on real earnings management. The sample includes 66,695 firm-year observations obtained from Compustat from 1989 to 2011. Real earnings management is measured by Gunny's Model (2010). Managerial ability is measured by the model of Demerjian et al. (2012). Variables are defined in detail in the (Variable Definitions) section above. Each coefficient represents the change in real earnings management based on a one unit change in the determinant. Hansen test for over-identifying restrictions takes the shape of χ^2 distribution and checks for the orthogonality between the instrumental variables (IV_{it}) and the error term (ε_{it}). The AB test (Arellano and Bond, 1991) examines the serial correlation in the first difference residual (v_{it}) by using the (m_j) statistic and takes the shape of a normal distribution with N(0,1). m_1 values represent the results of the 1st order serial correlation test, while m_2 values represent the results of the 2nd order serial correlation test. The Wald test (z_1) checks for the joint significance of the reported coefficients in the model. Intercept is included, but not tabulated. *, **, *** denotes significance at 0.10, 0.05, and 0.01, respectively.

Verichler	(1)	(2)	(3)
variables	Acc	rual earnings manager	ment
HHI _{it}	0.135***	-	-
<i>ProdSubstitutability</i> _{it}	-	0.169***	-
PricPower _{it}	-	-	0.000***
EarningsManagement _{i,t-1}	-0.115***	-0.118***	-0.121***
FirmSizeit	0.109***	0.129***	0.134***
SalesVolatility _{i,t-4,t}	0.339***	0.307***	0.307***
CashFlowVolatility _{i,t-4,t}	0.093***	0.124***	0.134***
$OperCycle_{i,t-4,t}$	0.001	0.002	0.005*
Losses _{i,t-4,t}	-0.096***	-0.031	-0.076***
<i>NationalAuditor</i> _{it}	0.104***	0.098***	0.129***
<i>LitigationInd</i> _{it}	-0.210***	-0.167***	-0.285***
MB _{it}	1.57e-05	1.61e-05	5.36e-06
SalesGrowthrk _{it}	-0.495***	-0.494***	-0.467***
<i>LnNumAnalyst</i> _{it}	0.089***	0.059***	0.075***
IndRev%	-0.342***	-0.416***	-0.418***
<i>Momentum_{it}</i>	0.516***	0.581***	0.526***
Segments _{it}	-0.080***	-0.097***	-0.105***
Foreign _{it}	-0.144***	-0.173***	-0.254***
GFC _{it}	0.056**	-0.238***	-0.925***
Dot-Com _{it}	0.232***	-0.169***	-0.790***
SOX _{it}	0.220***	-0.138***	-0.777***
Hansen	1,515.950	1,502.270	1,509.870
m_1	-17.840	-17.790	-17.770
m_2	-0.830	-1.020	-1.000
71	369.110	351,160	1.331.950

Appendix C Table VII The individual effect of market competition on accrual earnings management (AccrualEM)

Notes: This table presents the results from the system generalized method of moments regressions for the individual effects of market competition on accrual earnings management. The sample includes 66,695 firm-year observations obtained from Compustat from 1989 to 2011. Accrual earnings management is measured by Stubben's Model (2010). Each column represents the estimates from examining a dummy measure of market competition in the following order: (1) market concentration measured by the Herfindahl-Hirschman index (HHI), (2) product substitutability measured by the gross margin in the industry, and (3) pricing power measured by the industry adjusted Lerner Index (LI). HHI is a dummy variable that equals 1 if the firm is in an industry where market concentration is lower than the median and zero otherwise. ProdSubstitutability is a dummy variable that equals 1 if the firm is in an industry where product substitutability is lower than the median and zero otherwise. PricPower is a dummy variable that equals 1 if the firm has lower pricing power than the median and zero otherwise. Variables are defined in detail in the (Variable Definitions) section above. Each coefficient represents the change in accrual earnings management based on a one unit change in the determinant. Hansen test for over-identifying restrictions takes the shape of χ^2 distribution and checks for the orthogonality between the instrumental variables (IV_{il}) and the error term (ε_{il}) . The AB test (Arellano and Bond, 1991) examines the serial correlation in the first difference residual (v_{it}) by using the (m_i) statistic and takes the shape of a normal distribution with N(0,1). m₁ values represent the results of the 1st order serial correlation test, while m_2 values represent the results of the 2nd order serial correlation test. The Wald test (z_1) checks for the joint significance of the reported coefficients in the model. Intercept is included, but not tabulated. *, **, *** denotes significance at 0.10, 0.05, and 0.01, respectively.

Variables	(1)	(2)	(3)
	Real earnings management		
11111	0 176***		
ΠΠΙἰτ ProdSubstitutability	0.176****	- 0.000**	-
Proasubstitutability _{it}	-	0.088**	-
PricPower _{it}	-	-	0.000***
EarningsManagement _{i,t-1}	0.288***	0.28/***	0.290***
FirmSizeit	0.072***	0.080***	0.07/0***
$SalesVolatility_{i,t-4,t}$	0.255***	0.290***	0.217***
CashFlowVolatility _{i,t-4,t}	0.041***	0.035***	0.037***
<i>OperCycle</i> _{i,t-4,t}	0.009	0.002	0.002
Losses _{i,t-4,t}	0.024	0.044***	0.036***
<i>NationalAuditor</i> _{it}	0.184***	0.198***	0.192***
<i>LitigationInd</i> _{it}	0.298***	0.193***	0.252***
MB _{it}	3.24e-05***	2.95e-05***	2.82e-05***
SalesGrowthrk _{it}	-0.015*	-0.010	0.004
LnNumAnalyst _{it}	0.052***	0.047***	0.050***
IndRev%	-0.126***	-0.070*	-0.074**
<i>Momentum_{it}</i>	-0.163***	-0.157***	-0.138***
Segments _{it}	-0.010	-0.020	0.007
Foreign _{it}	0.216***	0.216***	0.201***
GFC _{it}	-0.230***	0.130***	0.110***
Dot-Com _{it}	-0.121***	0.220***	0.166***
SOX _{it}	-0.137***	0.192***	0.215***
Hansen	1,540.150	1,531.660	1,539.560
m_1	-14.390	-15.530	-14.310
m_2	1.030	1.570	0.97
7.1	1.086.850	879.310	1.932.970

Appendix C Table VIII The individual effect of market co	mpetition on real earnings management (l	RealEM)
--	--	---------

Notes: This table presents the results from the system generalized method of moments regressions for the individual effects of market competition on real earnings management. The sample includes 66,695 firm-year observations obtained from Compustat from 1989 to 2011. Real earnings management is measured by Gunny's Model (2010). Each column represents the estimates from examining a dummy measure of market competition in the following order: (1) market concentration measured by the Herfindahl-Hirschman index (HHI), (2) product substitutability measured by the gross margin in the industry, and (3) pricing power measured by the industry adjusted Lerner Index (LI). HHI is a dummy variable that equals 1 if the firm is in an industry where market concentration is lower than the median and zero otherwise. ProdSubstitutability is a dummy variable that equals 1 if the firm is in an industry where product substitutability is lower than the median and zero otherwise. PricPower is a dummy variable that equals 1 if the firm has lower pricing power than the median and zero otherwise. Variables are defined in detail in the (Variable Definitions) section above. Each coefficient represents the change in real earnings management based on a one unit change in the determinant. Hansen test for over-identifying restrictions takes the shape of χ^2 distribution and checks for the orthogonality between the instrumental variables (IV_{it}) and the error term (ϵ_{it}). The AB test (Arellano and Bond, 1991) examines the serial correlation in the first difference residual (v_{it}) by using the (m_i) statistic and takes the shape of a normal distribution with N(0,1). m₁ values represent the results of the 1st order serial correlation test, while m_2 values represent the results of the 2nd order serial correlation test. The Wald test (z_1) checks for the joint significance of the reported coefficients in the model. Intercept is included, but not tabulated. *, **, *** denotes significance at 0.10, 0.05, and 0.01, respectively.

Chapter 9 Conclusions

9.1 Introduction

Earnings management occurs as a result of the conflict in interests arising from the separation of ownership from control in public companies. Three groups of motives are expected to influence earnings management behaviour including contracting, capital market, and third-party motives. However, these three groups are not completely independent and may overlap in determining earnings management behaviour. In this thesis I have examined the effect of management compensation and managerial ability from the contracting group and market concentration and competition from the third-party group. Finally, I have studied how factors from both groups jointly influence earnings management.

While the prior research has considered management compensation as one of the contracting motives, this thesis views it as an outcome of earnings management behaviour because firms tend to reward/punish managers through their compensation. Accordingly, the information asymmetry and moral hazard drive managers to signal better performance, while shareholders screen these signals to determine management compensation. The thesis contributes to the literature by documenting that while firms reward/punish their managers for accrual/real earnings management because of its lower/higher subsequent adverse consequences, they react less intensively when high quality managers are involved in the manipulation because of the high contracting costs and the big knowledge gap that make their activities less likely to be detected or understood. Nevertheless, superior managers continue to prefer using/avoiding accrual/real earnings management because of the overall positive/negative effect on their future compensation.

At the external motives level, this thesis expects a non-linear effect of market concentration on earnings management based on the role of market concentration in increasing information asymmetry and decreasing communication costs at the same

time. The results show that while firms are more likely to use accrual earnings management as information quantity decreases in concentrated market, they prefer the sophisticated real earnings management as the information quality decreases. The thesis contributes to the literature by identifying an optimal level of market concentration where both accrual and real earnings management are declining at the same time.

Finally, as the different motives of earnings management are unlikely to be independent, market competition at the external level may interact with managerial ability at the contractual level in determining earnings management behaviour. While market competition increases the costs of communicating information in the market and thus may stimulate earnings management, the response of high quality managers is sophisticated as they possess the abilities to provide better performance which is difficult to be monitored due to their high contracting costs with the firms. The thesis contributes to the literature by documenting a different response of more able managers who prefer using accrual earnings management under the pressure of competition compared to other managers who continue using both accrual and real earnings management.

The remainder of this chapter is organized as follows. Section 9.2 provides a summary of the objectives and results of the three empirical chapters of this thesis. Section 9.3 explains the implications of the thesis to the regulators, policy makers, shareholders, investors, gatekeepers, and academics. Finally, section 9.4 presents the research limitations and suggests some recommendations for the future studies to take into consideration based on the results of this thesis.

9.2 Summary of Results

This thesis has focused on management compensation, managerial ability, market concentration and market competition as some of the contractual and external motives of earnings management. Chapter 2 has discussed the definitions, activities, and measurement of earnings management. Chapter 3 has established the theoretical framework of this thesis. Chapter 4 has summarized the three main groups of motives of earnings management. Chapter 5 has explained the data and research methods used in the thesis. Chapter 6 represents the first empirical chapter of this thesis and has

examined the impact of earnings management and managerial ability on management compensation. Chapter 7 is the second empirical chapter and has studied the non-linear relationship between market concentration and earnings management. Finally, chapter 8 is the third empirical chapter and has examined the role of market competition in the relation between managerial ability and earnings management. In this section I summarize the main objectives and findings of the three empirical chapters of this thesis.

9.2.1 Management Compensation: The Impact of Earnings Management and Managerial Ability

Although compensation has been extensively examined as a contractual motive of earnings management, the impact of earnings management on executives' compensation has not been considered in the literature to my knowledge. The objective of this empirical chapter is to examine the previous relation and use it to explain the behaviour of high quality managers in using more accrual and less real earnings management, taking into consideration the different economic consequences of each activity to the firm. To achieve this objective, I first examine the effect of managerial ability on management compensation. Second, I examine the impact of both accrual and real earnings management on management compensation. Finally, I examine whether the previous influence changes at the different levels of managerial ability.

First, I find that firms tend to attract or retain managers with better abilities by rewarding them with higher compensation. Due to the sophisticated nature of their decisions, superior managers receive better compensation as the positive consequences of their decisions are realized over time. The optimal pay-performance structure motivates managers to share risk and thus mitigates the information asymmetry and moral hazard problems arising from the separation of ownership and control.

Second, although accrual and real earnings management can be used by the managers to signal better performance, real earnings management is screened by the shareholders as a more costly activity when detected in the future. Accrual earnings management allows managers to extract better compensation because it enables them to improve their performance measures without influencing firms' future cash flows. In contrast, real earnings management results in a reduction of managers' future

compensation because it has serious economic consequences on the firm's future performance. As a result, while real earnings management is considered more costly than accrual earnings management from the firm's perspective because it severely influences firms' future performance, it is also considered more costly from management's perspective because it drives a reduction in future compensation.

Finally, high quality managers continue to prefer using accrual earnings management because it has an overall positive effect on their future compensation. In contrast, they avoid real earnings management due to its overall negative effect. In this sense, real earnings management is perceived more costly than accrual earnings management not only from the firm and management points of view, but also from the perspective of high quality executives.

9.2.2 Market Concentration and Earnings Management: A Non-Linear Relationship

The objective of this empirical chapter is to examine a non-linear relation between market concentration and earnings management based on two different perspectives in the agency theory and the revelation principle. While it is more difficult to monitor performance in concentrated markets because of the higher information asymmetry between the firms and the stakeholders, the presence of fewer competitors makes communicating information less costly. The literature supports the previous points of view and documents a positive linear effect of market concentration on earnings management in some instances (Dalia and Park, 2009; Markarian and Santalo, 2010; Karuna et al., 2012) and a negative linear effect in others (Datta et al., 2013). Therefore, I initially examine the non-linear effect of market concentration on earnings management. Then, I explain this effect through the asymmetry of information in the market.

First, I find a non-linear effect of market concentration on earnings management that takes an inverted U shape when accrual earnings management is the regressand and U shape in the case of real earnings management. Accrual earnings management increases at market concentration levels between 0 and 55% but starts decreasing after that level, while real earnings management decreases at market concentration levels between 0 and 60% but starts increasing afterwards. The results are in line with the contrasting effects of market concentration that involve high information asymmetry and low communication costs at the same time.

While the quantity of information decreases steadily in concentrated markets, the quality of information increases up to market concentration levels of 55% - 60% then starts decreasing. The reason is that firms get less threatened to lose their competitive advantages in concentrated markets and thus provide high quality information that allows analysts to make better forecasts. However, at high levels of market concentration, firms collude to hide information and thus make it difficult for the analysts to produce quality forecasts. These findings explain the increase in accrual earnings management at lower levels of market concentration because of the decrease in information quantity in the market, and the use of real earnings management at higher levels of market concentration to monitor firms' sophisticated activities.

9.2.3 Earnings Management, Managerial Ability and the Role of Competition

Managerial ability and market competition are not independent and thus they are likely to interact in determining the levels of earnings management. While managerial ability influences the contracting costs between managers and shareholders, competition changes the costs of communicating information to those shareholders. In this sense, the previously documented behaviour of superior managers using more accrual and less real earnings management may change under the pressure of market competition. Therefore, the objective of this empirical chapter is to examine the influence of market competition on accrual and real earnings managed by high quality managers.

I initially find that market competition has a positive effect on the relation between managerial ability and accrual earnings management. While market competition drives more adverse selection by increasing the cost of communicating information in the market, superior managers become more selective in their behaviour. Therefore, they choose more accrual earnings management because it is perceived less costly from the firm and management perspectives at the same time.

Finally, I find that managers use more real earnings management under the pressure of the high communication costs in competitive markets. However, superior managers select to avoid using real earnings management because they are able to realize its severe consequences to the firm and its management in the future. Therefore, I find that market competition decreases the amount of real earnings managed by high quality managers. The results imply that market competition and managerial ability together contribute to the welfare of the firm in the long run.

9.3 Implications

The previous findings have implications to be considered by regulators, policy makers, shareholders, investors, academics, and some of the gatekeepers e.g., auditors and analysts. First, while real earnings management involves some economic consequences to the firm because it contributes to decreasing future cash flows and sacrificing the long term projects, it results in lower compensation to managers at different levels of abilities. Therefore, real earnings management is considered a costly activity to the firms, managers, and even to the top quality executives. However, more effort should be invested by regulators in communicating the previous consequences of real earnings management to the different stakeholders e.g., shareholders, investors, and managers, in addition to emphasizing the role of the gatekeepers, e.g., auditors, in reporting such activities.

Second, market concentration cannot be considered absolutely good or bad to the economy. An optimal level has been identified in Chapter 7 at intermediate levels of market concentration where both accrual and real earnings management are decreasing. Therefore, while anti-competition laws attempt to mitigate the high levels of market concentration, the previous findings of this thesis suggest that both high and low levels of market concentration increase the likelihood of different types of earnings management activities. Therefore, it is more recommended that such laws emphasize the optimal level of market concentration that allows for optimal information and thus contributes to mitigating the agency conflict and avoiding more concentrated markets in the future. The optimal market concentration assists industry regulators to keep a balance between firms entering and exiting the market. Furthermore, it emphasizes the

role of analysts as gatekeepers that may contribute to enlarging the optimal level of market concentration by providing higher quality forecasts.

Finally, earnings management motives at the firm level cannot be considered independently from those at the industry level when explaining management decisions. Therefore, any regulation that attempts to mitigate one motive at a specific level has to consider it with respect to the other motives at a different level. For example, while market competition drives more real earnings management when examined individually, this thesis documents that it results in less real earnings management when considered together with managerial ability. In this sense, both factors serve as strong governance mechanisms in preventing the costly real earnings management. Explaining the interaction between earnings management motives at different levels assists regulators in identifying firms at higher risk of earnings management and ultimately helps investors and shareholders in making better decisions.

9.4 Limitations and Future Research

While the thesis has examined the costs of accrual and real earnings management from the perspective of managers with different abilities, the previous costs should be also considered at other contracting levels e.g., position in the executive team, tenure in the firm, years remaining to retirement, and the mobility of a manager to other firms. Distinguishing between the costs of earnings management strategies for the different types of managers would assist in further mitigating the costly earnings management activities. In addition, although the time period covered in chapter 6 of this thesis is still long enough to run a panel data methodology (15 years from 1992 to 2006), up to date data from new sources can be obtained to confirm the previous results or perform the suggested new ideas.

Although the literature has focused on market concentration measures when studying competition, the contradicting theoretical points of view regarding the impact of market concentration on earnings management can be also extended when considering the other measures of market competition. In this sense, competition is expected to play a double-edged role because of the increased communication costs and the decrease in information asymmetry at the same time. For future research, I recommend examining a non-linear impact of the other measures of market competition e.g., product substitutability and market pricing power, on earnings management as it may provide a new understanding of this area of research.

Finally, the contracting motives, capital market motives, and third-party motives of earnings management are unlikely to be independent in reality. While this thesis has examined the interaction of managerial ability as a contracting motive and market competition as a third-party motive, prior studies have almost ignored this area of research. To my knowledge, the study of Bushman et al. (2013) examines the influence of the voting rights as a corporate governance mechanism and market competition as an external motive on the loan loss provision in the banking sector of 46 countries. It documents that the role of weak governance in motivating earnings management can be mitigated by bank competition. Future studies should consider such related factors at different levels to extend research benefits beyond the firm borders.

Bibliography

- ABARBANELL, J. & LEHAVY, R. 2003. Biased Forecasts Or Biased Earnings? The Role of Reported Earnings in Explaining Apparent Bias and Over/Underreaction in Analysts' Earnings Forecasts. *Journal of Accounting and Economics*, 36, 105-146.
- ABDEL-KHALIK, A. R. 2002. Reforming Corporate Governance Post Enron: Shareholders' Board of Trustees and the Auditor. *Journal of Accounting and Public Policy*, 21, 97–103.
- ADUT, D., HOLDER, A. D. & ROBIN, A. 2013. Predictive Versus Opportunistic Earnings Management, Executive Compensation, and Firm Performance. *Journal of Accounting and Public Policy*, 32, 126-146.
- AGARWAL, S. & HAUSWALD, R. 2010. Distance and Private Information in Lending. *The Review of Financial Studies*, 23, 2757-2788.
- AGGARWAL, R. K. & NANDA, D. 2004. Access, Common Agency, and Board Size. *SSRN eLibrary*. Available at http://ssrn.com/abstract=571801.
- AGGARWAL, R. K. & SAMWICK, A. A. 1999. Executive Compensation, Strategic Competition, and Relative Performance Evaluation: Theory and Evidence. *Journal of Finance*, 54, 1999-2043.
- AGHION, P., BLOOM, N., BLUNDELL, R., GRIFFITH, R. & HOWITT, P. 2005. Competition and Innovation: An Inverted-U Relationship. *Quarterly Journal of Economics*, 120, 701-728.
- AHARONY, J., LEE, C.-W. J. & WONG, T. J. 2000. Financial Packaging of IPO Firms in China. *Journal of Accounting Research*, 38, 103-126.
- AHMED, A. S., BILLINGS, B. K., MORTON, R. M. & STANFORD-HARRIS, M. 2002. The Role of Accounting Conservatism in Mitigating Bondholder-Shareholder Conflicts over Dividend Policy and in Reducing Debt Costs.

Accounting Review, 77, 867-890.

- AHMED, A. S., NAINAR, S. M. K. & JIAN, Z. 2005. Do Analysts' Earnings Forecasts Fully Reflect the Information in Accruals? *Canadian Journal of Administrative Sciences (Canadian Journal of Administrative Sciences)*, 22, 329-342.
- AHMED, A. S., ZHOU, J. & LOBO, G. J. 2006. Job Security and Income Smoothing: An Empirical Test of the Fudenberg and Tirole (1995) Model. SSRN eLibrary. Available at http://ssrn.com/abstract=248288.
- ALBRECHT, W. D. & RICHARDSON, F. M. 1990. Income Smoothing by Economy Sector. *Journal of Business Finance & Accounting*, 17, 713-730.
- ALI, A., KLASA, S. & YEUNG, E. 2014. Industry Concentration and Corporate Disclosure Policy. *Journal of Accounting and Economics*, 58, 240-264.
- ALLAYANNIS, G. & WESTON, J. P. 2007. Earnings Volatility, Cash Flow Volatility, and Firm Value. Working paper. *University of Virginia and Rice University*.
- ALLEN, F. 1985. Repeated Principal–Agent Relationships with Lending and Borrowing. *Economics Letters*, 17, 27–31.
- ALONSO-BORREGO, C. & ARELLANO, M. 1999. Symmetrically Normalized Instrumental-Variable Estimation Using Panel Data. *Journal of Business & Economic Statistics*, 17, 36-49.
- ANDERSON, K. L., DELI, D. N. & GILLAN, S. L. 2006. Boards of Directors, Audit Committees, and the Information Content of Earnings. SSRN eLibrary. Available at http://ssrn.com/abstract=444241.
- ANTLE, R. & NALEBUFF, B. 1991. Conservatism and Auditor–Client Negotiations. Journal of Accounting Research, 29, 31–54.
- ARELLANO, M. & BOND, S. 1991. Some Tests of Specification for Panel Data: Monte Carlo Evidence and an Application to Employment Equations. *The Review of Economic Studies*, 58, 277-297.

ARROYO, M. R. 2007. Information Asymmetries, Credit Rationing and Banking

Concentration: The Argentinean case. Working paper, University Library of Munich.

- ATHANASAKOU, V., STRONG, N. C. & WALKER, M. 2011. The Market Reward for Achieving Analyst Earnings Expectations: Does Managing Expectations or Earnings Matter? *Journal of Business Finance & Accounting*, 38, 58-94.
- ATHANASAKOU, V. E., STRONG, N. C. & WALKER, M. 2009. Earnings Management or Forecast Guidance to Meet Analyst Expectations? Accounting & Business Research (Wolters Kluwer UK), 39, 3-35.
- AYERS, B. C., JIANG, J. & YEUNG, P. E. 2006. Discretionary Accruals and Earnings Management: An Analysis of Pseudo Earnings Targets. *The Accounting Review*, 81, 617-652.
- BABER, W. R., FAIRFIELD, P. M. & HAGGARD, J. A. 1991. The Effect of Concern about Reported Income on Discretionary Spending Decisions: The Case of Research and Development. *The Accounting Review*, 66, 818–829.
- BABER, W. R., KANG, S.-H. & LI, Y. 2011. Modeling Discretionary Accrual Reversal and the Balance Sheet as an Earnings Management Constraint. *The Accounting Review*, 86, 1189-1212.
- BADERTSCHER, B. A. 2011. Overvaluation and the Choice of Alternative Earnings Management Mechanisms. *The Accounting Review*, 86, 1491-1518.
- BADERTSCHER, B. A., BURKS, J. J. & EASTON, P. D. 2014. The Market Pricing of Other-Than-Temporary Impairments. *Accounting Review*, 89, 811-838.
- BADERTSCHER, B. A., PHILLIPS, J. D., PINCUS, M. & REGO, S. O. 2006. Tax Implications of Earnings Management Activities: Evidence from Restatements. *SSRN eLibrary*. Available at http://ssrn.com/abstract=888564.
- BAGNOLI, M., LEVINE, S. & WATTS, S. G. 2005. Analyst Estimation Revision Clusters and Corporate Events, Part I. *Annals of Finance*, 1, 245-265.
- BAGNOLI, M. & WATTS, S. G. 2000. The Effect of Relative Performance Evaluation on Earnings Management: a game-theoretic approach. *Journal of Accounting*

and Public Policy, 19, 377-397.

- BAGNOLI, M. & WATTS, S. G. 2010. Oligopoly, Disclosure, and Earnings Management. *The Accounting Review*, 85, 1191-1214.
- BAINBRIDGE, S. M. 2001. The Law and Economics of Insider Trading: A Comprehensive Primer. SSRN eLibrary. Available at http://ssrn.com/abstract=261277.
- BAKER, T., COLLINS, D. & REITENGA, A. 2003. Stock Option Compensation and Earnings Management Incentives. *Journal of Accounting, Auditing & Finance*, 18, 557-582.
- BALAKRISHNAN, K., VASHISHTHA, R. & VERRECCHIA, R. E. 2012. Aggregate Competition, Information Asymmetry and Cost of Capital: Evidence from Equity Market Liberalization. Working paper, Stern School of Business, New York University.
- BALL, R. & SHIVAKUMAR, L. 2008. Earnings Quality at Initial Public Offerings. *Journal of Accounting and Economics*, 45, 324-349.
- BALSAM, S., BARTOV, E. & MARQUARDT, C. 2002. Accruals Management, Investor Sophistication, and Equity Valuation: Evidence from 10-Q Filings. *Journal of Accounting Research*, 40, 987-1012.
- BALSAM, S., CHEN, L. H. & SANKARAGURUSWAMY, S. 2003. Earnings Management Prior to Stock Option Grants. SSRN eLibrary. Available at http://ssrn.com/abstract=378440.
- BAMBER, L. S., JOHN, J. & ISABEL YANYAN, W. 2010. What's My Style? The Influence of Top Managers on Voluntary Corporate Financial Disclosure. *Accounting Review*, 85, 1131-1162.
- BARANCHUK, N., MACDONALD, G. & YANG, J. 2011. The Economics of Super Managers. *The Review of Financial Studies*, 24, 3321-3368.
- BARCLAY, M. J., GODE, D. & KOTHARI, S. P. 2005. Matching Delivered Performance. *Journal of Contemporary Accounting & Economics*, 1, 1-25.

- BARNEY, J. B. 1990. The Debate between Traditional Management Theory and Organizational Economics: Substantive Differences or Intergroup Conflict? *Academy of Management Review*, 15, 382–393.
- BARTH, M. E., CRAM, D. P. & NELSON, K. K. 2001. Accruals and Prediction of Future Cash Flows. *The Accounting Review*, 76, 27–58.
- BARTH, M. E. & HUTTON, A. P. 2000. Differential Information Environments: Effects on Cost of Capital and Reflection in Share Prices of Recognized and Disclosed Accounting Amounts. SSRN eLibrary. Available at http://ssrn.com/abstract=55488.
- BARTH, M. E., LANDSMAN, W. R. & LANG, M. H. 2008. International Accounting Standards and Accounting Quality. *Journal of Accounting Research*, 46, 467– 498.
- BARTON, J. & MERCER, M. 2005. To Blame or Not To Blame: Analysts' Reactions to External Explanations for Poor Financial Performance. *Journal of Accounting and Economics*, 39, 509-533.
- BARTOV, E. 1993. The Timing of Asset Sales and Earnings Manipulation. *Accounting Review*, 68, 840–55.
- BARTOV, E., GIVOLY, D. & HAYN, C. 2002a. The rewards to meeting or beating earnings expectations. *Journal of Accounting and Economics*, 33, 173-204.
- BARTOV, E. & COHEN, D. A. 2009. The "Numbers Game" in the Pre- and Post-Sarbanes-Oxley Eras. *Journal of Accounting, Auditing & Finance,* 24, 505-534.
- BARTOV, E., MOHANRAM, P. & SEETHAMRAJU, C. 2002b. Valuation of Internet Stocks--An IPO Perspective. *Journal of Accounting Research*, 40, 321-346.
- BARUA, A., LEGORIA, J. & MOFFITT, J. S. 2006. Accruals Management to Achieve Earnings Benchmarks: A Comparison of Pre-managed Profit and Loss Firms. *Journal of Business Finance & Accounting*, 33, 653-670.
- BARUA, A., LIN, S. & SBARAGLIA, A. M. 2010. Earnings Management Using Discontinued Operations. *The Accounting Review*, 85, 1485-1509.

- BASU, S. 1997. The Conservatism Principle and the Asymmetric Timeliness of Earnings1. *Journal of Accounting and Economics*, 24, 3-37.
- BAUMAN, C. C., BAUMAN, M. P. & HALSEY, R. F. 2001. Do Firms Use the Deferred Tax Asset Valuation Allowance to Manage Earnings? *Journal of the American Taxation Association*, 23, 27-48.
- BAUMAN, M. P., BRASWELL, M. & SHAW, K. W. 2005. The Numbers Game: How do Managers Compensated with Stock Options meet Analysts' Earnings Forecasts? *Research in Accounting Regulation*, 18, 3-28.
- BECKER, C. L., DEFOND, M. L., JIAMBALVO, J. & SUBRAMANYAM, K. R. 1998. The Effect of Audit Quality on Earnings Management. *Contemporary Accounting Research*, 15, 1-24.
- BENEISH, M. D. & NICHOLS, C. D. 2005. Earnings Quality and Future Returns: The Relation between Accruals and the Probability of Earnings Manipulation. SSRN eLibrary. Available at http://ssrn.com/abstract=725162.
- BENEISH, M. D. & VARGUS, M. E. 2002. Insider Trading, Earnings Quality, and Accrual Mispricing. *Accounting Review*, 77, 755-791.
- BERGSTRESSER, D. & PHILIPPON, T. 2006. CEO Incentives and Earnings Management. *Journal of Financial Economics*, 80, 511-529.
- BERNARD, V. L. & SKINNER, D. J. 1996. What Motivates Managers' Choice of Discretionary Accruals? *Journal of Accounting and Economics*, 22, 313–325.
- BERTRAND, M. & SCHOAR, A. 2003. Managing With Style: The Effect of Managers on Firm Policies. *Quarterly Journal of Economics*, 118, 1169-1208.
- BEUSELINCK, C., DELOOF, M. & VANSTRAELEN, A. 2015. Cross-Jurisdictional Income Shifting and Tax Enforcement: Evidence from Public versus Private Multinationals. *Review of Accounting Studies*, 20, 710-746.
- BEYER, A. 2009. Capital Market Prices, Management Forecasts, and Earnings Management. *The Accounting Review*, 84, 1713-1747.

- BHATTACHARYA, N., DESAI, H. & VENKATARAMAN, K. 2013. Does Earnings Quality Affect Information Asymmetry? Evidence from Trading Costs. *Contemporary Accounting Research*, 30, 482-516.
- BLACK, L. K. 2009. Information Asymmetries between Lenders and the Availability of Competitive Outside Offers. Working paper, *Board of Governors of the Federal Reserve System*.
- BLAIR, M. M. 1995. *Ownership and Control: Rethinking Corporate Governance for the Twenty-First Century*. Washington, DC: Brookings Institution Press.
- BLUNDELL, R. & BOND, S. 1998. Initial Conditions and Moment Restrictions in Dynamic Panel Data Models. *Journal of Econometrics*, 87, 115-143.
- BOLTON, P. & SCHARFSTEIN, D. S. 1990. A Theory of Predation Based on Agency Problems. *American Economic Review*, 80, 93.
- BOLTON, P., SCHEINKMAN, J. O. S. & XIONG, W. E. I. 2006. Executive Compensation and Short-Termist Behaviour in Speculative Markets. *Review of Economic Studies*, 73, 577-610.
- BOONE, A. L., FIELD, L. C., KARPOFF, J. M. & RAHEJA, C. G. 2004. The Determinants of Corporate Board Size and Composition: An Empirical Analysis. AFA 2005 Philadelphia Meeting. SSRN eLibrary. Available at http://ssrn.com/abstract=605762.
- BORDEN, M. 2007. The Role of Financial Journalists in Corporate Governance. Fordham Journal of Corporate and Financial Law, 12, 311–369.
- BOUJELBENE, Y. & BESBES, L. 2012. The Determinants of Information Asymmetry between Managers and Investors: A Study on Panel Data. *IBIMA Business Review*, 2012, 3-5.
- BOWEN, R. M., DUCHARME, L. & SHORES, D. 1995. Stakeholders' Implicit Claims and Accounting Method Choice. *Journal of Accounting and Economics*, 20, 255-295.

BOYER, C. M., CICCONE, S. J. & ZHANG, W. 2003. Insider Trading and Earnings

Reporting: Evidence of Managerial Optimism or Opportunism. *SSRN eLibrary*. Available at http://ssrn.com/abstract=426500.

- BOYNTON, C. E., DOBBINS, P. S. & PLESKO, G. A. 1992. Earnings Management and the Corporate Alternative Minimum Tax. *Journal of Accounting Research*, 30, 131-153.
- BRADSHAW, M. T., RICHARDSON, S. A. & SLOAN, R. G. 2001. Do Analysts and Auditors Use Information in Accruals? *Journal of Accounting Research*, 39, 45-74.
- BRANSON, D. M. 2006. Too Many Bells? Too Many Whistles? Corporate Governance in the Post Enron, Post Worldcom Era. *South Carolina Law Review*, 58, 1-20.
- BRAZEL, J. F. & WEBB, E. 2006. CEO Compensation and the Seasoned Equity Offering Decision. *Managerial and Decision Economics*, 27, 363-378.
- BRENNAN, M. J. & HUGHES, P. J. 1991. Stock Prices and the Supply of Information. *Journal of Finance*, 46, 1665-1691.
- BRENNAN, M. J. & TAMAROWSKI, C. 2000. Investor Relations, Liquidity, and Stock Prices. *Journal of Applied Corporate Finance*, 12, 26-37.
- BRICKLEY, J. A. 2003. Empirical Research on CEO Turnover and Firm-Performance: A Discussion. *Journal of Accounting and Economics*, 36, 227-233.
- BROCHET, F. 2016. [Forthcoming]. Information Content of Insider Trades before and After the Sarbanes-Oxley Act. *The Accounting Review*. [Online]. [Accessed 4 November 2015]. Available at http://ssrn.com/abstract=1108731.
- BROWN, S. 2004. The Impact of Pension Assumptions on Firm Value. *SSRN eLibrary*. Available at http://ssrn.com/abstract=596666.
- BURGSTAHLER, D. & DICHEV, I. 1997. Earnings Management to Avoid Earnings Decreases and Losses. *Journal of Accounting and Economics*, 24, 99-126.
- BURGSTAHLER, D., HAIL, L. & LEUZ, C. 2006. The Importance of Reporting Incentives: Earnings Management in European Private and Public Firms. *The*

Accounting Review, 81, 983-1016.

- BURNS, N. & KEDIA, S. 2006. The Impact of Performance-Based Compensation on Misreporting. *Journal of Financial Economics*, 79, 35-67.
- BUSHEE, B. J. 1998. The Influence of Institutional Investors in Myopic R&D Investment Behavior. *Accounting Review*, 73, 305.
- BUSHMAN, R. M., WANG, Q. S. & WILLIAMS, C. D. 2013. Governance, Competition and Opportunistic Accounting Choices by Banks. Working paper, University of North Carolina at Chapel Hill.
- CAHAN, S. F. 1992. The Effect of Antitrust Investigations on Discretionary Accruals: A Refined Test of the Political-Cost Hypothesis. *Accounting Review*, 67, 77-95.
- CAHAN, S. F., CHAVIS, B. M. & ELMENDORF, R. G. 1997. Earnings Management of Chemical Firms in Response to Political Costs from Environmental Legislation. *Journal of Accounting, Auditing & Finance*, 12, 37-65.
- CALLAO, S. & JARNE, I. S. 2010. Have IFRS Affected Earnings Management in the European Union? *Accounting in Europe*, 7, 159–189.
- CAPKUN, V., COLLINS, D. W. & JEANJEAN, T. 2016. [Forthcoming]. The Effect of IAS/IFRS Adoption on Earnings Management (Smoothing): A Closer Look at Competing Explanations. *Journal of Accounting and Public Policy*. [Online]. [Accessed 7 June 2015]. Available at http://ssrn.com/abstract=1850228.
- CARAMANIS, C. & LENNOX, C. 2008. Audit Effort and Earnings Management. Journal of Accounting and Economics, 45, 116-138.
- CARTER, M., LYNCH, L. & ZECHMAN, S. C. 2009. Changes in Bonus Contracts in the Post-Sarbanes–Oxley Era. *Review of Accounting Studies*, 14, 480-506.
- CHAMBERS, D. J. 1999. Earnings Management and Capital Market Misallocation. SSRN eLibrary. Available at http://ssrn.com/abstract=198790.
- CHAN, K., CHAN, L., JEGADEESH, N. & LAKONISHOK, J. 2006. Earnings Quality and Stock Returns. *Journal of Business*, 79, 1041–1082.

- CHEN, K. C. W. & YUAN, H. 2004. Earnings Management and Capital Resource Allocation: Evidence from China's Accounting-Based Regulation of Rights Issues. *The Accounting Review*, 79, 645-665.
- CHEN, Q., KELLY, K. & SALTERIO, S. E. 2012. Do Changes in Audit Actions and Attitudes Consistent with Increased Auditor Scepticism Deter Aggressive Earnings Management? An Experimental Investigation. Accounting, Organizations and Society, 37, 95-115.
- CHENG, Q. & WARFIELD, T. 2005. Equity Incentives and Earnings Management. *The Accounting Review*, 80, 441-476.
- CHENG, Q., WARFIELD, T. & YE, M. 2011. Equity Incentives and Earnings Management: Evidence from the Banking Industry. *Journal of Accounting*, *Auditing & Finance*, 26, 317-349.
- CHENG, S. 2004. R&D Expenditures and CEO Compensation. *Accounting Review*, 79, 305–27.
- CHENG, T. Y. & FIRTH, M. 2000. An Empirical Analysis of the Bias and Rationality of Profit Forecasts Published in New Issue Prospectuses. *Journal of Business Finance & Accounting*, 27, 423-446.
- CHEUNG, C. S. & KRINSKY, I. 1994. Information Asymmetry and the Underpricing Of Initial Public Offerings: Further Empirical Evidence. *Journal of Business Finance & Accounting*, 21, 739-747.
- CHHAOCHHARIA, V., GRINSTEIN, Y., GRULLON, G. & MICHAELY, R. 2012. Product Market Competition and Internal Governance: Evidence from the Sarbanes Oxley Act. *Johnson School Research Paper Series No. 18*.
- CHI, W., LISIC, L. & PEVZNER, M. 2011. Is Enhanced Audit Quality Associated with Greater Real Earnings Management? *Accounting Horizons*, 25, 315–335.
- CHOI, Y., WALKER, M. & YOUNG, S. 2006. Earnings Reporting and Analysts' Earnings Forecasts: The Perceptions of UK Analysts and Financial Managers. Working paper, *Lancaster University*.

- CHRISTIE, A. A., JOYE, M. P. & WATTS, R. L. 2003. Decentralization of the Firm: Theory and Evidence. *Journal of Corporate Finance*, 9, 3-36.
- CHTOUROU, S. M., BEDARD, J. & COURTEAU, L. 2001. Corporate Governance and Earnings Management. SSRN eLibrary. Available at http://ssrn.com/abstract=275053.
- CLATWORHTY, M. & JONES, M. J. 2001. The Effect of Thematic Structure on the Variability of Annual Report Readability. *Accounting, Auditing and Accountability Journal,* 14, 311-326.
- COFFEE, J. C. 2003. Gatekeeper Failure and Reform: The Challenge of Fashioning Relevant Reforms. Columbia Law and Economics *SSRN eLibrary*. Available at: http://ssrn.com/abstract=447940.
- COHEN, D. A., DEY, A. & LYS, T. Z. 2008. Real and Accrual-Based Earnings Management in the Pre- and Post-Sarbanes-Oxley Periods. *The Accounting Review*, 83, 757-787.
- COHEN, D. A. & ZAROWIN, P. 2010. Accrual-Based and Real Earnings Management Activities around Seasoned Equity Offerings. *Journal of Accounting and Economics*, 50, 2-19.
- COLES, J. L., HERTZEL, M. & KALPATHY, S. 2006. Earnings Management Around Employee Stock Option Reissues. *Journal of Accounting and Economics*, 41, 173-200.
- COLLINS, D. W., GONG, G. & HRIBAR, P. 2003. Investor Sophistication and the Mispricing of Accruals. *Review of Accounting Studies*, 8, 2-3 (June): 251–276.
- CONYON, M. J. & READ, L. E. 2006. A Model of the Supply of Executives for Outside Directorships. *Journal of Corporate Finance*, 12, 645–659.
- CORE, J. & GUAY, W. 1999. The Use of Equity Grants to Manage Optimal Equity Incentive Levels. *Journal of Accounting and Economics*, 28, 151-184.
- CORE, J. & GUAY, W. 2001. Stock Option Plans for Non-Executive Employees. Journal of Financial Economics, 61, 253 - 287.

- CORE, J. & GUAY, W. 2002. Estimating the Value of Employee Stock Option Portfolios and Their Sensitivities to Price and Volatility. *Journal of Accounting Research*, 40, 613 - 630.
- CORE, J. E., GUAY, W. & VERDI, R. 2008. Is Accruals Quality a Priced Risk Factor? Journal of Accounting and Economics, 46, 2–22.
- DALIA, M. & PARK, J. 2009. Market Competition and Earnings Management. Working paper, *Harbert College of Business*.
- DANESHFAR, A., ZEGHAL, D. & SAEI, M. J. 2009. Earnings Management Effect in Different Stock Market Cycles. Journal of Business & Economics Research, 7, 1-9.
- DANIEL, N. D., DENIS, D. J. & NAVEEN, L. 2008. Do Firms Manage Earnings to Meet Dividend Thresholds? *Journal of Accounting and Economics*, 45, 2-26.
- DANOS, P. & EICHENSEHER, J. W. 1981. Audit Industry Dynamics: Factors Affecting Changes in Client-Industry Market Shares. *Journal of Accounting Research*, 20, 604–616.
- DARROUGH, M. & RANGAN, S. 2005. Do Insiders Manipulate Earnings When They Sell Their Shares in an Initial Public Offering? *Journal of Accounting Research*, 43, 1-33.
- DAS, S., KIM, K. & PATRO, S. 2011. An Analysis of Managerial Use and Market Consequences of Earnings Management and Expectation Management. *The Accounting Review*, 86, 1935-1967.
- DAS, S. & SHROFF, P. K. 2002. Fourth Quarter Reversals in Earnings Changes and Earnings Management. SSRN eLibrary. Available at http://ssrn.com/abstract=308441.
- DATTA, S., ISKANDAR-DATTA, M. & SINGH, V. 2013. Product Market Power, Industry Structure, and Corporate Earnings Management. *Journal of Banking & Finance*, 37, 3273-3285.

DAVIDSON, W. N., JIRAPORN, P. & DADALT, P. J. 2006. Causes and

Consequences of Audit Shopping: An Analysis of Auditor Opinions, Earnings Management, and Auditor Changes. *Quarterly Journal of Business and Economics*, 45, 69–87.

- DAVIS, J. H., SCHOORMAN, F. D. & DONALDSON, L. 1997. Toward a Stewardship Theory of Management. Academy of Management Review, 22, 20-47.
- DAVIS, L. R., SOO, B. & TROMPETER, G. 2006. Auditor Tenure and the Ability to Meet or Beat Earnings Forecasts. *Contemporary Accounting Research*, 26, 517-548.
- DEANGELO, H., DEANGELO, L. & SKINNER, D. J. 1994. Accounting Choice in Troubled Companies. *Journal of Accounting and Economics*, 17, 113-143.
- DEANGELO, H., DEANGELO, L. & SKINNER, D. J. 1996. Reversal of Fortune Dividend Signaling and the Disappearance of Sustained Earnings Growth. *Journal of Financial Economics*, 40, 341-371.
- DEANGELO, L. 1986. Accounting Numbers as Market Valuation Substitutes: A Study of Management Buyouts of Public Stockholders. *The Accounting Review*, 61, 400–420.
- DEANGELO, L. 1988. Discussion of Evidence of Earnings Management from the Provision for Bad Debts. *Journal of Accounting Research*, 26, 32-40.
- DECHOW, P. M. & DICHEV, I. D. 2002. The Quality of Accruals and Earnings: The Role of Accrual Estimation Errors. *Accounting Review*, 77, 35.
- DECHOW, P. M., GE, W., LARSON, C. R. & SLOAN, R. G. 2011. Predicting Material Accounting Misstatements. *Contemporary Accounting Research*, 28, 17-82.
- DECHOW, P. M., GE, W. & SCHRAND, C. 2010. Understanding Earnings Quality: A Review of the Proxies, their Determinants and their Consequences. *Journal of Accounting and Economics*, 50, 344-401.

DECHOW, P. M. & HUSON, M. R. 1994. The Effect of Restructuring Charges on

Executives' Cash Compensation. Accounting Review, 69, 138-156.

- DECHOW, P. M., RICHARDSON, S. A. & TUNA, I. 2003. Why Are Earnings Kinky? An Examination of the Earnings Management Explanation. *Review of Accounting Studies*, 8, 355-384.
- DECHOW, P. M. & SKINNER, D. J. 2000. Earnings Management: Reconciling the Views of Accounting Academics, Practitioners, and Regulators. *Accounting Horizons*, 14, 235-250.
- DECHOW, P. M. & SLOAN, R. G. 1991. Executive Incentives and the Horizon Problem: An Empirical Investigation. *Journal of Accounting and Economics*, 14, 51-89.
- DECHOW, P. M., SLOAN, R. G. & SWEENEY, A. P. 1995. Detecting Earnings Management. *The Accounting Review*, 70, 193-193.
- DECHUN, W. 2006. Founding Family Ownership and Earnings Quality. *Journal of Accounting Research*, 44, 619-656.
- DEFOND, M. L. & FRANCIS, J. R. 2005. Audit Research after Sarbanes-Oxley. Auditing: A Journal of Practice and Theory, 24, 5–30.
- DEFOND, M. L. & JIAMBALVO, J. 1994. Debt Covenant Violation and Manipulation of Accruals. *Journal of Accounting and Economics*, 17, 145-176.
- DEFOND, M. L. & SUBRAMANYAM, K. R. 1998. Auditor Changes and Discretionary Accruals. *Journal of Accounting and Economics*, 25, 35–67.
- DEGEORGE, F., DING, Y., JEANJEAN, T. & STOLOWY, H. 2004. Analyst Coverage, Earnings Management and Financial Development: An International Study. *Journal of Accounting and Public Policy*, 32, 1-25.
- DEJONG, A., MERTENS, G., VAN DER POEL, M. & VAN DIJK, R. 2014. How Does Earnings Management Influence Investor's Perceptions of Firm Value? Survey Evidence from Financial Analysts. *Review of Accounting Studies*, 19, 606-627.

- DELL'ARICCIA, G. 1998. Asymmetric Information and the Structure of the Banking Industry. *European Economic Review*, 45, 1957-1980.
- DEMERJIAN, P. R., LEV, B., LEWIS, M. F. & MCVAY, S. E. 2013a. Managerial Ability and Earnings Quality. *Accounting Review*, 88, 463-498.
- DEMERJIAN, P. R., LEV, B. & MCVAY, S. 2012. Quantifying Managerial Ability: A New Measure and Validity Tests. *Management Science*, 58, 1229-1248.
- DEMERJIAN, P. R., LEWIS, M. F. & MCVAY, S. E. 2013b. Managerial Ability and Earnings Management. Working paper. *Emory University*.
- DEMSETZ, H. & LEHN, K., 1985. The Structure of Corporate Ownership: Causes and Consequences. *Journal of Political Economy*, 93, 1155–1177.
- DEMSKI, J. S. 1998. Performance Measure Manipulation. Contemporary Accounting Research, 15, 261–285.
- DEMSKI, J. S. & FRIMOR, H. 1999. Performance Measure Garbling Under Renegotiation in Multi-Period Agencies. *Journal of Accounting Research*, 37, 187–214.
- DESAI, H., HOGAN, C. E. & WILKINS, M. S. 2006. The Reputational Penalty for Aggressive Accounting: Earnings Restatements and Management Turnover. *The Accounting Review*, 81, 83-112.
- DHALIWAL, D. S., GLEASON, C. A. & MILLS, L. 2004. Last-Chance Earnings Management: Using the Tax Expense to Meet Analysts' Forecasts. *Contemporary Accounting Research*, 21, 431–459.
- DHARAN, B. G. 2003. Earnings Management with Accruals and Financial Engineering. The Accounting World, Special Issue on Emerging Accounting Issues, February 2003, ICFAI.
- DONALDSON, L. 1990. The Ethereal Hand: Organizational Economics and Management Theory. *Academy of Management Review*, 15, 369–381.
- DUH, R.-R., LEE, W.-C. & LIN, C.-C. 2009. Reversing an Impairment Loss and

Earnings Management: The Role of Corporate Governance. *The International Journal of Accounting*, 44, 113-137.

- DURTSCHI, C. & EASTON, P. 2005. Earnings Management? The Shapes of the Frequency Distributions of Earnings Metrics are not Evidence Ipso Facto. *Journal of Accounting Research*, 43, 557-592.
- DUTTA, S. & FAN, Q. 2014. Equilibrium Earnings Management and Managerial Compensation in a Multi-period Agency Setting. *Review of Accounting Studies*, 19, 1047-1077.
- DYE, R. A. 1984. Inside Trading and Incentives. Journal of Business, 57, 295-313.
- DYE, R. A. 1985. Disclosure of Nonproprietary Information. *Journal of Accounting Research*, 23, 123–145.
- DYRENG, S. D., HANLON, M. & MAYDEW, E. L. 2010. The Effects of Executives on Corporate Tax Avoidance. *Accounting Review*, 85, 1163-1189.
- EASTERWOOD, C. M. 1997. Takeovers and Incentives for Earnings Management: An Empirical Analysis. *Journal of Applied Business Research*, 14, 29-47.
- EATON, J. & ROSEN, H. S. 1983. Agency, Delayed Compensation, and the Structure of Executive Remuneration. *Journal of Finance*, 38, 1489-1505.
- EFENDI, J., SRIVASTAVA, A. & SWANSON, E. P. 2007. Why Do Corporate Managers Misstate Financial Statements? The Role of Option Compensation and Other Factors. *Journal of Financial Economics*, 85, 667-708.
- EILIFSEN, A., KNIVSFLA, K. H. & SAETTEM, F. 1999. Earnings Manipulation: Cost of Capital versus Tax. *European Accounting Review*, 8, 481-491.
- ELBADRY, A., GOUNOPOULOS, D. & SKINNER, F. S. 2013. Governance Quality and Information Asymmetry. *Financial Markets, Institutions and Instruments*, 24 127-157.
- ELITZUR, R. 1995. A Classroom Exercise on Executive Incentive Compensation Schemes. *Managerial and Decision Economics*, 16, 649–652.

- ELITZUR, R. & YAARI, V. 1995. Executive Incentive Compensation and Earnings Manipulation in a Multi-Period Setting. *Journal of Economic Behavior & Organization*, 26, 201-219.
- ERICKSON, M. & WANG, S.-W. 1999. Earnings Management by Acquiring Firms in Stock for Stock Mergers. *Journal of Accounting and Economics*, 27, 149-176.
- ETTREDGE, M., SOO YOUNG, K. & SMITH, D. 2002. Competitive Harm and Companies' Positions on SFAS No. 131. *Journal of Accounting, Auditing & Finance*, 17, 93-109.
- EWERT, R. & WAGENHOFER, A. 2005. Economic Effects of Tightening Accounting Standards to Restrict Earnings Management. *The Accounting Review*, 80, 1101-1124.
- FAMA, E. F. 1980. Agency Problems and the Theory of the Firm. Journal of Political Economy, 88, 288-307.
- FAMA, E. F. & FRENCH, K. R. 1997. Industry Costs of Equity. Journal of Financial Economics, 43, 153-193.
- FAN, Q. 2007. Earnings Management and Ownership Retention for Initial Public Offering Firms: Theory and Evidence. *The Accounting Review*, 82, 27-64.
- FELTHAM, G. A. & JINHAN, P. 2000. Analysis of the Impact of Accounting Accruals on Earnings Uncertainty and Response Coefficients. *Journal of Accounting*, *Auditing & Finance*, 15, 199-220.
- FENG, M., GE, W., LUO, S. & SHEVLIN, T. 2011. Why Do CFOs Become Involved in Material Accounting Manipulations? *Journal of Accounting and Economics*, 51, 21-36.
- FENG, M., GRAMLICH, J. D. & GUPTA, S. 2009. Special Purpose Vehicles: Empirical Evidence on Determinants and Earnings Management. *The Accounting Review*, 84, 1833-1876.
- FISCHER, P. E. & VERRECCHIA, R. E. 2004. Disclosure Bias. *Journal of Accounting* and Economics, 38, 223–250.

- FLOROU, A. & CONYON, M.J. 2004. Does Governance Quality Mitigate Horizon Effects? Investment Patterns Surrounding CEO Departures. SSRN eLibrary. Available at http://ssrn.com/abstract=545982.
- FOSTER, G. 1987. Rambo IX: Briloff and the Capital Market. *Journal of Accounting, Auditing and Finance*, 2, 409–430.
- FRANCIS, J., HUANG, A. H., RAJGOPAL, S. & ZANG, A. Y. 2008. CEO Reputation and Earnings Quality. *Contemporary Accounting Research*, 25, 109-147.
- FRANCIS, J., LAFOND, R., OLSSON, P. & SCHIPPER, K. 2004. Cost of Equity and Earnings Attributes. *The Accounting Review*, 79, 967–1010.
- FRANCIS, J., LAFOND, R., OLSSON, P. & SCHIPPER, K. 2005. The Market Pricing of Accruals Quality. *Journal of Accounting and Economics*, 39, 295-327.
- FRANCIS, J. & PHILBRICK, D. 1993. Analysts' Decisions as Products of a Multi-Task Environment. *Journal of Accounting Research*, 31, 216–230.
- FRANCIS, J., PHILBRICK, D. & SCHIPPER, K. 1994. Shareholder Litigation and Corporate Disclosures. *Journal of Accounting Research*, 32, 137-164.
- FRANCIS, J. R. & WANG, D. 2004. Investor Protection, Auditor Conservatism and Earnings Quality: Are Big 4 Auditors Conservative Only in the United States? Working Paper, University of Missouri-Columbia.
- FRANCIS, J. R. & WANG, D. 2008. The Joint Effect of Investor Protection And Big 4 Audits On Earnings Quality around the World. *Contemporary Accounting Research*, 25, 157–191.
- FRANKEL, R. M., JOHNSON, M. F., NELSON, K. K., KINNEY, W. R. & LIBBY, R. 2002. The Relation between Auditors' Fees for Nonaudit Services and Earnings Management / Discussion of the Relation between Auditors' Fees for Nonaudit Services and Earnings Management. *The Accounting Review*, 77, 71-105,107-114.
- FUDENBERG, D. & TIROLE, J. 1986. A "Signal-Jamming" Theory of Predation. RAND Journal of Economics (RAND Journal of Economics), 17, 366-376.

- FUDENBERG, D. & TIROLE, J. 1995. A Theory Of Income And Dividend Smoothing Based On Incumbency Rents. *Journal of Political Economy*, 103, 75.
- GAIO, C. 2010. The Relative Importance of Firm and Country Characteristics for Earnings Quality around the World. *European Accounting Review*, 19, 693–738.
- GAO, P. & SHRIEVES, R. 2002. Earnings Management and Executive Compensation: A Case of Overdose of Option and Underdose of Salary? Working paper, Northwester University and University of Tennessee.
- GARCÍA LARA, J. M., GARCÍA OSMA, B. & MORA, A. 2005. The Effect of Earnings Management on the Asymmetric Timeliness of Earnings. *Journal of Business Finance & Accounting*, 32, 691-726.
- GARCÍA LARA, J. M., GARCÍA OSMA, B. & PENALVA, F. 2007. Board of Directors' Characteristics and Conditional Accounting Conservatism: Spanish Evidence. *European Accounting Review*, 16, 727-755.
- GARCIA-OSMA, B. & POPE, P. F., 2011. Strategic Balance Sheet Adjustments under First-Time IFRS Adoption and the Consequences for Earnings Quality. Working paper, *Lancaster University*.
- GAREN, J. E. 1994. Executive Compensation and Principal-Agent Theory. *Journal of Political Economy*, 102, 1175.
- GASSEN, J., FULBIER, R. U. & SELLHORN, T. 2006. International Differences in Conditional Conservatism: The Role of Unconditional Conservatism and Income Smoothing. *European Accounting Review*, 15, 527–564.
- GAVER, J. J., GAVER, K. M. & AUSTIN, J. R. 1995. Additional Evidence on Bonus Plans and Income Management. *Journal of Accounting and Economics*, 19, 3-28.
- GAVER, J. J. & PATERSON, J. S. 2000. Earnings Management under Changing Regulatory Regimes: State Accreditation in the Insurance Industry. *Journal of Accounting and Public Policy*, 19, 399-420.
- GE, W., MATSUMOTO, D. & ZHANG, J. L. 2011. Do CFOs Have Style? An Empirical Investigation of the Effect of Individual CFOs on Accounting
Practices. Contemporary Accounting Research, 28, 1141-1179.

- GEIGER, M. A. & NORTH, D. S. 2006. Does Hiring A New CFO Change Things? An Investigation of Changes in Discretionary Accruals. *The Accounting Review*, 81, 781–809.
- GELB, D. S. & ZAROWIN, P. 2002. Corporate Disclosure Policy and the Informativeness of Stock Prices. *Review of Accounting Studies*, 7, 33-52.
- GHOSH, A., MARRA, A. & MOON, D. 2010. Corporate Boards, Audit Committees, and Earnings Management: Pre- and Post-SOX Evidence. *Journal of Business Finance & Accounting*, 37, 1145–1176.
- GILSON, S. C. 1989. Management Turnover and Financial Distress. *Journal of Financial Economics*, 25, 241-262.
- GIROUX, G. 2003. Financial Analysis: A User Approach. New York: Wiley & Sons Inc.
- GIROUX, G. 2004. Detecting Earnings Management. New Jersey: Wiley & Sons Inc.
- GIVOLY, D., HAYN, C. K. & KATZ, S. P. 2010. Does Public Ownership of Equity Improve Earnings Quality? *Accounting Review*, 85, 195-225.
- GODFREY, J., MATHER, P. & RAMSAY, A. 2003. Earnings and Impression Management in Financial Reports: The Case of CEO Changes. *Abacus*, 39, 95-123.
- GOEL, A. M. & THAKOR, A. V. 2003. Why Do Firms Smooth Earnings? *The Journal* of Business, 76, 151–192.
- GOLDMAN, E. & SLEZAK, S. L. 2006. An Equilibrium Model of Incentive Contracts in the Presence of Information Manipulation. *Journal of Financial Economics*, 80, 603-626.
- GONG, G., LOUIS, H. & SUN, A. X. 2008. Earnings Management, Lawsuits, and Stock-For-Stock Acquirers' Market Performance. *Journal of Accounting and Economics*, 46, 62-77.

- GOPALAN, R. & JAYARAMAN, S. 2012. Private Control Benefits and Earnings Management: Evidence from Insider Controlled Firms. *Journal of Accounting Research*, 50, 117-157.
- GRAHAM, J. R., HARVEY, C. R. & RAJGOPAL, S. 2005. The Economic Implications of Corporate Financial Reporting. *Journal of Accounting and Economics*, 40, 3-73.
- GRAHAM, J. R., LI, S. & QIU, J. 2012. Managerial Attributes and Executive Compensation. *The Review of Financial Studies*, 25, 144-186.
- GRAMLICH, J. D. 1991. The Effect of the Alternative Minimum Tax Book Income Adjustment on Accrual Decisions. *Journal of the American Taxation Association*, 13, 36.
- GRAY, R. P. & CLARKE, F. L. 2004. A Methodology for Calculating the Allowance for Loan Losses in Commercial Banks. *Abacus*, 40, 321-341.
- GRIFFIN, P. A. 2003. A League of Their Own? Financial Analysts' Responses to Restatements and Corrective Disclosures. *Journal of Accounting, Auditing & Finance*, 18, 479-518.
- GU, Z. & CHEN, T. 2004. Analysts' Treatment of Nonrecurring Items in Street Earnings. *Journal of Accounting and Economics*, 38, 129-170.
- GU, Z., CHI-WEN JEVONS, L. & ROSETT, J. G. 2005. What Determines the Variability of Accounting Accruals? *Review of Quantitative Finance and Accounting*, 24, 313-334.
- GUAY, W. R., KOTHARI, S. P. & WATTS, R. L. 1996. A Market-Based Evaluation of Discretionary Accrual Models. *Journal of Accounting Research*, 34, 83-105.
- GUL, F. A., FUNG, S. Y. K. & JAGGI, B. 2009. Earnings Quality: Some Evidence on the Role of Auditor Tenure and Auditors' Industry Expertise. *Journal of Accounting and Economics*, 47, 265-287.
- GUNNY, K. A. 2010. The Relation between Earnings Management Using Real Activities Manipulation and Future Performance: Evidence from Meeting

Earnings Benchmarks. Contemporary Accounting Research, 27, 855-888.

- GUO, Y., JUNG, B. & YANG, Y. S. 2015. Product Market Competition and Earnings Quality: A Non-Linear Relationship. Working paper, *Shidler College of Business*.
- HALL, M. & TIDEMAN, N. 1967. Measures of Concentration. *Journal of the American Statistical Association*, 62, 162-8.
- HAN, J. C. Y. & SHIING-WU, W. 1998. Political Costs and Earnings Management of Oil Companies during the 1990 Persian Gulf Crisis. *The Accounting Review*, 73, 103-117.
- HANSEN, G. A. 1999. Bias and Measurement Error in Discretionary Accrual Models. SSRN eLibrary. Available at http://ssrn.com/abstract=192748.
- HARRIS, M. & HOLMSTOM, B. 1982. A Theory of Wage Dynamics. *Review of Economic Studies*, 49, 315-333.
- HARRIS, M. & RAVIV, A. 1979. Optimal Incentive Contracts with Imperfect Information. *Journal of Economic Theory*, 20, 231-259.
- HART, O. D. 1983. The Market Mechanism as an Incentive Scheme. *The Bell Journal* of Economics, 14, 366-382.
- HART, O. D. 1995. Firms, Contracts, and Financial Structure. Oxford: Clarendon Press.
- HASSABELNABY, H. R., MOSEBACH, M. & WHISENANT, S. 2005. The Effect of Technical Default Cost on Discretionary Accounting Decisions. SSRN eLibrary. Available at http://ssrn.com/abstract=660802.
- HAW, I. M., HO, S. S. M. & LI, A. Y. 2011. Corporate Governance and Earnings Management by Classification Shifting. *Contemporary Accounting Research*, 28, 517-553.
- HAYES, R. M. & SCHAEFER, S. 1999. How Much Are Differences In Managerial Ability Worth? *Journal of Accounting and Economics*, 27, 125-148.

- HAZARIKA, S., KARPOFF, J. M. & NAHATA, R. 2012. Internal Corporate Governance, CEO Turnover, and Earnings Management. *Journal of Financial Economics*, 104, 44–69.
- HE, F., SRINIDHI, B., SU, X. & GUL, F. A. 2003. Earnings Management by Changing R&D Expenditure: Evidence on the Role of CEO Stock Compensation. SSRN eLibrary. Available at http://ssrn.com/abstract=459040.
- HEALY, P. M. 1985. The Effect of Bonus Schemes on Accounting Decisions. *Journal* of Accounting and Economics, 7, 85-107.
- HEALY, P. M. & PALEPU, K. G. 2001. Information Asymmetry, Corporate Disclosure, and the Capital Markets: A Review of the Empirical Disclosure Literature. *Journal of Accounting and Economics*, 31, 405-440.
- HEALY, P. M. & WAHLEN, J. M. 1999. A Review of the Earnings Management Literature and Its Implications for Standard Setting. Accounting Horizons, 13, 365-383.
- HEFLIN, F. & HSU, C. 2008. The Impact of the SEC's Regulation of Non-GAAP Disclosures. *Journal of Accounting and Economics*, 46, 349–365.
- HELLMAN, N. 1999. Earnings Manipulation: Cost of Capital versus Tax: A Commentary. *European Accounting Review*, 8, 493–497.
- HERMALIN, B. E. 1992. The Effects of Competition on Executive Behavior. *The RAND Journal of Economics*, 23, 350-365.
- HERMALIN, B. E. & WEISBACH, M. S. 2003. Boards of Directors as an Endogenously Determined Institution: A Survey of the Economic Literature. *Economic Policy Review*, 9, 1–20.
- HILLIER, D., PINDADO, J., DE QUEIROZ, V. & DE LA TORRE, C. 2011. The Impact of Country-Level Corporate Governance on Research and Development. *Journal of International Business Studies*, 42, 76-98.
- HOCHBERG, Y. V., NEWMAN, Y. S. & RIERSON, M. A. 2003. Information in the Time-Series Dynamics of Earnings Management: Evidence from Insider

Trading and Firm Returns. Working Paper, Northwestern University.

- HOLMSTROM, B. 1982. Moral Hazard in Teams. *The Bell Journal of Economics*, 13, 324-340.
- HOLMSTROM, B. & MILGROM, P. 1987. Aggregation and Linearity in the Provision of Intertemporal Incentives. *Econometrica*, 55, 303-328.
- HOLTHAUSEN, R. W., LARCKER, D. F. & SLOAN, R. G. 1995. Annual Bonus Schemes and the Manipulation of Earnings. *Journal of Accounting and Economics*, 19, 29-74.
- HOOGHIEMSTRA, R. 2000. Corporate Communication and Impression Management
 New Perspectives Why Companies Engage in Social Reporting. *Journal of Business Ethics*, 27, 55-68.
- HOSSAIN, M., MITRA, S., REZAEE, Z. & SARATH, B. 2011. Corporate Governance and Earnings Management in the Pre– and Post–Sarbanes-Oxley Act Regimes: Evidence from Implicated Option Backdating Firms. *Journal of Accounting, Auditing & Finance*, 26, 279-315.
- HOUQE, M. N., VAN ZIJL, T., DUNSTAN, K. & KARIM, A. K. M. W. 2012. The Effect of IFRS Adoption and Investor Protection on Earnings Quality around the World. *The International Journal of Accounting*, 47, 333-355.
- HOUSTON, J. F., LEV, B. & TUCKER, J. W. 2008. To Guide or Not to Guide? Causes and Consequences of Stopping Quarterly Earnings Guidance. *Contemporary Accounting Research*, 27, 143-185.
- HRIBAR, P. & COLLINS, D. 2002. Errors in Estimating Accruals: Implications for Empirical Research. *Journal of Accounting Research*, 40, 105–134.
- HRIBAR, P., JENKINS, N. T. & JOHNSON, W. B. 2006. Stock Repurchases as an Earnings Management Device. *Journal of Accounting and Economics*, 41, 3-27.
- HRIBAR, P. & NICHOLS, D. 2007. The Use of Unsigned Earnings Quality Measures in Tests of Earnings Management. *Journal of Accounting Research*, 45, 1017-1053.

- HUBBARD, R. G. & PALIA, D. 1995. Executive Pay and Performance: Evidence from the U.S. Banking Industry. *Journal of Financial Economics*, 38, 105-130.
- HUNT, A., MOYER, S. & SHEVLIN, T. 1997. Earnings Volatility, Earnings Management, and Equity Value. Working paper, *University of Washington at Seattle*.
- HUTTON, A. P., MARCUS, A. J. & TEHRANIAN, H. 2009. Opaque Financial Reports, R2, and Crash Risk. *Journal of Financial Economics*, 94, 67–86.
- IATRIDIS, G. & KADORINIS, G. 2009. Earnings Management and Firm Financial Motives: A Financial Investigation of UK Listed Firms. *International Review of Financial Analysis*, 18, 164-173.
- IMPERATORE, C. & TROMBETTA, M. 2015. Earnings Management in Tough Times: An International Comparison of Banking and Equity Crises. Working paper, *IE Business School*.
- IPINO, E. & PARBONETTI, A. 2011. Mandatory IFRS Adoption: The Trade-Off between Accrual and Real-Based Earnings Management. SSRN eLibrary. Available at http://ssrn.com/abstract=2039711.
- IQBAL, A., ESPENLAUB, S. & STRONG, N. 2009. Earnings Management around UK Open Offers. *The European Journal of Finance*, 15, 29-51.
- JAGGI, B., LEUNG, S. & GUL, F. 2009. Family Control, Board Independence and Earnings Management: Evidence Based on Hong Kong Firms. *Journal of Accounting and Public Policy*, 28, 281-300.
- JAGGI, B. & PICHENG, L. 2002. Earnings Management Response to Debt Covenant Violations and Debt Restructuring. *Journal of Accounting, Auditing & Finance*, 17, 295-324.
- JEANJEAN, T. & STOLOWY, H. 2008. Do Accounting Standards Matter? An Exploratory Analysis of Earnings Management before and After IFRS Adoption. *Journal of Accounting and Public Policy*, 27, 480-494.

JENSEN, M. C. & MECKLING, W. H. 1976. Theory of the Firm: Managerial Behavior,

Agency Costs and Ownership Structure. *Journal of Financial Economics*, 3, 305-360.

- JENSEN, M. C. & MURPHY, K. J. 1990. Performance Pay and Top-Management Incentives. *Journal of Political Economy*, 98, 225-264.
- JENSEN, M. C., MURPHY, K. J. & WRUCK, E. G. 2004. Remuneration: Where we've Been, How We Got to Here, What are the Problems, and How to Fix Them? SSRN eLibrary. Available at http://ssrn.com/abstract=561305.
- JEONG-BON, K. I. M. & CHEONG, H. Y. I. 2006. Ownership Structure, Business Group Affiliation, Listing Status, and Earnings Management: Evidence from Korea. *Contemporary Accounting Research*, 23, 427-464.
- JETER, D. C. & SHIVAKUMAR, L. 1999. Cross-Sectional Estimation of Abnormal Accruals Using Quarterly and Annual Data: Effectiveness in Detecting Event-Specific Earnings Management. Accounting & Business Research (Wolters Kluwer UK), 29, 299-319.
- JIAN, M. & LEE, K. W. 2011. Does CEO Reputation Matter For Capital Investments? *Journal of Corporate Finance*, 17, 929-946.
- JIANG, H., HABIB, A. & HU, B. 2011. Ownership Concentration, Voluntary Disclosures and Information Asymmetry in New Zealand. *The British* Accounting Review, 43, 39-53.
- JIRAPORN, P., DADALT, P., DAVIDSON, W. & NING, Y. 2007. Too Bust to Show Up? An Analysis of Directors' Absences. *The Quarterly Review of Economics* and Finance, 49, 1159-1171.
- JIRAPORN, P., KIM, Y. S. & MATHUR, I. 2006. Does Corporate Diversification Exacerbate or Mitigate Earnings Management? An Empirical Analysis. SSRN eLibrary. Available at http://ssrn.com/abstract=879803.
- JONES, J. J. 1991. Earnings Management during Import Relief Investigations. *Journal* of Accounting Research, 29, 193-228.

KALLUNKI, J.-P. & MARTIKAINEN, M. 1999. Do Firms Use Industry-Wide Targets

When Managing Earnings? Finnish evidence. *The International Journal of Accounting*, 34, 249-259.

- KANAGARETNAM, K., CHEE YEOW, L. & LOBO, G. J. 2014. Influence of National Culture on Accounting Conservatism and Risk-Taking in the Banking Industry. *Accounting Review*, 89, 1115-1149.
- KANG, S. H. & SIVARAMAKRISHNAN, K. 1995. Issues in Testing Earnings Management and an Instrumental Variable Approach. *Journal of Accounting Research*, 33, 353–367.
- KAPLAN, S. N., KLEBANOV, M. M. & SORENSEN, M. 2012. Which CEO Characteristics and Abilities Matter? *Journal of Finance*, 67, 973-1007.
- KARPOFF, J. M., LEE, D. S. & MARTIN, G. S. 2008. The Cost to Firms of Cooking the Books. *Journal of Financial and Quantitative Analysis*. 43, 581-612.
- KARUNA, C. 2007. Industry Product Market Competition and Managerial Incentives. Journal of Accounting and Economics, 43, 275-297.
- KARUNA, C., SUBRAMANYAM, K. R. & TIAN, F. 2012. Industry Product Market Competition and Earnings Management. Working paper, *University of Houston*.
- KE, B. 2001. Why Do CEOs of Publicly Traded Firms Prefer Reporting Small Increases in Earnings and Long Duration of Consecutive Earnings Increases? SSRN eLibrary. Available at http://ssrn.com/abstract=250308.
- KERSTEIN, J. & RAI, A. 2007. Intra-Year Shifts in the Earnings Distribution and Their Implications for Earnings Management. *Journal of Accounting and Economics*, 44, 399-419.
- KIM, Y., LIU, C. & RHEE, S. G. 2003. The Relation of Earnings Management to Firm Size. Working paper, Sung Kyun Kwan University.
- KIM, Y., PARK, M. S. & WIER, B. 2012. Is Earnings Quality Associated with Corporate Social Responsibility? *The Accounting Review*, 87, 761-796.
- KLEIN, A. 2002. Audit Committee, Board of Director Characteristics, and Earnings

Management. Journal of Accounting and Economics, 33, 375-400.

- KOH, P.-S. 2003. On The Association Between Institutional Ownership And Aggressive Corporate Earnings Management In Australia. *The British* Accounting Review, 35, 105-128.
- KOH, P.-S. 2005. Institutional Ownership and Income Smoothing: Australian Evidence. Accounting Research Journal, 18, 93-110.
- KOH, P.-S. 2007. Institutional Investor Type, Earnings Management and Benchmark Beaters. *Journal of Accounting and Public Policy*, 26, 267-299.
- KOLE, S. & LEHN, K. 1999. Deregulation and the Adaptation of Governance Structure: The Case of the Us Airline Industry. *Journal of Financial Economics*, 52, 79-117.
- KOLEV, K., MARQUARDT, C. A. & MCVAY, S. E. 2008. SEC Scrutiny and the Evolution of Non-GAAP Reporting. *The Accounting Review*, 83, 157–184.
- KOTHARI, S. P. 2001. Capital Markets Research in Accounting. *Journal of Accounting* and Economics, 31, 105-231.
- KOTHARI, S. P., LEONE, A. J. & WASLEY, C. E. 2005. Performance Matched Discretionary Accrual Measures. *Journal of Accounting and Economics*, 39, 163-197.
- KOTHARI, S. P., MIZIK, N. & ROYCHOWDHURY, S. 2016. [Forthcoming]. Managing for the Moment: The Role of Real Activity versus Accruals Earnings Management in SEO Valuation. *The Accounting Review*. [Online]. [Accessed 17 March 2015]. Available at http://ssrn. com/abstract= 1982826.
- KRISHNAN, G. V., RAMAN, K. K., KE, Y. & WEI, Y. 2011. CFO/CEO-Board Social Ties, Sarbanes- Oxley, and Earnings Management. Accounting Horizons, 25, 537-557.
- KROSS, W. J., RO, B. T. & SUK, I. 2011. Consistency in Meeting or Beating Earnings Expectations and Management Earnings Forecasts. *Journal of Accounting and Economics*, 51, 37-57.

- KUMAR, K. R. & VISVANATHAN, G. 2003. The Information Content of the Deferred Tax Valuation Allowance. *Accounting Review*, 78, 471.
- KWON, S. S., YIN, Q. J. & HAN, J. 2006. The Effect of Differential Accounting Conservatism on the "Over-Valuation" of High-Tech Firms Relative to Low-Tech Firms. *Review of Quantitative Financial Accounting*, 27, 143-173.
- LAKONISHOK, J. & LEE, I. 2001. Are Insider Trades Informative? The Review of Financial Studies, 14, 79-111.
- LAKONISHOK, J., SHLEIFER, A., THALER, R. H. & VISHNY, R. W. 1991. Window Dressing by Pension Fund Managers. SSRN eLibrary. Available at http://ssrn.com/abstract=471558.
- LAMBERT, R. A. & LARCKER, D. F. 1987. An Analysis of the Use of Accounting and Market Measures of Performance in Executive Compensation Contracts. *Journal of Accounting Research*, 25, 85-125.
- LAMBERT, R. A., LEUZ, C. & VERRECCHIA, R. 2007. Accounting Information, Disclosure, and the Cost of Capital. *Journal of Accounting Research*, 45, 385– 419.
- LANG, M., SMITH RAEDY, J. & WILSON, W. 2006. Earnings Management and Cross Listing: Are Reconciled Earnings Comparable to US Earnings? *Journal of Accounting and Economics*, 42, 255-283.
- LARCKER, D. F. & RICHARDSON, S. A., 2004. Fees Paid to Audit Firms, Accrual Choices, and Corporate Governance. *Journal of Accounting Research*, 42, 625– 658.
- LAUX, C. & LAUX, V. 2009. Board Committees, CEO Compensation, and Earnings Management. *The Accounting Review*, 84, 869-891.
- LEAN, J. & TUCKER, J. 2001. Information Asymmetry, Small Firm Finance and the Role of Government. *Journal of Finance and Management in Public Services*, 1, 43-62.

LEE, B.B. & CHOI, B. 2002. Company Size, Auditor type and Earnings Management.

Journal of Forensic Accounting, 3, 27-50.

- LEHN, K. & ZHAO, M. 2006. CEO Turnover after Acquisitions: Are Bad Bidders Fired? *Journal of Finance*, 61, 1759-1811.
- LENNOX, C. S. 2005. Audit Quality and Executive Officers' Affiliations with CPA Firms. *Journal of Accounting and Economics*, 39, 201–231.
- LEVITT, A. 1998. *The Numbers Game*. [Online]. 28 September, NYU Center for Law and Business, New York. [Accessed 19 May 2015]. Available from: http://www.sec.gov/spch220.txt.
- LEUZ, C., NANDA, D. & WYSOCKI, P. D. 2003. Earnings Management and Investor Protection: An International Comparison. *Journal of Financial Economics*, 69, 505–527.
- LEV, B. 1989. On the Usefulness of Earnings and Earnings Research: Lessons and Directions from Two Decades of Empirical Research. *Journal of Accounting Research*, 27, 153-192.
- LI, J. & ZHOU, J. 2006. Earnings Management and Delisting Risk of Initial Public Offerings. *SSRN eLibrary*. Available at http://ssrn.com/abstract=641021.
- LIBERTY, S. E. & ZIMMERMAN, J. L. 1986. Labor Union Contract Negotiations and Accounting Choices. *Accounting Review*, 61, 692.
- LIEBERMAN, M. B. & ASABA, S. 2006. Why Do Firms Imitate Each Other? Academy of Management Review, 31, 366-385.
- LIN, J. P. 2003. The Long-Run Underperformance of Post-Listing Stock Returns: The Evidence of Earnings Management. SSRN eLibrary. Available at http://ssrn.com/abstract=341740.
- LIN, Z. X. & SHIH, M. S. H. 2006. Does the Stock Market See a Zero or Small Positive Earnings Surprise as a Red Flag? SSRN eLibrary. Available at http://ssrn.com/abstract=929943.

LINCK, J. S., NETTER, J. M. & YANG, T. 2006. Effects and Unintended

Consequences of the Sarbanes-Oxley Act on Corporate Boards. *SSRN eLibrary*. Available at http://ssrn.com/abstract=687496.

- LIPTON, M. & LORSCH, J. 1992. A Modest Proposal for Improved Corporate Governance. *Business Lawyer*, 48, 59–77.
- LO, K. 2008. Earnings Management and Earnings Quality. *Journal of Accounting and Economics*, 45, 350-357.
- MAGNAN, M., MENINI, A. & PARBONETTI, A. 2015. Fair Value Accounting: Information or Confusion for Financial Markets? *Review of Accounting Studies*, 20, 559-591.
- MALMENDIER, U., TATE, G. & YAN, J. O. N. 2011. Overconfidence and Early-Life Experiences: The Effect of Managerial Traits on Corporate Financial Policies. *Journal of Finance*, 66, 1687-1733.
- MANNE, H. G. 1966. Insider Trading and the Stock Market. New York. Free Press.
- MARKARIAN, G. & GILL-DE-ALBORNOZ, B. 2012. Income Smoothing and Idiosyncratic Volatility. Working paper, *IE Business School*.
- MARKARIAN, G. & SANTALO, J. 2010. Product Market Competition, Information and Earnings Management. *Journal of Business Finance & Accounting*, 41, 572-599.
- MARTINEZ, J. A., DESENDER, K. & LAMY, M. L. 2015. Market Discipline and Bank Income Smoothing: Analyzing the Role of Accounting Disclosure Regulation and Competition. Working paper, Universidad Carlos III.
- MARTINI, J. 2015. The Optimal Focus of Transfer Prices: Pre-Tax Profitability versus Tax Minimization. *Review of Accounting Studies*, 20, 866-898.
- MAURITZEN, J. 2014. Sun and Lemons: Getting over Information Asymmetries in the California Solar Market. Working paper, *Department of Business and Management Science, NHH Norwegian School of Economics*.

MAYDEW, E. L. 1997. Tax-Induced Earnings Management by Firms with Net

Operating Losses. Journal of Accounting Research, 35, 83-96.

- MCANALLY, M. L., SRIVASTAVA, A. & WEAVER, C. D. 2008. Executive Stock Options, Missed Earnings Targets, and Earnings Management. *The Accounting Review*, 83, 185-216.
- MCGAHAN, A. M. & PORTER, M. E. 1997. How Much Does Industry Matter, Really? *Strategic Management Journal*, 18, 15-30.
- MCINNIS, J. 2010. Earnings Smoothness, Average Returns, and Implied Cost of Equity Capital. *Accounting Review*, 85, 315-341.
- MCNICHOLS, M. 2000. Research Design Issues in Earnings Management Studies. Journal of Accounting and Public Policy, 19, 313-345.
- MCNICHOLS, M. 2002. Discussion of: The Quality of Accruals and Earnings The Role of Accrual Estimation Errors. *Accounting Review*, 77, 61.
- MCNICHOLS, M. & WILSON, G. P. 1988. Evidence of Earnings Management from the Provision for Bad Debts. *Journal of Accounting Research*, 26, 1-31.
- MCVAY, S. E. 2006. Earnings Management Using Classification Shifting: An Examination of Core Earnings and Special Items. *The Accounting Review*, 81, 501-531.
- MCVAY, S. E., NAGAR, V. & TANG, V. W. 2006. Trading Incentives to Meet the Analyst Forecast. *Review of Accounting Studies*, 11, 575-598.
- MERKL-DAVIES, D. M. & BRENNAN, N. M. 2007. Discretionary Disclosure Strategies in Corporate Narratives: Incremental Information or Impression Management? *Journal of Accounting Literature*, 26, 116-194.
- MICHELSON, S. E., JORDAN-WAGNER, J. & WOOTTON, C. H. 1995. A Market Based Analysis of Income Smoothing. *Journal of Business Finance and Accounting*, 22, 1179–1193.
- MIGUEL, A. D., PINDADO, J. & TORRE, C. D. L. 2004. Ownership Structure and Firm Value: New Evidence from Spain. *Strategic Management Journal*, 25,

1199-1207.

- MILGROM, P. R. & ROBERTS, J. 1992. *Economics, Organization, and Management*. Englewood Cliffs. New Jersey: Prentice-Hall.
- MONKS, R. A. G. & MINOW, N. 2004. Corporate Governance. Oxford: Blackwell.
- MOSES, D. O. 1987. Income Smoothing and Incentives: Empirical Tests Using Accounting Changes. *The Accounting Review*, 62, 358–377.
- MULFORD, C.W. & COMISKEY, E.E. 2002. *The Financial Numbers Game: Detecting Creative Accounting Practices*. New York: John Wiley & Sons.
- MURPHY, K. J. & ZIMMERMAN, J. L. 1993. Financial Performance Surrounding CEO Turnover. *Journal of Accounting and Economics*, 16, 273-315.
- MYERS, J. N., MYERS, L. A. & SKINNER, D. J. 2007. Earnings Momentum and Earnings Management. *Journal of Accounting, Auditing & Finance*, 22, 249-284.
- NALEBUFF, B. J. & STIGLITZ, J. E. 1983. Prizes and Incentives: Towards A General Theory of Compensation and Competition. *Bell Journal of Economics*, 14, 21-43.
- NARAYANAN, M. P. 1985. Managerial Incentives for Short-term Results. *Journal of Finance*, 40, 1469-1484.
- NDUBIZU, G. A. 2007. Do Cross-Border Listing Firms Manage Earnings or Seize a Window of Opportunity? *The Accounting Review*, 82, 1009-1030.
- NELSON, M. W., ELLIOTT, J. A. & TARPLEY, R. L. 2003. How Are Earnings Managed: Examples from Auditors. *Accounting Horizons*, 17, 17–35.
- NELSON, M. W., ELLIOTT, J. A., TARPLEY, R. L. & GIBBINS, M. 2002. Evidence from Auditors about Managers' and Auditors' Earnings Management Decisions / Discussion of Evidence from Auditors about Managers' and Auditors' Earnings Management Decisions. *The Accounting Review*, 77, 175-211.
- NICKELL, S. 1981. Biases in Dynamic Models with Fixed Effects. *Econometrica*, 49, 1417-1426.

- NISKANEN, W. A. 2005. *After Enron: Lessons for Public Policy*. Lanham, MD: Rowman & Littlefield Publishers, Inc.
- NORDBERG, D. & MCNULTY, T. 2013. Creating Better Boards through Codification: Possibilities and Limitations In UK Corporate Governance, 1992–2010. *Business History*, 55, 348-374.
- NOE, T. H. 1997. Insider Trading and the Problem of Corporate Agency. *Journal of Law, Economics, & Organization,* 13, 287-318.
- O'BRIEN, P. C. 2005. Discussion of Earnings Management through Transaction Structuring: Contingent Convertible Debt and Diluted Earnings per Share. *Journal of Accounting Research*, 43, 245-250.
- O'BRIEN, P. C. & BHUSHAN, R. 1990. Analyst Following and Institutional Ownership. *Journal of Accounting Research*, 28, 55-82.
- O'BRIEN, P. C., MCNICHOLS, M. F. & LIN, H.-W. 2005. Analyst Impartiality and Investment Banking Relationships. *Journal of Accounting Research*, 43, 623-650.
- O'CONNELL, B. T. 2004. Enron. Con: "He that filches from me my good name ... makes me poor indeed". *Critical Perspectives on Accounting*, 15, 733-749.
- OBERHOLZER-GEE, F. & WULF, J. 2012. Earnings Management from the Bottom Up: An Analysis of Managerial Incentives below the CEO. Working paper, *Harvard Business School*.
- OECD. 2010. Information Exchanges between Competitors under Competition Law. Paris: Organisation for Economic Co-operation and Development.
- OGNEVA, M. 2012. Accrual Quality, Realized Returns, and Expected Returns: The Importance of Controlling for Cash Flow Shocks. *The Accounting Review*, 87, 1415–1444.
- OSMA, B. 2008. Board Independence and Real Earnings Management: The Case of R&D Expenditure. *Corporate Governance: An International Review*, 16, 116–131.

- OSMA, B. & YOUNG, S. 2009. R&D Expenditure and Earnings Targets. *European* Accounting Review, 18, 7–32.
- OTHMAN, H. B. & ZEGHAL, D. 2006. A Study of Earnings-Management Motives in the Anglo-American and Euro-Continental Accounting Models: The Canadian and French Cases. *The International Journal of Accounting*, 41, 406-435.
- OYER, P. 1998. Fiscal Year Ends and Non-Linear Incentive Contracts: The Effect on Business Seasonality. *Quarterly Journal of Economics*, 113, 149–185.
- PARK, M. S. & PARK, T. 2004. Insider Sales and Earnings Management. *Journal of* Accounting and Public Policy, 23, 381-411.
- PATTEN, D. M. & TROMPETER, G. 2003. Corporate Responses to Political Costs: An Examination of the Relation between Environmental Disclosure and Earnings Management. *Journal of Accounting and Public Policy*, 22, 83-94.
- PEASNELL, K. V., POPE, P. F. & YOUNG, S. 2005. Board Monitoring And Earnings Management: Do Outside Directors Influence Abnormal Accruals? *Journal of Business Finance and Accounting*, 32, 1311–1346.
- PELTIER-RIVEST, D. 1999. The Determinants of Accounting Choices in Troubled Companies. *Quarterly Journal of Business & Economics*, 38, 28.
- PENG, E. Y. 2011. Accruals Quality and the Incentive Contracting Role of Earnings. Journal of Accounting and Public Policy, 30, 460-480.
- PERRY, S. E. & WILLIAMS, T. H. 1994. Earnings Management Preceding Management Buyout Offers. *Journal of Accounting and Economics*, 18, 157-179.
- PERRY, T. 2000. Incentive Compensation for Outside Directors and CEO Turnover. SSRN eLibrary. Available at http://ssrn.com/abstract=236033.
- PERRY, T. & SHIVDASANI, A. 2005. Do Boards Affect Performance? Evidence from Corporate Restructuring. *The Journal of Business*, 78, 1403-1432.
- PERRY, T. & ZENNER, M. 2001. Pay For Performance? Government Regulation and the Structure of Compensation Contracts. *Journal of Financial Economics*, 62,

453-488.

- PETRONI, K. R. 1992. Optimistic Reporting in the Property- Casualty Insurance Industry. *Journal of Accounting and Economics*, 15, 485-508.
- PETROVITS, C. M. 2006. Corporate-Sponsored Foundations and Earnings Management. *Journal of Accounting and Economics*, 41, 335-362.
- PHILLIPS, J., PINCUS, M. & REGO, S. O. 2003. Earnings management: New Evidence Based on Deferred Tax Expense. *The Accounting Review*, 78, 491-521.
- PINCUS, M. & RAJGOPAL, S. 2002. The Interaction between Accrual Management and Hedging: Evidence from Oil and Gas Firms. *Accounting Review*, 77, 127-160.
- PINCUS, M., RAJGOPAL, S. & VENKATACHALAM, M. 2007. The Accrual Anomaly: International Evidence. *The Accounting Review*, 82, 169–203.
- PINDADO, J. & REQUEJO, I. 2014. Panel Data: A Methodology for Model Specification and Testing. Wiley Encyclopedia of Management. John Wiley & Sons, Ltd.
- POPP, A., TOMS, S. & WILSON, J. F. 2003. Strategy, Governance and Industrial Organization: Historical Perspectives on the Dynamics of Industrial Clustering in England. Working Paper, University of Nottingham Business School.
- POURCIAU, S. 1993. Earnings Management and Nonroutine Executive Changes. Journal of Accounting and Economics, 16, 317-336.
- PRAWITT, D. F., SMITH, J. L. & WOOD, D. A. 2009. Internal Audit Quality and Earnings Management. *The Accounting Review*, 84, 1255-1280.
- PRENCIPE, A. & BAR-YOSEF, S. 2011. Corporate Governance and Earnings Management in Family-Controlled Companies. *Journal of Accounting, Auditing* & Finance, 26, 199-227.
- PRENDERGAST, C. 2000. What Trade-off of Risk and Incentives? A.E.R. Papers and Proc., 90, 421-25.

- PRENDERGAST, C. 2002. Incentives and Uncertainty. *Journal of Labor Economics*, 20, 115-S137.
- RAMAKRISHNAN, R. T. S. & THOMAS, J. K. 1998. Valuation of Permanent, Transitory, and Price-Irrelevant Components of Reported Earnings. *Journal of Accounting, Auditing & Finance,* 13, 301-336.
- RAMAN, K. & SHAHRUR, H. 2008. Relationship-Specific Investments and Earnings Management: Evidence on Corporate Suppliers and Customers. *The Accounting Review*, 83, 1041-1081.
- REITENGA, A. L. & TEARNEY, M. G. 2003. Mandatory CEO Retirements, Discretionary Accruals, and Corporate Governance Mechanisms. *Journal of Accounting, Auditing & Finance*, 18, 255-280.
- RICHARDSON, V. J. & WAEGELEIN, J. F. 2002. The Influence of Long-Term Performance Plans on Earnings Management and Firm Performance. *Review of Quantitative Finance and Accounting*, 18, 161-183.
- RONEN, J. & BERMAN, A. 2004. Musing on Post Enron Reforms. *Journal of Accounting, Auditing and Finance*, 19, 331–342.
- RONEN, J., RONEN, T. & YAARI, V. 2003. The Effect of Voluntary Disclosure and Preemptive Preannouncements on Earnings Response Coefficients (ERC) When Firms Manage Earnings. *Journal of Accounting, Auditing & Finance*, 18, 379-409.
- RONEN, J. & SADAN, S. 1981. Smoothing Income Numbers: Objectives, Means, and Implications. Reading, MA: Addison-Wesley.
- RONEN, J., TZUR, J. & YAARI, V. 2006. The Effect of Directors' Equity Incentives on Earnings Management. *Journal of Accounting and Public Policy*, 25, 359– 389.
- RONEN, J. & YAARI, V. 1993. The Disclosure Policy of the Firm in an Efficient Market. *Review of Quantitative Finance and Accounting*, 3, 311-324.

RONEN, J. & YAARI, V. 2008. Earnings Management: Emerging Insights in Theory,

Practice, and Research. New York: Springer Science and Business Media.

- ROSNER, R. L. 2003. Earnings Manipulation in Failing Firms. *Contemporary Accounting Research*, 20, 361-408.
- ROYCHOWDHURY, S. 2006. Earnings Management through Real Activities Manipulation. *Journal of Accounting and Economics*, 42, 335-370.
- SALOP, S. C. 2015. The Evolution and Vitality of Merger Presumptions: A Decision-Theoretic Approach. Working paper, *Georgetown University*.
- SANKAR, M. R. 1999. The Impact of Alternative Forms of Earnings Management on the Returns-Earnings Relation. SSRN eLibrary. Available at http://ssrn.com/abstract=146732.
- SANKAR, M. R. & SUBRAMANYAM, K. R. 2001. Reporting Discretion and Private Information Communication through Earnings. *Journal of Accounting Research*, 39, 365–386.
- SAWICKI, J. & SHRESTHA, K. 2016. [Forthcoming]. Insider Trading and Earnings Management. *Journal of Business Finance & Accounting*. [Online]. [Accessed 10 December 2015]. Available at http://ssrn.com/abstract=1019889.
- SCHERER, F. 1967. Market Structure and the Employment of Scientists and Engineers. *American Economic Review*, 57, 524.
- SCHIPPER, K. 1989. COMMENTARY on Earnings Management. Accounting *Horizons*, 3, 91-102.
- SCHMIDT, K. M. 1997. Managerial Incentives and Product Market Competition. *Review of Economic Studies*, 64, 191-213.
- SCOTT, W. R. 1997. Financial Accounting Theory. New Jersey: Prentice Hall.
- SEYHUN, H. N. 1992. The Effectiveness of the Insider-Trading Sanctions. *Journal of Law and Economics*, 35, 149-182.
- SHANG, A. 2003. Earnings Management and Institutional Ownership. Working paper,

Harvard University.

- SHIVAKUMAR, L. 2000. Do Firms Mislead Investors by Overstating Earnings Before Seasoned Equity Offerings? *Journal of Accounting and Economics*, 29, 339-371.
- SHROFF, P. K., VENKATARAMAN, R. & XIN, B. 2004. Leaders and Followers among Security Analysts: Analysis of Impact and Accuracy. SSRN eLibrary. Available at http://ssrn.com/abstract=487902.
- SINGER, Z. V. I. 2007. Discretionary Financial Reporting: Items Manipulated by IPO Firms, and Investors' Increased Awareness. Working paper, University of California, Berkeley.
- SIREGAR, S. V. & UTAMA, S. 2008. Type of Earnings Management and the Effect of Ownership Structure, Firm Size, and Corporate-Governance Practices: Evidence from Indonesia. *The International Journal of Accounting*, 43, 1-27.
- SLOAN, R. G., 1996. Do Stock Prices Fully Reflect Information in Accruals and Cash Flows About Future Earnings? *The Accounting Review*, 71 (3), 289–315.
- SMITH JR, C. W. 1993. A Perspective on Accounting- Based Debt Covenant Violations. Accounting Review, 68, 289-303.
- SOARES, N. & STARK, A.W. 2009. The Accruals Anomaly Can Implementable Portfolio Strategies be Developed that are Profitable Net of Transactions Costs in the UK? Accounting and Business Research, 39, 321–345.
- SRIVARDHAN, M. 2009. Role of Institutional Investors in Corporate Governance. *SSRN eLibrary*. Available at http://ssrn.com/abstract=1391803.
- STATHOPOULOS, K., ESPENLAUB, S. & WALKER, M. 2007. Agency Theory versus Managerial Ownership Theories: Understanding the Non-Linear Relationship between Managerial Incentives and Firm Risk. Working paper, University of Manchester.
- STEIN, J. C. 1989. Efficient Capital Markets, Inefficient Firms: A Model of Myopic Corporate Behaviour. *Quarterly Journal of Economics*, 104, 655–669.

- STRONG, N. & WALKER, M. 1987. *Information and Capital Markets*. Oxford: Basil Blackwell Ltd.
- STUBBEN, S. R. 2010. Discretionary Revenues as a Measure of Earnings Management. *The Accounting Review*, 85, 695-717.
- SUBRAMANYAM, K. R. 1996. The Pricing of Discretionary Accruals. Journal of Accounting and Economics, 22, 249-281.
- SUN, B. 2014. Executive Compensation and Earnings Management under Moral Hazard. *Journal of Economic Dynamics and Control*, 41, 276-290.
- SUN, L. 2012. Executive Compensation and Contract-Driven Earnings Management. Asian Academy of Management Journal of Accounting and Finance, 8, 111–127.
- SUN, L. & RATH, S. 2009. An Empirical Analysis of Earnings Management in Australia. *International Journal of Human and Social Sciences*, 4, 1069-1085.
- SUNDVIK, D. 2013. Industry Differences in Earnings Management: Evidence from Finnish Private Firms. Working Paper, *Hanken School of Economics*.
- TINAIKAR, S. & XUE, S. 2009. Product Market Competition and Earnings Management: Some International Evidence. Working paper, *Fisher School of Accounting*.
- TEOH, S. H., WELCH, I. & WONG, T. J. 1998. Earnings Management and the Underperformance of Seasoned Equity Offerings1. *Journal of Financial Economics*, 50, 63-99.
- THOMAS, J. K. 1989. Unusual Patterns in Reported Earnings. *The Accounting Review*, 64, 773–787.
- THOMAS, J. K. & ZHANG, X. J. 2000. Identifying Unexpected Accruals: A Comparison of Current Approaches. *Journal of Accounting and Public Policy*, 19, 347–376.
- TONIATO, J., ALMEIDA, J. E. F. & MORAES, F. 2006. Earnings Management and Industry Classification in Brazil: an Exploratory Investigation. *Corporate*

Ownership and Control, 3, 144-149.

- TURNER, L. E. 2001. The State of Financial Reporting Today: An Unfinished Chapter III. [Online]. 21 June, the Third Annual SEC Disclosure and Accounting Conference. [Accessed 10 November 2015]. Available from: http://www.sec.gov/news/speech/spch508.htm.
- VANCIL, R. F. 1987. *Passing the Baton: Managing the Process of CEO Succession*. Boston: Harvard Business School Press.
- VERRECCHIA, R. E. 1983. Discretionary Disclosure. Journal of Accounting and Economics, 5, 179–194.
- VISVANATHAN, G. 2008. Corporate Governance and Real Earnings Management. Academy of Accounting and Financial Studies Journal, 12, 9-22.
- VLAD, A. P. 2015. The Impact of the 2010 EU Bank Stress-Test Results Disclosure on Banks' Earnings Management. Working paper, Université de Cergy-Pontoise.
- WALKER, M. 2013. How Far Can We Trust Earnings Numbers? What Research Tells us About Earnings Management. Accounting and Business Research, 43, 445-481.
- WARNER, J. B., WATTS, R. L. & WRUCK, K. H. 1988. Stock Prices and Top Management Changes. *Journal of Financial Economics*, 20, 461-492.
- WEISBACH, M. S. 1988. Outside Directors and CEO Turnover. Journal of Financial Economics, 20, 431–460.
- WIEDMAN, C. I. & MARQUARDT, C. A. 2002. How are Earnings Managed? An Examination of Specific Accruals. SSRN eLibrary. Available at http://ssrn.com/abstract=375660.
- WINTOKI, M. B., LINCK, J. S. & NETTER, J. M. 2012. Endogeneity and the Dynamics of Internal Corporate Governance. *Journal of Financial Economics*, 105, 581-606.
- WU, Y. W. 1997. Management Buyouts and Earnings Management. Journal of

Accounting, Auditing & Finance, 12, 373-389.

- XIE, B., DAVIDSON, W. N. & DADALT, P. J. 2003. Earnings Management and Corporate Governance: The Roles of the Board and the Audit Committee. *Journal of Corporate Finance*, 9, 295–316.
- XIE, H. 2001. The Mispricing of Abnormal Accruals. *The Accounting Review*, 76, 357–373.
- XUE, Y. 2003. Information Content of Earnings Management: Evidence from Managing Earnings to Exceed Thresholds. SSRN eLibrary. Available at http://ssrn.com/abstract=582601.
- YAARI, H., DADALT, P., RONEN, J. & YAARI, V. 2007. An Accruals Conundrum in Earnings Management Research. Working paper, New York University and Morgan State University.
- YAARI, V. 2005. Smoothing, Conservative Smoothing, and Truth Telling. *SSRN eLibrary*. Available at http://ssrn.com/abstract=754886.
- YE, J. 2006. Accounting Accruals and Tests of Earnings Management. Working Paper, *Baruch College*.
- YERMACK, D. 1996. Higher Market Valuation of Companies with a Small Board of Directors. *Journal of Financial Economics*, 40, 185–211.
- YERMACK, D. 2004. Remuneration, Retention, and Reputation Incentives for Outside Directors. *Journal of Finance*, 59, 2281–2308.
- YOUNG, S. 1999. Systematic Measurement Error in the Estimation of Discretionary Accruals: An Evaluation of Alternative Modelling Procedures. *Journal of Business Finance and Accounting*, 26, 833–862.
- YU, F. 2008. Analyst Coverage and Earnings Management. Journal of Financial Economics, 88, 245–271.
- YUN, F., BARUA, A., CREADY, W. M. & THOMAS, W. B. 2010. Managing Earnings Using Classification Shifting: Evidence from Quarterly Special Items.

Accounting Review, 85, 1303-1323.

- ZANG, A. Y. 2012. Evidence on the Trade-Off between Real Activities Manipulation and Accrual-Based Earnings Management. *The Accounting Review*, 87, 675-703.
- ZHANG, Y., ZHOU, J. & ZHOU, N. 2006. Audit Committee Quality, Auditor Independence, and Internal Control Weaknesses. SSRN eLibrary. Available at http://ssrn.com/abstract=925732.
- ZHAO, Y. & CHEN, K. H. 2008. Staggered Boards and Earnings Management. *The Accounting Review*, 83, 1347-1381.